IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

-----X

In re : Chapter 11

:

SEMCRUDE, L.P., et al., : Case No. 08-11525 (BLS)

:

Debtors. : Jointly Administered

:

FOURTH AMENDED JOINT PLAN OF AFFILIATED DEBTORS PURSUANT TO CHAPTER 11 OF THE BANKRUPTCY CODE

RICHARDS, LAYTON & FINGER, P.A. One Rodney Square P.O. Box 551 Wilmington, Delaware 19899 Mark D. Collins John H. Knight (302) 651-7700

- and -

WEIL, GOTSHAL & MANGES LLP 200 Crescent Court, Suite 300 Dallas, Texas 75201 Martin A. Sosland Sylvia A. Mayer (214) 746-7700

ATTORNEYS FOR THE DEBTORS AND DEBTORS IN POSSESSION

Dated: October 27, 2009

US_ACTIVE:\43164410\12\43164410_12.DOC\73451.0005

	Page
Article I	DEFINITIONS1
1.1	Active States
1.2	Administrative Expense Claim
1.3	Alberta Court1
1.4	Allowed1
1.5	Alon Claim2
1.6	Auriga Revolver/Term Lender Distribution
1.7	Avoidance Actions 2
1.8	Bankruptcy Code
1.9	Bankruptcy Court
1.10	Bankruptcy Rules
1.11	BNPP
1.12	BNPP Claim2
1.13	Board
1.14	Business Day
1.15	Canadian Distribution
1.16	Canadian Plans
1.17	Cash
1.18	Cash Equivalent
1.19	Catsimatidis Group
1.20	Catsimatidis Settlement Order
1.21	Causes of Action
1.22	CCAA
1.23	Chapter 11 Cases
1.24	Claim3
1.25	Class
1.26	Class A New Common Stock
1.27	Class B New Common Stock
1.28	Collateral4

/	4 •	1
l rn	ntın	ued)
ιcυ	111111	ucu

	Page
1.29	Compensation and Benefits Plans
1.30	Confirmation Date
1.31	Confirmation Order
1.32	Contributing Lender Assignment
1.33	Contributing Lenders
1.34	Contributing Lenders' Claims
1.35	Creditor
1.36	Creditors' Committee
1.37	Creditors' Settlement
1.38	Cure Schedule
1.39	Debtors5
1.40	Debtors in Possession
1.41	Disbursement Account(s)
1.42	Disbursing Agent
1.43	Disclosure Statement
1.44	Disclosure Statement Order
1.45	Disputed5
1.46	Disputed Claim Amount
1.47	Disputed Claims Reserve 6
1.48	Disputed Production Receivable
1.49	Downstream Claims 6
1.50	Downstream Purchasers 6
1.51	DTC6
1.52	Eaglwing6
1.53	Eaglwing First Purchaser Producer Twenty-Day Claims
1.54	Effective Date
1.55	Entity
1.56	Equity Interest
1.57	Examiner

(continued)

		Page
1.58	Exit Facility	7
1.59	Fee Auditor	7
1.60	Fee Auditor Order	7
1.61	Final Order	7
1.62	First Purchaser Producer Claims	8
1.63	First Purchaser Producer Twenty-Day Claims	8
1.64	First Purchaser Producers	8
1.65	Fourth Amended Schedules	8
1.66	General Unsecured Claim	8
1.67	Gross Production Receivable	8
1.68	Inactive States	8
1.69	Initial Petition Date	8
1.70	Intercompany Claim	8
1.71	Intercompany Equity Interest	8
1.72	Investigative Order	8
1.73	IRS	8
1.74	June Decisions	8
1.75	Lender Cash	8
1.76	Lender Deficiency Claim	9
1.77	Lender Steering Committee	9
1.78	Lien	9
1.79	Litigation Trust	9
1.80	Litigation Trust Agreement	9
1.81	Litigation Trust Assets	9
1.82	Litigation Trust Board	9
1.83	Litigation Trust Claims	9
1.84	Litigation Trust Fund Reserve Amount	10
1.85	Litigation Trust Funds	10
1.86	Litigation Trust Interests	10

,	4 •	•
(co	ntinı	ied)
(60	11 (1111	icu,

	Page
1.87	Litigation Trustee
1.88	Management Committee
1.89	Management Incentive Plan
1.90	Management Stock
1.91	Minimum Operating Cash
1.92	New Common Stock
1.93	New Holdco
1.94	New Holdco Bylaws
1.95	New Holdco Certificate of Incorporation
1.96	Non-Settling Party
1.97	Notice
1.98	Operators
1.99	Other Proceedings
1.100	Other Secured Claim
1.101	Other Twenty-Day Claim
1.102	Other Twenty-Day Claims Settlement
1.103	Owners
1.104	Person
1.105	Petition Date
1.106	Plan
1.107	Plan Cash
1.108	Plan Currency
1.109	Plan Supplement
1.110	Postpetition Administrative Agent
1.111	Postpetition Financing Agreement
1.112	Postpetition Financing Claim
1.113	Postpetition Financing Order
1.114	Postpetition Lenders
1.115	Prepetition Administrative Agent

/	4 •	1
(cor	ıtını	164)
(COL		ucu,

	Page
1.116	Prepetition Credit Agreement
1.117	Prepetition Lenders
1.118	Priority Non-Tax Claim
1.119	Priority Tax Claim
1.120	Pro Rata Share
1.121	Producer Decisions
1.122	Producer Deficiency Claims
1.123	Producer Plaintiffs
1.124	Producer Representative
1.125	Producer State-Specific Adversary Proceedings
1.126	Producers
1.127	Producers' Committee
1.128	Producers' Committee Professional Fees
1.129	Producers' Committee Retention Order
1.130	Producers' Settlement
1.131	Professional Compensation and Reimbursement Claim
1.132	Record Date
1.133	Released Actions 14
1.134	Reorganized Debtors
1.135	Reorganized SemGroup Companies
1.136	Restricted Cash
1.137	Retained Causes of Action
1.138	Revolver/Term Lender Claim
1.139	Revolver/Term Lender Effective Date Cash
1.140	Schedules
1.141	Second Lien Term Loan Facility
1.142	Second Lien Term Loan Interest
1.143	Secured Claim
1.144	Secured First Purchaser Producer Claims

(- 4	1\
(coi	านท	ued)

	Page
1.145	Secured Lender Claims
1.146	Secured Revolver/Term Lender Claim
1.147	Secured Tax Claim
1.148	Secured Working Capital Lender Claim
1.149	SemCAMS ULC
1.150	SemCAMS ULC Plan
1.151	SemCanada Energy
1.152	SemCanada Energy Claim
1.153	SemCanada Energy Group
1.154	SemCanada Energy Plan
1.155	SemCanada Group
1.156	SemCanada Nova Scotia
1.157	SemCanada Nova Scotia Plan
1.158	SemCrude Pipeline
1.159	SemGroup
1.160	SemGroup Equity Interest
1.161	SemGroup Finance
1.162	Senior Notes 17
1.163	Senior Notes Claim
1.164	Senior Notes Indenture
1.165	Senior Notes Indenture Charging Lien
1.166	Senior Notes Indenture Trustee
1.167	Senior Notes Indenture Trustee Fees
1.168	September 15 Order
1.169	Settling Party
1.170	Swap Claim
1.171	Tax Code
1.172	Termination Procedures Motion
1.173	Third Circuit Appeals

/	4 •	1
LOOI	1tini	I DAI
(CUI	ւաու	ıed)
	-	,

		Page
1.174	Twenty-Day Claim	18
1.175	Undisputed Production Receivable	18
1.176		
1.177	Unsecured First Purchaser Producer Claims	18
1.178	US Term Lender Group	18
1.179	US Term Lender Group Fees	18
1.180	Warrant Agreement	18
1.181	Warrants	19
1.182	White Cliffs Credit Agreement	19
1.183	White Cliffs Credit Agreement Claim	19
1.184	Working Capital Lender Claim	19
1.185	Working Capital Lender Effective Date Cash	19
1.186	Interpretation; Application of Definitions; Rules of Construction	19
Article II	TREATMENT OF ADMINISTRATIVE EXPENSE CLAIMS, POSTPETITION FINANCING CLAIMS, PROFESSIONAL COMPENSATION AND REIMBURSEMENT CLAIMS, AND PRIORITY TAX CLAIMS; PAYMENT OF SENIOR NOTES INDENTURE TRUSTEE FEES AND US TERM LENDER GROUP	
2.1	FEES	
2.1	Administrative Expense Claims	
2.2	Postpetition Financing Claims	
2.3	Professional Compensation and Reimbursement Claims	
2.4	Priority Tax Claims	
2.5	Senior Notes Indenture Trustee Fees	
2.6	US Term Lender Group Fees	
Article III	SETTLEMENTS	
3.1	Producers' Settlement	
3.2	Other Twenty-Day Claims Settlement	
Article IV	CLASSIFICATION OF CLAIMS AND EQUITY INTERESTS	
Article V	TREATMENT OF CLAIMS AND EQUITY INTERESTS	
5.1	Classes 1 through 26 – Priority Non-Tax Claims	3/

/	4 •	1
(COY	ıtın'	ued)
(001		ucu,

		Page
5.2	Classes 27 through 52 – Secured Tax Claims	37
5.3	Classes 53 through 55 – Secured First Purchaser Producer Claims	37
5.4	Classes 70 through 95 – Secured Working Capital Lender Claims	38
5.5	Classes 96 through 121 – Secured Revolver/Term Lender Claims	38
5.6	Class 122 – White Cliffs Credit Agreement Claim	39
5.7	Classes 123 through 148 – Other Secured Claims	39
5.8	Classes 149 through 174 – Senior Notes Claims	40
5.9	Classes 175 through 200 – Lender Deficiency Claims	41
5.10	Classes 201 through 226 – General Unsecured Claims	41
5.11	Classes 227 through 252 – Intercompany Claims	42
5.12	Classes 253 through 278 – Intercompany Equity Interests	42
5.13	Class 279 – SemGroup Equity Interests	42
Article VI	IDENTIFICATION OF CLAIMS AND EQUITY INTERESTS IMPAIRED AND NOT IMPAIRED BY THE PLAN	43
6.1	Impaired and Unimpaired Classes	43
6.2	Controversy Concerning Impairment	43
Article VII	ACCEPTANCE OR REJECTION OF PLAN; EFFECT OF REJECTION BY ONE OR MORE CLASSES OF CLAIMS	
7.1	Impaired Classes to Vote on Plan	43
7.2	Acceptance by Class of Creditors and Holders of Equity Interests	43
7.3	Cramdown	43
Article VIII	IMPLEMENTATION OF THE PLAN	44
8.1	Non-Substantive Consolidation	44
8.2	Restructuring Transactions	44
8.3	Producer Representative	45
8.4	Litigation Trust Arrangements	45
8.5	Section 1145 Securities	45
8.6	Corporate Action	45
8.7	Existence	46
8.8	Vesting of Assets in the Reorganized Debtors	47

(00	ntin	ued)
(CU	111111	ucu,

		Page
8.9	Cancellation of Debt and Equity Securities and Related Obligations	47
8.10	Effectuating Documents and Further Transactions	47
8.11	Compensation and Benefit Plans	48
Article IX	PRESERVATION AND PROSECUTION OF CAUSES OF ACTION HELD BY THE DEBTORS	48
9.1	Preservation and Prosecution of Causes of Action	48
Article X	PROVISION FOR TREATMENT OF DISPUTED CLAIMS OR ASSERTED ADMINISTRATIVE EXPENSES UNDER THE PLAN	49
10.1	Objections to Claims; Prosecution of Disputed Claims or Asserted Administrative Expenses	49
10.2	No Distributions Pending Allowance	50
10.3	Estimation of Claims	50
10.4	Payments and Distributions on Disputed Claims	51
Article XI	THE LITIGATION TRUST	53
11.1	Establishment of the Litigation Trust	53
11.2	Purpose of the Litigation Trust	53
11.3	Funding Expenses of the Litigation Trust	54
11.4	Transfer of Assets	54
11.5	Valuation of Assets	54
11.6	Litigation; Responsibilities of Litigation Trustee	55
11.7	Investment Powers	55
11.8	Annual Distribution; Withholding	56
11.9	Reporting Duties	56
11.10	Trust Implementation	57
11.11	Registry of Beneficial Interests	58
11.12	Termination	58
11.13	Net Litigation Trust Recovery	58
Article XII	PROVISIONS FOR THE ESTABLISHMENT AND MAINTENANCE OF DISBURSEMENT ACCOUNT(S)	58
12.1	Establishment of Disbursement Account(s)	58

/	4 •	•
$I \cap \cap$	ntin	илла і
	ntin	ucu
(-0		,

			Page
	12.2	Maintenance of Disbursement Account(s)	59
	12.3	Establishment of Producer Disbursement Account(s)	59
Article	XIII	PROVISIONS REGARDING DISTRIBUTIONS UNDER THE PLAN	59
	13.1	Time and Manner of Distributions	59
	13.2	Timeliness of Payments	61
	13.3	Distributions by the Disbursing Agent	61
	13.4	Manner of Payment under the Plan	61
	13.5	Delivery of Distributions	61
	13.6	Fractional New Common Stock / Warrants	62
	13.7	Undeliverable Distributions	62
	13.8	Time Bar to Cash Payments	62
	13.9	Distributions After Effective Date	63
	13.10	Setoffs	63
	13.11	Allocation of Plan Distributions Between Principal and Interest	63
	13.12	Record Date	63
	13.13	Senior Notes Indenture Trustee as Claim Holder	63
	13.14	Limited Recoveries	63
Article	XIV	RIGHTS AND POWERS OF DISBURSING AGENT	64
	14.1	Powers of the Disbursing Agent	64
	14.2	Fees and Expenses Incurred From and After the Effective Date	64
	14.3	Exculpation	64
Article :	XV	RIGHTS AND POWERS OF PRODUCER REPRESENTATIVE AND THE PREPETITION ADMINISTRATIVE AGENT	64
	15.1	Powers of the Producer Representative and the Prepetition Administrative Agent	64
	15.2	Exculpation	65
Article	XVI	EXECUTORY CONTRACTS AND UNEXPIRED LEASES	65
	16.1	Assumption or Rejection of Executory Contracts and Unexpired Leases	65
	16.2	Approval of Assumption or Rejection of Executory Contracts and Unexpired Leases	65

TABLE OF CONTENTS (continued)

		I	Page
1	6.3	Inclusiveness	. 66
1	6.4	Cure of Defaults for Assumed Executory Contracts and Unexpired Leases	. 66
1	6.5	Bar Date for Filing Proofs of Claim Relating to Executory Contracts and Unexpired Leases Rejected Pursuant to the Plan	. 66
1	6.6	Insurance Policies	. 66
Article X	KVII	COMMITTEES, EXAMINER, AND FEE AUDITOR	. 67
1	7.1	Dissolution of Creditors' Committee and Producers' Committee	. 67
1	7.2	Examiner	. 67
1	7.3	Fee Auditor	. 68
Article X	KVIII	CONDITIONS PRECEDENT TO EFFECTIVE DATE OF THE PLAN; IMPLEMENTATION PROVISIONS	. 68
1	8.1	Conditions Precedent to Effective Date of the Plan	. 68
1	8.2	Failure of Conditions Precedent	. 70
1	8.3	Waiver of Conditions Precedent	. 70
Article X	XIX	PROVISIONS REGARDING CORPORATE GOVERNANCE AND MANAGEMENT OF THE REORGANIZED DEBTORS AND NEW HOLD CO.	70
1	0.1	HOLDCO	
_	9.1	General	
_	9.2	Directors and Officers of New Holdco	
	.9.3	Issuance of New Securities	
_	9.4	Listing of New Common Stock	
	.9.5	Management Incentive Plan	
Article X		EFFECT OF CONFIRMATION	
	20.1	Title to and Vesting of Assets	
	20.2	Discharge of Claims and Termination of Equity Interests	
	20.3	Discharge of Debtors	
2	20.4	Injunction on Claims	. 72
2	20.5	Term of Existing Injunctions or Stays	
2	20.6	Exculpation	. 73
2	00.7	Preservation of Causes of Action / Reservation of Rights	74

/	4 •	1
(COY	ıtın'	ued)
(001		ucu,

		Page
20.8	Injunction on Causes of Action	74
20.9		
20.1		
20.1	1 Releases by Holders of Claims and Equity Interests	75
20.1	2 Releases by Members of Creditors' Committee	76
20.1	3 Releases by Members of Producers' Committee	76
20.1	4 Release of Guarantors	76
Article XXI	RETENTION OF JURISDICTION	76
21.1	Retention of Jurisdiction	76
Article XXI	I MODIFICATION, REVOCATION, OR WITHDRAWAL OF THE PLAN	78
22.1	Modification of the Plan	78
22.2	Revocation or Withdrawal of the Plan	79
Article XXI	II MISCELLANEOUS PROVISIONS	79
23.1	Effectuating Documents and Further Transactions	79
23.2	Withholding and Reporting Requirements	79
23.3	Plan Supplement	80
23.4	Payment of Statutory Fees	80
23.5	Expedited Tax Determination	80
23.6	Post-Confirmation Date Fees and Expenses	80
23.7	Substantial Consummation	80
23.8	Severability	80
23.9	Governing Law	81
23.1	0 Time	81
23.1	1 Binding Effect	81
23.1	2 Exhibits/Schedules	81
23.1	3 Notices	81
23.1	4 Closing of the Chapter 11 Cases	82
23.1	5 Section Headings	82
23.1	6 Exemption from Registration	82

TABLE OF CONTENTS (continued)

		Page
23.17	Exemption from Transfer Taxes	82
23.18	No Effect on Downstream Claims	82
23.19	Inconsistencies	83

SemCrude, L.P., Chemical Petroleum Exchange, Incorporated, Eaglwing, L.P., Grayson Pipeline, L.L.C., Greyhawk Gas Storage Company, L.L.C., K.C. Asphalt L.L.C., SemCanada II, L.P., SemCanada L.P., SemCrude Pipeline, L.L.C., SemFuel Transport LLC, SemFuel, L.P., SemGas Gathering LLC, SemGas Storage, L.L.C., SemGas, L.P., SemGroup Asia, L.L.C., SemGroup Finance Corp., SemGroup, L.P., SemKan, L.L.C., SemManagement, L.L.C., SemMaterials Vietnam, L.L.C., SemMaterials, L.P., SemOperating G.P., L.L.C., SemStream, L.P., SemTrucking, L.P., Steuben Development Company, L.L.C., and SemCap, L.L.C. hereby propose the following joint chapter 11 plan pursuant to section 1121(a) of the Bankruptcy Code.

ARTICLE I

DEFINITIONS

As used in the Plan, the following terms shall have the respective meanings specified below:

- 1.1 <u>Active States</u> means the states of Kansas, New Mexico, Oklahoma, Texas, and Wyoming.
- 1.2 <u>Administrative Expense Claim</u> means (a) any Claim constituting a cost or expense of administration of the Chapter 11 Cases asserted or authorized to be asserted in accordance with sections 503(b) and 507(a)(2) of the Bankruptcy Code during the period up to and including the Effective Date and (b) any fees or charges assessed against the Debtors' estates pursuant to section 1930, chapter 123, Title 28, United States Code.
 - 1.3 **Alberta Court** means the Court of Queen's Bench of Alberta.
- 1.4 <u>Allowed</u> means, with reference to any Claim, (i) any Claim that has been listed by the Debtors in their Schedules, as such Schedules may be amended by the Debtors from time to time in accordance with Bankruptcy Rule 1009, as liquidated in amount and not disputed or contingent and for which no contrary proof of claim has been filed or objection thereto interposed, (ii) any Claim that is not Disputed, (iii) any Claim that is compromised, settled, or otherwise resolved pursuant to the authority granted to the Debtors or the Reorganized Debtors, as the case may be, pursuant to a Final Order of the Bankruptcy Court, or (iv) any Claim that has been allowed hereunder or by Final Order; provided, however, that Claims allowed solely for the purpose of voting to accept or reject the Plan pursuant to an order of the Bankruptcy Court shall not be considered "Allowed Claims" hereunder; provided, further, that First Purchaser Producer Twenty-Day Claims and Secured First Purchaser Producer Claims shall solely be Allowed in accordance with Section 3.1 of the Plan. With respect to any Other Twenty-Day Claim, such Claim shall solely become Allowed pursuant to (a) the Other Twenty-Day Claims Settlement, (b) a settlement or compromise between the Prepetition Administrative Agent and the holder of any such Other Twenty-Day Claim or (c) a Final Order of the Bankruptcy Court, and shall not be considered Allowed under clause (i) or (ii) above. Unless otherwise specified herein or by order of the Bankruptcy Court, an "Allowed Administrative Expense Claim" (other than of the kind specified in section 503(b)(1)(B) of the Bankruptcy Code which shall be afforded interest at a rate determined under applicable non-bankruptcy law in accordance with section 511 of the

Bankruptcy Code) or "Allowed Claim" shall not, for any purpose under the Plan, include interest on such Administrative Expense Claim or Claim from and after the Petition Date.

- 1.5 <u>Alon Claim</u> means the Twenty-Day Claim in the amount of \$10.5 million for Alon USA LP, which was Allowed pursuant to a settlement agreement approved by the Bankruptcy Court.
- 1.6 <u>Auriga Revolver/Term Lender Distribution</u> means that portion, if any, of a Canadian Distribution resulting from the settlement agreement between SemCAMS ULC and Auriga Energy Inc., dated July 2, 2009, which was approved by an order of the Alberta Court, dated July 14, 2009, that constitutes Revolver/Term Lender Effective Date Cash as set forth in clause (iv) of Section 5.5(b)(y) hereof.
- 1.7 <u>Avoidance Actions</u> means Causes of Action arising under chapter 5 of the Bankruptcy Code, including, but not limited to, Causes of Action arising under sections 502(d), 510, 542, 543, 544, 547, 548, 549, 550 and 553 of the Bankruptcy Code.
- 1.8 **Bankruptcy Code** means chapter 11 of title 11 of the United States Code, as amended from time to time, as applicable to the Chapter 11 Cases.
- 1.9 **Bankruptcy Court** means the United States Bankruptcy Court for the District of Delaware having jurisdiction over the Chapter 11 Cases.
- 1.10 <u>Bankruptcy Rules</u> means the Federal Rules of Bankruptcy Procedure, as promulgated by the United States Supreme Court under section 2075 of title 28 of the United States Code, and any local rules of the Bankruptcy Court, as amended, as applicable to the Chapter 11 Cases.
 - 1.11 **BNPP** means BNP Paribas.
- 1.12 <u>BNPP Claim</u> means the claim of BNPP against SemCanada Energy or any non-Debtor third party relating to a foreign exchange transaction between BNPP and SemCanada Energy that was to settle on July 21, 2008 and pursuant to which SemCanada Energy was obligated to transfer CAD \$5,561,500 to BNPP.
 - 1.13 **Board** means the board of directors of New Holdco.
- 1.14 **Business Day** means any day other than a Saturday, a Sunday, or any other day on which commercial banks in New York, New York are required or authorized to close by law or executive order.
- 1.15 <u>Canadian Distribution</u> means Cash in respect of Claims under the Prepetition Credit Agreement that is distributed to (i) any of the Debtors or the Reorganized Debtors by the Canadian subsidiaries of SemGroup, or (ii) the Prepetition Lenders or the Prepetition Administrative Agent on behalf of the Prepetition Lenders by the Canadian subsidiaries of SemGroup, in each case pursuant to the Canadian Plans or a bankruptcy, receivership or other proceeding of SemCanada Energy, A.E. Sharp Ltd. and CEG Energy Options, Inc.

- 1.16 <u>Canadian Plans</u> means, collectively, the SemCanada Nova Scotia Plan, the SemCAMS ULC Plan and, if approved by the required majority of the creditors of SemCanada Energy, A.E. Sharp Ltd. and CEG Energy Options, Inc. and sanctioned by the Alberta Court, the SemCanada Energy Plan.
- 1.17 <u>Cash</u> means the lawful currency of the United States of America except that Cash distributed pursuant to a Canadian Distribution may include the lawful currency of Canada.
- 1.18 <u>Cash Equivalent</u> means securities or instruments of the type permitted under section 345 of the Bankruptcy Code.
- 1.19 <u>Catsimatidis Group</u> means Messrs. John A. Catsimatidis, J. Nelson Happy, Martin R. Bring, James C. Hansel, Myron L. Turfitt, Matthew Coughlin, Tulsa Energy Acquisitions, L.L.C., United Refining Energy Corporation, and United Refining Company, and each of such Person's affiliates, attorneys, consultants, advisors, and agents.
- 1.20 <u>Catsimatidis Settlement Order</u> means the order of the Bankruptcy Court, approving the Stipulation of Settlement between the Debtors, Terrence Ronan, SemGroup, G.P., L.L.C., the Prepetition Administrative Agent, the Creditors' Committee and the Catsimatidis Group.
- 1.21 <u>Causes of Action</u> means, without limitation, any and all actions, causes of action, proceedings, controversies, liabilities, obligations, rights, suits, damages, judgments, Claims, objections to Claims, benefits of subordination of Claims, and demands whatsoever, whether known or unknown, reduced to judgment, liquidated or unliquidated, fixed or contingent, matured or unmatured, disputed or undisputed, secured or unsecured, assertable directly or derivatively, existing or hereafter arising, in law, equity, or otherwise, based in whole or in part upon any act or omission or other event occurring prior to the Petition Date or during the course of the Chapter 11 Cases, including through the Effective Date.
- 1.22 <u>CCAA</u> means the Canadian Companies' Creditors Arrangement Act, R.S.C. 1985 c. C-36, as amended.
- 1.23 <u>Chapter 11 Cases</u> means the cases commenced under chapter 11 of the Bankruptcy Code by the Debtors on or after the Initial Petition Date, styled <u>In re SemCrude</u>, <u>L.P., et al.</u>, Chapter 11 Case No. 08-11525 (BLS) (Jointly Administered), currently pending before the Bankruptcy Court.
 - 1.24 <u>Claim</u> shall have the meaning set forth in section 101(5) of the Bankruptcy Code.
- 1.25 <u>Class</u> means a category of Claims or Equity Interests as set forth in Article IV of the Plan.
- 1.26 <u>Class A New Common Stock</u> means the Class A New Common Stock of New Holdco authorized under the New Holdco Certificate of Incorporation.
- 1.27 <u>Class B New Common Stock</u> means the Class B New Common Stock of New Holdco authorized under the New Holdco Certificate of Incorporation.

- 1.28 <u>Collateral</u> means any property or interest in property of the estates of the Debtors subject to a Lien to secure the payment or performance of a Claim, which Lien is not subject to avoidance or otherwise invalid under the Bankruptcy Code or applicable state law.
- 1.29 <u>Compensation and Benefits Plans</u> means employee related plans including 401(k) plans, employee insurance benefits, and flexible spending accounts.
- 1.30 <u>Confirmation Date</u> means the date on which the Clerk of the Bankruptcy Court enters the Confirmation Order on the docket of the Bankruptcy Court with respect to the Chapter 11 Cases.
- 1.31 <u>Confirmation Order</u> means the order of the Bankruptcy Court confirming the Plan pursuant to section 1129 of the Bankruptcy Code.
- 1.32 <u>Contributing Lender Assignment</u> means a written assignment of a Contributing Lender's Claims to the Litigation Trust, in substantially the form which shall be included in the Plan Supplement.
- 1.33 <u>Contributing Lenders</u> means (i) holders of Secured Lender Claims or Lender Deficiency Claims who vote to approve the Plan or who execute a Contributing Lender Assignment and (ii) the Prepetition Administrative Agent.
- Contributing Lenders' Claims means any and all Causes of Action held by the Contributing Lenders, solely in their capacity as holders of Secured Lender Claims or Lender Deficiency Claims, or as Prepetition Administrative Agent, as applicable, against (i) SemGroup Energy Partners, G.P., L.L.C. and its subsidiaries, (ii) any Debtor, (iii) all current and former officers, directors, or employees of any Debtor or non-Debtor affiliate, (iv) all affiliates of persons described in clause (iii) hereof except for such Entities which constitute a direct or indirect investment or wholly or partially-owned subsidiary of SemGroup (other than SemGroup Energy Partners, G.P., L.L.C and its subsidiaries as provided in clause (i) hereof) and (v) all Entities that provided services to or conducted transactions with any Debtor or non-Debtor affiliate, including, without limitation, all attorneys, accountants, financial advisors, trading counterparties, and customers or vendors, in each case solely as the provider of services or goods to any Debtor or non-Debtor affiliate; provided, however, that the BNPP Claim will be excluded. For the avoidance of doubt, Contributing Lenders' Claims do not include (a) Released Actions or (b) any Causes of Action asserted against a Prepetition Lender with a Claim under a Swap Contract (as defined in the Prepetition Credit Agreement) or a holder of a Swap Claim to determine whether or not such Swap Contract Claim qualifies as a Lender Swap Obligation under the Prepetition Credit Agreement.
- 1.35 <u>Creditor</u> means any Entity holding a Claim against the Debtors' estates or, pursuant to section 102(2) of the Bankruptcy Code, against property of the Debtors, that arose or is deemed to have arisen prior to or as of the Petition Date.
- 1.36 <u>Creditors' Committee</u> means the statutory committee of creditors holding Unsecured Claims appointed in the Chapter 11 Cases pursuant to section 1102(a)(1) of the Bankruptcy Code, as reconstituted from time to time.

- 1.37 <u>Creditors' Settlement</u> means the global compromise and settlement of certain intercreditor disputes embodied in the Plan and supported by the Creditors' Committee, the Lender Steering Committee and the Debtors, providing for, among other things, (i) an increase in the potential recoveries for the holders of Claims in Classes 149 through 174 from (a) (1) provided certain assumptions in favor of the Prepetition Lenders are made, 0.26% of New Common Stock and (2) 30% of the Litigation Trust Interests to (b) (1) 3.75% of New Common Stock, (2) Warrants to purchase 3.75% of New Common Stock, and (3) 30% of the Litigation Trust Interests and (ii) an increase in the potential recoveries for the holders of Claims in Classes 201 through 226 from (a) (1) provided certain assumptions in favor of the Prepetition Lenders are made, 0.06% of New Common Stock and (2) 10% of the Litigation Trust Interests to (b) (1) 1.25% of New Common Stock, (2) Warrants to purchase 1.25% of New Common Stock, and (3) 10% of the Litigation Trust Interests.
- 1.38 <u>Cure Schedule</u> means the Schedule of Executory Contracts to be Assumed included in Tab 1(A) to the Second Supplement to Plan Supplement to Fourth Amended Joint Plan of Affiliated Debtors Pursuant to Chapter 11 of the Bankruptcy Code, filed by the Debtors on October 23, 2009.
- 1.39 <u>Debtors</u> means SemCrude, L.P., Chemical Petroleum Exchange, Incorporated, Eaglwing, L.P., Grayson Pipeline, L.L.C., Greyhawk Gas Storage Company, L.L.C., K.C. Asphalt L.L.C., SemCanada II, L.P., SemCanada L.P., SemCrude Pipeline, L.L.C., SemFuel Transport LLC, SemFuel, L.P., SemGas Gathering LLC, SemGas Storage, L.L.C., SemGas, L.P., SemGroup Asia, L.L.C., SemGroup Finance Corp., SemGroup, L.P., SemKan, L.L.C., SemManagement, L.L.C., SemMaterials Vietnam, L.L.C., SemMaterials, L.P., SemOperating G.P., L.L.C., SemStream, L.P., SemTrucking, L.P., Steuben Development Company, L.L.C., and SemCap, L.L.C.
- 1.40 <u>Debtors in Possession</u> means the Debtors as debtors in possession pursuant to sections 1101(1) and 1107(a) of the Bankruptcy Code.
- 1.41 <u>Disbursement Account(s)</u> means the account(s) to be established by the Debtors or the Reorganized Debtors, as the case may be, on the Effective Date in accordance with Section 12.1 of the Plan, together with any interest earned thereon.
- 1.42 <u>Disbursing Agent</u> means, solely to effectuate distributions pursuant to the Plan, the Reorganized Debtors or such other Entity as may be designated in the Confirmation Order.
- 1.43 <u>Disclosure Statement</u> means the disclosure statement for the Plan approved by the Bankruptcy Court in accordance with section 1125 of the Bankruptcy Code.
- 1.44 <u>Disclosure Statement Order</u> means the Final Order of the Bankruptcy Court approving the Disclosure Statement in accordance with section 1125 of the Bankruptcy Code.
- 1.45 <u>Disputed</u> means, with reference to (a) any Claim, proof of which was timely and properly filed, or an Administrative Expense Claim, which is disputed under the Plan or as to which a timely objection has been filed pursuant to section 502(d) or 510 of the Bankruptcy Code, or otherwise, and/or request for estimation in accordance with section 502(c) of the Bankruptcy Code and Bankruptcy Rule 3018 has been interposed, and which objection and/or

request for estimation has not been withdrawn or determined by a Final Order or (b) any Claim, proof of which was required to be filed by order of the Bankruptcy Court but as to which a proof of claim was not timely or properly filed. A Claim that is Disputed by the Debtors as to its amount only shall be deemed Allowed in the amount the Debtors admit owing, if any, and Disputed as to the excess.

- 1.46 **Disputed Claim Amount** means the lesser of (a) the liquidated amount set forth in the proof of claim filed with the Bankruptcy Court relating to a Disputed Claim, (b) if the Bankruptcy Court has estimated such Disputed Claim pursuant to section 502(c) of the Bankruptcy Code, the amount of a Disputed Claim as estimated by the Bankruptcy Court, and (c) the amount of such Disputed Claim allowed by the Bankruptcy Court pursuant to section 502 of the Bankruptcy Code, or zero, if such Disputed Claim is disallowed by the Bankruptcy Court pursuant to such section, in either case, regardless of whether the order or judgment allowing or disallowing such Claim has become a Final Order; provided, however, that, in the event that such Claim has been disallowed, but the order of disallowance has not yet become a Final Order, the Bankruptcy Court may require the Disbursing Agent to reserve Plan Currency in an amount that would be attributed to such Claim if it were an Allowed Claim, or a lesser amount, to the extent that the Bankruptcy Court, in its sole and absolute discretion, determines such reserve is necessary to protect the rights of such holder under all of the facts and circumstances relating to the order of disallowance and the appeal of such holder from such order.
 - 1.47 <u>Disputed Claims Reserve</u> means the reserve on account of Disputed Claims.
- 1.48 <u>Disputed Production Receivable</u> means, for any counterparty, the difference between such counterparty's Gross Production Receivable and its Undisputed Production Receivable.
- 1.49 <u>Downstream Claims</u> means claims asserted or to be asserted by one or more First Purchaser Producers against Downstream Purchasers arising from such Downstream Purchasers' purchases (including receipt, exchanges, delivery of or control over) of crude oil and/or gas from Debtors, that Debtors purchased from the First Purchaser Producers prior to the Petition Date, which oil and gas First Purchaser Producers assert Debtors transferred to the Downstream Purchasers and for which Debtors have not been paid in full (without setoff, netting, or other adjustment); <u>provided</u>, that Downstream Claims shall be limited to claims against those Entities referred to on Schedule 1.49 and their successor or subsequent purchasers and, except as may be identified on Schedule 1.49, Downstream Claims shall in no event include any Claims against Prepetition Lenders or their respective affiliates (other than any such claims against Goldman Sachs & Co., J. Aron & Company, and their respective affiliates) who were such at 5:00 p.m. Eastern Daylight Time, on May 14, 2009.
- 1.50 <u>Downstream Purchasers</u> means non-Debtor third parties against which the Downstream Claims are asserted or to be asserted.
 - 1.51 **DTC** means The Depository Trust Company.
 - 1.52 **Eaglwing** means Eaglwing, L.P., an Oklahoma limited partnership.

- 1.53 <u>Eaglwing First Purchaser Producer Twenty-Day Claims</u> means any Twenty-Day Claims asserted against Eaglwing by a First Purchaser Producer as set forth on Schedule 1 entitled "First Purchaser Producer Twenty-Day Claims (Listed by Debtor)" that was derived from the Fourth Amended Schedules.
- 1.54 <u>Effective Date</u> means the first (1st) Business Day following the Confirmation Date on which (a) the conditions to effectiveness of the Plan set forth in Section 18.1 of the Plan have been satisfied or otherwise waived in accordance with Section 18.3 of the Plan and (b) the effectiveness of the Confirmation Order shall not be stayed.
- 1.55 **Entity** means a Person, a corporation, a general partnership, a limited partnership, a limited liability company, a limited liability partnership, an association, a joint stock company, a joint venture, an estate, a trust, an unincorporated organization, a governmental unit or any subdivision thereof, including, without limitation, the Office of the United States Trustee, or any other entity.
 - 1.56 **Equity Interest** means any ownership interest in any of the Debtors.
- 1.57 **Examiner** means Louis J. Freeh, appointed as examiner of the Debtors pursuant to an order of the Bankruptcy Court, dated October 14, 2008.
- 1.58 **Exit Facility** means the working capital financing to be entered into by the Reorganized Debtors and the lenders party thereto in connection with the consummation of the Plan and effective on the Effective Date on such terms as are contained in the commitment letter included in the Plan Supplement.
- 1.59 <u>Fee Auditor</u> means Warren H. Smith of Warren H. Smith & Associates, P.C., appointed as fee auditor pursuant to the Fee Auditor Order.
- 1.60 <u>Fee Auditor Order</u> means the order of the Bankruptcy Court, dated October 22, 2008, appointing the Fee Auditor.
- 1.61 <u>Final Order</u> means an order or judgment of the Bankruptcy Court or any other court of competent jurisdiction as to which the time to appeal, petition for certiorari, or move for reargument or rehearing has expired and as to which no appeal, petition for certiorari, or other proceedings for reargument or rehearing shall then be pending or as to which any appeal, petition for certiorari, reargue, or rehear shall have been waived in writing in form and substance satisfactory to the Debtors or, on and after the Effective Date, the Reorganized Debtors, or, in the event that an appeal, writ of certiorari, or reargument or rehearing thereof has been sought, such order of the Bankruptcy Court or other court of competent jurisdiction shall have been determined by the highest court to which such order was appealed, or certiorari, reargument, or rehearing shall have been denied or resulted in no modification of such order and the time to take any further appeal, petition for certiorari, or move for reargument or rehearing shall have expired; provided, however, that the possibility that a motion under section 502(j) of the Bankruptcy Code, Rule 59 or Rule 60 of the Federal Rules of Civil Procedure, or any analogous rule under the Bankruptcy Rules or applicable state court rules of civil procedure, may be, but has not been, filed with respect to such order shall not cause such order not to be a Final Order.

- 1.62 <u>First Purchaser Producer Claims</u> means the Secured First Purchaser Producer Claims and the Unsecured First Purchaser Producer Claims.
- 1.63 <u>First Purchaser Producer Twenty-Day Claims</u> means the Twenty-Day Claims asserted by First Purchaser Producers as set forth on Schedule 1 entitled "First Purchaser Producer Twenty-Day Claims (Listed By Debtor)" that was derived from the Fourth Amended Schedules.
- 1.64 <u>First Purchaser Producers</u> means Producers with respect to which first sales from oil and gas wells located in the Active States or the Inactive States were made to a Debtor.
- 1.65 **Fourth Amended Schedules** means the amended Schedules filed by the Debtors on August 13, 2009 (Docket Nos. 5204-5212).
- 1.66 <u>General Unsecured Claim</u> means an Unsecured Claim (including a Producer Deficiency Claim), other than a Senior Notes Claim, a Lender Deficiency Claim, or an Intercompany Claim.
- 1.67 <u>Gross Production Receivable</u> means any account receivable of the Debtors arising from the sale of crude oil or natural gas before application of any asserted right of setoff or defense other than cash payment.
- 1.68 <u>Inactive States</u> means the states of Colorado, Louisiana, Missouri, Montana, Nebraska, and North Dakota.
- 1.69 <u>Initial Petition Date</u> means July 22, 2008, the date on which SemGroup, L.P. and 24 of its direct and indirect subsidiaries filed their voluntary petitions for relief commencing certain of the Chapter 11 Cases.
- 1.70 <u>Intercompany Claim</u> means any Unsecured Claim held by any Debtor against any other Debtor.
- 1.71 <u>Intercompany Equity Interest</u> means any Equity Interest held by any of the other Debtors.
- 1.72 <u>Investigative Order</u> means the Bankruptcy Court order, dated September 10, 2008, authorizing and directing the Examiner to conduct an investigation of certain of the Debtors' pre-Petition Date transactions.
- 1.73 <u>IRS</u> means the Internal Revenue Service, an agency of the United States Department of Treasury.
- 1.74 <u>June Decisions</u> means the three Bankruptcy Court decisions issued on June 19, 2009 with respect to the Producer State-Specific Adversary Proceedings for the states of Kansas, Oklahoma and Texas.
- 1.75 <u>Lender Cash</u> means Plan Cash reduced by (a) Cash used to pay or reserved to pay Administrative Expense Claims (including, without duplication, Twenty-Day Claims), (b)

Cash used to pay the Postpetition Financing Claims, (c) Cash used to pay or reserved to pay the Professional Compensation and Reimbursement Claims, (d) Cash used to pay or reserved to pay the portion of any Priority Non-Tax Claims and Priority Tax Claims paid on the Effective Date, (e) the Senior Notes Indenture Trustee Fees, (f) the US Term Lender Group Fees, and (g) the Litigation Trust Funds.

- 1.76 <u>Lender Deficiency Claim</u> means (i) an Unsecured Claim in respect of the unsecured portion of a Revolver/Term Lender Claim or a Working Capital Lender Claim or (ii) a Swap Claim.
- 1.77 <u>Lender Steering Committee</u> means the unofficial steering committee of certain of the Prepetition Lenders, as reconstituted from time to time.
 - 1.78 <u>Lien</u> shall have the meaning set forth in section 101(37) of the Bankruptcy Code.
- 1.79 <u>Litigation Trust</u> means the Entity to be created on the Effective Date in accordance with Section 11.1 of the Plan and the Litigation Trust Agreement for the benefit of holders of Allowed Senior Notes Claims, Allowed Lender Deficiency Claims, and Allowed General Unsecured Claims.
- 1.80 <u>Litigation Trust Agreement</u> means the trust agreement, substantially in the form contained in the Plan Supplement.
- 1.81 <u>Litigation Trust Assets</u> means the Litigation Trust Claims, the Contributing Lenders' Claims, the Litigation Trust Funds, and any other assets acquired by the Litigation Trust after the Effective Date or pursuant to the Plan.
- 1.82 <u>Litigation Trust Board</u> means the group of Persons approved prior to the Effective Date by the Bankruptcy Court, or any replacements thereafter selected in accordance with the provisions of the Litigation Trust Agreement, who shall have the authority set forth in the Litigation Trust Agreement, consisting of three (3) Persons selected by the Lender Steering Committee and three (3) Persons selected by the Creditors' Committee, with decisions to require the approval of at least four (4) of such Persons or, with the consent of the Creditors' Committee, the Prepetition Administrative Agent and the Lender Steering Committee, a group consisting of two (2) Persons selected by the Lender Steering Committee and two (2) Persons selected by the Creditors' Committee, with decisions to require the approval of at least three (3) of such Persons.
- 1.83 <u>Litigation Trust Claims</u> means all Causes of Action asserted, or which may be asserted, by or on behalf of the Debtors or the Debtors' estates, in respect of matters arising prior to the Effective Date, including, but not limited to, Causes of Action in respect of Disputed Production Receivables and Avoidance Actions, but specifically excluding (i) other Causes of Action arising in the ordinary course of the Debtors' business; (ii) Released Actions; and (iii) any Causes of Action against a Prepetition Lender with a Claim under a Swap Contract (as defined in the Prepetition Credit Agreement) or a holder of a Swap Claim to determine whether or not such Swap Contract Claim qualifies as a Lender Swap Obligation under the Prepetition Credit Agreement.

- 1.84 <u>Litigation Trust Fund Reserve Amount</u> means, initially, a reserve of \$15 million, and, thereafter, an amount to be fixed from time to time by the Litigation Trust Board, which reserve shall be in place to fund all expenses of the Litigation Trust, including, but not limited to, the fees and expenses of the professionals selected pursuant to the Litigation Trust Agreement and the costs related to any valuations; <u>provided</u>, <u>however</u>, that the Litigation Trust Fund Reserve Amount shall not exceed \$15 million until the Litigation Trust Funds have been repaid in full to the holders of the Allowed Secured Working Capital Lender Claims as provided in Section 11.3 of the Plan.
- 1.85 <u>Litigation Trust Funds</u> means the \$15 million of Plan Cash used to initially fund the Litigation Trust pursuant to Section 11.3 of the Plan.
- 1.86 <u>Litigation Trust Interests</u> means the beneficial interests in the Litigation Trust to be deemed distributed to holders of Allowed Senior Notes Claims, Allowed Lender Deficiency Claims, and Allowed General Unsecured Claims.
- 1.87 <u>Litigation Trustee</u> means the Entity, solely in its capacity as Litigation Trustee, approved prior to the Effective Date by the Bankruptcy Court to administer the Litigation Trust in accordance with the terms and provisions of Article XI hereof and the Litigation Trust Agreement.
- 1.88 <u>Management Committee</u> means the management committee of SemGroup G.P., L.L.C., the general partner of SemGroup.
- 1.89 <u>Management Incentive Plan</u> means the management incentive plan to be adopted by New Holdco, which shall be in substantially the form contained in the Plan Supplement.
- 1.90 <u>Management Stock</u> means the Class A New Common Stock to be issued by New Holdco to employees and Board members in accordance with the Management Incentive Plan.
- 1.91 <u>Minimum Operating Cash</u> means \$50 million of Cash as of the Effective Date, whether such Cash is held by the Debtors or a Canadian subsidiary of SemGroup.
- 1.92 <u>New Common Stock</u> means the Class A New Common Stock and Class B New Common Stock of New Holdco authorized under the New Holdco Certificate of Incorporation.
- 1.93 <u>New Holdco</u> means the new parent company of the Reorganized Debtors established under Delaware law.
- 1.94 <u>New Holdco Bylaws</u> means the bylaws of New Holdco, substantially in the form contained in the Plan Supplement.
- 1.95 <u>New Holdco Certificate of Incorporation</u> means the certificate of incorporation of New Holdco, substantially in the form contained in the Plan Supplement.
- 1.96 <u>Non-Settling Party</u> means any Creditor who has asserted an Other Twenty-Day Claim that does not elect to accept the Other Twenty-Day Claims Settlement.

- 1.97 <u>Notice</u> means the notice(s) of certain objections to First Purchaser Producer Twenty-Day Claims and Other Twenty-Day Claims issued by the Debtors pursuant to the September 15 Order on July 17, 2009.
- 1.98 <u>Operators</u> means the operators of oil and gas wells located in the Active States or the Inactive States.
- 1.99 <u>Other Proceedings</u> means any action filed by a Producer against any Prepetition Lenders or their respective affiliates, including without limitation Merrill Lynch and Co. or any of its affiliates, and directly or indirectly arising under, in connection with, or related to the provision of services to, or transactions conducted with, any Debtor, but specifically excluding any Downstream Claims.
- 1.100 <u>Other Secured Claim</u> means any Secured Claim other than a Secured Tax Claim, a Secured Lender Claim, a White Cliffs Credit Agreement Claim, or a Secured First Purchaser Producer Claim.
- 1.101 Other Twenty-Day Claim means a Twenty-Day Claim asserted by a Creditor that does not constitute a First Purchaser Producer Twenty-Day Claim as set forth on Schedule 3 entitled "Other Twenty-Day Claims (Listed by Debtor)" that was derived from the Notice or the Fourth Amended Schedules, as applicable.
- 1.102 <u>Other Twenty-Day Claims Settlement</u> means the settlement of the Other Twenty-Day Claims asserted against the Debtors as set forth in Section 3.2 of this Plan.
- 1.103 <u>Owners</u> means working interest, royalty, and overriding royalty interest owners in oil and gas wells located in the Active States or the Inactive States.
- 1.104 **Person** shall have the meaning set forth in section 101(41) of the Bankruptcy Code.
- 1.105 <u>Petition Date</u> means the Initial Petition Date; <u>provided</u>, <u>however</u>, that with respect to that Debtor which commenced its Chapter 11 Case subsequent to July 22, 2008, "Petition Date" shall refer to the date on which such Chapter 11 Case was commenced.
- 1.106 <u>Plan</u> means this Fourth Amended Joint Plan of Affiliated Debtors Pursuant to Chapter 11 of the Bankruptcy Code (including, without limitation, the Plan Supplement and all exhibits, supplements, appendices, and schedules hereto or thereto), either in its present form or as the same may be altered, amended, modified, or supplemented from time to time in accordance with the terms and provisions hereof.
- 1.107 <u>Plan Cash</u> means all Cash and Cash Equivalents (including Restricted Cash) of the Debtors on the Effective Date, other than Minimum Operating Cash.
- 1.108 <u>Plan Currency</u> means the mixture of Plan Cash, Lender Cash, Second Lien Term Loan Interests, New Common Stock, Warrants (if any), and Litigation Trust Interests to be distributed to holders of Allowed Claims pursuant to the Plan.

- 1.109 **Plan Supplement** means the document containing the forms of documents specified in Section 23.3 of the Plan.
- 1.110 **Postpetition Administrative Agent** means Bank of America, N.A., as administrative agent to the Postpetition Lenders under the Postpetition Financing Agreement.
- 1.111 <u>Postpetition Financing Agreement</u> means the Debtor in Possession Credit Agreement, dated as of August 8, 2008, by and among SemCrude, L.P., as borrower, SemGroup, L.P., as a guarantor, SemOperating G.P., L.L.C., as a guarantor, Bank of America, N.A., as administrative agent and L/C issuer, Banc of America LLC, as sole lead arranger and sole book manager, and each lender from time to time party thereto, as entered into pursuant to the Postpetition Financing Order and as modified, amended, or extended from time to time during the Chapter 11 Cases and any of the documents and instruments related thereto.
- 1.112 **Postpetition Financing Claim** means any Claim against the Debtors arising under, in connection with, or related to the Postpetition Financing Agreement.
- 1.113 Postpetition Financing Order means, collectively, (a) the Interim Order (1) Authorizing Debtors to Obtain Postpetition Financing, (2) Authorizing Debtors to Use Cash Collateral, (3) Granting Adequate Protection to Prepetition Secured Parties, and (4) Scheduling a Final Hearing, entered by the Bankruptcy Court on August 8, 2008 and (b) the Final Order (1) Authorizing Debtors to Obtain Postpetition Financing, (2) Authorizing Debtors to Use Cash Collateral, and (3) Granting Adequate Protection to Prepetition Secured Parties, entered by the Bankruptcy Court on September 17, 2008, as each of the foregoing is modified, amended, supplemented or extended from time to time during the Chapter 11 Cases.
- 1.114 **Postpetition Lenders** means, collectively, the banks and other Entities that are parties to the Postpetition Financing Agreement, as lenders thereunder, and their successors and assigns.
- 1.115 <u>Prepetition Administrative Agent</u> means Bank of America, N.A., as administrative agent to the Prepetition Lenders under the Prepetition Credit Agreement.
- 1.116 Prepetition Credit Agreement means that certain Amended and Restated Credit Agreement, dated as of October 18, 2005 (as amended, modified, and supplemented from time to time through and including the Petition Date), among SemCrude, L.P., as U.S. borrower, and SemCams Midstream Company, as Canadian borrower, SemGroup, as a guarantor, SemOperating G.P., L.L.C., as a guarantor, Bank of America, N.A., as administrative agent and L/C issuer, Bank of America Securities, LLC, as joint lead arranger and sole book manager, BNPP as joint lead arranger and co-syndication agent, Bank of Montreal d/b/a "Harris Nesbitt," as co-syndication agent, Bank of Oklahoma, N.A. and The Bank of Nova Scotia, as co-documentation agents, and the lenders party thereto, and any of the documents and instruments related thereto.
- 1.117 <u>Prepetition Lenders</u> means, collectively, the banks and other Entities that are parties to the Prepetition Credit Agreement or hold a security interest in collateral under the Prepetition Credit Agreement, as lenders or holders of swap obligations that constitute Lender

Swap Obligations (as defined in the Prepetition Credit Agreement), and their successors and assigns.

- 1.118 **Priority Non-Tax Claim** means any Claim against the Debtors, other than an Administrative Expense Claim or a Priority Tax Claim, entitled to priority in payment in accordance with sections 507(a)(4), (5), (7), or (9) of the Bankruptcy Code, but only to the extent entitled to such priority.
- 1.119 **Priority Tax Claim** means any Claim of a governmental unit against the Debtors entitled to priority of payment pursuant to sections 502(i) and 507(a)(8) of the Bankruptcy Code.
- 1.120 **Pro Rata Share** means the proportion that a Claim bears to the sum of all Claims within such Class or group of Classes for which an allocation is being determined.
- 1.121 **Producer Decisions** means the June Decisions and any Bankruptcy Court decision issued after September 20, 2009 with respect to the Producer State-Specific Adversary Proceedings for the states of New Mexico or Wyoming.
- 1.122 <u>Producer Deficiency Claims</u> means Claims by First Purchaser Producers with respect to which first sales were made to a Debtor in an Inactive State for amounts in excess of the related First Purchaser Producer Twenty-Day Claims.
- 1.123 **Producer Plaintiffs** means the Producers named as plaintiffs in the Producer State-Specific Adversary Proceedings or in the Other Proceedings.
- 1.124 **Producer Representative** means the person appointed under Section 8.3 of this Plan to resolve any disputes as to the Allowed amount of First Purchaser Producer Twenty-Day Claims, Secured First Purchaser Producer Claims and allowance of any fees and expense reimbursement requests of Producer Plaintiffs.
- 1.125 <u>Producer State-Specific Adversary Proceedings</u> means the declaratory judgment actions filed in the Bankruptcy Court by certain Producers to adjudicate the threshold questions of law related to lien or trust statutes alleged to be applicable in the Active States, Colorado, Missouri, and North Dakota.
- 1.126 **Producers** means the operators of, and working interest, royalty, and overriding royalty interest owners in, oil and gas wells located in the Active or the Inactive States.
- 1.127 <u>Producers' Committee</u> means the committee of certain of the Producers appointed in the Chapter 11 Cases pursuant to section 1102(a)(2) of the Bankruptcy Code, as reconstituted from time to time.
- 1.128 **Producers' Committee Professional Fees** means an amount of up to \$5.5 million for reasonable fees and expenses incurred from the appointment of the Producers' Committee through the Effective Date by the professionals for the Producers' Committee, whose retention has been approved by order of the Bankruptcy Court.

- 1.129 **Producers' Committee Retention Order** means the order entered by the Bankruptcy Court on December 16, 2008, authorizing the retention and employment of Andrews Kurth LLP, as counsel to the Producers' Committee.
- 1.130 **Producers' Settlement** means the compromise and settlement reached on September 14, 2009 among the Debtors, the Prepetition Administrative Agent, the Lender Steering Committee, the Producers' Committee and the Producer Plaintiffs that is embodied in the Plan and resolves, among other things, treatment of the First Purchaser Producer Twenty-Day Claims, the Secured First Purchaser Producer Claims, and the Unsecured First Purchaser Producer Claims.
- 1.131 <u>Professional Compensation and Reimbursement Claim</u> means a Claim for services rendered or reimbursement of expenses incurred through and including the Effective Date pursuant to sections 503(b)(2), (3), (4), or (5) of the Bankruptcy Code.
 - 1.132 **Record Date** means October 30, 2009.
- 1.133 Released Actions means (a) Causes of Action, if any, against the Prepetition Lenders, the Postpetition Lenders, the Prepetition Administrative Agent, the Postpetition Administrative Agent and/or the holders of Swap Claims (other than Bank of Oklahoma and its affiliates) based in whole or in part on any act, omission, transaction, event, or other circumstance and arising under, in connection with, or related to the Prepetition Credit Agreement, the Postpetition Financing Agreement or otherwise arising under, in connection with, or related to the provision of services to, or transactions conducted with, any Debtor or non-Debtor affiliate and (b) any Causes of Action released pursuant to the Catsimatidis Settlement Order; provided, however, that solely for purposes of this definition, "Prepetition Lenders" and "holders of Swap Claims" shall be limited to those Entities who were Prepetition Lenders and/or holders of Swap Claims at 5:00 p.m., Eastern Daylight Time, on May 14, 2009.
 - 1.134 **Reorganized Debtors** means the Debtors on and after the Effective Date.
- 1.135 **Reorganized SemGroup Companies** means New Holdco and its direct and indirect subsidiaries, other than any direct or indirect subsidiaries of SemGroup Holdings, L.P.
- 1.136 **Restricted Cash** means, collectively, the following amounts deposited into a segregated account of the Debtors pursuant to a turnover motion: (i) approximately \$89.8 million deposited by J. Aron & Company, (ii) approximately \$10.7 million deposited by BP Oil Supply Company, and (iii) approximately \$21.6 deposited by ConocoPhillips Company.
- 1.137 <u>Retained Causes of Action</u> means any Causes of Action retained by the Reorganized Debtors and not transferred to the Litigation Trust. For the avoidance of doubt, the Released Actions are not Retained Causes of Action.
- 1.138 **Revolver/Term Lender Claim** means a Claim of a Prepetition Lender under the Prepetition Credit Agreement arising under, in connection with, or related to the Revolver Obligations (as defined in the Prepetition Credit Agreement) or the U.S. Term Obligations (as defined in the Prepetition Credit Agreement).

- 1.139 Revolver/Term Lender Effective Date Cash means (i) \$74 million, plus (ii) net Cash proceeds, if any, in excess of \$51 million received prior to the Effective Date from the sale of assets of SemFuel, L.P. other than inventory and receivables, plus (iii) net Cash proceeds, if any, received after July 21, 2009 and prior to the Effective Date from the sales or assignments of the Revolver/Term Priority Collateral (as defined in the Prepetition Credit Agreement), plus (iv) the portion of net Cash proceeds, if any, received after July 21, 2009 and prior to the Effective Date from the sales or assignments of the Pari Passu Collateral (as defined in the Prepetition Credit Agreement) allocable to the Revolver Obligations (as defined in the Prepetition Credit Agreement) or the U.S. Term Obligations (as defined in the Prepetition Credit Agreement) in accordance with the provisions of the Prepetition Credit Agreement.
- 1.140 <u>Schedules</u> means the schedules of assets and prepetition liabilities, the lists of holders of Equity Interests, and the statements of financial affairs filed by the Debtors in accordance with section 521 of the Bankruptcy Code, Bankruptcy Rule 1007, and the Official Forms of the Bankruptcy Rules, as such schedules and statements have been or may be amended or supplemented on or prior to the Confirmation Date.
- 1.141 <u>Second Lien Term Loan Facility</u> means the secured second lien term loan facility to be entered into by certain of the Reorganized Debtors and the Prepetition Lenders in connection with the consummation of the Plan and effective on the Effective Date, in the aggregate principal amount of \$300 million, substantially in the form contained in the Plan Supplement.
- 1.142 **Second Lien Term Loan Interest** means a participation interest in the Second Lien Term Loan Facility.
- 1.143 Secured Claim means a Claim against the Debtors (a) secured by a Lien on Collateral or (b) subject to setoff under sections 553, 555, 556, 559, 560, and 561 of the Bankruptcy Code, in each case to the extent of the value of the Collateral or to the extent of the amount subject to setoff, as applicable, as determined in accordance with section 506(a) of the Bankruptcy Code or as otherwise agreed to, in writing, by the Debtors or the Reorganized Debtors, as the case may be, and the holder of such Claim; provided, however, that, to the extent that the value of such interest is less than the amount of the Claim which has the benefit of such security, the unsecured portion of such Claim shall be treated as an Unsecured Claim unless, in any such case, the Class of which such Claim is a part makes a valid and timely election in accordance with section 1111(b) of the Bankruptcy Code to have such Claim treated as a Secured Claim to the extent allowed.
- 1.144 <u>Secured First Purchaser Producer Claims</u> means Claims asserted by First Purchaser Producers against Debtors with respect to which first sales were made to a Debtor in an Active State for amounts in excess of such related First Purchaser Producer Twenty-Day Claims. The total amount of such Claims, including any Claims with respect to First Purchaser Producer Twenty-Day Claims and suspense amounts related to Claims of First Purchaser Producers with respect to wells that had first sales to the Debtors prior to December 1, 2007, is as set forth on Schedule 2 entitled "Secured First Purchaser Producer Claims (Listed By Debtor)" that was derived from the Fourth Amended Schedules.

- 1.145 <u>Secured Lender Claims</u> means Secured Working Capital Lender Claims and Secured Revolver/Term Lender Claims.
- 1.146 <u>Secured Revolver/Term Lender Claim</u> means a Revolver/Term Lender Claim to the extent of the value of the Prepetition Lender's Collateral that is allocable to the respective claim in accordance with the provisions of the Prepetition Credit Agreement.
- 1.147 <u>Secured Tax Claim</u> means any Secured Claim that, absent its secured status, would be entitled to priority in right of payment under section 507(a)(8) of the Bankruptcy Code.
- 1.148 <u>Secured Working Capital Lender Claim</u> means a Working Capital Lender Claim to the extent of the value of the Prepetition Lender's Collateral that is allocable to the respective claim in accordance with the provisions of the Prepetition Credit Agreement.
- 1.149 <u>SemCAMS ULC</u> means SemCAMS ULC, a privately-held unlimited liability company incorporated under the Nova Scotia Companies Act.
- 1.150 <u>SemCAMS ULC Plan</u> means the Plan of Arrangement and Reorganization filed by SemCAMS ULC under the CCAA on July 24, 2009, as such plan may be amended, varied or supplemented by SemCAMS ULC from time to time.
- 1.151 <u>SemCanada Energy</u> means SemCanada Energy Company, a privately-held unlimited liability company incorporated under the Nova Scotia Companies Act.
- 1.152 <u>SemCanada Energy Claim</u> means, in the event that the SemCanada Energy Plan is not approved by the requisite majorities of creditors of SemCanada Energy, A.E. Sharp Ltd. and CEG Energy Options, Inc. or the SemCanada Energy Plan is not sanctioned by the Alberta Court, an unsecured principal amount of \$200,000,000 of the Secured Lender Claims against the Debtors, which shall be solely collected from and enforced against SemCanada Energy, A.E. Sharp Ltd. and CEG Energy Options, Inc. (including any security interest granted by SemCanada Energy, A.E. Sharp Ltd. and CEG Energy Options, Inc.) and without recourse to the Debtors, SemCAMS ULC and SemCanada Nova Scotia.
- 1.153 **SemCanada Energy Group** means SemCanada Energy, A.E. Sharp Ltd. and CEG Energy Options, Inc.
- 1.154 <u>SemCanada Energy Plan</u> means the Consolidated Plan of Distribution filed by SemCanada Energy, A.E. Sharp Ltd. and CEG Energy Options, Inc. under the CCAA on July 24, 2009, as such plan may be amended, varied or supplemented by SemCanada Energy, A.E. Sharp Ltd. and CEG Energy Options, Inc. from time to time.
- 1.155 **SemCanada Group** means SemCanada Nova Scotia, SemCAMS ULC, and the SemCanada Energy Group.
- 1.156 <u>SemCanada Nova Scotia</u> means SemCanada Crude Company, a privately-held unlimited liability company incorporated under the Nova Scotia Companies Act.

- 1.157 <u>SemCanada Nova Scotia Plan</u> means the Plan of Arrangement and Reorganization filed by SemCanada Nova Scotia under the CCAA on July 24, 2009, as such plan may be amended, varied or supplemented by SemCanada Nova Scotia from time to time.
- 1.158 **SemCrude Pipeline** means SemCrude Pipeline, L.L.C., a Delaware limited liability company.
 - 1.159 **SemGroup** means SemGroup, L.P., an Oklahoma limited partnership.
 - 1.160 **SemGroup Equity Interest** means an Equity Interest in SemGroup.
- 1.161 <u>SemGroup Finance</u> means SemGroup Finance Corp., a Delaware corporation, which will be renamed SemGroup Holdings Inc. and become New Holdco.
- 1.162 <u>Senior Notes</u> means the 8.75% senior unsecured notes in the original principal amount of \$600 million issued pursuant to the Senior Notes Indenture.
- 1.163 <u>Senior Notes Claim</u> means any Claim against the Debtors and their non-Debtor affiliates arising under, in connection with, or related to the Senior Notes Indenture, including, without limitation, any Claims arising from any guarantees under the Senior Notes Indenture.
- 1.164 <u>Senior Notes Indenture</u> means that certain indenture, dated as of November 18, 2005 (as amended, modified, and supplemented from time to time through and including the Petition Date), by and among SemGroup and SemGroup Finance, as issuers, and the Senior Notes Indenture Trustee.
- 1.165 <u>Senior Notes Indenture Charging Lien</u> means any Lien or other priority in payment or right available to the Senior Notes Indenture Trustee pursuant to the Senior Notes Indenture or otherwise available to the Senior Notes Indenture Trustee under applicable law for, among other things, the payment of the Senior Notes Indenture Trustee Fees.
- 1.166 <u>Senior Notes Indenture Trustee</u> means HSBC Bank USA, N.A., in its capacity as successor to Wells Fargo Bank, National Association, as indenture trustee under the Senior Notes Indenture.
- 1.167 <u>Senior Notes Indenture Trustee Fees</u> means an amount of up to \$750,000 for the Senior Notes Indenture Trustee's reasonable fees and expenses incurred prior to the Effective Date, including the reasonable fees and expenses of the Senior Notes Indenture Trustee's attorneys and agents.
- 1.168 **September 15 Order** means the Order Establishing Procedures for the Resolution of Administrative Claims Asserted Pursuant to Section 503(b)(9) of the Bankruptcy Code and Regarding Payments for Post-Petition Purchases (Docket No. 1376), entered by the Bankruptcy Court on September 15, 2008.
- 1.169 **Settling Party** means any Creditor who elects to accept the Other Twenty-Day Claims Settlement.

- 1.170 **Swap Claim** means an Unsecured Claim of a Prepetition Lender or an Affiliate (as defined in the Prepetition Credit Agreement) of a Prepetition Lender in respect of a Swap Contract (as defined in the Prepetition Credit Agreement) that is not a Lender Swap Obligation (as defined in the Prepetition Credit Agreement).
- 1.171 <u>Tax Code</u> means the Internal Revenue Code of 1986, as amended from time to time.
- 1.172 <u>Termination Procedures Motion</u> means the Examiner's Motion for Entry of an Order Regarding Certain Procedural Issues in Connection with the Termination of the Examination of SemCrude, L.P., et al., filed on May 29, 2009 (Docket No. 4149).
- 1.173 **Third Circuit Appeals** means the appeals pending in the Third Circuit Court of Appeals with respect to the June Decisions.
- 1.174 <u>Twenty-Day Claim</u> means any Claim (whether secured or unsecured) for the value of any goods received by a Debtor within twenty (20) days before the Petition Date in which the goods have been sold to the Debtor in the ordinary course of such Debtor's business.
- 1.175 <u>Undisputed Production Receivable</u> means any account receivable of the Debtors arising from the sale of crude oil or natural gas after application of any counterparty's asserted right of setoff or other defense.
- 1.176 <u>Unsecured Claim</u> means any Claim against the Debtors, other than an Administrative Expense Claim, a Secured Claim, a Professional Compensation and Reimbursement Claim, or a Priority Tax Claim.
- 1.177 <u>Unsecured First Purchaser Producer Claims</u> means (a) Claims by First Purchaser Producers with respect to which first sales were made to a Debtor in an Inactive State for amounts in excess of such First Purchaser Producers' related Twenty-Day Claims and (b) suspense amounts related to Claims of First Purchaser Producers with respect to wells that had first sales to the Debtors prior to December 1, 2007.
- 1.178 <u>US Term Lender Group</u> means the ad hoc group of holders of US Term Loans and Revolver Loans (each as defined in the Prepetition Credit Agreement) formed in July 2008 and represented throughout the Chapter 11 Cases by Ropes & Gray LLP and Saul Ewing LLP, as constituted from time to time, and currently comprised of those holders of US Term Loans and Revolver Loans (each as defined in the Prepetition Credit Agreement) set forth in the Rule 2019 Statement (Ropes & Gray LLP and Saul Ewing LLP) filed by SemCrude US Term Lender Group (Docket No. 4335).
- 1.179 <u>US Term Lender Group Fees</u> means an amount of up to \$930,000 for reasonable fees and expenses incurred from the Petition Date through July 15, 2009 by the professionals for the US Term Lender Group, consisting of Ropes & Gray LLP, Alvarez & Marsal, Saul Ewing LLP, and Duff & Phelps.
- 1.180 <u>Warrant Agreement</u> means the agreement governing the issuance of the Warrants, substantially in the form contained in the Plan Supplement.

- 1.181 <u>Warrants</u> means warrants to purchase shares of New Common Stock issued by New Holdco pursuant to the Warrant Agreement.
- 1.182 White Cliffs Credit Agreement means that certain credit agreement, dated as of June 17, 2008 (as amended, modified, and supplemented from time to time through and including the Petition Date), among SemCrude Pipeline, as borrower, General Electric Capital Corporation, as administrative agent, and the lenders party thereto, consisting of (i) a \$60 million revolving credit facility and (ii) a \$60 million term loan facility, and any of the documents and instruments related thereto.
- 1.183 White Cliffs Credit Agreement Claim means any Claim against the Debtors arising under, in connection with, or related to the White Cliffs Credit Agreement.
- 1.184 Working Capital Lender Claim means a Claim of a Prepetition Lender (or an Affiliate (as defined in the Prepetition Credit Agreement) thereof) arising under, in connection with, or related to the Working Capital Obligations (as defined in the Prepetition Credit Agreement) under the Prepetition Credit Agreement (or, in the case of a Working Capital Obligation that is also a Lender Swap Obligation (as defined in the Prepetition Credit Agreement), the related Swap Contract (as defined in the Prepetition Credit Agreement)).
- 1.185 <u>Working Capital Lender Effective Date Cash</u> means (i) all Lender Cash as of the Effective Date, <u>minus</u> (ii) Revolver/Term Lender Effective Date Cash.
- 1.186 <u>Interpretation; Application of Definitions; Rules of Construction</u> Unless the context otherwise requires, any capitalized term used and not defined herein or elsewhere in the Plan that is defined in the Bankruptcy Code shall have the meaning assigned to that term in the Bankruptcy Code. Wherever from the context it appears appropriate, each term stated in either the singular or the plural shall include both the singular and the plural and pronouns stated in the masculine, feminine, or neuter gender shall include the masculine, feminine, and neuter. Unless otherwise specified, (a) all article, section, schedule, or exhibit references in the Plan are to the respective article of, section in, schedule to, or exhibit to the Plan, as the same may be altered, amended, modified, or supplemented from time to time in accordance with the terms and provisions hereof and (b) all references to dollars are to the lawful currency of the United States of America. The words "herein," "hereof," "hereto," "hereunder," and other words of similar import refer to the Plan as a whole and not to any particular section, subsection, or clause contained in the Plan. The rules of construction contained in section 102 of the Bankruptcy Code shall apply to the construction of the Plan. In computing any period of time prescribed or allowed by the Plan, unless otherwise expressly provided, the provisions of Bankruptcy Rule 9006(a) shall apply. The headings in the Plan are for convenience of reference only and shall not limit or otherwise affect the provisions of the Plan.

ARTICLE II

TREATMENT OF ADMINISTRATIVE EXPENSE CLAIMS, POSTPETITION FINANCING CLAIMS, PROFESSIONAL COMPENSATION AND REIMBURSEMENT CLAIMS, AND PRIORITY TAX CLAIMS; PAYMENT OF SENIOR NOTES INDENTURE TRUSTEE FEES AND US TERM LENDER GROUP FEES

- 2.1 Administrative Expense Claims. On the later to occur of (a) the Effective Date and (b) the date on which an Administrative Expense Claim (including an Unsecured Claim entitled to priority under section 503(b)(9) of the Bankruptcy Code) shall become an Allowed Claim, the Reorganized Debtors, or in the case of the First Purchaser Producer Twenty-Day Claims, the Producer Representative, shall (i) pay to each holder of an Allowed Administrative Expense Claim, in Cash, the full amount of such Allowed Administrative Expense Claim or (ii) satisfy and discharge such Allowed Administrative Expense Claim in accordance with such other terms no more favorable to the claimant than as may be agreed upon by and between the holder thereof and the Debtors or the Reorganized Debtors, as the case may be; provided, however, that the First Purchaser Producer Twenty-Day Claims shall be Allowed and paid by the Producer Representative on the Effective Date in accordance with Section 3.1 hereof; provided, further, that at the Effective Date the Reorganized Debtors shall reserve in Cash an amount equal to the aggregate amount of the Other Twenty-Day Claims for any Non-Settling Parties (at that date) set forth on Schedule 3 hereto or such lesser amount as shall be approved by the Bankruptcy Court and hold such reserve for any such Disputed Other Twenty-Day Claim until the final adjudication or resolution of such Claim; provided, further, that Allowed Administrative Expense Claims representing liabilities incurred by the Debtors in Possession during the Chapter 11 Cases shall be paid by the Reorganized Debtors in accordance with the terms and conditions of the particular transaction and any agreements relating thereto. Notwithstanding the provisions of the Producers' Committee Retention Order, the amount of Allowed Administrative Expense Claims for Producers' Committee Professional Fees shall not reduce any payment under Section 2.1 hereof.
- Postpetition Financing Claims. On the Effective Date, (a) all outstanding Postpetition Financing Claims shall be indefeasibly paid and satisfied, in full, in Cash by the Debtors, (b) all commitments under the Postpetition Financing Agreement will terminate, (c) all letters of credit outstanding under the Postpetition Financing Agreement shall either (A) be returned to the issuer undrawn and marked "cancelled," or (B) be cash collateralized (with funds borrowed under the Exit Facility) as and to the extent required by the terms of the Postpetition Financing Agreement or (C) be collateralized by back-to-back letters of credit provided to the issuer in an amount equal to 105% of the face amount of the outstanding letters of credit, in form and substance and from a financial institution acceptable to such issuer, and (d) all money posted by the Debtors in accordance with the Postpetition Financing Agreement and the agreements and instruments executed in connection therewith shall be released to the applicable Reorganized Debtors for distribution in accordance with the terms and provisions of the Plan. Nothing in this Plan or in the Confirmation Order, whether under section 1141 of the Bankruptcy Code or otherwise, shall discharge any remaining Postpetition Financing Claims.
- 2.3 **Professional Compensation and Reimbursement Claims.** All Entities seeking an award by the Bankruptcy Court of compensation for services rendered or reimbursement of

expenses incurred through and including the Effective Date pursuant to sections 503(b)(2), (3), (4), or (5) of the Bankruptcy Code shall (i) file their respective applications for allowances of compensation for services rendered and reimbursement of expenses incurred through the Effective Date by no later than the date that is sixty (60) days after the Effective Date or such other date as may be fixed by the Bankruptcy Court and (ii) if granted such an award by the Bankruptcy Court, be paid in full in such amounts as are Allowed by the Bankruptcy Court (A) on the date that such Professional Compensation and Reimbursement Claim becomes an Allowed Professional Compensation and Reimbursement Claim, or as soon thereafter as is reasonably practicable or (B) upon such other terms as may be mutually agreed upon between such holder of a Professional Compensation and Reimbursement Claim and the Reorganized Debtors. Objections to Professional Compensation and Reimbursement Claims shall be filed no later than thirty (30) days after an application or request for such Claim is filed with the Bankruptcy Court.

- 2.4 **Priority Tax Claims.** Except to the extent that a holder of an Allowed Priority Tax Claim has been paid by the Debtors prior to the Effective Date or agrees to a different treatment, each holder of an Allowed Priority Tax Claim shall receive, at the sole option and discretion of the Reorganized Debtors, (i) Cash in an amount equal to such Allowed Priority Tax Claim on the Effective Date, (ii) in accordance with section 1129(a)(9)(C) of the Bankruptcy Code, equal semi-annual Cash payments in an aggregate amount equal to such Allowed Priority Tax Claim, together with interest at a rate determined under applicable non-bankruptcy law in accordance with section 511 of the Bankruptcy Code, over a period ending not later than five (5) years after the Petition Date, or (iii) upon such other terms determined by the Bankruptcy Court to provide the holder of such Allowed Priority Tax Claim deferred Cash payments having a value, as of the Effective Date, equal to such Allowed Priority Tax Claim.
- 2.5 <u>Senior Notes Indenture Trustee Fees.</u> The Senior Notes Indenture Fees shall be paid within ten (10) Business Days after the Effective Date to the Senior Notes Indenture Trustee as part of the distribution to holders of Senior Notes Claims; <u>provided, however, that the Senior Notes Indenture Trustee shall, on or prior to the Effective Date, provide to the Reorganized Debtors and the Lender Steering Committee (both of which preserve their right to dispute the payment of any portion of the invoiced fees and expenses if such fees and expenses are deemed to be unreasonable) fee statements (including reasonable documentation) with respect thereto, which may be reviewed by the Reorganized Debtors and the Lender Steering Committee for a period of up to ten (10) Business Days before payment is made. For the avoidance of doubt, any portion of the Senior Notes Indenture Trustee Fees not paid as part of the distribution to the Senior Notes Claims within ten (10) Business Days after the Effective Date may be satisfied pursuant to the Senior Notes Indenture Charging Lien.</u>
- 2.6 <u>US Term Lender Group Fees</u>. The US Term Lender Group Fees shall be paid within ten (10) Business Days after the Effective Date to the relevant professional of the US Term Lender Group as part of the distribution to holders of Secured Revolver/Term Lender Claims; <u>provided</u>, <u>however</u>, the US Term Lender Group shall, on or prior to the Effective Date, provide to the Lender Steering Committee (which preserves its right to dispute the payment of any portion of the invoiced fees and expenses if such fees and expenses are deemed to be unreasonable) fee statements (including reasonable documentation) with respect thereto, which

may be reviewed by the Lender Steering Committee for a period of up to ten (10) Business Days before payment is made.

ARTICLE III

SETTLEMENTS

- 3.1 **Producers' Settlement**. The Producers' Settlement among the Debtors, the Prepetition Administrative Agent, the Lender Steering Committee, the Producers' Committee and the Producer Plaintiffs provides for the following payments under the Plan and the resolution of certain ongoing litigation as described below:
- (a) Payment to Producer Representative. On the Effective Date, Cash in the total amount of \$172.5 million less the aggregate amount of First Purchaser Producer Twenty-Day Claims included in the cure amounts listed in the Cure Schedule shall be distributed to the Producer Representative who shall be responsible for making distributions to Producers in accordance with the Plan, including this Section 3.1. Notwithstanding anything to the contrary in this Plan or elsewhere, the Producers shall not be entitled to any further payment from the Debtors, the Prepetition Administrative Agent or the Prepetition Lenders on account of their Claims other than (i) the cure amounts listed on the Cure Schedule and (ii) the distributions payable to the holders of Unsecured First Purchaser Producer Claims as holders of General Unsecured Claims.
- First Purchaser Producer Twenty-Day Claims. On the Effective Date, the (b) Producer Representative shall make distributions to holders of Allowed First Purchaser Producer Twenty-Day Claims in accordance with Section 2.1 hereof. Each holder of a First Purchaser Producer Twenty-Day Claim shall be deemed, as of the Effective Date, to have an Allowed First Purchaser Producer Twenty-Day Claim equal to (i) the amount listed on Schedule 1 hereto plus (ii) any additional amount, if any, agreed to in writing by the Producers' Committee or the Producer Representative or Allowed by Final Order less (iii) the amount of First Purchaser Producer Twenty-Day Claims for such holder included in the cure amounts listed in the Cure Schedule. Pursuant to the September 15 Order, only holders of First Purchaser Producer Twenty-Day Claims which timely filed objections to the Notice may seek allowance of its Claim in an amount different than listed on Schedule 1 hereto which disputes shall be resolved pursuant to the September 15 Order by the Producer Representative in accordance with Section 8.3 hereof. Notwithstanding anything to the contrary in this Plan or elsewhere, the sum of (i) 100% of the aggregate amounts of the actual payments of the First Purchaser Producer Twenty-Day Claims against Debtors other than Eaglwing plus (ii) 65% of the aggregate amount of the actual payments for the First Purchaser Producer Twenty-Day Claims against Eaglwing shall not exceed (x) \$125.5 million less (y) the aggregate amount of First Purchaser Producer Twenty-Day Claims included in the cure amounts listed in the Cure Schedule. Notwithstanding anything to the contrary herein, the Producer Representative shall not be required to pay the amount of any First Purchaser Producer Twenty-Day Claims listed on Schedule 1 for a Producer to the extent the amount thereof exceeds the amount of the First Purchaser Producer Twenty-Day Claims actually asserted by such Producer in its proof of claim.

- <u>Secured First Purchaser Producer Claims</u>. As soon as practicable after the Effective Date, the Producer Representative shall pay each holder of an Allowed Secured First Purchaser Producer Claim its Pro Rata Share of Cash in an amount equal to \$47 million less the sum of (i) the aggregate amount of Cash paid or to be paid in respect of Allowed First Purchaser Producer Twenty-Day Claims against Eaglwing in excess of 65% of the aggregate amount of the First Purchaser Producer Twenty-Day Claims against Eaglwing on Schedule 1 hereto until the Allowed First Purchaser Producer Twenty-Day Claims against Eaglwing are paid in full, (ii) the amount of Cash paid or to be paid pursuant to Section 3.1(d) of the Plan, (iii) the fees and expenses incurred by the Producer Representative and its professionals paid or to be paid and (iv) the amount of Cash, if any, paid or to be paid to the holders of Allowed First Purchaser Twenty-Day Claims in excess of the sum of (x) \$125.5 million less (y) the aggregate amount of First Purchaser Producer Twenty-Day Claims included in the cure amounts listed in the Cure Schedule, exclusive of Cash distributed pursuant to this Section 3.1(c)(i) (the "Adjusted Settlement Amount"). The amount of the Secured First Purchaser Producer Claim for each holder of such Claim to be used in determining such holder's Pro Rata Share of the Adjusted Settlement Amount shall be calculated as follows: the amount of the Secured First Purchaser Producer Claim shown for such holder on the "Total" column on Schedule 2 hereto less (i) the amount, if any, of the Secured First Purchaser Producer Claim included in the cure amounts listed in the Cure Schedule, and if any remains, less (ii) any suspense amounts included in the "Total" column on Schedule 2 hereto for such holder with respect to wells that had first sales to the Debtors prior to December 1, 2007, and if any remains, less (iii) the total actual amount paid to such holder for First Purchaser Producer Twenty-Day Claims applicable to oil and gas sold from leases in the Active States which was included in the payment made to such holder pursuant to Section 3.1(b) hereof. No portion of any Secured First Purchaser Producer Claim shall be an Unsecured Claim against the Debtors for any purpose under this Plan. No portion of any unpaid Secured First Purchaser Producer Claim is released, compromised, or discharged under this Plan solely for the purposes of any Downstream Claims against Downstream Purchasers, which Downstream Claims Producers shall be free to assert in courts of competent jurisdiction.
- Producers' Committee or by the Producer Representative, as soon as practicable after the Effective Date, the Producer Representative shall reimburse the Producers named as plaintiffs in the Producer State-Specific Adversary Proceedings for the Active States in Cash for payment of reasonable fees and out-of-pocket costs incurred in connection with the Producer State-Specific Adversary Proceedings for the Active States or in the Other Proceedings, as applicable. In addition, the Producer Representative shall pay any professional fees and expenses of the Producers' Committee to the extent that the fees and expenses exceed the cap amount contained in Section 1.127 of the Plan. For the avoidance of doubt, all amounts payable pursuant to this Section 3.1(d) shall be payable solely from the \$172.5 million less the aggregate amount of First Purchaser Producer Twenty-Day Claims included in the cure amounts listed in the Cure Schedule distributed to the Producer Representative pursuant to Section 3.1(a) hereof.
- (e) <u>Producer Deficiency Claims</u>. No portion of any Secured First Purchaser Producer Claim shall be an Unsecured Claim for any purpose under this Plan. Each holder of an Unsecured First Purchaser Producer Claim shall vote as a holder of a General Unsecured Claim

under this Plan and shall be treated for all purposes as the holder of a General Unsecured Claim under this Plan.

- (f) <u>Producer State-Specific Adversary Proceedings</u>. The Debtors, the Prepetition Administrative Agent, the Lender Steering Committee, the Producers' Committee and the Producer Plaintiffs have agreed to take, and in some cases have already taken, the following actions with respect to the Producer State-Specific Adversary Proceedings:
 - (i) Third Circuit Appeals. The Debtors, the Prepetition Administrative Agent and the Producer Plaintiffs jointly submitted a request on September 16, 2009 to the Third Circuit Court of Appeals to adjourn the oral argument on the Third Circuit Appeals to a date after the anticipated Effective Date, which request was granted on September 17, 2009. To the extent the Third Circuit Appeals are not remanded pursuant to Section 3.1(f)(iii) hereof, the Debtors, the Prepetition Administrative Agent and the Producer Plaintiffs shall, no later than the Effective Date, voluntarily dismiss with prejudice the Third Circuit Appeals, against, but only against, the Debtors, the Prepetition Administrative Agent and the Prepetition Lenders.
 - Bankruptcy Court Proceedings. On the Effective Date, to the extent that (ii) the Third Circuit Appeals are not remanded to the Bankruptcy Court pursuant to Section 3.1(f)(iii) hereof, the Producer Plaintiffs shall file all papers necessary to voluntarily dismiss with prejudice the Producer State-Specific Adversary Proceedings with respect to the Debtors, the Prepetition Administrative Agent and the Prepetition Lenders. With the exception of (i) any Bankruptcy Court proceedings with respect to the Bankruptcy Court decision on global legal objections to Twenty-Day Claims argued on September 9, 2009 and (ii) any Bankruptcy Court proceedings with respect to Other Twenty-Day Claims, all Producer-related litigation commenced in the Bankruptcy Court or to which the Debtors or the Prepetition Administrative Agent is a party shall be abated as to all parties until November 11, 2009. Each applicable Producer Plaintiff shall file all papers necessary to voluntarily dismiss with prejudice the Debtors, the Prepetition Administrative Agent and the Prepetition Lenders from all such Producer-related litigation commenced in the Bankruptcy Court on the Effective Date concurrently with the payment by the Debtors to the Producer Representative described in Section 3.1(a) hereof. Each of the Producer Plaintiffs, the Debtors, the Prepetition Administrative Agent and the applicable Prepetition Lenders shall take all such actions, including executing stipulations, as are reasonably required to effect the foregoing dismissals with prejudice.
 - (iii) <u>Producer Decisions</u>. If the Plan is confirmed and the Producer Plaintiffs submit an application to the Third Circuit Court of Appeals requesting that it remand to the Bankruptcy Court and an application to the Bankruptcy Court requesting that the Bankruptcy Court vacate the Producer Decisions, then neither the Debtors nor the Prepetition Administrative Agent shall oppose any such applications.

- (iv) Other Proceedings. Each of the Debtors, the Producers named as plaintiffs in the Producer State-Specific Adversary Proceedings, the Prepetition Administrative Agent, the Prepetition Lenders and their respective affiliates, as applicable, shall abate all Other Proceedings to which it is a party until November 11, 2009. On the Effective Date, the Producers named as plaintiffs in the Producer State-Specific Adversary Proceedings shall file all papers necessary to voluntarily dismiss with prejudice the Other Proceedings with respect to the Debtors, the Prepetition Administrative Agent, and the applicable Prepetition Lenders and/or their affiliates, as applicable. Each of the Producers named as plaintiffs in the Producer State-Specific Adversary Proceedings, the Debtors, the Prepetition Administrative Agent and the applicable Prepetition Lenders and/or their affiliates shall take all such actions, including executing stipulations, as are reasonably required to effect the foregoing dismissals with prejudice.
- (g) <u>Twenty-Day Claims</u>. The Debtors, the Prepetition Administrative Agent, the Lender Steering Committee, the Producers' Committee and the Producer Plaintiffs have agreed as follows with respect to the Twenty-Day Claims:
 - (i) <u>Global Legal Objections</u>. The Producers' Settlement resolves the global legal objections with respect to the First Purchaser Producer Twenty-Day Claims. The Producers' Settlement shall have no effect on any legal objections with respect to Other Twenty-Day Claims.
 - Factual-Based Objections. The Debtors, the Prepetition Administrative (ii) Agent and the Lender Steering Committee, as applicable, shall (A) abate their fact-based objections to First Purchaser Producer Twenty-Day Claims until November 11, 2009 and (B) voluntarily dismiss with prejudice any fact-based objections to the First Purchaser Producer Twenty-Day Claims on the Effective Date. After the Effective Date, the Producer Representative shall be permitted to pursue the disputes against any First Purchaser Producer regarding certain Twenty-Day Claims that were set forth in the Notice in accordance with the Producer claims resolution process described in Section 8.3 hereof; provided, however that the expenses incurred by the Producer Representative in connection with pursuing such disputes shall be paid out of the \$47 million referred to in Section 3.1(c) of the Plan. The Producers' Settlement shall have no effect on the ability of the Debtors, the Prepetition Administrative Agent and the Prepetition Lenders to pursue any and all fact-based objections with respect to the Other Twenty-Day Claims.
- (h) <u>Payments by Operators</u>. Operators receiving payments under the Plan shall be responsible for disbursing payment to Owners entitled to payment thereto, if any. The Debtors shall provide data in their possession related to certain Owners to the Producer Representative, on which the Operators can rely in making distributions pursuant to the Plan.
- (i) <u>No Impact on Litigation Trust</u>. Notwithstanding anything in this Section 3.1 or the Plan to the contrary, and without the express written consent of the Litigation Trust Board, no Entity or Creditor (including but not limited to the Prepetition Administrative Agent,

the Postpetition Administrative Agent, the Producer Representative, the holder of a Secured Lender Claim, any Operator, or any Owner), shall be permitted to assert, bring, institute, commence, or participate in any Litigation Trust Claim (other than distributions, if any, in its capacity as a holder of Litigation Trust Interests).

3.2 <u>Other Twenty-Day Claims Settlement</u>. The holders of the Other Twenty-Day Claims are not party to the Producers' Settlement. The Debtors shall seek agreement of certain holders of Claims and Administrative Expense Claims to the Other Twenty-Day Claims Settlement as provided below:

(a) <u>Settling Parties</u>.

- (i) Each holder of an Other Twenty-Day Claim that elects to participate in the Other Twenty-Day Claims Settlement shall be deemed to have an Allowed Claim equal to 66% of the aggregate amount of such holder's Other Twenty-Day Claims as scheduled on Schedule 3 hereto and shall receive on account of its Claims against the Debtors and their estates, Cash on the Effective Date equal to such proposed settlement amount.
- (ii) As part of the Other Twenty-Day Claims Settlement, the Debtors and the Prepetition Administrative Agent shall not pursue any objections to the Other Twenty-Day Claims of such Settling Party, and any decisions that the Bankruptcy Court reaches with regard to the global legal objections shall not affect a Settling Parties' entitlement to payment on the Effective Date as described above.
- (iii) In exchange for this payment, the Other Twenty-Day claimant shall release certain parties as described in Section 3.2(d) hereof. The claimant shall not receive a mutual release from the parties who the claimant is releasing under the Plan.
- (b) Non-Settling Parties. On the Effective Date, the Debtors shall reserve in Cash an amount equal to the amount of the Other Twenty-Day Claims of each Non-Settling Party (at that date) set forth on Schedule 3 hereto (or such lesser amount as shall be approved by the Bankruptcy Court) and hold such reserve until the status of such Claim has been Allowed by agreement of the Prepetition Administrative Agent and the holder of such Other Twenty-Day Claim or Final Order. Each of the Debtors (solely prior to the Effective Date), the Creditors' Committee (solely prior to the Effective Date and solely as to Avoidance Actions) and the Prepetition Administrative Agent shall be free to pursue any objections, including both fact-based objections and global legal objections, with respect to such Other Twenty-Day Claims of Non-Settling Parties.
- (c) <u>Election Notices</u>. Each holder of an Other Twenty-Day Claim shall receive an Election Notice concurrently with the Debtors' solicitation of votes to accept the Plan. The Election Notice shall contain information regarding the terms of the Other Twenty-Day Claims Settlement, including the deadline for returning such notice.

TO CONSENT AND AGREE TO THE OTHER TWENTY-DAY CLAIMS SETTLEMENT, A HOLDER MUST COMPLETE ITS ELECTION NOTICE AND RETURN IT TO THE BALLOTING AGENT BY THE SPECIFIED DEADLINE. IF A HOLDER ELECTS TO ACCEPT THE OTHER TWENTY-DAY CLAIMS SETTLEMENT, SUCH HOLDER SHALL BE DEEMED TO HAVE CONSENTED AND AGREED TO RECEIVE TREATMENT FOR SUCH CLAIM THAT IS DIFFERENT FROM THAT SET FORTH IN SECTION 503(B)(9) OF THE BANKRUPTCY CODE AND SHALL BE BOUND BY THE TERMS OF SUCH SETTLEMENT, INCLUDING THOSE SET FORTH ABOVE.

IF AN ELECTION NOTICE IS NOT RECEIVED BY THE BALLOTING AGENT BY THE VOTING DEADLINE, OR IF THE HOLDER ELECTS NOT TO PARTICIPATE ON ITS ELECTION NOTICE, SUCH HOLDER WILL BE A NON-SETTLING PARTY AND SUBJECT TO THE TREATMENT DESCRIBED IN SECTION 3.2(B) HEREIN.

Release of Litigation. In addition to the provisions of Section 20.11 of (d) this Plan, upon the Effective Date, each Creditor that elects to participate in the Other Twenty-Day Claims Settlement forever discharges the Debtors, the Prepetition Lenders (excluding J. Aron & Company and its affiliates), and the Prepetition Administrative Agent, jointly and as to each of them, from and against any and all liability that they now have, had, or may have arising out of or relating to such Creditor's Other Twenty-Day Claim. The Creditor agrees to voluntarily dismiss with prejudice any motions, objections, or other pleadings regarding such Other Twenty-Day Claim from the Bankruptcy Court's consideration. The Creditor also agrees not to aid, assist, support, or otherwise participate with any other party in prosecuting Other Twenty-Day Claims against the Debtors, the Reorganized Debtors, Prepetition Administrative Agent and/or the Prepetition Lenders (excluding J. Aron & Company and its affiliates) or to take any positions contrary to the Debtors, the Reorganized Debtors, Prepetition Administrative Agent and/or Prepetition Lenders (excluding J. Aron & Company and its affiliates). This release of Other Twenty-Day Claims is specifically intended to include and does include Other Twenty-Day Claims that the Creditor might not now know or expect to exist in their favor at the Effective Date, even if knowledge of such claims might have otherwise materially affected the granting of this release. The Creditor understands and agrees that this release of Other Twenty-Day Claims shall be treated as a full and complete defense to, and will forever be a complete bar to the commencement or prosecution of, any and all Claims released herein. The Creditor intends that this release of Other Twenty-Day Claims and the release contained in Section 20.11 hereof be complete and not subject to a claim of mistake of fact and that such release and delivery according to the terms of the Other Twenty-Day Claims Settlement present a FULL **AND COMPLETE SETTLEMENT** of the Claims released herein. Regardless of the adequacy or inadequacy of the consideration paid, the release included herein and in Section 20.11 hereof is intended to settle or avoid litigation and/or settle the claims released herein, and to be final and complete.

ARTICLE IV

CLASSIFICATION OF CLAIMS AND EQUITY INTERESTS

Claims (other than Administrative Expense Claims, Postpetition Financing Claims, Professional Compensation and Reimbursement Claims, and Priority Tax Claims) and Equity Interests are classified for all purposes, including voting, confirmation, and distribution pursuant to the Plan, as follows:

Class	Debtor	Designation	Impairment	Entitled to Vote
Class 1	SemCrude, L.P.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 2	Chemical Petroleum Exchange, Incorporated	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 3	Eaglwing, L.P.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 4	Grayson Pipeline, L.L.C.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 5	Greyhawk Gas Storage Company, L.L.C.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 6	K.C. Asphalt L.L.C.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 7	SemCanada II, L.P.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 8	SemCanada L.P.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 9	SemCrude Pipeline, L.L.C.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 10	SemFuel Transport LLC	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 11	SemFuel, L.P.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 12	SemGas Gathering LLC	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 13	SemGas Storage, L.L.C.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 14	SemGas, L.P.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 15	SemGroup Asia, L.L.C.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)

Class	Debtor	Designation	Impairment	Entitled to Vote
Class 16	SemGroup Finance Corp.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 17	SemGroup, L.P.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 18	SemKan, L.L.C.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 19	SemManagement, L.L.C.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 20	SemMaterials Vietnam, L.L.C.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 21	SemMaterials, L.P.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 22	SemOperating G.P., L.L.C.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 23	SemStream, L.P.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 24	SemTrucking, L.P.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 25	Steuben Development Company, L.L.C.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 26	SemCap, L.L.C.	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 27	SemCrude, L.P.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 28	Chemical Petroleum Exchange, Incorporated	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 29	Eaglwing, L.P.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 30	Grayson Pipeline, L.L.C.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 31	Greyhawk Gas Storage Company, L.L.C.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 32	K.C. Asphalt L.L.C.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 33	SemCanada II, L.P.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 34	SemCanada L.P.	Secured Tax Claims	Unimpaired	No (deemed to

Class	Debtor	Designation	Impairment	Entitled to Vote
				accept)
Class 35	SemCrude Pipeline, L.L.C.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 36	SemFuel Transport LLC	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 37	SemFuel, L.P.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 38	SemGas Gathering LLC	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 39	SemGas Storage, L.L.C.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 40	SemGas, L.P.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 41	SemGroup Asia, L.L.C.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 42	SemGroup Finance Corp.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 43	SemGroup, L.P.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 44	SemKan, L.L.C.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 45	SemManagement, L.L.C.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 46	SemMaterials Vietnam, L.L.C.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 47	SemMaterials, L.P.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 48	SemOperating G.P., L.L.C.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 49	SemStream, L.P.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 50	SemTrucking, L.P.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 51	Steuben Development Company, L.L.C.	Secured Tax Claims	Unimpaired	No (deemed to accept)
Class 52	SemCap, L.L.C.	Secured Tax Claims	Unimpaired	No (deemed to accept)

Class	Debtor	Designation	Impairment	Entitled to Vote	
Class 53	Eaglwing, L.P.	Secured First Purchaser Producer Claims	Impaired	Yes	
Class 54	SemCrude, L.P.	Secured First Purchaser Producer Claims	Impaired	Yes	
Class 55	SemGas, L.P.	Secured First Purchaser Producer Claims	Impaired	Yes	
Class 56		*INTENTIONALLY OMITTED*			
Class 57		*INTENTIONALLY OMITTED*			
Class 58		*INTENTIONALLY OMITTED*			
Class 59		*INTENTIONALLY OMITTED*			
Class 60		*INTENTIONALLY OMITTED*			
Class 61		*INTENTIONALLY OMITTED*			
Class 62		*INTENTIONALLY OMITTED*			
Class 63		*INTENTIONALLY OMITTED*			
Class 64		*INTENTIONALLY OMITTED*			
Class 65		*INTENTIONALLY OMITTED*			
Class 66		*INTENTIONALLY OMITTED*			
Class 67		*INTENTIONALLY OMITTED*			
Class 68		*INTENTIONALLY OMITTED*			
Class 69		*INTENTIONALLY OMITTED*			
Class 70	SemCrude, L.P.	Secured Working Capital Lender Claims	Impaired	Yes	
Class 71	Chemical Petroleum Exchange, Incorporated	Secured Working Capital Lender Claims	Impaired	Yes	
Class 72	Eaglwing, L.P.	Secured Working Capital Lender Claims	Impaired	Yes	
Class 73	Grayson Pipeline, L.L.C.	Secured Working Capital Lender Claims	Impaired	Yes	
Class 74	Greyhawk Gas Storage Company, L.L.C.	Secured Working Capital Lender Claims	Impaired	Yes	
Class 75	K.C. Asphalt L.L.C.	Secured Working Capital Lender Claims	Impaired	Yes	

Class	Debtor	Designation	Impairment	Entitled to Vote
Class 76	SemCanada II, L.P.	Secured Working Capital Lender Claims	Impaired	Yes
Class 77	SemCanada L.P.	Secured Working Capital Lender Claims	Impaired	Yes
Class 78	SemCrude Pipeline, L.L.C.	Secured Working Capital Lender Claims	Impaired	Yes
Class 79	SemFuel Transport LLC	Secured Working Capital Lender Claims	Impaired	Yes
Class 80	SemFuel, L.P.	Secured Working Capital Lender Claims	Impaired	Yes
Class 81	SemGas Gathering LLC	Secured Working Capital Lender Claims	Impaired	Yes
Class 82	SemGas Storage, L.L.C.	Secured Working Capital Lender Claims	Impaired	Yes
Class 83	SemGas, L.P.	Secured Working Capital Lender Claims	Impaired	Yes
Class 84	SemGroup Asia, L.L.C.	Secured Working Capital Lender Claims	Impaired	Yes
Class 85	SemGroup Finance Corp.	Secured Working Capital Lender Claims	Impaired	Yes
Class 86	SemGroup, L.P.	Secured Working Capital Lender Claims	Impaired	Yes
Class 87	SemKan, L.L.C.	Secured Working Capital Lender Claims	Impaired	Yes
Class 88	SemManagement, L.L.C.	Secured Working Capital Lender Claims	Impaired	Yes
Class 89	SemMaterials Vietnam, L.L.C.	Secured Working Capital Lender Claims	Impaired	Yes
Class 90	SemMaterials, L.P.	Secured Working Capital Lender Claims	Impaired	Yes
Class 91	SemOperating G.P., L.L.C.	Secured Working Capital Lender Claims	Impaired	Yes
Class 92	SemStream, L.P.	Secured Working Capital Lender Claims	Impaired	Yes
Class 93	SemTrucking, L.P.	Secured Working Capital Lender Claims	Impaired	Yes
Class 94	Steuben Development	Secured Working Capital Lender	Impaired	Yes

Class	Debtor	Designation	Impairment	Entitled to Vote
	Company, L.L.C.	Claims		
Class 95	SemCap, L.L.C.	Secured Working Capital Lender Claims	Impaired	Yes
Class 96	SemCrude, L.P.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 97	Chemical Petroleum Exchange, Incorporated	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 98	Eaglwing, L.P.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 99	Grayson Pipeline, L.L.C.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 100	Greyhawk Gas Storage Company, L.L.C.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 101	K.C. Asphalt L.L.C.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 102	SemCanada II, L.P.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 103	SemCanada L.P.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 104	SemCrude Pipeline, L.L.C.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 105	SemFuel Transport LLC	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 106	SemFuel, L.P.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 107	SemGas Gathering LLC	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 108	SemGas Storage, L.L.C.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 109	SemGas, L.P.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 110	SemGroup Asia, L.L.C.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 111	SemGroup Finance Corp.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 112	SemGroup, L.P.	Secured Revolver/Term Lender Claims	Impaired	Yes

Class	Debtor	Designation	Impairment	Entitled to Vote
Class 113	SemKan, L.L.C.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 114	SemManagement, L.L.C.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 115	SemMaterials Vietnam, L.L.C.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 116	SemMaterials, L.P.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 117	SemOperating G.P., L.L.C.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 118	SemStream, L.P.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 119	SemTrucking, L.P.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 120	Steuben Development Company, L.L.C.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 121	SemCap, L.L.C.	Secured Revolver/Term Lender Claims	Impaired	Yes
Class 122	SemCrude Pipeline, L.L.C.	White Cliffs Credit Agreement Claim	Impaired	Yes
Class 123	SemCrude, L.P.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 124	Chemical Petroleum Exchange, Incorporated	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 125	Eaglwing, L.P.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 126	Grayson Pipeline, L.L.C.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 127	Greyhawk Gas Storage Company, L.L.C.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 128	K.C. Asphalt L.L.C.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 129	SemCanada II, L.P.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 130	SemCanada L.P.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 131	SemCrude Pipeline, L.L.C.	Other Secured Claims	Unimpaired	No (deemed to

Class	Debtor	Designation	Impairment	Entitled to Vote
				accept)
Class 132	SemFuel Transport LLC	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 133	SemFuel, L.P.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 134	SemGas Gathering LLC	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 135	SemGas Storage, L.L.C.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 136	SemGas, L.P.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 137	SemGroup Asia, L.L.C.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 138	SemGroup Finance Corp.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 139	SemGroup, L.P.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 140	SemKan, L.L.C.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 141	SemManagement, L.L.C.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 142	SemMaterials Vietnam, L.L.C.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 143	SemMaterials, L.P.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 144	SemOperating G.P., L.L.C.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 145	SemStream, L.P.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 146	SemTrucking, L.P.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 147	Steuben Development Company, L.L.C.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 148	SemCap, L.L.C.	Other Secured Claims	Unimpaired	No (deemed to accept)
Class 149	SemCrude, L.P.	Senior Notes Claims	Impaired	Yes

Class	Debtor	Designation	Impairment	Entitled to Vote
Class 150	Chemical Petroleum Exchange, Incorporated	Senior Notes Claims	Impaired	Yes
Class 151	Eaglwing, L.P.	Senior Notes Claims	Impaired	Yes
Class 152	Grayson Pipeline, L.L.C.	Senior Notes Claims	Impaired	Yes
Class 153	Greyhawk Gas Storage Company, L.L.C.	Senior Notes Claims	Impaired	Yes
Class 154	K.C. Asphalt L.L.C.	Senior Notes Claims	Impaired	Yes
Class 155	SemCanada II, L.P.	Senior Notes Claims	Impaired	Yes
Class 156	SemCanada L.P.	Senior Notes Claims	Impaired	Yes
Class 157	SemCrude Pipeline, L.L.C.	Senior Notes Claims	Impaired	Yes
Class 158	SemFuel Transport LLC	Senior Notes Claims	Impaired	Yes
Class 159	SemFuel, L.P.	Senior Notes Claims	Impaired	Yes
Class 160	SemGas Gathering LLC	Senior Notes Claims	Impaired	Yes
Class 161	SemGas Storage, L.L.C.	Senior Notes Claims	Impaired	Yes
Class 162	SemGas, L.P.	Senior Notes Claims	Impaired	Yes
Class 163	SemGroup Asia, L.L.C.	Senior Notes Claims	Impaired	Yes
Class 164	SemGroup Finance Corp.	Senior Notes Claims	Impaired	Yes
Class 165	SemGroup, L.P.	Senior Notes Claims	Impaired	Yes
Class 166	SemKan, L.L.C.	Senior Notes Claims	Impaired	Yes
Class 167	SemManagement, L.L.C.	Senior Notes Claims	Impaired	Yes
Class 168	SemMaterials Vietnam, L.L.C.	Senior Notes Claims	Impaired	Yes
Class 169	SemMaterials, L.P.	Senior Notes Claims	Impaired	Yes
Class 170	SemOperating G.P., L.L.C.	Senior Notes Claims	Impaired	Yes
Class 171	SemStream, L.P.	Senior Notes Claims	Impaired	Yes
Class 172	SemTrucking, L.P.	Senior Notes Claims	Impaired	Yes
Class 173	Steuben Development Company, L.L.C.	Senior Notes Claims	Impaired	Yes
Class 174	SemCap, L.L.C.	Senior Notes Claims	Impaired	Yes
Class 175	SemCrude, L.P.	Lender Deficiency Claims	Impaired	Yes

Class	Debtor	Designation	Impairment	Entitled to Vote
Class 176	Chemical Petroleum Exchange, Incorporated	Lender Deficiency Claims	Impaired	Yes
Class 177	Eaglwing, L.P.	Lender Deficiency Claims	Impaired	Yes
Class 178	Grayson Pipeline, L.L.C.	Lender Deficiency Claims	Impaired	Yes
Class 179	Greyhawk Gas Storage Company, L.L.C.	Lender Deficiency Claims	Impaired	Yes
Class 180	K.C. Asphalt L.L.C.	Lender Deficiency Claims	Impaired	Yes
Class 181	SemCanada II, L.P.	Lender Deficiency Claims	Impaired	Yes
Class 182	SemCanada L.P.	Lender Deficiency Claims	Impaired	Yes
Class 183	SemCrude Pipeline, L.L.C.	Lender Deficiency Claims	Impaired	Yes
Class 184	SemFuel Transport LLC	Lender Deficiency Claims	Impaired	Yes
Class 185	SemFuel, L.P.	Lender Deficiency Claims	Impaired	Yes
Class 186	SemGas Gathering LLC	Lender Deficiency Claims	Impaired	Yes
Class 187	SemGas Storage, L.L.C.	Lender Deficiency Claims	Impaired	Yes
Class 188	SemGas, L.P.	Lender Deficiency Claims	Impaired	Yes
Class 189	SemGroup Asia, L.L.C.	Lender Deficiency Claims	Impaired	Yes
Class 190	SemGroup Finance Corp.	Lender Deficiency Claims	Impaired	Yes
Class 191	SemGroup, L.P.	Lender Deficiency Claims	Impaired	Yes
Class 192	SemKan, L.L.C.	Lender Deficiency Claims	Impaired	Yes
Class 193	SemManagement, L.L.C.	Lender Deficiency Claims	Impaired	Yes
Class 194	SemMaterials Vietnam, L.L.C.	Lender Deficiency Claims	Impaired	Yes
Class 195	SemMaterials, L.P.	Lender Deficiency Claims	Impaired	Yes
Class 196	SemOperating G.P., L.L.C.	Lender Deficiency Claims	Impaired	Yes
Class 197	SemStream, L.P.	Lender Deficiency Claims	Impaired	Yes
Class 198	SemTrucking, L.P.	Lender Deficiency Claims	Impaired	Yes
Class 199	Steuben Development Company, L.L.C.	Lender Deficiency Claims	Impaired	Yes
Class 200	SemCap, L.L.C.	Lender Deficiency Claims	Impaired	Yes
Class 201	SemCrude, L.P.	General Unsecured Claims	Impaired	Yes

Class	Debtor	Designation	Impairment	Entitled to Vote
Class 202	Chemical Petroleum Exchange, Incorporated	General Unsecured Claims	Impaired	Yes
Class 203	Eaglwing, L.P.	General Unsecured Claims	Impaired	Yes
Class 204	Grayson Pipeline, L.L.C.	General Unsecured Claims	Impaired	Yes
Class 205	Greyhawk Gas Storage Company, L.L.C.	General Unsecured Claims	Impaired	Yes
Class 206	K.C. Asphalt L.L.C.	General Unsecured Claims	Impaired	Yes
Class 207	SemCanada II, L.P.	General Unsecured Claims	Impaired	Yes
Class 208	SemCanada L.P.	General Unsecured Claims	Impaired	Yes
Class 209	SemCrude Pipeline, L.L.C.	General Unsecured Claims	Impaired	Yes
Class 210	SemFuel Transport LLC	General Unsecured Claims	Impaired	Yes
Class 211	SemFuel, L.P.	General Unsecured Claims	Impaired	Yes
Class 212	SemGas Gathering LLC	General Unsecured Claims	Impaired	Yes
Class 213	SemGas Storage, L.L.C.	General Unsecured Claims	Impaired	Yes
Class 214	SemGas, L.P.	General Unsecured Claims	Impaired	Yes
Class 215	SemGroup Asia, L.L.C.	General Unsecured Claims	Impaired	Yes
Class 216	SemGroup Finance Corp.	General Unsecured Claims	Impaired	Yes
Class 217	SemGroup, L.P.	General Unsecured Claims	Impaired	Yes
Class 218	SemKan, L.L.C.	General Unsecured Claims	Impaired	Yes
Class 219	SemManagement, L.L.C.	General Unsecured Claims	Impaired	Yes
Class 220	SemMaterials Vietnam, L.L.C.	General Unsecured Claims	Impaired	Yes
Class 221	SemMaterials, L.P.	General Unsecured Claims	Impaired	Yes
Class 222	SemOperating G.P., L.L.C.	General Unsecured Claims	Impaired	Yes
Class 223	SemStream, L.P.	General Unsecured Claims	Impaired	Yes
Class 224	SemTrucking, L.P.	General Unsecured Claims	Impaired	Yes
Class 225	Steuben Development Company, L.L.C.	General Unsecured Claims	Impaired	Yes
Class 226	SemCap, L.L.C.	General Unsecured Claims	Impaired	Yes
Class 227	SemCrude, L.P.	Intercompany Claims	Impaired	No (deemed to

Class	Debtor	Designation	Impairment	Entitled to Vote
				accept)
Class 228	Chemical Petroleum Exchange, Incorporated	Intercompany Claims	Impaired	No (deemed to accept)
Class 229	Eaglwing, L.P.	Intercompany Claims	Impaired	No (deemed to accept)
Class 230	Grayson Pipeline, L.L.C.	Intercompany Claims	Impaired	No (deemed to accept)
Class 231	Greyhawk Gas Storage Company, L.L.C.	Intercompany Claims	Impaired	No (deemed to accept)
Class 232	K.C. Asphalt L.L.C.	Intercompany Claims	Impaired	No (deemed to accept)
Class 233	SemCanada II, L.P.	Intercompany Claims	Impaired	No (deemed to accept)
Class 234	SemCanada L.P.	Intercompany Claims	Impaired	No (deemed to accept)
Class 235	SemCrude Pipeline, L.L.C.	Intercompany Claims	Impaired	No (deemed to accept)
Class 236	SemFuel Transport LLC	Intercompany Claims	Impaired	No (deemed to accept)
Class 237	SemFuel, L.P.	Intercompany Claims	Impaired	No (deemed to accept)
Class 238	SemGas Gathering LLC	Intercompany Claims	Impaired	No (deemed to accept)
Class 239	SemGas Storage, L.L.C.	Intercompany Claims	Impaired	No (deemed to accept)
Class 240	SemGas, L.P.	Intercompany Claims	Impaired	No (deemed to accept)
Class 241	SemGroup Asia, L.L.C.	Intercompany Claims	Impaired	No (deemed to accept)
Class 242	SemGroup Finance Corp.	Intercompany Claims	Impaired	No (deemed to accept)
Class 243	SemGroup, L.P.	Intercompany Claims	Impaired	No (deemed to accept)
Class 244	SemKan, L.L.C.	Intercompany Claims	Impaired	No (deemed to accept)
Class 245	SemManagement, L.L.C.	Intercompany Claims	Impaired	No (deemed to accept)

Class	Debtor	Designation	Impairment	Entitled to Vote
Class 246	SemMaterials Vietnam, L.L.C.	Intercompany Claims	Impaired	No (deemed to accept)
Class 247	SemMaterials, L.P.	Intercompany Claims	Impaired	No (deemed to accept)
Class 248	SemOperating G.P., L.L.C.	Intercompany Claims	Impaired	No (deemed to accept)
Class 249	SemStream, L.P.	Intercompany Claims	Impaired	No (deemed to accept)
Class 250	SemTrucking, L.P.	Intercompany Claims	Impaired	No (deemed to accept)
Class 251	Steuben Development Company, L.L.C.	Intercompany Claims	Impaired	No (deemed to accept)
Class 252	SemCap, L.L.C.	Intercompany Claims	Impaired	No (deemed to accept)
Class 253	SemCrude, L.P.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 254	Chemical Petroleum Exchange, Incorporated	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 255	Eaglwing, L.P.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 256	Grayson Pipeline, L.L.C.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 257	Greyhawk Gas Storage Company, L.L.C.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 258	K.C. Asphalt L.L.C.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 259	SemCanada II, L.P.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 260	SemCanada L.P.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 261	SemCrude Pipeline, L.L.C.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 262	SemFuel Transport LLC	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 263	SemFuel, L.P.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 264	SemGas Gathering LLC	Intercompany Equity Interests	Unimpaired	No (deemed to

Class	Debtor	Designation	Impairment	Entitled to Vote
				accept)
Class 265	SemGas Storage, L.L.C.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 266	SemGas, L.P.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 267	SemGroup Asia, L.L.C.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 268	SemGroup Finance Corp.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 269	SemGroup, L.P.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 270	SemKan, L.L.C.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 271	SemManagement, L.L.C.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 272	SemMaterials Vietnam, L.L.C.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 273	SemMaterials, L.P.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 274	SemOperating G.P., L.L.C.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 275	SemStream, L.P.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 276	SemTrucking, L.P.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 277	Steuben Development Company, L.L.C.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 278	SemCap, L.L.C.	Intercompany Equity Interests	Unimpaired	No (deemed to accept)
Class 279	SemGroup, L.P.	SemGroup Equity Interests	Impaired	No (deemed to reject)

ARTICLE V

TREATMENT OF CLAIMS AND EQUITY INTERESTS

5.1 Classes 1 through 26 – Priority Non-Tax Claims

- (a) <u>Impairment and Voting</u>. Classes 1 through 26 are unimpaired by the Plan. Each holder of an Allowed Priority Non-Tax Claim is conclusively presumed to have accepted the Plan and is not entitled to vote to accept or reject the Plan.
- (b) <u>Distributions</u>. Unless otherwise mutually agreed upon by the holder of an Allowed Priority Non-Tax Claim and the Reorganized Debtors, each holder of an Allowed Priority Non-Tax Claim shall receive, on account of their Claims against the Debtors and their estates, Cash in an amount equal to such Allowed Priority Non-Tax Claim on the later of the Effective Date and the date such Allowed Priority Non-Tax Claim becomes an Allowed Priority Non-Tax Claim, or as soon thereafter as is practicable.

5.2 <u>Classes 27 through 52 – Secured Tax Claims</u>

- (a) <u>Impairment and Voting</u>. Classes 27 through 52 are unimpaired by the Plan. Each holder of an Allowed Secured Tax Claim is conclusively presumed to have accepted the Plan and is not entitled to vote to accept or reject the Plan.
- (b) <u>Distributions</u>. Except to the extent that a holder of an Allowed Secured Tax Claim has been paid by the Debtors prior to the Effective Date or agrees to a different treatment, each holder of an of an Allowed Secured Tax Claim shall receive, at the sole option of the Reorganized Debtors, (i) Cash in an amount equal to such Allowed Secured Tax Claim on the Effective Date, including any interest on such Allowed Secured Tax Claim required to be paid pursuant to section 506(b) of the Bankruptcy Code, or (ii) equal semi-annual Cash payments in an aggregate amount equal to such Allowed Secured Tax Claim, together with interest at a rate determined under applicable non-bankruptcy law in accordance with section 511 of the Bankruptcy Code, over a period ending not later than five (5) years after the Petition Date, or (iii) upon such other terms determined by the Bankruptcy Court to provide the holder of such Allowed Secured Tax Claim deferred Cash payments having a value, as of the Effective Date, equal to such Allowed Secured Tax Claim.

5.3 Classes 53 through 55 – Secured First Purchaser Producer Claims

- (a) <u>Impairment and Voting</u>. Classes 53 through 55 are impaired by the Plan. Each holder of a Secured First Purchaser Producer Claim is entitled to vote to accept or reject the Plan.
- (b) <u>Distributions</u>. On the Effective Date, or as soon thereafter as is practicable, each holder of an Allowed Secured First Purchaser Producer Claim shall receive, on account of their Claims against the Debtors and their estates, its Pro Rata Share of Cash in accordance with Section 3.1(c) of the Plan.

5.4 Classes 70 through 95 – Secured Working Capital Lender Claims

(a) <u>Impairment and Voting</u>. Classes 70 through 95 are impaired by the Plan. Each holder of a Secured Working Capital Lender Claim is entitled to vote to accept or reject the Plan. The vote by each holder of a Secured Working Capital Lender Claim in favor of or against the Plan is deemed to be a vote in favor of or against the Canadian Plans, respectively.

Distributions. The Secured Working Capital Lender Claims are hereby Allowed Claims, not subject to offset, defense, counterclaim, reduction, or credit of any kind whatsoever. On the Effective Date, or as soon thereafter as is practicable, each holder of an Allowed Secured Working Capital Lender Claim shall receive, on account of their Claims against the Debtors and their estates, its Pro Rata Share of (x) (i) the Working Capital Lender Effective Date Cash, (ii) \$174 million in principal amount of the Second Lien Term Loan Interests, and (iii) 23,306,753 shares of the New Common Stock (subject to dilution of ownership percentage from the Warrants and the Management Stock), which distribution of New Common Stock is inclusive of the New Common Stock that holders of Allowed Intercompany Claims are deemed to be entitled to but is being redistributed to holders of Allowed Secured Working Capital Lender Claims pursuant to Section 5.11 of the Plan; and (y) subsequent distributions in accordance with Section 13.1 of this Plan to the extent, if any, the Reorganized Debtors (or the Prepetition Administrative Agent, in the case of certain Canadian Distributions) receive, after the Effective Date, (i) a Canadian Distribution (other than an Auriga Revolver/Term Lender Distribution), (ii) net Cash proceeds from the realization of receivables or inventory of SemFuel, L.P. or SemMaterials, L.P. (other than net Cash proceeds referred to in Section 5.5(b)(y) of this Plan), (iii) Cash distributions from SemGroup Holdings, L.P., or net Cash proceeds from the sale of Equity Interests in SemGroup Holdings, L.P. (to the extent allocable to the Working Capital Obligations (as defined in the Prepetition Credit Agreement)), (iv) any Cash proceeds of Undisputed Production Receivables, (v) the Litigation Trust Funds, or (vi) any Cash released from reserves for Administrative Expense Claims, Professional Compensation and Reimbursement Claims, Priority Non-Tax Claims, or Priority Tax Claims.

5.5 Classes 96 through 121 – Secured Revolver/Term Lender Claims

- (a) <u>Impairment and Voting</u>. Classes 96 through 121 are impaired by the Plan. Each holder of a Secured Revolver/Term Lender Claim is entitled to vote to accept or reject the Plan. The vote by each holder of a Secured Revolver/Term Lender Claim in favor of or against the Plan is deemed to be a vote in favor of or against the Canadian Plans, respectively.
- Distributions. The Secured Revolver/Term Lender Claims are hereby (b) Allowed Claims, not subject to offset, defense, counterclaim, reduction, or credit of any kind whatsoever. On the Effective Date, or as soon thereafter as is practicable, each holder of an Allowed Secured Revolver/Term Lender Claim shall receive, on account of their Claims against the Debtors and their estates, its Pro Rata Share of (x) (i) the Revolver/Term Lender Effective Date Cash, (ii) \$126 million principal amount of the Second Lien Term Loan Interests, (iii) 16,023,247 shares of the New Common Stock (subject to dilution of ownership percentage from the Warrants and the Management Stock) and (iv) the US Term Lender Group Fees; provided, however, that the US Term Lender Group Fees shall be paid to the professionals of the US Term Lender Group in accordance with Section 2.6 of this Plan; and (y) subsequent distributions in accordance with Section 13.1 of this Plan to the extent, if any, the Reorganized Debtors (or the Prepetition Administrative Agent, in the case of the Auriga Revolver/Term Lender Distribution) receive, after the Effective Date, (i) net Cash proceeds from the sale of any property, plant and/or equipment of SemMaterials, L.P., (ii) net Cash proceeds in excess of \$51 million from the sale of assets of SemFuel, L.P. other than inventory and receivables, (iii) net Cash proceeds from the sale of Equity Interests in SemGroup Holdings, L.P. (to the extent allocable to the Revolver Obligations (as defined in the Prepetition Credit Agreement) or the U.S. Term Obligations (as

defined in the Prepetition Credit Agreement)) or (iv) the Auriga Revolver/Term Lender Distribution.

5.6 <u>Class 122 – White Cliffs Credit Agreement Claim</u>

- (a) <u>Impairment and Voting</u>. Class 122 is impaired by the Plan. Each holder of a White Cliffs Credit Agreement Claim is entitled to vote to accept or reject the Plan.
- (b) <u>Distributions</u>. On the Effective Date, the White Cliffs Credit Agreement will be amended, extended, or refinanced on terms to be agreed to by the holders of the White Cliffs Credit Agreement Claims and to be contained in the Plan Supplement.

5.7 <u>Classes 123 through 148 – Other Secured Claims</u>

- (a) <u>Impairment and Voting</u>. Classes 123 through 148 are unimpaired by the Plan. Each holder of an Other Secured Claim is conclusively presumed to have accepted the Plan and is not entitled to vote to accept or reject the Plan.
- (b) <u>Distributions</u>. On the Effective Date, or as soon thereafter as is practicable, each holder of an Allowed Other Secured Claim shall receive, on account of their Claims against the Debtors and their estates, one of the following distributions: (i) the payment of such holder's Allowed Other Secured Claim in full in Cash; (ii) the sale or disposition proceeds of the property securing any Allowed Other Secured Claim to the extent of the value of its interest in such property; (iii) the surrender to the holder or holders of any Allowed Other Secured Claim of the property securing such Claim; or (iv) such other distributions as shall be necessary to satisfy the requirements of chapter 11 of the Bankruptcy Code. The manner and treatment of each Allowed Other Secured Claim shall be determined by the Debtors and transmitted, in writing, to the holder of such Other Secured Claim on or prior to the deadline to vote to accept or reject the Plan.

5.8 Classes 149 through 174 – Senior Notes Claims

(a) <u>Impairment and Voting</u>. Classes 149 through 174 are impaired by the Plan. Each holder of a Senior Notes Claim is entitled to vote to accept or reject the Plan. The vote by each holder of a Senior Notes Claim in favor of or against the Plan is deemed to be a vote in favor of or against the Canadian Plans, respectively.

(b) Distributions.

(i) <u>Distributions If Any of Classes 149 Through 174 Accept the Plan.</u> The Senior Notes Claims are Allowed Claims in the aggregate amount of \$609,770,833.33, not subject to offset, defense, counterclaim, reduction, or credit of any kind whatsoever. On the Effective Date, or as soon thereafter as is practicable, if any of Classes 149 through 174 accept the Plan, each holder of an Allowed Senior Notes Claim shall be entitled to receive, on account of their Claims against the Debtors and their estates and any related claim(s) under the Canadian Plans, its Pro Rata Share of (a) 1,552,500 shares of the New Common Stock (subject to dilution of

ownership percentage from the Warrants and the Management Stock), (b) Warrants to purchase 1,634,210 shares of New Common Stock (subject to dilution of ownership percentage from the Management Stock), (c) 30% of the Litigation Trust Interests, and (d) the Senior Notes Indenture Trustee Fees; <u>provided</u>, <u>however</u>, that the Senior Notes Indenture Trustee Fees shall be paid to the Senior Notes Indenture Trustee in accordance with Section 2.5 of this Plan.

- (ii) <u>Distributions If All of Classes 149 Through 174 Reject the Plan.</u> On the Effective Date, or as soon thereafter as is practicable, if all Classes 149 through 174 reject the Plan, each holder of an Allowed Senior Notes Claim shall be entitled to receive, on account of their Claims against the Debtors and their estates and any related claim(s) under the Canadian Plans, its Pro Rata Share of (a) 106,498 shares of the New Common Stock (subject to dilution of ownership percentage from the Warrants and the Management Stock), (b) 30% of the Litigation Trust Interests, and (c) the Senior Notes Indenture Trustee Fees; provided, however, that the Senior Notes Indenture Trustee Fees shall be paid to the Senior Notes Indenture Trustee in accordance with Section 2.5 of this Plan.
- (iii) Potential Additional Distribution. In addition, if all Classes in Classes 201 through 226 (General Unsecured Claims) vote to reject the Plan and any of Classes 149 through 174 accept the Plan, each holder of an Allowed Senior Notes Claim shall be entitled to receive its Pro Rata Share of the New Common Stock and Warrants that would have been distributed to the holders of Claims in Classes 201 through 226 as a result of the Creditors' Settlement.
- Distribution Mechanics. All distributions to holders of Allowed (iv) Senior Notes Claims shall be made (a) to the Senior Notes Indenture Trustee or (b) with the prior written consent of the Senior Notes Indenture Trustee, through the facilities of the DTC. The Senior Notes Indenture Trustee shall administer the distributions in accordance with the Plan and the Senior Notes Indenture Trustee and shall be compensated in accordance with Section 2.5 of this Plan, without further Bankruptcy Court approval, for all services related to distributions pursuant to the Plan (and for the related reasonable fees and expenses of counsel or professionals engaged by the Senior Notes Indenture Trustee with respect to administering or implementing such distributions in accordance with Section 2.5 of this Plan). The Senior Notes Indenture Trustee shall not be required to give any bond, surety, or other security for the performance of its duties with respect to the administration and implementation of distributions.

5.9 Classes 175 through 200 – Lender Deficiency Claims

- (a) <u>Impairment and Voting</u>. Classes 175 through 200 are impaired by the Plan. Each holder of a Lender Deficiency Claim is entitled to vote to accept or reject the Plan. The vote by each holder of a Lender Deficiency Claim in favor or against the Plan is deemed to be a vote in favor of or against the Canadian Plans, respectively.
- Swap Contract (as defined in the Prepetition Credit Agreement) that is not a Lender Swap Obligation (as defined in the Prepetition Credit Agreement) are Allowed Claims, not subject to offset, defense, counterclaim, reduction, or credit of any kind whatsoever. On the Effective Date, or as soon thereafter as is practicable, each holder of an Allowed Lender Deficiency Claim shall be entitled to receive, on account of their Claims against the Debtors and their estates and any related claim(s) under the Canadian Plans, its Pro Rata Share of 60% of the Litigation Trust Interests. In addition, if all of Classes 149 through 174 (Senior Notes Claims) and all Classes 201 through 226 (General Unsecured Claims) vote to reject the Plan, each holder of an Allowed Lender Deficiency Claim shall be entitled to receive its Pro Rata Share of the additional New Common Stock (but not the Warrants) that would have been distributed to the holders of Claims in Classes 149 through 174 and Classes 201 through 226 as a result of the Creditors' Settlement.

5.10 Classes 201 through 226 – General Unsecured Claims

- (a) <u>Impairment and Voting</u>. Classes 201 through 226 are impaired by the Plan. Each holder of a General Unsecured Claim is entitled to vote to accept or reject the Plan.
- (b) <u>Distributions If Any Class in Classes 201 Through 226 Accepts the Plan.</u> On the Effective Date, or as soon thereafter as is practicable, each holder of an Allowed General Unsecured Claim against a Debtor in a Class of General Unsecured Claims which accepts the Plan shall be entitled to receive on account of its Claims against the Debtors and their estates, its Pro Rata Share (calculated among all Classes of General Unsecured Claims which accept the Plan) of (a) 517,500 shares of the New Common Stock (subject to dilution of ownership percentage from the Warrants and the Management Stock), (b) Warrants to purchase 544,737 shares of New Common Stock (subject to dilution of ownership percentage from the Management Stock), and (c) 10% of the Litigation Trust Interests.
- (c) <u>Distributions If Any Class in Classes 201 Through 226 Rejects the Plan.</u>
 On the Effective Date, or as soon thereafter as is practicable, if any Class in Classes 201 through 226 rejects the Plan, each holder of an Allowed General Unsecured Claim in such rejecting Class shall be entitled to receive, on account of their Claims against the Debtors and their estates, its Pro Rata Share (calculated among all Classes of General Unsecured Classes) of (a) 25,327 shares of the New Common Stock (subject to dilution of ownership percentage from the Warrants and the Management Stock) and (b) 10% of the Litigation Trust Interests. In such event, each holder of an Allowed General Unsecured Claim in any accepting Classes in Classes 201 through 226 shall be entitled to receive its Pro Rata Share of the additional New Common Stock and the Warrants that would have been distributed to the holders of Claims in Classes 201 through 226 that rejected the Plan as provided in Section 5.10(b).
- (d) <u>Potential Additional Distribution</u>. In addition, if all Classes in Classes 149 through 174 (the Senior Notes Claims) vote to reject the Plan, each holder of an Allowed Claim

in any of the Classes 201 through 226 that votes to accept the Plan shall be entitled to receive its Pro Rata Share (calculated among all Classes of General Unsecured Claims which accept the Plan) of the additional New Common Stock and the Warrants that would have been distributed to the holders of Claims in Classes 149 through 174 as a result of the Creditors' Settlement.

5.11 Classes 227 through 252 – Intercompany Claims

- (a) <u>Impairment and Voting</u>. Classes 227 through 252 are impaired by the Plan. Notwithstanding the foregoing, each holder of an Allowed Intercompany Claim is conclusively presumed to have accepted the Plan by virtue of proposing the Plan and is not required to submit a ballot accepting the Plan.
- (b) <u>Treatment</u>. On the Effective Date, or as soon thereafter as is practicable, each Debtor which is a holder of an Allowed Intercompany Claim shall be deemed to be entitled to receive on account of such Allowed Intercompany Claim the New Common Stock it would receive if such Allowed Intercompany Claim were an Allowed General Unsecured Claim, which New Common Stock shall be redistributed to holders of Allowed Secured Working Capital Lender Claims in accordance with the provisions of the Plan.

5.12 Classes 253 through 278 – Intercompany Equity Interests

- (a) <u>Impairment and Voting</u>. Classes 253 through 278 are unimpaired by the Plan. Each holder of an Allowed Intercompany Equity Interest is conclusively presumed to have accepted the Plan and is not entitled to vote to accept or reject the Plan.
- (b) <u>Distributions</u>. Subject to Article VII of the Plan, on the Effective Date or as soon thereafter as is practicable, each Allowed Intercompany Equity Interest shall be retained.

5.13 Class 279 – SemGroup Equity Interests

- (a) <u>Impairment and Voting</u>. Class 279 is impaired by the Plan. Notwithstanding the foregoing, each holder of an Allowed SemGroup Equity Interest is conclusively presumed to have rejected the Plan and is not entitled to vote to accept or reject the Plan.
- (b) <u>Treatment</u>. Each holder of an Allowed SemGroup Equity Interest shall receive no distribution for and on account of such SemGroup Equity Interest and such SemGroup Equity Interest shall be cancelled on the Effective Date.

ARTICLE VI

IDENTIFICATION OF CLAIMS AND EQUITY INTERESTS IMPAIRED AND NOT IMPAIRED BY THE PLAN

6.1 <u>Impaired and Unimpaired Classes</u>. Claims and Equity Interests in Classes 1 through 52, 123 through 148, and 253 through 278 are not impaired under the Plan. Claims and Equity Interests in Classes 53 through 55, 70 through 122, 149 through 252, and 279 are impaired under the Plan.

6.2 <u>Controversy Concerning Impairment</u>. In the event of a controversy as to whether any Class of Claims or Equity Interests is impaired under the Plan, the Bankruptcy Court shall, after notice and a hearing, determine such controversy.

ARTICLE VII

ACCEPTANCE OR REJECTION OF PLAN; EFFECT OF REJECTION BY ONE OR MORE CLASSES OF CLAIMS

- 7.1 <u>Impaired Classes to Vote on Plan</u>. Each holder of a Claim or Equity Interest in an impaired Class, not otherwise deemed to have rejected the Plan, shall be entitled to vote separately to accept or reject the Plan. The Claims included in Classes 53 through 55, 70 through 122 and 149 through 226 are impaired and therefore are entitled to vote to accept or reject the Plan. The Classes of Intercompany Claims and Intercompany Equity Interests are deemed to have accepted the Plan by virtue of proposing the Plan.
- 7.2 Acceptance by Class of Creditors and Holders of Equity Interests. An impaired Class of holders of Claims shall have accepted the Plan if the Plan is accepted by at least two-thirds (2/3) in dollar amount and more than one-half (1/2) in number of the Allowed Claims of such Class that have voted to accept or reject the Plan. An impaired Class of Holders of Equity Interests shall have accepted the Plan if the Plan is accepted by at least two-thirds (2/3) in amount of the Allowed Equity Interests of such Class that have voted to accept or reject the Plan.
- 7.3 <u>Cramdown</u>. In the event that any impaired Class of Claims or Equity Interests fails to accept the Plan in accordance with section 1129(a) of the Bankruptcy Code, the Debtors reserve the right to request that the Bankruptcy Court confirm the Plan in accordance with section 1129(b) of the Bankruptcy Code.

ARTICLE VIII

IMPLEMENTATION OF THE PLAN

- 8.1 Non-Substantive Consolidation. On the Effective Date, the Debtors' estates shall not be deemed to be substantively consolidated for purposes of the Plan. Any Claims against one or more of the Debtors based upon a guaranty, indemnity, co-signature, surety, or otherwise, of Claims against another Debtor shall be treated as separate and distinct Claims against the estates of the respective Debtors and shall be entitled to the treatment provided for under the Plan's provisions concerning distributions.
- 8.2 **Restructuring Transactions.** On the Effective Date, the following transactions shall be effectuated in the following order:
- (a) <u>Transfer of Obligations</u>. The Debtors shall transfer to SemGroup, and SemGroup shall assume, all of the Debtors' outstanding obligations related to Secured Claims and Unsecured Claims that are being discharged pursuant to the Plan.

- (b) New Holdco. SemGroup Finance shall adopt the New Holdco Certificate of Incorporation pursuant to which it shall change its name to SemGroup Corporation and increase its authorized number of shares of capital stock. The adoption of the New Holdco Certificate of Incorporation and increase in capital stock shall be hereby authorized without any further need for any corporate action.
- (c) <u>Contribution of New Entities to New Holdco</u>. SemGroup shall contribute all of its ownership interests in its directly-owned subsidiaries to SemGroup Finance in exchange for (i) 41,400,000 shares of New Common Stock, (ii) Warrants to purchase 2,178,948 shares of New Common Stock, and (iii) the Second Lien Term Loan Interests. The issuance of the New Common Stock, Warrants and Second Lien Term Loan Interests by New Holdco to SemGroup shall be hereby authorized without any further need for any corporate action.
- (d) <u>Distributions to Holders of Allowed Claims</u>. Following SemGroup's receipt of the New Common Stock, the Warrants and the Second Lien Term Loan Interests, SemGroup shall distribute the Plan Currency to the Disbursing Agent for the holders of Allowed Claims.
- (e) <u>Vesting of SemGroup's Assets in New Holdco</u>. Except as otherwise provided in the Plan or any agreement, instrument or other document incorporated herein, all assets in SemGroup's estate shall vest in New Holdco and New Holdco may operate its business and use, acquire or dispose of property free of any restrictions of the Bankruptcy Code or the Bankruptcy Rules and in all respects as if there were no pending case under any chapter or provision of the Bankruptcy Code.
- (f) <u>Tax Matters</u>. In connection with the transfer of assets from SemGroup to New Holdco, New Holdco shall prepare and deliver to SemGroup a copy of Form 8594, along with any required exhibits thereto, allocating the purchase price of the transferred assets among such assets in accordance with Section 1060 of the Tax Code and the Treasury Regulations promulgated thereunder. SemGroup and New Holdco shall prepare and file all tax returns and reports in a manner consistent with such allocation.
- 8.3 **Producer Representative**. Prior to the Confirmation Hearing, the Producers' Committee shall select an individual to serve as the Producer Representative, who shall be an estate representative for the purpose of objecting to: (i) First Purchaser Producer Twenty-Day Claims, (ii) proofs of claim filed in respect of Secured First Purchaser Producer Claims, and (iii) requests for reimbursement by Producer Plaintiffs for the Active States for reimbursement of fees and out-of-pocket costs incurred in connection with the Producer State-Specific Adversary Proceedings for the Active States or in the Other Proceedings. The Producer Representative shall be entitled to employ professionals of its choosing. The fees and expenses of the Producer Representative and any professionals hired by the Producer Representative shall be paid solely in accordance with Section 3.1(c) and 3.1(d) hereof and Debtors shall have no responsibility therefor. The Producer Representative shall have any and all rights of the Debtors and the Reorganized Debtors: (i) to object to First Purchaser Producer Twenty-Day Claims in accordance with the Notices filed by the Debtors on July 17, 2009, (ii) to object to the Secured First Purchaser Producer Claims pursuant to Section 10.1 of this Plan, and (iii) to object to professional fee reimbursement requests of Producer Plaintiffs. The Producers' Committee shall

endeavor to reach agreement in writing with Producer Plaintiffs for reimbursement of fees and out-of-pocket costs. In the event agreement is not reached with a specific Producer Plaintiff, that party shall be required to file a request for reimbursement in accordance with section 503(b) of the Bankruptcy Code and the Producer Representative shall have the right to object to such applications. The Reorganized Debtors shall cooperate with and provide reasonable assistance to representatives of the Producer Representative for the purpose of prosecuting the objections herein to Claims. Notwithstanding anything in the Plan to the contrary, the Producer Representative shall not be permitted to commence or participate in any Claim or Cause of Action that shall be transferred to the Litigation Trust pursuant to the Plan or reserved to the Prepetition Lenders or the Prepetition Administrative Agent as provided in Section 10.1(b) of the Plan.

- 8.4 <u>Litigation Trust Arrangements</u>. On the Effective Date, New Holdco will enter into the Litigation Trust Agreement pursuant to which the Litigation Trust Funds will be advanced to the Litigation Trust. The Litigation Trust Funds will be secured by all of the assets of the Litigation Trust and will be paid to the holders of the Secured Working Capital Lender Claims before the holders of the Litigation Trust Interests receive any distributions on account of such interests.
- 8.5 <u>Section 1145 Securities</u>. To the extent provided in section 1145 of the Bankruptcy Code and under applicable nonbankruptcy law, the issuance under the Plan of the Plan Currency will be exempt from registration under the Securities Act of 1933, as amended, and all rules and regulations promulgated thereunder.
- 8.6 <u>Corporate Action</u>. Upon the Effective Date, the following transactions shall be deemed to occur:
- General. All actions contemplated by the Plan shall be deemed authorized (a) and approved in all respects, including (i) the execution and entry into the Litigation Trust Agreement, (ii) the execution and entry into the Exit Facility, (iii) the distribution of the New Common Stock, (iv) the distribution of the Warrants, (v) the execution and entry of the Second Lien Term Loan Facility, (vi) adoption of the Management Incentive Plan, (vii) selection of the Board and the officers of New Holdco, and (viii) all other actions contemplated by the Plan (whether to occur before, on, or after the Effective Date), including without limitation, actions in connection with the sale or disposal of the remaining assets of SemMaterials, L.P. or SemFuel, L.P. and wind-down of their respective affairs. All matters provided for in the Plan involving the structure of the Debtors or the Reorganized Debtors and any action required by the Debtors or the Reorganized Debtors in connection with the Plan shall be deemed to have occurred and shall be in effect, without any requirement of further action by the security holders, directors, or officers of the Debtors, the Reorganized Debtors, or New Holdco. On or prior (as applicable) to the Effective Date, the appropriate officers of the Debtors, the Reorganized Debtors, or New Holdco, as applicable, shall be authorized and directed to issue, execute, and deliver the agreements, documents, securities, and instruments contemplated by the Plan (or necessary or desirable to effect the transactions contemplated by the Plan) in the name of and on behalf of the Reorganized Debtors, including (i) the Litigation Trust Agreement, (ii) the Exit Facility, (iii) the Second Lien Term Loan Facility, (iv) the Warrant Agreement, and (v) any and all other agreements, documents, securities, and instruments relating to the foregoing, including, without

limitation, in connection with the sale, disposal or wind-down of SemMaterials, L.P. Acceptance of the Plan by the holders of Claims will be deemed to constitute approval of the Management Incentive Plan for purposes of Sections 162(m) and 422 of the Internal Revenue Code of 1986, as amended, as well as Section 16 of the Exchange Act and any stock exchange listing requirement.

- (b) New Holdco Certificate of Incorporation and New Holdco Bylaws. On the Effective Date, SemGroup Finance shall adopt the New Holdco Certificate of Incorporation and the New Holdco Bylaws and shall file the New Holdco Certificate of Incorporation with the Secretary of State of Delaware. In addition, on or before the Effective Date, pursuant to and only to the extent required by section 1123(a)(6) of the Bankruptcy Code, the New Holdco Certificate of Incorporation shall satisfy the provisions of the Bankruptcy Code and shall include, among other things, pursuant to section 1123(a)(6) of the Bankruptcy Code, (i) a provision prohibiting the issuance of non-voting equity securities and (ii) a provision setting forth an appropriate distribution of voting power among classes of equity securities possessing voting power, including, in the case of any class of equity securities having a preference over another class of equity securities with respect to dividends, adequate provisions for the election of directors representing such preferred class in the event of default in the payment of such dividends. On the Effective Date, the boards of directors of each Reorganized Debtor shall be deemed to have adopted the restated bylaws for such Debtor.
- 8.7 **Existence.** Except as otherwise provided in the Plan, each Reorganized Debtor shall continue to exist after the Effective Date as a separate corporation, limited liability company, partnership, or other form, as the case may be, with all the powers of a corporation, limited liability company, partnership, or other form, as the case may be, pursuant to the applicable law in the jurisdiction in which each applicable Reorganized Debtor is incorporated or formed and pursuant to the respective certificate of incorporation and bylaws (or other formation documents) in effect prior to the Effective Date, except to the extent such certificate of incorporation and bylaws (or other formation documents) are amended by the Plan or otherwise, and to the extent such documents are amended, such documents are deemed to be pursuant to the Plan and require no further action or approval.
- 8.8 Vesting of Assets in the Reorganized Debtors. Except as otherwise provided in the Plan or any agreement, instrument, or other document incorporated therein, on the Effective Date, all property in the Debtors' estates, the Litigation Trust Assets, and any property acquired by any of the Debtors pursuant to the Plan shall vest in the Reorganized Debtors or the Litigation Trust, as the case may be, free and clear of all Liens, Claims, charges, or other encumbrances (except for Liens, if any, granted to secure the repayment of the Litigation Trust Funds, the Exit Facility, the Second Lien Term Loan Interests, and Claims pursuant to the Postpetition Financing Agreement that by their terms survive termination of the Postpetition Financing Agreement). On and after the Effective Date, each Reorganized Debtor may operate its business and may use, acquire, or dispose of property and compromise or settle any Retained Causes of Action or interests without supervision or approval by the Bankruptcy Court and free of any restrictions of the Bankruptcy Code or Bankruptcy Rules.
- 8.9 <u>Cancellation of Debt and Equity Securities and Related Obligations</u>. Except (a) as otherwise expressly provided in the Plan, (b) with respect to executory contracts or

unexpired leases that have been assumed by the Debtors, (c) for purposes of evidencing a right to distributions under the Plan, or (d) with respect to any Claim that is Allowed under the Plan, on the Effective Date, any instruments or documents evidencing any Claims or Equity Interests shall be deemed automatically cancelled and deemed surrendered without further act or action under any applicable agreement, law, regulation, order, or rule and the obligations of the Debtors under the agreements, instruments, and other documents, indentures, and certificates of designations governing such Claims and Equity Interests, as the case may be, shall be discharged; provided, however, that such instruments or documents shall continue in effect solely for the purpose of (x) allowing the holders of such Claims to receive their distributions under the Plan and (y) allowing the Disbursing Agent to make such distributions to be made on account of such Allowed Claims; provided, further, that if the Senior Notes Indenture Trustee Fees have not been paid on the Effective Date, this Section 8.9 shall be of no force and effect with respect to the Senior Notes Indenture Charging Lien; provided, further, that the provisions of the Prepetition Credit Agreement governing the relationship of Bank of America, N.A., in its capacity as administrative agent to the Prepetition Lenders under the Prepetition Credit Agreement, and the Prepetition Lenders, including but not limited to those provisions relating to the rights of the Prepetition Administrative Agent to expense reimbursement, indemnification and other similar amounts, shall not be affected by and shall survive the Plan, entry of the Confirmation Order and the occurrence of the Effective Date. SemGroup shall recognize and report on its final federal, state, local and foreign income tax returns any and all cancellation of indebtedness or similar income arising as a result of the Plan. In addition, none of SemGroup G.P., L.L.C., the general partner of SemGroup, L.P., nor any of the Debtors shall make an election to treat any of the Debtors (or their subsidiaries) as an association taxable as a corporation for any tax purposes unless such election is effective no earlier than the day after the Effective Date.

- 8.10 Effectuating Documents and Further Transactions. On and after the Effective Date, the Reorganized Debtors, the Board, and the officers of New Holdco are authorized to and may issue, execute, deliver, file, or record such contracts, securities, instruments, releases, and other agreements or documents and take such actions as may be necessary or appropriate to effectuate, implement, and further evidence the terms and conditions of the Plan and the securities issued pursuant to the Plan in the name of and on behalf of the Reorganized Debtors, without the need for any approvals, authorizations, or consents except for those expressly required pursuant to the Plan.
- 8.11 <u>Compensation and Benefit Plans.</u> Notwithstanding anything contained in the Plan to the contrary, unless specifically rejected by order of the Bankruptcy Court, or unless subject to a motion for approval of rejection that has been filed and served prior to the Confirmation Date, the Compensation and Benefit Plans shall be deemed to be, and shall be treated as though they are, executory contracts that are deemed assumed under the Plan on the same terms, and the Debtors' obligations under the Compensation and Benefit Plans shall be deemed assumed pursuant to section 365(a) of the Bankruptcy Code, shall survive confirmation of the Plan, shall remain unaffected thereby, and shall not be discharged in accordance with section 1141 of the Bankruptcy Code. Any default existing under the Compensation and Benefit Plans shall be cured promptly after it becomes known by the Reorganized Debtors.

ARTICLE IX

PRESERVATION AND PROSECUTION OF CAUSES OF ACTION HELD BY THE DEBTORS

9.1 Preservation and Prosecution of Causes of Action. In accordance with section 1123(b) of the Bankruptcy Code, (a) subject to the provisions of Section 10.1(b), the Reorganized Debtors shall retain and may enforce all rights to commence and pursue, as appropriate, any and all Retained Causes of Action, and (b) the Litigation Trust may enforce all rights to commence and pursue, as appropriate, any and all Litigation Trust Claims, and the Reorganized Debtors' rights to commence, prosecute, or settle their Retained Causes of Action shall be preserved notwithstanding the occurrence of the Effective Date. Subject to the provisions of Section 10.1(b), the Reorganized Debtors may pursue their Retained Causes of Action, as appropriate, in accordance with the best interests of the Reorganized Debtors. No Entity may rely on the absence of a specific reference in the Plan or the Disclosure Statement to any Cause of Action against them as any indication that the Debtors, the Litigation Trustee, the Prepetition Lenders or the Prepetition Administrative Agent, as applicable, will not pursue any and all available Causes of Action against them. The Reorganized Debtors expressly reserve all rights to prosecute any and all Retained Causes of Action against any Entity, except as otherwise expressly provided herein. Unless any Causes of Action against an Entity are expressly waived, relinquished, exculpated, released, compromised, or settled in the Plan or a Bankruptcy Court order, the Debtors, the Litigation Trustee, the Prepetition Lenders or the Prepetition Administrative Agent, as the case may be, expressly reserve all Causes of Action, for later adjudication, and, therefore, no preclusion doctrine, including the doctrines of res judicata, collateral estoppel, issue preclusion, claim preclusion, estoppel (judicial, equitable, or otherwise), or laches, shall apply to such Causes of Action upon, after, or as a consequence of the Confirmation Order. The Reorganized Debtors reserve and shall retain the Retained Causes of Action notwithstanding the rejection of any executory contract or unexpired lease during the Chapter 11 Cases or pursuant to the Plan. In accordance with section 1123(b)(3) of the Bankruptcy Code, any Causes of Action that a Debtor may hold against any Entity shall vest in the Reorganized Debtors or the Litigation Trustee, as the case may be. As of the Effective Date, the Reorganized Debtors shall have assigned the Litigation Trust Claims to the Litigation Trust. Subject to the provisions of Section 10.1(b), the Reorganized Debtors shall have the exclusive right, authority, and discretion to determine and to initiate, file, prosecute, enforce, abandon, settle, compromise, release, withdraw, or litigate to judgment any Retained Causes of Action and to decline to do any of the foregoing without the consent or approval of any third party or further notice to or action, order, or approval of the Bankruptcy Court.

ARTICLE X

PROVISION FOR TREATMENT OF DISPUTED CLAIMS OR ASSERTED ADMINISTRATIVE EXPENSES UNDER THE PLAN

10.1 <u>Objections to Claims; Prosecution of Disputed Claims or Asserted</u> **Administrative Expenses.**

- The Reorganized Debtors, the Prepetition Administrative Agent (solely with respect to Other Twenty-Day Claims) or the Producer Representative (solely with respect to First Purchaser Producer Twenty-Day Claims and Secured First Purchaser Producer Claims) or the Prepetition Lenders (solely with respect to Causes of Action referred to in Section 10.1(b)(ii)), shall object to the allowance of Claims or Equity Interests filed with the Bankruptcy Court with respect to which they dispute liability, priority, and/or amount; provided, however, that the foregoing shall not in any way limit the ability or the right of the Litigation Trustee to assert, commence, or prosecute any Cause of Action that is a Litigation Trust Claim against any holder of such Claim. All objections, affirmative defenses, and counterclaims shall be litigated to Final Order; provided, however, that except as set forth in Section 10.1(b) hereof, the Reorganized Debtors (within such parameters as may be established by the Board), or the Prepetition Administrative Agent (solely with respect to Other Twenty-Day Claims) or the Producer Representative (solely with respect to First Purchaser Producer Twenty-Day Claims and Secured First Purchaser Producer Claims), shall have the authority to file, settle, compromise, or withdraw any objections to Claims or Equity Interests; provided, further, that the foregoing shall not in any way limit the ability or the right of the Litigation Trustee to assert, commence, or prosecute any Cause of Action that is a Litigation Trust Claim against any holder of such Claim. Unless otherwise ordered by the Bankruptcy Court, the Reorganized Debtors, the Prepetition Administrative Agent (solely with respect to Other Twenty-Day Claims) or the Producer Representative (solely with respect to First Purchaser Producer Twenty-Day Claims and Secured First Purchaser Producer Claims) or the Prepetition Lenders (solely with respect to Causes of Action referred to in Section 10.1(b)(ii)), shall file and serve all objections to Claims as soon as practicable, but, in each instance, not later than 180 days following the Confirmation Date or such later date as may be approved by the Bankruptcy Court or 60 days in the case of the Prepetition Lenders for Causes of Action referred to in Section 10.1(b)(ii). All orders of the Bankruptcy Court regarding procedures to resolve certain Claims or Administrative Expenses, including the Twenty-Day Claims, that have been entered prior to the Effective Date shall continue to govern the resolution of such Claims and Administrative Expenses after the Effective Date.
 - (b) Notwithstanding anything to the contrary in Section 10.1(a):
 - (i) The Prepetition Administrative Agent shall have the authority to file, settle, compromise, or withdraw any objections to Other Twenty-Day Claims. With respect to any period after the Effective Date, the Prepetition Administrative Agent shall be reimbursed solely from the net savings resulting from the successful prosecution of such objections prior to the distribution of any additional monies to the holders of the Secured Working Capital Lender Claims under this Plan; provided, however, that \$3 million of Plan Cash shall be distributed to the Prepetition Administrative Agent on the Effective Date and used to initially fund the Prepetition Administrative Agent for purposes of this Section 10.1(b)(i).
 - (ii) Notwithstanding anything to the contrary in this Plan, any Prepetition Lender shall, at its own expense, have the authority to file, not later than 60 days following the Confirmation Date, a Cause of Action against a Prepetition Lender with a Claim under a Swap Contract (as

defined in the Prepetition Credit Agreement) or a holder of a Swap Claim solely to determine whether or not such Swap Contract Claim qualifies as a Lender Swap Obligation under the Prepetition Credit Agreement, provided, however, such Swap Contract Claim (x) has not been Allowed in whole or in part as a Lender Swap Obligation on or before the Effective Date or (y) is not subject to an objection filed by the Debtors on or before the Effective Date, and any such Cause of Action shall be settled, compromised or otherwise resolved or determined pursuant to a Final Order of the Bankruptcy Court. Any Cause of Action described by this Section 10.1(b)(ii) shall not constitute a Released Action and is not subject to the release found in Section 20.11 of the Plan.

- 10.2 No Distributions Pending Allowance. Notwithstanding any other provision hereof, if any portion of a Claim is a Disputed Claim, no payment or distribution provided hereunder shall be made on account of such Claim unless and until such Disputed Claim becomes an Allowed Claim. Notwithstanding the foregoing, on the Effective Date, distributions will be made to each Prepetition Lender in respect of its Secured Working Capital Lender Claims that are not derived from Lender Swap Obligations (as defined in the Prepetition Credit Agreement) based on a ratio where the numerator is the amount of such Prepetition Lender's Secured Working Capital Lender Claims and the denominator is the aggregate amount of all Secured Working Capital Lender Claims, including any derived from Lender Swap Obligations (as defined in the Prepetition Credit Agreement), regardless of whether any Claims with respect to Lender Swap Obligations (as defined in the Prepetition Credit Agreement) are Disputed on the Effective Date. If a Claim with respect to any Lender Swap Obligation (as defined in the Prepetition Credit Agreement) is not Allowed on the Effective Date, distributions with respect to such Lender Swap Obligation (as defined in the Prepetition Credit Agreement) will be held in reserve by the Disbursing Agent until such Claim becomes Allowed or disallowed.
- Estimation of Claims. Unless otherwise limited by an order of the Bankruptcy Court, the Reorganized Debtors, the Prepetition Administrative Agent (solely with respect to Other Twenty-Day Claims) or the Producer Representative (solely with respect to First Purchaser Producer Twenty-Day Claims and Secured First Purchaser Producer Claims), may at any time request that the Bankruptcy Court estimate for final distribution purposes any contingent, unliquidated, or Disputed Claim pursuant to section 502(c) of the Bankruptcy Code, regardless of whether the Debtors or the Reorganized Debtors previously objected to such Claim. The Bankruptcy Court will retain jurisdiction to consider any request to estimate any Claim at any time during litigation concerning any objection to any Claim, including, without limitation, during the pendency of any appeal relating to any such objection. Unless otherwise provided in an order of the Bankruptcy Court, in the event that the Bankruptcy Court estimates any contingent, unliquidated, or Disputed Claim, the estimated amount shall constitute either the allowed amount of such Claim or a maximum limitation on such Claim, as determined by the Bankruptcy Court; provided, however, that if the estimate constitutes the maximum limitation on such Claim, the Debtors, the Reorganized Debtors, the Prepetition Administrative Agent or the Producer Representative, as the case may be, may elect to pursue supplemental proceedings to object to any ultimate allowance of such Claim, and; provided, further, that the foregoing is not intended to limit the rights granted by section 502(j) of the Bankruptcy Code. All of the

aforementioned Claims objection, estimation, and resolution procedures are cumulative and not necessarily exclusive of one another.

10.4 **Payments and Distributions on Disputed Claims**.

- <u>Disputed Claims Reserve</u>. From and after the Effective Date, and until such time as all Disputed Claims have been compromised and settled or determined by Final Order, the Disbursing Agent or, as applicable, the Producer Representative, shall reserve and hold in escrow for the benefit of each holder of a Disputed Claim and any dividends, gains, or income attributable thereto, in an amount equal to distributions which would have been made to the holder of such Disputed Claim if it were an Allowed Claim in an amount equal to the lesser of (i) the Disputed Claim Amount, (ii) the amount in which the Disputed Claim shall be estimated by the Bankruptcy Court pursuant to section 502 of the Bankruptcy Code for purposes of allowance, which amount, unless otherwise ordered by the Bankruptcy Court, shall constitute and represent the maximum amount in which such Claim ultimately may become an Allowed Claim, or (iii) such other amount as may be agreed upon by the holder of such Disputed Claim and the Reorganized Debtors, the Prepetition Administrative Agent (solely with respect to Other Twenty-Day Claims) or the Producer Representative (solely with respect to First Purchaser Producer Twenty-Day Claims and Secured First Purchaser Producer Claims). Any Plan Currency reserved and held for the benefit of a holder of a Disputed Claim shall be treated as a payment and reduction on account of such Disputed Claim for purposes of computing any additional amounts to be paid in Cash or distributed in other Plan Currency in the event the Disputed Claim ultimately becomes an Allowed Claim. In the event that a Disputed Claim is not Allowed, in whole or in part, the holders of Allowed Claims in the same Class as the holder of the Claim that is not Allowed shall receive their Pro Rata Share of any Plan Currency reserved on account of the Claim that is not Allowed. Such Cash and any dividends, gains, or income paid on account of other Plan Currency reserved for the benefit of holders of Disputed Claims shall be either (x) held by the Disbursing Agent in an interest-bearing account or (y) invested in interest-bearing obligations issued by the United States Government and guaranteed by the United States Government, and having (in either case) a maturity of not more than thirty (30) days, for the benefit of such holders pending determination of their entitlement thereto under the terms of the Plan. No payments or distributions shall be made with respect to all or any portion of any Disputed Claim pending the entire resolution thereof by Final Order.
- (b) Allowance of Disputed Claims. At such time as a Disputed Claim becomes, in whole or in part, an Allowed Claim, the Disbursing Agent, or as applicable, the Producer Representative, shall distribute to the holder thereof the distributions, if any, to which such holder is then entitled under the Plan, together with any interest which has accrued on the amount of Cash and any dividends or distributions attributable to the Plan Currency so reserved (net of any expenses, including any taxes on the escrow, relating thereto), but only to the extent that such interest is attributable to the amount of the Allowed Claim. Such distribution, if any, shall be made as soon as practicable after the date that the order or judgment of the Bankruptcy Court allowing or disallowing such Disputed Claim becomes a Final Order but in no event more than ninety (90) days thereafter; provided, however, for matters affecting Disputed Claims that are First Purchaser Producer Twenty-Day Claims or Secured First Purchaser Producer Claims, the Producer Representative may in its sole discretion waive the requirement of a Final Order. The balance of any Cash previously reserved shall be included in Plan Cash and the balance of

any Plan Currency previously reserved shall be included in future calculations of Plan Currency to holders of Allowed Claims in accordance with the terms and provisions of the Plan. For the avoidance of doubt, any Cash released from reserves for Administrative Expense Claims (other than amounts distributed to the Producer Representative pursuant to Section 3.1(a) hereof), Professional Compensation and Reimbursement Claims, Priority Non-Tax Claims or Priority Tax Claims shall be paid to the Prepetition Administrative Agent for distribution to holders of Allowed Secured Working Capital Lender Claims.

- (c) <u>Tax Treatment of Escrow.</u> Subject to the receipt of contrary guidance from the IRS or a court of competent jurisdiction (including the receipt by the Disbursing Agent of a private letter ruling requested by the Disbursing Agent, or the receipt of an adverse determination by the IRS upon audit if not contested by the Disbursing Agent, or a condition imposed by the IRS in connection with a private letter ruling requested by the Debtor), the Disbursing Agent shall (i) treat the escrow as one or more discrete trusts (which may be composed of separate and independent shares) for federal income tax purposes in accordance with the trust provisions of the Tax Code (Sections 641 et seq.) and (ii) to the extent permitted by applicable law, report consistent with the foregoing for state and local income tax purposes. All holders of Allowed Claims shall report, for tax purposes, consistent with the foregoing.
- Funding of Escrow's Tax Obligation. If the Disputed Claims Reserve created in accordance with Section 10.4(a) hereof has insufficient funds to pay any applicable taxes imposed upon it or its assets, subject to the other provisions contained herein, the Reorganized Debtors shall advance to the escrow the funds necessary to pay such taxes (a "Tax Advance"), with such Tax Advances repayable from future amounts otherwise receivable by the escrow pursuant to this Section 10.4 or otherwise. If and when a distribution is to be made from the escrow, the distributee will be charged its pro rata portion of any outstanding Tax Advance (including accrued interest). If a Cash distribution is to be made to such distributee, the Disbursing Agent shall be entitled to withhold from such distributee's distribution the amount required to pay such portion of the Tax Advance (including accrued interest). If such Cash is insufficient to satisfy the respective portion of the Tax Advance, the distributee shall, as a condition to receiving such other assets, pay in Cash to the Disbursing Agent an amount equal to the unsatisfied portion of the Tax Advance (including accrued interest). Failure to make such payment shall entitle the Disbursing Agent to reduce and permanently adjust the amounts that would otherwise be distributed to such distributee to fairly compensate the Disputed Claims reserve created in accordance with Section 10.4(a) hereof for the unpaid portion of the Tax Advance (including accrued interest).

ARTICLE XI

THE LITIGATION TRUST

11.1 <u>Establishment of the Litigation Trust</u>. On the Effective Date, the Debtors or the Reorganized Debtors, as the case may be, on their own behalf and on behalf of the holders of Allowed Senior Notes Claims, Allowed Lender Deficiency Claims, and Allowed General Unsecured Claims shall execute the Litigation Trust Agreement and shall take all other steps necessary to establish the Litigation Trust in accordance with and pursuant to the terms of the Plan. The Debtors or the Reorganized Debtors shall transfer to the Litigation Trust all of their

right, title, and interest in the Litigation Trust Assets, including any Litigation Trust Claims being prosecuted by the Creditors' Committee on behalf of the Debtors' estates prior to the Effective Date. The Litigation Trust shall also be responsible for, and respond on behalf of the Debtors or the Reorganized Debtors to, the subpoenas and requests for production of documents substantially in the form attached hereto as Exhibit 11.1. The Plan Supplement will include a list of all Causes of Action being prosecuted as of the date thereof that will be transferred to the Litigation Trust. On the Effective Date, all of the rights, title to, and interests in the Contributing Lenders' Claims shall be deemed to be transferred to the Litigation Trust by the Contributing Lenders who voted to approve the Plan, without any further act or writing, and such Contributing Lenders' Claims shall be included as part of the Litigation Trust Assets; provided, however, that any Prepetition Lender or holder of a Swap Claim who did not vote to approve the Plan shall not be deemed to have transferred any rights until it has executed a Contributing Lender Assignment transferring its rights, title to, and interests in the Contributing Lenders' Claims in connection with its receipt of any distributions from the Litigation Trust. No Contributing Lender shall have the right to bring any Cause of Action with respect to any Contributing Lenders' Claims, and only the Litigation Trust shall have such right. Any recoveries on account of the Litigation Trust Assets shall be distributed to holders of Litigation Trust Interests in accordance with the Plan and the Litigation Trust Agreement. Any attorney-client privilege, work-product privilege, or other privilege or immunity attaching to any documents or communications (whether written or oral) associated with the above-described rights and Causes of Action shall be transferred to the Litigation Trust and shall vest in the Litigation Trustee and its representatives. The Debtors or the Reorganized Debtors, as the case may be, the Litigation Trustee and the Creditors' Committee are authorized to take all necessary actions to effectuate the transfer of such privileges. The Confirmation Order shall provide that the Litigation Trustee's receipt of transferred privileges shall be without waiver in recognition of the joint and/or successorship interest in prosecuting claims on behalf of the Debtors' estates. Notwithstanding any agreement or order entered by the Bankruptcy Court to the contrary, the Creditors' Committee shall share any discovery obtained prior to and after the Effective Date with the Litigation Trustee and the Litigation Trust Board.

- 11.2 <u>Purpose of the Litigation Trust</u>. The Litigation Trust shall be established for the sole purpose of liquidating the Litigation Trust Assets, in accordance with Treasury Regulation Section 301.7701-4(d), with no objective to continue or engage in the conduct of a trade or business.
- Trust Agreement, upon the creation of the Litigation Trust, the Debtors or the Reorganized Debtors, as the case may be, shall transfer the Litigation Trust Funds to finance the operations of the Litigation Trust, which funding, subject to the Litigation Trust Fund Reserve Amount, shall be repaid to the holders of the Secured Working Capital Lender Claims from the first Cash received by the Litigation Trust. The Debtors, the Debtors in Possession and the Reorganized Debtors shall have no further obligation to provide any funding with respect to the Litigation Trust. After payment in full of the Litigation Trust Funds pursuant to the preceding sentence, any Cash received in respect of any Litigation Trust Assets (excluding the Litigation Trust Funds themselves) shall be first allocated to replenish the Litigation Trust Fund Reserve Amount prior to being distributed to holders of Litigation Trust Interests in accordance with the Plan and the Litigation Trust Agreement.

- **Transfer of Assets.** The transfer of the Litigation Trust Assets to the Litigation Trust shall be made, as provided herein, for the benefit of the holders of Allowed Senior Notes Claims, Allowed Lender Deficiency Claims, and Allowed General Unsecured Claims. Immediately thereafter, on behalf of the holders of Allowed Senior Notes Claims, Allowed Lender Deficiency Claims, and Allowed General Unsecured Claims, the Debtors or the Reorganized Debtors, as the case may be, shall transfer such Litigation Trust Assets to the Litigation Trust in exchange for Litigation Trust Interests for the benefit of holders of Allowed Senior Notes Claims, Allowed Lender Deficiency Claims, and Allowed General Unsecured Claims in accordance with the Plan. Upon the transfer of the Litigation Trust Assets, the Debtors, the Debtors in Possession or the Reorganized Debtors, as the case may be, shall have no interest in or with respect to the Litigation Trust Assets or the Litigation Trust. Notwithstanding the foregoing, for purposes of section 553 of the Bankruptcy Code, the transfer of the Litigation Trust Assets to the Litigation Trust shall not affect the mutuality of obligations which otherwise may have existed prior to the effectuation of such transfer. To the extent that any Litigation Trust Assets cannot be transferred to the Litigation Trust because of a restriction on transferability under applicable non-bankruptcy law that is not superseded or preempted by section 1123 of the Bankruptcy Code or any other provision of the Bankruptcy Code, such Litigation Trust Assets shall be deemed to have been retained by the Reorganized Debtors or the Contributing Lenders, as applicable, and the Litigation Trustee shall be deemed to have been designated as a representative of the Reorganized Debtors or the Contributing Lenders, as applicable, pursuant to section 1123(b)(3)(B) of the Bankruptcy Code to enforce and pursue such Litigation Trust Assets on behalf of the Reorganized Debtors or the Contributing Lenders, as applicable. Notwithstanding the foregoing, all net proceeds of such Litigation Trust Assets shall be transferred to the Litigation Trust to be distributed to holders of the Secured Working Capital Lender Claims and holders of the Litigation Trust Interests consistent with the terms of this Plan and the Litigation Trust Agreement.
- Trust, but in no event later than 60 days thereafter, the Litigation Trust Board shall inform, in writing, the Litigation Trustee of the value of the assets transferred to the Litigation Trust, based on the good faith determination of the Litigation Trust Board, and the Litigation Trustee shall apprise, in writing, the beneficiaries of the Litigation Trust and the holders of the Secured Working Capital Lender Claims of such valuation. The valuation shall be used consistently by all parties (including the Debtors, the Reorganized Debtors, the Litigation Trustee, and the beneficiaries of the Litigation Trust) for all federal income tax purposes.

11.6 Litigation; Responsibilities of Litigation Trustee.

(a) The Litigation Trustee, upon direction by the Litigation Trust Board and in the exercise of their collective reasonable business judgment, shall, in an expeditious but orderly manner, liquidate and convert to Cash the assets of the Litigation Trust, make timely distributions, and not unduly prolong the duration of the Litigation Trust. The liquidation of the Litigation Trust Assets may be accomplished either through the prosecution, compromise and settlement, abandonment, or dismissal of any or all claims, rights, or causes of action, or otherwise. The Litigation Trustee, upon direction by the Litigation Trust Board, shall have the absolute right to pursue or not to pursue any and all Litigation Trust Assets as it determines is in the best interests of the beneficiaries of the Litigation Trust, and consistent with the purposes of

the Litigation Trust, and shall have no liability for the outcome of its decision except for any damages caused by willful misconduct or gross negligence. The Litigation Trustee may incur any reasonable and necessary expenses in liquidating and converting the Litigation Trust Assets to Cash and shall be reimbursed in accordance with the provisions of the Litigation Trust Agreement.

- (b) No later than fifteen (15) days prior to the date the hearing to confirm the Plan is commenced, the Litigation Trustee shall be selected by the Creditors' Committee and the Lender Steering Committee, named in the Confirmation Order or in the Litigation Trust Agreement, and have the power to (i) prosecute for the benefit of the Litigation Trust all Claims, rights, and Causes of Action transferred to the Litigation Trust (whether such suits are brought in the name of the Litigation Trust or otherwise) and (ii) otherwise perform the functions and take the actions provided for or permitted herein or in any other agreement executed by the Litigation Trustee pursuant to the Plan. Any and all proceeds generated from the Litigation Trust Assets shall be the property of the Litigation Trust.
- 11.7 **Investment Powers.** The right and power of the Litigation Trustee to invest assets transferred to the Litigation Trust, the proceeds thereof, or any income earned by the Litigation Trust shall be limited to the right and power to invest such assets (pending periodic distributions in accordance with Section 11.8 of this Plan) only in Cash and Government securities as defined in section 2(a)(16) of the Investment Company Act of 1940, as amended; provided, however, that (a) the scope of any such permissible investments shall be further limited to include only those investments that a liquidating trust, within the meaning of Treasury Regulation Section 301.7701-4(d), may be permitted to hold, pursuant to the Treasury Regulations, or any modification in the IRS guidelines, whether set forth in IRS rulings, other IRS pronouncements, or otherwise, and (b) the Litigation Trustee may expend the assets of the Litigation Trust (i) as reasonably necessary to meet contingent liabilities and maintain the value of the assets of the Litigation Trust during liquidation, (ii) to pay reasonable administrative expenses (including, but not limited to, any taxes imposed on the Litigation Trust or reasonable fees and expenses in connection with litigation), and (iii) to satisfy other liabilities incurred or assumed by the Litigation Trust (or to which the assets are otherwise subject) in accordance with the Plan or the Litigation Trust Agreement.
- 11.8 Annual Distribution; Withholding. The Litigation Trustee shall distribute at least semi-annually to the holders of the Secured Working Capital Lender Claims until the Litigation Trust Funds have been repaid and thereafter to the holders of the Litigation Trust Interests, all net Cash income plus all net Cash proceeds from the liquidation of the Litigation Trust Assets (including as Cash for this purpose, all Cash Equivalents); provided, however, that the Litigation Trust may retain such amounts (i) as are reasonably necessary to meet contingent liabilities and to maintain the value of the assets of the Litigation Trust during liquidation, (ii) to pay reasonable administrative expenses (including any taxes imposed on the Litigation Trust or in respect of the assets of the Litigation Trust), (iii) to satisfy other liabilities incurred or assumed by the Litigation Trust (or to which the assets are otherwise subject) in accordance with the Plan or the Litigation Trust Agreement, and (iv) as determined by the Litigation Trust Board, to fund the operations of the Litigation Trust. All such distributions to be made under the Litigation Trust Agreement shall be made to the Disbursing Agent who shall distribute them to the holders of the Secured Working Capital Lenders Claims until the Litigation Trust Funds have been

repaid and thereafter to the holders of the Litigation Trust Interests based on the number of Litigation Trust Interests held by a holder compared with the aggregate number of Litigation Trust Interests outstanding, in each case subject to the terms of the Plan and the Litigation Trust Agreement. The Litigation Trustee may withhold from amounts distributable to any Person any and all amounts, determined in the Litigation Trustee's reasonable sole discretion, to be required to be withheld by any law, regulation, rule, ruling, directive, or other governmental requirement.

11.9 **Reporting Duties**.

- (a) Federal Income Tax Treatment of the Litigation Trust.
 - Litigation Trust Assets Treated as Owned by Creditors. For all federal income tax purposes, all parties (including, without limitation, the Debtors, the Reorganized Debtors, the Litigation Trustee, and the beneficiaries of the Litigation Trust) shall treat the transfer of the Litigation Trust Assets to the Litigation Trust for the benefit of the beneficiaries of the Litigation Trust, whether their Claims are Allowed on or after the Effective Date, as (a) a transfer of the Litigation Trust Assets directly to those holders of Allowed Claims receiving Litigation Trust Interests (other than to the extent allocable to Disputed Claims), followed by (b) the transfer by such Persons to the Litigation Trust of the Litigation Trust Assets in exchange for beneficial interests in the Litigation Trust (and in respect of the Litigation Trust Assets allocable to the Disputed Claims Reserve, as a transfer to the Disputed Claims Reserve). Accordingly, those holders of Allowed Claims receiving Litigation Trust Interests shall be treated for federal income tax purposes as the grantors and owners of their respective shares of the Litigation Trust Assets. The foregoing treatment also shall apply, to the extent permitted by applicable law, for state and local income tax purposes.

(ii) Tax Reporting.

(1) The Litigation Trustee shall file returns for the Litigation Trust as a grantor trust pursuant to Treasury Regulation section 1.671-4(a) and in accordance with this Section 11.9(a)(ii). The Litigation Trustee also shall annually send to each holder of a Secured Working Capital Lender Claim or holder of a Litigation Trust Interest a separate statement setting forth the holder's share of items of income, gain, loss, deduction, or credit and will instruct all such holders to report such items on their federal income tax returns. The Litigation Trustee also shall file (or cause to be filed) any other statements, returns, or disclosures relating to the Litigation Trust that are required by any governmental unit.

(2) As soon as possible after the Effective Date, the Litigation Trustee shall make the valuation of the Litigation Trust Assets prepared by the Litigation Trust Board under Section 11.5 hereof available from time to time, to the extent relevant, and such valuation shall be used consistently by all parties (including, without limitation, the Debtors, the Reorganized Debtors, the Litigation Trustee, and the beneficiaries of the Litigation Trust) for all federal income tax purposes.

- Allocations of Litigation Trust taxable income (3) among the holders of the Secured Working Capital Lender Claims or the holders of the Litigation Trust Interests shall be determined by reference to the manner in which an amount of Cash equal to such taxable income would be distributed (without regard to any restrictions on distributions described herein) if, immediately prior to such deemed distribution, the Litigation Trust had distributed all of its other assets (valued at their tax book value) to the holders of the Secured Working Capital Lender Claims or the holders of the Litigation Trust Interests, in each case up to the tax book value of the assets treated as contributed by such holders, adjusted for prior taxable income and loss and taking into account all prior and concurrent distributions from the Litigation Trust. Similarly, taxable loss of the Litigation Trust shall be allocated by reference to the manner in which an economic loss would be borne immediately after a liquidating distribution of the remaining Litigation Trust Assets. The tax book value of the Litigation Trust Assets for this purpose shall equal their fair market value on the Effective Date, adjusted in accordance with tax accounting principles prescribed by the Tax Code, and applicable tax regulations, and other applicable administrative and judicial authorities and pronouncements.
- (4) The Litigation Trustee shall be responsible for payments, out of the Litigation Trust Assets, of any taxes imposed on the Litigation Trust or its assets.
- (5) The Litigation Trustee may request an expedited determination of taxes of the Litigation Trust under section 505(b) of the Bankruptcy Code for all returns filed for, or on behalf of, the Litigation Trust for all taxable periods through the dissolution of the Litigation Trust.
- 11.10 <u>Trust Implementation</u>. On the Effective Date, the Litigation Trust shall be established and become effective for the benefit of the holders of Allowed Senior Notes Claims, Allowed Lender Deficiency Claims, and Allowed General Unsecured Claims. The Litigation Trust Agreement shall be included in the Plan Supplement and shall contain provisions similar to those contained in trust agreements utilized in comparable circumstances, including, but not limited to, any and all provisions necessary to ensure the continued treatment of the Litigation Trust as a grantor trust for federal income tax purposes. All parties (including the Debtors or the Reorganized Debtors, as the case may be, the Litigation Trustee, and holders of Allowed Senior Notes Claims, Allowed Lender Deficiency Claims, and Allowed General Unsecured Claims) shall execute any documents or other instruments as necessary to cause title to the applicable assets to be transferred to the Litigation Trust.
- 11.11 **Registry of Beneficial Interests.** The Litigation Trustee shall maintain a registry of the holders of the Secured Working Capital Lender Claims and the holders of Litigation Trust Interests. The Litigation Trust Interests may not be transferred or assigned, except by operation of law or by will or the laws of descent and distribution.
- 11.12 <u>Termination</u>. The Litigation Trust shall terminate no later than the fifth (5th) anniversary of the Effective Date; <u>provided</u>, <u>however</u>, that, on or prior to the date that is ninety (90) days prior to such termination, the Bankruptcy Court, upon motion by a party in interest, may extend the term of the Litigation Trust if it is necessary to the liquidation of the Litigation Trust Assets. Notwithstanding the foregoing, multiple extensions can be obtained so long as

Bankruptcy Court approval is obtained at least 90 days prior to the expiration of each extended term; <u>provided</u>, <u>however</u>, that in no event shall the term of the Litigation Trust extend past the tenth anniversary of the Effective Date.

11.13 Net Litigation Trust Recovery. Notwithstanding anything contained herein to the contrary, in the event that a defendant in a litigation brought by the Litigation Trustee for and on behalf of the Litigation Trust (i) is required by a Final Order to make payment to the Litigation Trust (the "Judgment Amount") and (ii) is permitted by a Final Order to assert a right of setoff under sections 553, 555, 556, 559, 560, and 561 of the Bankruptcy Code or applicable non-bankruptcy law against the Judgment Amount (a "Valid Setoff"), (y) such defendant shall be obligated to pay only the excess, if any, of the amount of the Judgment Amount over the Valid Setoff and (z) none of the Litigation Trust or the holders or beneficiaries of the Litigation Trust Interests shall be entitled to assert a claim against the Debtors or the Reorganized Debtors with respect to the Valid Setoff.

ARTICLE XII

PROVISIONS FOR THE ESTABLISHMENT AND MAINTENANCE OF DISBURSEMENT ACCOUNT(S)

- 12.1 **Establishment of Disbursement Account(s).** On or prior to the Effective Date, the Debtors or the Reorganized Debtors, as the case may be, shall establish one or more segregated bank accounts in the name of the Disbursing Agent under the Plan, which accounts shall be trust accounts for the benefit of Creditors and holders of Claims pursuant to the Plan and utilized solely for the investment and distribution of Plan Cash and for distributions received from the Litigation Trust (in respect of the Litigation Trust Funds or otherwise) by the Disbursing Agent for further distribution to holders of Allowed Secured Working Capital Lender Claims and Litigation Trust Interests, in each case consistent with the terms and conditions of the Plan. On or prior to the Effective Date, and periodically to the extent received thereafter, the Debtors or Reorganized Debtors shall deposit into such Disbursement Account(s) all Plan Cash and Cash subsequently received and to be distributed pursuant to Section 5.4(b)(y) or 5.5(b)(y)of this Plan. The Debtors shall hold in a separate Disbursement Account any Cash reserved to pay (i) any Administrative Expense Claims (including, without duplication, Twenty-Day Claims), (ii) any Professional Compensation and Reimbursement Claims and (iii) the portion of any Priority Non-Tax Claims and Priority Tax Claims not paid on the Effective Date; the Cash held in such Disbursement Account shall not be subject to any rights of set-off by the Disbursing Agent.
- 12.2 <u>Maintenance of Disbursement Account(s)</u>. The Reorganized Debtors shall invest Cash in Disbursement Account(s) in Cash Equivalents; <u>provided</u>, <u>however</u>, that sufficient liquidity shall be maintained in such account or accounts to (a) make promptly when due all payments upon Disputed Claims if, as, and when they become Allowed Claims and (b) make promptly when due the other payments provided for in the Plan.
- 12.3 <u>Establishment of Producer Disbursement Account(s)</u>. On or prior to the Effective Date, the Producer Representative shall establish one or more segregated bank accounts in the name of the Producer Representative under the Plan, which accounts shall be

trust accounts for the benefit of First Purchaser Producers who hold Claims pursuant to the Plan and utilized solely for the investment and distribution of funds transferred to the Producer Representative pursuant to Section 3.1 of the Plan for further distribution to holders of Allowed First Purchaser Producer Twenty-Day Claims and Allowed First Purchaser Producer Secured Claims, and the others entitled to distributions from the Producer Representative under Section 3.1 of the Plan, in each case consistent with the terms and conditions of the Plan. The Producer Representative shall hold in a separate account any Cash reserved to pay (i) any First Purchaser Producer Twenty-Day Claims and (ii) First Purchaser Producer Secured Claims not paid on the Effective Date. Notwithstanding anything to the contrary in this Plan, Section 10.4(d) shall not apply to any account established pursuant to this Section 12.3, and all income taxes payable with respect to such account shall be paid solely out of such account.

ARTICLE XIII

PROVISIONS REGARDING DISTRIBUTIONS UNDER THE PLAN

- 13.1 <u>Time and Manner of Distributions</u>. Distributions under the Plan shall be made as follows:
- (a) <u>Initial Distributions of Cash</u>. On or as soon as practicable after the Effective Date, the Disbursing Agent shall distribute, or cause to be distributed (i) to the Producer Representative as provided in Section 3.1 hereof on behalf of holders of Disputed and Allowed First Purchaser Producer Twenty-Day Claims and Disputed and Allowed Secured First Purchaser Producer Claims, (ii) to the Disputed Claims Reserve on behalf of holders of Disputed Claims, (iii) to each holder of Allowed Priority Non-Tax Claims, Allowed Secured Tax Claims, Allowed Secured Lender Claims, and Allowed Other Secured Claims and (iv) to each Settling Party, such Creditor's share, if any, of Cash as determined pursuant to the Plan.
- (b) <u>Subsequent Distributions of Cash</u>. On the first (1st) Business Day that is after the close of two (2) full calendar quarters following the date of the initial Effective Date distributions and, thereafter, on each first (1st) Business Day following the close of two (2) full calendar quarters, the Disbursing Agent shall distribute, or cause to be distributed, to the Disputed Claims Reserve on behalf of holders of Disputed Claims, and to each holder of Allowed Priority Non-Tax Claims, Allowed Secured Tax Claims, Allowed Secured Lender Claims and Allowed Other Secured Claims and to each Settling Party, an amount equal to such Creditor's share, if any, of Cash as determined pursuant to the Plan, until such time as there is no longer any potential Cash.
- (c) <u>Distributions of New Common Stock and Warrants</u>. Notwithstanding anything contained herein to the contrary, commencing on or as soon as practicable after the Effective Date, the Disbursing Agent shall commence distributions, or cause to be distributed, to the Disputed Claims Reserve on behalf of holders of Disputed Claims, and to each holder of Allowed Secured Lender Claims, Allowed Senior Notes Claims, Allowed Lender Deficiency Claims, and Allowed General Unsecured Claims, an amount equal to such Creditor's share, if any, of New Common Stock and Warrants, as determined pursuant to the Plan, and semi-annually thereafter until such time as there is no longer any potential New Common Stock and Warrants to distribute; <u>provided</u>, <u>however</u>, that during the period of retention of the New

Common Stock and the Warrants, the Disbursing Agent shall distribute, or cause to be distributed, to the Disputed Claims Reserve on behalf of holders of Disputed Claims, and to each holder of Allowed Secured Lender Claims, Allowed Senior Notes Claims, Allowed Lender Deficiency Claims, and Allowed General Unsecured Claims, an amount equal to such Creditor's share, if any, of dividends declared and distributed with respect to the New Common Stock and the Warrants; and, provided, further, that until such time as all Disputed Claims have been resolved by Final Order, in whole or in part, the Disbursing Agent shall hold in reserve at least one percent (1%) of the New Common Stock and the Warrants to be distributed in accordance with Article IV of the Plan. The Class A New Common Stock and the Class B New Common Stock shall be identical in all respects, except that the Class B New Common Stock shall not be eligible for trading on a national securities exchange or a national market system. The Class B New Common Stock shall convert automatically into Class A New Common Stock upon the transfer of such stock to an Entity which is permitted to hold margin securities or at the request of the holder.

- (d) <u>Distributions of Second Lien Term Loan Interests</u>. The Disbursing Agent shall distribute to each holder of an Allowed Secured Lender Claim, such Creditor's share, if any, of Second Lien Term Loan Interests as determined pursuant to Article IV hereof.
- (e) <u>Distributions of the Litigation Trust Interests</u>. The Disbursing Agent shall commence distributions, or cause to be distributed, to the Disputed Claims Reserve on behalf of holders of Disputed Claims, and to each holder of Allowed Senior Notes Claims, Allowed Lender Deficiency Claims, and Allowed General Unsecured Claims, such Creditor's share, if any, of Litigation Trust Interests as determined pursuant to Article IV hereof, and semi-annually thereafter until such time as there are no longer any Litigation Trust Interests to distribute. All Litigation Trust Interests shall be deemed to have been issued as of the Effective Date, whether or not held in reserve.
- all distributions to be made under the Litigation Trust to the Prepetition Administrative Agent to repay the Litigation Trust Funds to the holders of Allowed Secured Working Capital Lender Claims and/or in respect of the Litigation Trust Interests to the Disbursing Agent. Promptly after receipt of any such distribution (x) in respect of the repayment of the Litigation Trust Funds, the Prepetition Administrative Agent shall distribute the same to the holders of Allowed Secured Working Capital Lender Claims (or to the Disputed Claims Reserve on behalf of holders of Disputed Claims that are Secured Working Capital Lender Claims) and (y) in respect of distributions to holders of the Litigation Trust Interests, the Disbursing Agent shall distribute the same to the holders of the Litigation Trust Interests, in each case in accordance with this Plan; provided, however, that any Prepetition Lender or holder of a Swap Claim who did not vote to accept the Plan shall not be entitled to receive any distributions until it has executed a Contributing Lender Assignment.
- 13.2 <u>Timeliness of Payments</u>. Any payments or distributions to be made pursuant to the Plan shall be deemed to be made timely if made within thirty (30) days after the dates specified in the Plan. Whenever any distribution to be made under this Plan shall be due on a day other than a Business Day, such distribution shall instead be made, without interest, on the

immediately succeeding Business Day, and shall not be deemed to have been made on the date due.

- 13.3 <u>Distributions by the Disbursing Agent</u>. Except as provided in Section 13.1(f), all distributions under the Plan shall be made by the Disbursing Agent. All distributions made to the holders of the Secured Lender Claims shall be paid by the Disbursing Agent to the Prepetition Administrative Agent which shall make all distributions to the holders of the Secured Lender Claims. All distributions made to the Producers (other than distributions with respect to Unsecured First Purchaser Producer Claims which are provided for in Section 5.10) shall be paid by the Disbursing Agent to the Producer Representative who shall make all distributions (other than distributions with respect to Unsecured First Purchaser Producer Claims which are provided for in Section 5.10) to the Producers. All distributions made to the holders of the Senior Notes Claims shall be paid by the Disbursing Agent to the Indenture Trustee which shall make all distributions to the holders of the Senior Notes Claims. The Disbursing Agent and, as applicable, the Producer Representative, shall be deemed to hold all property to be distributed hereunder in trust for the Persons entitled to receive the same. The Disbursing Agent, and as applicable, the Producer Representative, shall not hold an economic or beneficial interest in such property.
- 13.4 <u>Manner of Payment under the Plan</u>. Unless the Entity receiving a payment agrees otherwise, any payment in Cash to be made by the Disbursing Agent shall be made, at the election of the Disbursing Agent, by check drawn on a domestic bank or by wire transfer from a domestic bank; <u>provided</u>, <u>however</u>, that no Cash payments shall be made to a holder of an Allowed Claim until such time as the amount payable thereto is equal to or greater than Ten Dollars (\$10.00); <u>provided</u>, <u>further</u>, that all payments in Cash to the Prepetition Administrative Agent shall be made by wire transfer.
- 13.5 <u>Delivery of Distributions</u>. Subject to the provisions of Bankruptcy Rule 9010, distributions and deliveries to holders of Allowed Claims shall be made at the address of such holders as set forth on the Schedules filed with the Bankruptcy Court unless superseded by the address set forth on proofs of claim filed by such holders, or at the last known address of such holders if no proof of claim is filed or if the Debtors have been notified in writing of a change of address.
- 13.6 <u>Fractional New Common Stock / Warrants</u>. No fractional shares of New Common Stock or Warrants shall be issued. Fractional shares of New Common Stock and Warrants shall be rounded to the next greater or next lower number of shares in accordance with the following method: (a) fractions of one-half (1/2) or greater shall be rounded to the next higher whole number and (b) fractions of less than one-half (1/2) shall be rounded to the next lower whole number. The total number of shares or interests of New Common Stock and Warrants to be distributed to a Class hereunder shall be adjusted as necessary to account for the rounding provided for in this Section 13.6.

13.7 <u>Undeliverable Distributions</u>.

(a) <u>Holding of Undeliverable Distributions</u>. If any distribution to any holder is returned to the Reorganized Debtors or, as applicable, the Prepetition Administrative Agent or

the Producer Representative, as undeliverable, no further distributions shall be made to such holder unless and until the Reorganized Debtors or, as applicable, the Prepetition Administrative Agent or the Producer Representative, are notified, in writing, of such holder's then-current address. Undeliverable distributions shall remain in the possession of the Reorganized Debtors until such time as a distribution becomes deliverable. All Entities ultimately receiving undeliverable Cash shall not be entitled to any interest or other accruals of any kind. Nothing contained in the Plan shall require the Reorganized Debtors or, as applicable, the Prepetition Administrative Agent or the Producer Representative, to attempt to locate any holder of an Allowed Claim.

- (b) <u>Failure to Claim Undeliverable Distributions</u>. Any holder of an Allowed Claim that does not assert its rights pursuant to the Plan to receive a distribution within five (5) years from and after the Effective Date shall have its entitlement to such undeliverable distribution discharged and shall be forever barred from asserting any entitlement pursuant to the Plan against the Reorganized Debtors or, as applicable, the Prepetition Administrative Agent or the Producer Representative, or their respective property. In such case, any consideration held for distribution on account of such Claim shall revert to the Reorganized Debtors.
- 13.8 Time Bar to Cash Payments. Checks issued by the Disbursing Agent or, as applicable, the Prepetition Administrative Agent or the Producer Representative, on account of Allowed Claims shall be null and void if not negotiated within 180 days from and after the date of issuance thereof. Requests for re-issuance of any check shall be made directly to the Disbursing Agent or, as applicable, the Prepetition Administrative Agent or the Producer Representative, by the holder of the Allowed Claim with respect to which such check originally was issued. Any claim in respect of such a voided check shall be made on or before the later of (a) the fifth (5th) anniversary of the Effective Date or (b) 180 days after the date of issuance of such check, if such check represents a final distribution hereunder on account of such Claim. After such date, all Claims in respect of voided checks shall be discharged and forever barred and the Reorganized Debtors or the Producer Representative, as applicable, shall retain all monies related thereto.
- 13.9 <u>Distributions After Effective Date</u>. Distributions made after the Effective Date to holders of Claims that are not Allowed Claims as of the Effective Date, but which later become Allowed Claims, shall be deemed to have been made in accordance with the terms and provisions of Section 13.1 of the Plan.
- Postpetition Financing Claims, and the Senior Notes Claims (as to which any and all rights of setoff or recoupment have been waived), the Reorganized Debtors may, pursuant to applicable non-bankruptcy law, set off against any Allowed Claim and the distributions to be made pursuant to the Plan on account thereof (before any distribution is made on account of such Claim), the claims, rights, and causes of action of any nature the Debtors or the Reorganized Debtors may hold against the holder of such Allowed Claim; provided, however, that neither the failure to effect such a setoff nor the allowance of any Claim hereunder shall constitute a waiver or release by the Debtors, the Debtors in Possession, the Reorganized Debtors, or the Litigation Trustee of any such claims, rights, and causes of action that the Debtors, the Debtors in Possession, or the Reorganized Debtors or the Litigation Trustee may possess against such holder, and; provided,

<u>further</u>, that, unless otherwise provided by order of the Bankruptcy Court, nothing contained in the Plan is intended to limit the ability of any Creditor to effectuate rights of setoff or recoupment preserved or permitted by the provisions of sections 553, 555, 556, 559, 560, or 561 of the Bankruptcy Code or pursuant to the common law right of recoupment.

- 13.11 Allocation of Plan Distributions Between Principal and Interest. To the extent that any Allowed Claim entitled to a distribution under the Plan is comprised of indebtedness and accrued but unpaid interest thereon, such distribution shall be allocated first to the principal amount of the Claim (as determined for federal income tax purposes) and then, to the extent the consideration exceeds the principal amount of the Claim, to accrued but unpaid interest.
- 13.12 **Record Date.** With respect to all holders of Claims against the Debtors in Classes 1 through 148 and Classes 175 through 226, the Record Date shall be the date for the purposes of determining the holders of Allowed Claims entitled to receive distributions pursuant to the Plan and the claims register shall be closed on the Record Date. The Debtors and the Reorganized Debtors shall have no obligation to recognize any transfer of any Claims occurring after 5:00 p.m. (prevailing Eastern Time) on the Record Date with respect to such Classes.
- 13.13 <u>Senior Notes Indenture Trustee as Claim Holder</u>. Consistent with Bankruptcy Rule 3003(c), the Debtors shall recognize the master proof of claim (Claim No. 2614) filed by the Senior Notes Indenture Trustee in respect of the Senior Notes Claims. Accordingly, any proof of claim filed by a holder of a Senior Notes Claim may be disallowed as duplicative of the Senior Notes Indenture Trustee master proof of claim, without further action or Bankruptcy Court order.
- 13.14 <u>Limited Recoveries</u>. Notwithstanding anything contained herein to the contrary, in the event that the sum of distributions from Plan Currency, including distributions from Litigation Trust Interests and distributions under the Canadian Plans, are equal to or in excess of one hundred percent (100%) of any holder's Allowed Claim, then distributions from the Litigation Trust to be distributed to such holder in excess of such one hundred percent (100%) shall be deemed redistributed to holders of Litigation Trust Interests for and on behalf of holders of other Litigation Trust Interests and accordingly shall be distributed in accordance with the provisions of the documents, instruments and agreements governing such Litigation Trust Interests and the Bankruptcy Code.

ARTICLE XIV

RIGHTS AND POWERS OF DISBURSING AGENT

14.1 **Powers of the Disbursing Agent.** The Disbursing Agent shall be empowered to (a) take all steps and execute all instruments and documents necessary to effectuate the Plan, (b) make distributions contemplated by the Plan, (c) comply with the Plan and the obligations hereunder, (d) file all tax returns and pay taxes in connection with the reserves created pursuant to the Plan, and (e) exercise such other powers as may be vested in the Disbursing Agent pursuant to order of the Bankruptcy Court, pursuant to the Plan, or as deemed by the Disbursing Agent to be necessary and proper to implement the provisions of the Plan.

- 14.2 <u>Fees and Expenses Incurred From and After the Effective Date</u>. Except as otherwise ordered by the Bankruptcy Court, the amount of any reasonable fees and expenses incurred by the Disbursing Agent from and after the Effective Date, including, without limitation, reasonable fees and expenses of counsel, shall be paid in Cash by the Reorganized Debtors without further order of the Bankruptcy Court within twenty (20) days of receipt of an invoice by the Reorganized Debtors. In the event that the Reorganized Debtors object to the payment of such invoice for post-Effective Date fees and expenses, in whole or in part, and the parties cannot resolve such objection after good faith negotiation, the Bankruptcy Court shall retain jurisdiction to make a determination as to the extent to which the invoice shall be paid by the Reorganized Debtors.
- 14.3 <u>Exculpation</u>. From and after the Effective Date, the Disbursing Agent shall be exculpated by all Persons and Entities, including, without limitation, holders of Claims and Equity Interests and other parties in interest, from any and all Claims, Causes of Action, and other assertions of liability arising out of the discharge of the powers and duties conferred upon such Disbursing Agent by the Plan or any order of the Bankruptcy Court entered pursuant to or in furtherance of the Plan, or applicable law, except for actions or omissions to act arising out of the gross negligence or willful misconduct of such Disbursing Agent. No holder of a Claim or an Equity Interest or other party in interest shall have or pursue any Claim or Cause of Action against the Disbursing Agent for making payments in accordance with the Plan or for implementing the provisions of the Plan.

ARTICLE XV

RIGHTS AND POWERS OF PRODUCER REPRESENTATIVE AND THE PREPETITION ADMINISTRATIVE AGENT

- Agent. The Producer Representative and the Prepetition Administrative Agent shall be empowered to (a) take all steps and execute all instruments and documents necessary to effectuate the Plan, (b) make distributions contemplated by the Plan, (c) comply with the Plan and the obligations hereunder, (d) file all tax returns and pay taxes in connection with the reserves created pursuant to the Plan, and (e) exercise such other powers as may be vested in the Producer Representative or the Prepetition Administrative Agent pursuant to order of the Bankruptcy Court, pursuant to the Plan, or as deemed by the Producer Representative to be necessary and proper to implement the provisions of the Plan.
- 15.2 **Exculpation.** From and after the Effective Date, the Producer Representative and the Prepetition Administrative Agent shall be exculpated by all Persons and Entities, including, without limitation, holders of Claims and Equity Interests and other parties in interest, from any and all Claims, Causes of Action, and other assertions of liability arising out of the discharge of the powers and duties conferred upon the Producer Representative or the Prepetition Administrative Agent by the Plan or any order of the Bankruptcy Court entered pursuant to or in furtherance of the Plan, or applicable law, except for actions or omissions to act arising out of the gross negligence or willful misconduct of the Producer Representative or the Prepetition Administrative Agent. No holder of a Claim or an Equity Interest or other party in interest shall have or pursue any Claim or Cause of Action against the Producer Representative or the

Prepetition Administrative Agent for making payments in accordance with the Plan or for implementing the provisions of the Plan.

ARTICLE XVI

EXECUTORY CONTRACTS AND UNEXPIRED LEASES

- Assumption or Rejection of Executory Contracts and Unexpired Leases. Pursuant to sections 365(a) and 1123(b)(2) of the Bankruptcy Code, all executory contracts and unexpired leases that exist between the Debtors and any Person or Entity shall be deemed rejected by the Debtors, as of the Effective Date, except for any executory contract or unexpired lease (i) that has been assumed pursuant to an order of the Bankruptcy Court entered prior to the Effective Date and for which the motion was filed prior to the Confirmation Date, (ii) as to which a motion for approval of the assumption of such executory contract or unexpired lease has been filed and served prior to the Confirmation Date, or (iii) that is specifically designated as a contract or lease to be assumed on Schedule 1(A) (executory contracts) or Schedule 1(B) (unexpired leases), which Schedules shall be contained in the Plan Supplement; provided, however, that the Debtors reserve the right, on or prior to the Confirmation Date, to amend Schedules 1(A) and 1(B) to delete any executory contract or unexpired lease therefrom or add any executory contract or unexpired lease thereto, in which event such executory contract(s) or unexpired lease(s) shall be deemed to be, respectively, assumed or rejected. The Debtors shall provide notice of any amendments to Schedules 1(A) and 1(B) to the parties to the executory contracts and unexpired leases affected thereby. The listing of a document on Schedule 1(A) or 1(B) shall not constitute an admission by the Debtors that such document is an executory contract or an unexpired lease or that the Debtors have any liability thereunder.
- Leases. Entry of the Confirmation Order shall, subject to and upon the occurrence of the Effective Date, constitute (i) the approval, pursuant to sections 365(a) and 1123(b)(2) of the Bankruptcy Code, of the rejection of the executory contracts and unexpired leases rejected pursuant to Section 16.1 of this Plan, (ii) the extension of time, pursuant to section 365(d)(4) of the Bankruptcy Code, within which the Debtors may assume, assume and assign, or reject the unexpired leases specified in Section 16.1 of this Plan through the date of entry of an order approving the assumption, assumption and assignment, or rejection of such unexpired leases, and (iii) the approval, pursuant to sections 365(a) and 1123(b)(2) of the Bankruptcy Code, of the assumption of the executory contracts and unexpired leases assumed pursuant to Section 16.1 of this Plan.
- 16.3 <u>Inclusiveness</u>. Unless otherwise specified on Schedules 1(A) and 1(B), each executory contract and unexpired lease listed or to be listed on Schedules 1(A) and 1(B) shall include modifications, amendments, supplements, restatements, or other agreements made directly or indirectly by any agreement, instrument, or other document that in any manner affects such executory contract or unexpired lease, without regard to whether such agreement, instrument, or other document is listed on Schedules 1(A) and 1(B).
- 16.4 <u>Cure of Defaults for Assumed Executory Contracts and Unexpired Leases.</u> Except as may otherwise be agreed to by the parties, within thirty (30) days after the Effective

Date, the Reorganized Debtors shall cure any and all undisputed defaults under any executory contract or unexpired lease assumed by the Debtors pursuant to the Plan, in accordance with section 365(b) of the Bankruptcy Code. All disputed defaults that are required to be cured shall be cured either within thirty (30) days of the entry of a Final Order determining the amount, if any, of the Reorganized Debtors' liability with respect thereto, or as otherwise may be agreed to by the parties.

- 16.5 Bar Date for Filing Proofs of Claim Relating to Executory Contracts and Unexpired Leases Rejected Pursuant to the Plan. Claims arising out of the rejection of an executory contract or unexpired lease pursuant to Section 16.1 of the Plan must be filed with the Bankruptcy Court and served upon the Debtors (or, on and after the Effective Date, Reorganized Debtors) no later than thirty (30) days after the later of (i) notice of entry of an order approving the rejection of such executory contract or unexpired lease, (ii) notice of entry of the Confirmation Order, and (iii) notice of an amendment to Schedule 1(A) or 1(B). All such Claims not filed within such time will be forever barred from assertion against the Debtors and their estates or the Reorganized Debtors and their property.
- 16.6 <u>Insurance Policies</u>. Notwithstanding anything contained in the Plan to the contrary, unless subject to a motion for approval or rejection that has been filed and served prior to the Confirmation Date, all of the Debtors' insurance policies and any agreements, documents or instruments relating thereto, shall be treated as executory contracts under the Plan and shall be assumed pursuant to the Plan, effective as of the Effective Date. Nothing contained in this Section 16.6 shall constitute or be deemed a waiver of any Cause of Action that the Debtors may hold against any Entity, including, without limitation, the insurer, under any of the Debtors' insurance policies.

ARTICLE XVII

COMMITTEES, EXAMINER, AND FEE AUDITOR

Dissolution of Creditors' Committee and Producers' Committee. On the Effective Date, the Creditors' Committee and the Producers' Committee shall be dissolved and the members thereof and the professionals retained by the Creditors' Committee and the Producers' Committee in accordance with section 1103 of the Bankruptcy Code shall be released and discharged from their respective fiduciary obligations; provided, however, that the Creditors' Committee shall not be dissolved for the following limited purposes: (i) seeking approval of any application of a Professional Compensation and Reimbursement Claim; (ii) objecting to any application of a Professional Compensation and Reimbursement Claim; and (iii) any appeals of the Confirmation Order; provided, further, that the Producers' Committee shall not be dissolved for the limited purpose of prosecuting any application of a Professional Compensation and Reimbursement Claim submitted by the Producers' Committee. All reasonable fees and expenses of members of the Lender Steering Committee, the reasonable fees and expenses of the Prepetition Administrative Agent's professionals for post-Effective Date activities referred to in clauses (i), (ii) and (iii) of the preceding sentence, the reasonable fees and expenses of members of the Creditors' Committee, the reasonable fees and expenses of the Creditors' Committee's professionals for post-Effective Date activities authorized under this Section 17.1 and the reasonable fees and expenses of the Producers' Committee's professionals for post-Effective

Date activities authorized under this Section 17.1 shall be paid without further Bankruptcy Court approval upon the submission of invoices to the Reorganized Debtors. Following the Effective Date, none of the Creditors' Committee's professionals shall be precluded from representing any Entity acting for the Litigation Trust or other Entities created by this Plan, including, without limitation, the Litigation Trustee or the Litigation Trust.

- **Examiner.** To the extent not discharged and released on or prior to the Confirmation Date, on the tenth (10th) day following the Confirmation Date, the Examiner and the professionals retained by the Examiner shall be released and discharged from their respective obligations outstanding pursuant to the Investigative Order. On or prior to the thirtieth (30th) day following the Confirmation Date, or such other date as shall be agreed upon, in writing, by the Examiner and the Debtors or the Reorganized Debtors, as the case may be, and except as (w) otherwise provided in Bankruptcy Court orders relating to the Termination Procedures Motion, (x) previously delivered, (y) otherwise available on a centralized, coded filing system available to the Debtors or (z) prohibited by any existing confidentiality order entered by the Bankruptcy Court or other confidentiality agreement executed by the Examiner, the Examiner shall deliver to the Reorganized Debtors and the Litigation Trustee (i) one copy of the report filed by the Examiner in the Chapter 11 Cases, (ii) all material cited in the footnotes of the report, (iii) any other materials and work product, including, without limitation, transcripts, interview memoranda, witness folders, expert reports and analyses, and transactional documents and summaries thereof, produced, developed, or compiled by the Examiner and his consultants and experts in connection with the Investigative Order, and (iv) a schedule of all materials which the Examiner is, or claims to be, precluded from delivering to the Debtors or the Litigation Trustee, in each case in connection with the Investigative Order. Notwithstanding the foregoing, nothing contained in the Plan is intended, nor shall it be deemed, to modify, affect, or amend the terms or provisions of Bankruptcy Court orders relating to the appointment of the Examiner or the Termination Procedures Motion.
- 17.3 <u>Fee Auditor</u>. To the extent not discharged and released on or prior to the Confirmation Date, on the tenth (10th) day following the entry of a Final Order in respect of the last of any outstanding fee applications, the Fee Auditor shall be released and discharged from its obligations outstanding pursuant to the Fee Auditor Order. The amount of any reasonable fees and expenses of the Fee Auditor incurred after the Effective Date shall be paid in Cash by the Reorganized Debtors without further order of the Bankruptcy Court within thirty (30) days of receipt of an invoice by the Reorganized Debtors. In the event that the Reorganized Debtors object to payment of such invoice for post-Effective Date fees and expenses, in whole or in part, and the parties cannot resolve such objection after good faith negotiation, the Bankruptcy Court shall retain jurisdiction to make a determination as to the extent to which the invoice shall be paid by the Reorganized Debtors.

ARTICLE XVIII

CONDITIONS PRECEDENT TO EFFECTIVE DATE OF THE PLAN; IMPLEMENTATION PROVISIONS

- 18.1 <u>Conditions Precedent to Effective Date of the Plan</u>. The occurrence of the Effective Date and the substantial consummation of the Plan are subject to satisfaction of the following conditions precedent:
- (a) <u>Entry of the Confirmation Order</u>. The Clerk of the Bankruptcy Court shall have entered the Confirmation Order in form and substance acceptable to the Debtors, the Lender Steering Committee, and the Creditors' Committee and the effectiveness of which shall not have been stayed ten (10) days following the entry thereof.
- (b) Producers' Settlement Payments. (i) The sum of (A) the total payments with respect to the First Purchaser Producer Twenty-Day Claims <u>plus</u> (B) the total payments with respect to the Secured First Purchaser Producer Claims (including payments pursuant to Sections 3.1(c)(i), (ii), (iii) and (iv) of this Plan) shall not exceed \$172.5 million in the aggregate. (ii) The sum of (A) the total payments with respect to the Other Twenty-Day Claims paid by the Debtors to the Settling Parties under the Other Twenty-Day Claims Settlement, or otherwise, <u>plus</u> (B) the total reserve funded by the Debtors with respect to Other Twenty-Day Claims held by Non-Settling Parties shall not exceed \$154 million in the aggregate. (iii) The sum of (A) the amount of the total payments referred to in Section 18.1(b)(i) hereof <u>plus</u> (B) the amount of the total payments and reserves referred to in Section 18.1(b)(ii) hereof <u>plus</u> (C) the payment with respect to the Alon Claim shall not exceed \$337 million in the aggregate.
- (c) Adversary Proceedings. (i) The Producer Plaintiffs shall have voluntarily dismissed with prejudice the Producer State-Specific Adversary Proceedings with respect to the Debtors, the Prepetition Administrative Agent and the Prepetition Lenders by no later than the Effective Date. (ii) The parties to the Producer State-Specific Adversary Proceedings shall have filed papers, in a form acceptable to the Debtors and the Prepetition Administrative Agent, voluntarily dismissing with prejudice the Third Circuit Appeals by no later than the Effective Date. (iii) Each of the Producers named as plaintiffs in the Producer State-Specific Adversary Proceedings shall have voluntarily dismissed with prejudice the Other Proceedings to which it is a party with respect to the Debtors, the Prepetition Administrative Agent and the Prepetition Lenders and/or their affiliates, as applicable, by no later than the Effective Date.
- (d) <u>Twenty-Day Claims</u>. The Debtors and the Prepetition Administrative Agent shall have voluntarily dismissed with prejudice all fact-based objections to First Purchaser Producer Twenty-Day Claims by no later than the Effective Date of the Plan; <u>provided</u>, <u>however</u>, that the Producers' Committee or the Producer Representative is permitted to pursue the disputes regarding certain First Purchaser Producer Twenty-Day Claims that were set forth in the Notices filed by the Debtors on July 17, 2009.
- (e) <u>Other Producer-Related Litigation</u>. Other than the legal and fact-based objections to the Other Twenty-Day Claims, all Producer-related litigation commenced in the Bankruptcy Court and to which the Debtors or the Prepetition Administrative Agent are currently

parties shall have been voluntarily dismissed with prejudice as to the Debtors, the Prepetition Administrative Agent, and the Prepetition Lenders.

- (f) <u>Release of Restricted Cash</u>. The Clerk of the Bankruptcy Court shall have entered an order, in form and substance acceptable to counsel for the Prepetition Administrative Agent and counsel for the Producers' Committee, authorizing the release from escrow of the Restricted Cash for use to fund the Plan.
- (g) <u>Execution of Documents; Other Actions</u>. All other actions and documents necessary to implement the Plan shall have been effected or executed and shall be reasonably acceptable to the Lender Steering Committee and the Creditors' Committee.
- (h) <u>Consents Obtained</u>. The Debtors shall have received all authorizations, consents, regulatory approvals, rulings, letters, no-action letters, opinions, or documents that are necessary to implement the Plan and that are required by law, regulation, or order.
- (i) <u>Exit Facility</u>. An Exit Facility, the terms and conditions of which shall be reasonably satisfactory to the Debtors, the Lender Steering Committee, and the Creditors' Committee, shall have been provided.
- (j) <u>Approval and Consummation of Certain Canadian Plans</u>. The SemCAMS ULC Plan and the SemCanada Nova Scotia Plan shall have been approved and consummated concurrently with the effectiveness of the Plan and the aggregate amount of Canadian Distributions made on or prior to the Effective Date shall have been no less than \$160 million.
- (k) <u>Litigation Trust</u>. The Litigation Trust Agreement shall have been executed and all steps necessary to establish the Litigation Trust in accordance with and pursuant to the terms of the Plan shall have occurred in a manner reasonably satisfactory to the Lender Steering Committee and the Creditors' Committee.
- (l) <u>New Securities</u>. The New Common Stock and Warrants shall have been authorized in the amounts set forth in the Plan and on terms reasonably satisfactory to the Lender Steering Committee and the Creditors' Committee.
- 18.2 Failure of Conditions Precedent. In the event that one or more of the conditions specified in Section 18.1 hereof have not occurred or been waived as permitted by Section 18.3 hereof on or before November 15, 2009, (i) the Confirmation Order shall be vacated, (ii) no distributions under the Plan shall be made, (iii) the Debtors and all holders of Claims and Equity Interests shall be restored to the status quo ante as of the day immediately preceding the Confirmation Date as though the Confirmation Date never occurred, and (iv) the Debtors' obligations with respect to Claims and Equity Interests shall remain unchanged and nothing contained herein shall constitute or be deemed to be a waiver or release of any Claims or Equity Interests by or against the Debtors or any other person or to prejudice in any manner the rights of the Debtors or any person in any further proceedings involving the Debtors. For the avoidance of doubt, and notwithstanding anything in the Disclosure Statement or the Plan to the contrary, if the Plan is not confirmed or does not become effective, nothing in the Plan or the Disclosure Statement shall be construed as a waiver of any rights or Claims of the Debtors, the Lender

Steering Committee, the Creditors' Committee, the Producers' Committee or the Producer Plaintiffs.

18.3 <u>Waiver of Conditions Precedent</u>. The Debtors, subject to receipt of consent of the Lender Steering Committee and the Creditors' Committee, which consent shall not be unreasonably withheld, and to the extent not prohibited by applicable law, may waive one (1) or more of the conditions precedent to effectiveness of the Plan set forth in Section 18.1 hereof.

ARTICLE XIX

PROVISIONS REGARDING CORPORATE GOVERNANCE AND MANAGEMENT OF THE REORGANIZED DEBTORS AND NEW HOLDCO

19.1 <u>General</u>. On the Effective Date, the management, control, and operation of the Reorganized Debtors and New Holdco shall become the general responsibility of the Board.

19.2 Directors and Officers of New Holdco.

- (a) New Holdco Board. The initial Board shall be selected by a committee comprised of members of the Lender Steering Committee and the Creditors' Committee and shall be disclosed not later than ten (10) days prior to the Confirmation Hearing. The Board shall consist of seven (7) directors, composed of (i) the Chief Executive Officer of New Holdco, (ii) five (5) directors nominated by the Lender Steering Committee, and (iii) one (1) director nominated by the Creditors' Committee. A majority of the members of the Board shall be independent directors with knowledge of or experience in the Reorganized SemGroup Companies' industry. Each of the members of the initial Board shall serve in accordance with applicable nonbankruptcy law and the New Holdco Certificate of Incorporation and New Holdco Bylaws, as the same may be amended from time to time.
- (b) New Holdco Officers. The identity of the Persons who will serve as executive officers of New Holdco will be disclosed in the Plan Supplement if known by the date the Plan Supplement is filed.
- 19.3 <u>Issuance of New Securities</u>. The issuance of the following securities by New Holdco is hereby authorized without further act or action under applicable law, regulation, order, or rule:
 - (a) 41,400,000 shares of New Common Stock;
 - (b) 2,178,947 Warrants; and
 - (c) the Management Stock.
- 19.4 <u>Listing of New Common Stock.</u> New Holdco shall use commercially reasonable efforts to cause the shares of the Class A New Common Stock to be listed on a national securities exchange or market system for trading on or as soon as practicable after the Effective Date.

19.5 <u>Management Incentive Plan</u>. New Holdco shall, on the Effective Date, adopt the Management Incentive Plan for certain of its employees and Board members, pursuant to which such employees and Board members shall be eligible to receive Management Stock. The terms of the Management Incentive Plan shall be contained in the Plan Supplement.

ARTICLE XX

EFFECT OF CONFIRMATION

- 20.1 <u>Title to and Vesting of Assets</u>. On the Effective Date, pursuant to sections 1141(b) and (c) of the Bankruptcy Code, all property of the estates of the Debtors shall vest in the Reorganized Debtors or the Litigation Trust, as the case may be, free and clear of all Claims, Liens, encumbrances, and other interests, except as provided herein, and the Confirmation Order shall be a judicial determination of discharge of the liabilities of the Debtors and the Debtors in Possession except as provided in the Plan. From and after the Effective Date, the Reorganized Debtors may operate their businesses and may use, acquire, and dispose of property free and clear of any restrictions of the Bankruptcy Code or the Bankruptcy Rules and in all respects as if there were no pending cases under any chapter or provision of the Bankruptcy Code, except as provided herein.
- 20.2 **Discharge of Claims and Termination of Equity Interests.** Except as otherwise provided in the Plan or in the Confirmation Order, the rights afforded in the Plan and the payments and distributions to be made hereunder shall be in exchange for and in complete satisfaction and discharge of all existing debts and Claims other than the SemCanada Energy Claim, and shall terminate all Equity Interests, of any kind, nature, or description whatsoever, including any interest accrued on such Claims from and after the Petition Date, against or in the Debtors or any of their assets or properties to the fullest extent permitted by section 1141 of the Bankruptcy Code. Except as provided in the Plan, on the Effective Date, all existing Claims against the Debtors other than the SemCanada Energy Claim and Equity Interests in the Debtors, shall be, and shall be deemed to be satisfied and discharged, and all holders of Claims other than the SemCanada Energy Claim and Equity Interests shall be precluded and enjoined from asserting against the Reorganized Debtors or the Litigation Trust, or any of their respective assets or properties, any other or further Claim or Equity Interest based upon any act or omission, transaction, or other activity of any kind or nature that occurred prior to the Effective Date, whether or not such holder has filed a proof of Claim or proof of Equity Interest.
- Distributions to be made hereunder, except as otherwise expressly provided herein, the Debtors shall be discharged of all Claims, Equity Interests, rights, and liabilities that arose prior to the Effective Date, to the fullest extent permitted by section 1141 of the Bankruptcy Code, other than the SemCanada Energy Claim. Upon the Effective Date, all Persons and Entities shall be forever precluded and enjoined, pursuant to section 524 of the Bankruptcy Code, from asserting against the Debtors, the Debtors in Possession, the Litigation Trust, or their respective successors or assigns, including, without limitation, the Reorganized Debtors, the Litigation Trust, or their respective asset properties or interests in property, any discharged Claim or Equity Interest in the Debtors, any other or further Claims, other than the SemCanada Energy Claim, based upon any act or omission, transaction, or other activity of any kind or nature that occurred prior to the

Confirmation Date, whether or not the facts or legal bases therefore were known or existed prior to the Confirmation Date regardless of whether a proof of Claim or Equity Interest was filed, whether the holder thereof voted to accept or reject the Plan, or whether the Claim or Equity Interest is an Allowed Claim or an Allowed Equity Interest.

Injunction on Claims. Except as otherwise expressly provided in the Plan, the Confirmation Order, or such other order of the Bankruptcy Court that may be applicable, all Persons or Entities who have held, hold, or may hold Claims, or other debt or liability that is discharged or Equity Interests or other right of equity interest that is discharged pursuant to the Plan are permanently enjoined, from and after the Effective Date, from (a) commencing or continuing in any manner any action or other proceeding of any kind on any such Claim, or other debt or liability or Equity Interest or other right of equity interest that is terminated or cancelled pursuant to the Plan against the Debtors, the Debtors in Possession, or the Reorganized Debtors, the Debtors' estates, or properties or interests in properties of the Debtors, the Reorganized Debtors, the Disbursing Agent, the Prepetition Administrative Agent, the Producer Representative or the Litigation Trust, (b) the enforcement, attachment, collection, or recovery by any manner or means of any judgment, award, decree, or order against the Debtors, the Debtors in Possession, or the Reorganized Debtors, the Debtors' estates or properties, the Disbursing Agent, the Prepetition Administrative Agent, the Producer Representative, the Litigation Trust, or interests in properties of the Debtors, the Debtors in Possession, the Reorganized Debtors, or the Litigation Trust, (c) creating, perfecting, or enforcing any encumbrance of any kind against the Debtors, the Debtors in Possession, or the Reorganized Debtors, the Debtors' estates or properties, the Disbursing Agent, the Prepetition Administrative Agent, the Producer Representative, the Litigation Trust, or interests in properties of the Debtors, the Debtors in Possession, the Reorganized Debtors, or the Litigation Trust, (d) except to the extent provided, permitted, or preserved by sections 553, 555, 556, 559, 560 or 561 of the Bankruptcy Code or pursuant to the common law right of recoupment, asserting any right of setoff, subrogation, or recoupment of any kind against any obligation due from the Debtors, the Debtors in Possession, or the Reorganized Debtors, the Debtors' estates or properties, the Disbursing Agent, the Prepetition Administrative Agent, the Producer Representative, the Litigation Trust, or interests in properties of the Debtors, the Debtors in Possession, the Reorganized Debtors, or the Litigation Trust with respect to any such Claim or other debt or liability that is discharged or Equity Interest or other right of equity interest that is terminated or cancelled pursuant to the Plan, and (e) taking any actions to interfere with the implementation or consummation of the Plan; provided, however, that such injunction shall not preclude the United States of America, any State, or any of their respective police or regulatory agencies from enforcing their police or regulatory powers; and, provided, further, that except in connection with a properly filed proof of claim, the foregoing proviso does not permit the United States of America, any state, or any of their respective police or regulatory agencies from obtaining any monetary recovery from the Debtors, the Debtors in Possession, or the Reorganized Debtors, or their respective property or interests in property with respect to any such Claim or other debt or liability that is discharged or Equity Interest or other right of equity interest that is terminated or cancelled pursuant to the Plan, including, without limitation, any monetary claim or penalty in furtherance of a police or regulatory power. Such injunction shall extend to all successors of the Debtors and Debtors in Possession, including the Litigation Trust, the Creditors' Committee and its respective members, the Producers' Committee and its respective members, the Lender Steering Committee and its respective members, the Prepetition Administrative Agent, the

Postpetition Administrative Agent, the Disbursing Agent, the Producer Representative and the respective properties and interests in property of all of the foregoing; provided, however, that such injunction shall not extend to or protect members of the Creditors' Committee, the Producers' Committee, and the Lender Steering Committee, and their respective properties and interests in property for actions based upon acts outside the scope of service on the Creditors' Committee, the Producers' Committee, or the Lender Steering Committee, and is not intended, nor shall it be construed, to extend to the assertion, the commencement, or the prosecution of any claim or cause of action against any present or former member of the Creditors' Committee, Producers' Committee, or the Lender Steering Committee, and their respective properties and interests in property arising from or relating to such member's pre-Petition Date acts or omissions or any current or former directors, officers, professionals, agents, financial advisors, underwriters, lenders, members of the Management Committee or auditors relating to acts or omissions occurring prior to the Petition Date.

- 20.5 <u>Term of Existing Injunctions or Stays</u>. Unless otherwise provided in the Plan, the Confirmation Order, or a separate order of the Bankruptcy Court, all injunctions or stays arising under or entered during the Chapter 11 Cases pursuant to sections 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until the later of the Effective Date and the date indicated in such applicable order.
- **Exculpation.** None of the Debtors, the Reorganized Debtors, the Lender Steering 20.6 Committee and its members, the Prepetition Administrative Agent, the Postpetition Administrative Agent, the Creditors' Committee and its members, the Producers' Committee and its members, the Examiner (other than those functions defined by the Investigative Order), the Litigation Trustee, the Disbursing Agent, the Producer Representative and any of their respective directors, officers, employees, members, attorneys, consultants, advisors, and agents (but solely in their capacities as such), shall have or incur any liability to any holder of a Claim or Equity Interest of any other Entity for any act taken or omitted to be taken in connection with, related to, or arising out of, the Chapter 11 Cases, the formulation, preparation, dissemination, implementation, confirmation, approval, or administration of the Plan or any compromises or settlements contained therein, the Disclosure Statement related thereto, the property to be distributed under the Plan, or any contract, instrument, release, or other agreement or document provided for or contemplated in connection with the consummation of the transactions set forth in the Plan; provided, however, that the foregoing provisions of this Section 20.6 shall not affect the liability of (a) any Entity that otherwise would result from any such act or omission to the extent that such act or omission is determined in a Final Order to have constituted gross negligence or willful misconduct, including, without limitation, fraud and criminal misconduct, (b) the professionals of the Debtors, the Reorganized Debtors, the Lender Steering Committee, the Prepetition Administrative Agent, the Postpetition Administrative Agent, the Creditors' Committee, the Producers' Committee, the Examiner, the Litigation Trustee, the Producer Representative or the Disbursing Agent to their respective clients pursuant to applicable codes of professional conduct, (c) any of such Persons with respect to any act or omission prior to the Petition Date, except as otherwise expressly set forth elsewhere in the Plan or (d) any Entity with respect to their obligations pursuant to the Plan. Any of the foregoing parties in all respects shall be entitled to rely upon the advice of counsel with respect to their duties and responsibilities under the Plan.

20.7 <u>Preservation of Causes of Action / Reservation of Rights.</u>

- (a) Except with respect to Released Actions, nothing contained in the Plan or the Confirmation Order shall be deemed to be a waiver, release, or the relinquishment of any rights or Causes of Action that the Debtors, the Reorganized Debtors, or the Litigation Trust may have or which the Reorganized Debtors or the Litigation Trust may choose to assert on behalf of their respective estates under any provision of the Bankruptcy Code or any applicable nonbankruptcy law, including, without limitation, (i) any and all Claims against any Person or Entity, to the extent such Person or Entity asserts a crossclaim, counterclaim, and/or Claim for setoff which seeks affirmative relief against the Debtors, the Reorganized Debtors, their officers, directors, or representatives, (ii) the turnover of any property of the Debtors' estates, and (iii) Causes of Action against current or former directors, officers, professionals, agents, financial advisors, underwriters, lenders, members of the Management Committee, or auditors relating to acts or omissions occurring prior to the Petition Date.
- (b) Except with respect to Released Actions, nothing contained in the Plan or the Confirmation Order shall be deemed to be a waiver, release, or relinquishment of any Cause of Action, right of setoff, or other legal or equitable defense which the Debtors had immediately prior to the Petition Date, against or with respect to any Claim left unimpaired by the Plan. The Reorganized Debtors or the Litigation Trust, as the case may be, shall have, retain, reserve, and be entitled to assert all such Claims, Causes of Action, rights of setoff, and other legal or equitable defenses which they had immediately prior to the Petition Date fully as if the Chapter 11 Cases had not been commenced or the Litigation Trust Claims had not been transferred to the Litigation Trust in accordance with the Plan, and all of the Reorganized Debtors' legal and equitable rights respecting any Claim left unimpaired by the Plan may be asserted after the Confirmation Date to the same extent as if the Chapter 11 Cases had not been commenced.
- 20.8 <u>Injunction on Causes of Action</u>. Except as provided in the Plan, as of the Effective Date, all non-Debtor entities are permanently enjoined from commencing or continuing in any manner, any Causes of Action, whether directly, derivatively, on account of or respecting any debt or Cause of Action (i) of the Debtors, the Debtors in Possession, or the Reorganized Debtors which the Debtors, the Debtors in Possession, the Reorganized Debtors, or the Litigation Trust, as the case may be, retain sole and exclusive authority to pursue in accordance with Section 10.1 of the Plan or (ii) which has been released pursuant to the Plan, including, without limitation, pursuant to Sections 20.9, 20.10, 20.11, 20.12, 20.13 or 20.14 of the Plan and the Confirmation Order shall provide for such injunctions.
- 20.9 <u>Limited Release of Officers and Employees</u>. No claims of the Debtors' estates against their present and former officers, Management Committee members, employees, consultants, and agents and arising from or relating to the period prior to the Petition Date are released by this Plan. As of the Effective Date, the Debtors and the Debtors in Possession shall be deemed to have waived and released its officers, employees, consultants, and agents who were officers, employees, consultants, or agents, respectively, at any time during the Chapter 11 Cases, from any and all claims of the Debtors' estates arising from or relating to the period from and after the Petition Date; provided, however, that, except as otherwise provided by prior or subsequent Final Order of the Bankruptcy Court, this provision shall not operate as a waiver or release of (a) any Person (i) named or subsequently named as a defendant in any action

commenced by or on behalf of the Debtors in Possession, including any actions prosecuted by the Creditors' Committee and the Prepetition Administrative Agent on behalf of the Prepetition Lenders or any Litigation Trust Claim prosecuted by the Litigation Trust, (ii) identified or subsequently identified in a report by the Examiner as having engaged in acts of dishonesty or willful misconduct detrimental to the interests of the Debtors, or (iii) adjudicated or subsequently adjudicated by a court of competent jurisdiction to have engaged in acts of dishonesty or willful misconduct detrimental to the interests of the Debtors or (b) any claim (i) with respect to any loan, advance, or similar payment by the Debtors to any such Person, (ii) with respect to any contractual obligation owed by such Person to the Debtors, (iii) relating to such Person's knowing fraud, or (iv) to the extent based upon or attributable to such Person gaining in fact a personal profit to which such Person was not legally entitled; and, provided, further, that the foregoing is not intended, nor shall it be construed, to release any of the Debtors' claims that may exist against the Debtors' directors and officers liability insurance.

- 20.10 <u>Releases by the Debtors</u>. On the Effective Date, effective as of the Confirmation Date, to the fullest extent permissible under applicable law, the Debtors shall release and be permanently enjoined from any prosecution or attempted prosecution of the Released Actions; <u>provided, however,</u> that the foregoing shall not operate as a waiver of or release from any Causes of Action filed and unresolved as of the Effective Date against any Prepetition Lender with a Claim under a Swap Contract (as defined in the Prepetition Credit Agreement) or a holder of a Swap Claim solely to determine whether or not such Swap Contract Claim qualifies as a Lender Swap Obligation under the Prepetition Credit Agreement.
- 20.11 Releases by Holders of Claims and Equity Interests. On the Effective Date, effective as of the Confirmation Date, to the fullest extent permissible under applicable law, each Person who votes to accept the Plan, any Person who receives a distribution under the Plan and each Settling Party shall be deemed to (i) consensually forever release and be permanently enjoined from any prosecution or attempted prosecution of any Released Actions which such Person has or may have against (A) the Prepetition Lenders or holders of Swap Claims (excluding Bank of Oklahoma and its affiliates), the Prepetition Administrative Agent, the Postpetition Administrative Agent, and the Postpetition Lenders under the Postpetition Financing Agreement or (B) the Catsimatidis Group and (ii) agree not to aid, assist, support, or otherwise participate with any other party in prosecuting Twenty-Day Claims against the Debtors, the Reorganized Debtors, the Prepetition Administrative Agent and/or the Prepetition Lenders or to take any positions contrary to the Debtors, the Reorganized Debtors, the Prepetition Administrative Agent and/or the Prepetition Lenders.
- 20.12 Releases by Members of Creditors' Committee. On the Effective Date, effective as of the Confirmation Date, to the fullest extent permissible under applicable law, the members of the Creditors' Committee, in their individual capacity as such, shall be deemed to consensually forever release and be permanently enjoined from any prosecution or attempted prosecution of any Released Actions which such Person has or may have had against (i) the Prepetition Lenders or holders of Swap Claims (excluding Bank of Oklahoma and its affiliates), the Prepetition Administrative Agent, the Postpetition Administrative Agent, and the Postpetition Lenders under the Postpetition Financing Agreement or (ii) the Catsimatidis Group.

- 20.13 Releases by Members of Producers' Committee. On the Effective Date, effective as of the Confirmation Date, to the fullest extent permissible under applicable law, the members of the Producers' Committee, in their individual capacity as such, shall be deemed to consensually forever release and be permanently enjoined from any prosecution or attempted prosecution of any Released Actions which such Person has or may have had against the Prepetition Lenders or holders of Swap Claims, the Prepetition Administrative Agent, the Postpetition Administrative Agent, and the Postpetition Lenders under the Postpetition Financing Agreement. For the avoidance of doubt, this release shall not affect, prejudice, or release any claims of a Producer with respect to any Claims or Causes of Action relating to or with respect to any parties other than the Prepetition Lenders or holders of Swap Claims, the Prepetition Administrative Agent, the Postpetition Administrative Agent, and the Postpetition Lenders under the Postpetition Financing Agreement.
- 20.14 **Release of Guarantors**. This Plan shall operate as a full release of all Entities (regardless of whether such Entities are Debtors) that are Guarantors (as such term is defined in the Senior Notes Indenture) other than any Entity in the SemCanada Group or that are Guarantors (as such term is defined in the Prepetition Credit Agreement) other than any Entity in the SemCanada Group from any liability arising out of or relating to the Senior Notes Indenture or Prepetition Credit Agreement, respectively, not otherwise discharged by and pursuant to this Plan.

ARTICLE XXI

RETENTION OF JURISDICTION

- 21.1 <u>Retention of Jurisdiction</u>. The Bankruptcy Court shall retain and have exclusive jurisdiction over any matter arising under the Bankruptcy Code, arising in or related to the Chapter 11 Cases or the Plan, or that relates to the following purposes:
- (a) to resolve any matters related to the assumption, assumption and assignment, or rejection of any executory contract or unexpired lease to which a Debtor is a party or with respect to which a Debtor may be liable and to hear, determine and, if necessary, liquidate, any Claims arising therefrom, including those matters related to the amendment after the Effective Date of the Plan, to add any executory contracts or unexpired leases to the list of executory contracts and unexpired leases to be rejected;
- (b) to enter such orders as may be necessary or appropriate to implement or consummate the provisions of the Plan and all contracts, instruments, releases, and other agreements or documents created in connection with the Plan;
- (c) to determine any and all motions, adversary proceedings, applications, and contested or litigation matters that may be pending on the Effective Date or that, pursuant to the Plan, may be instituted by the Reorganized Debtors or the Litigation Trust prior to or after the Effective Date (which jurisdiction shall be non-exclusive as to any non-core matters);

- (d) to ensure that distributions to holders of Allowed Claims are accomplished as provided herein, including any disputes between Owners and Operators with regard to distributions required under Section 3.1(h) of the Plan;
- (e) to hear and determine any timely objections to Administrative Expense Claims or to proofs of Claim and Equity Interests filed, both before and after the Confirmation Date, including any objections to the classification of any Claim or Equity Interest, and to allow, disallow, determine, liquidate, classify, estimate, or establish the priority of or secured or unsecured status of any Claim, in whole or in part;
- (f) to resolve any Disputed Claims, including objections or estimation motions prosecuted by the Producer Representative;
- (g) to enter and implement such orders as may be appropriate in the event the Confirmation Order is for any reason stayed, revoked, modified, reversed, or vacated;
- (h) to issue such orders in aid of execution and consummation of the Plan, to the extent authorized by section 1142 of the Bankruptcy Code;
- (i) to consider any modifications of the Plan, to cure any defect or omission, or reconcile any inconsistency in any order of the Bankruptcy Court, including, without limitation, the Confirmation Order;
- (j) to hear and determine all applications for awards of compensation for services rendered and reimbursement of expenses of professionals under sections 330, 331, and 503(b) of the Bankruptcy Code;
- (k) to hear and determine disputes arising in connection with or relating to the Plan or the interpretation, implementation, or enforcement of the Plan or the extent of any Entity's obligations incurred in connection with or released under the Plan;
- (l) to issue injunctions, enter and implement other orders, or take such other actions as may be necessary or appropriate to restrain interference by any Entity with the consummation, implementation, or enforcement of the Plan, the Confirmation Order, or any other order of the Bankruptcy Court;
- (m) to determine any other matters that may arise in connection with or are related to the Plan, the Disclosure Statement, the Confirmation Order, or any other contract, instrument, release, or other agreement or document created in connection with the Plan or the Disclosure Statement;
- (n) to recover all assets of the Debtors and property of the Debtors' estates, wherever located;
- (o) to hear and determine matters concerning state, local, and federal taxes in accordance with sections 346, 505, and 1146 of the Bankruptcy Code (including the expedited determination of tax under section 505(b) of the Bankruptcy Code);

- (p) to determine the scope of any discharge of any Debtor under the Plan or the Bankruptcy Code;
- (q) to hear any other matter or for any purpose specified in the Confirmation Order that is not inconsistent with the Bankruptcy Code;
- (r) to resolve any disputes regarding whether a Cause of Action constitutes a Retained Cause of Action or a Litigation Trust Claim;
- (s) to enforce the terms of the Litigation Trust Agreement and to decide any claims or disputes which may arise or result from, or be connected with, the Litigation Trust Agreement, any breach or default under the Litigation Trust Agreement, or the transactions contemplated by the Litigation Trust Agreement; and
 - (t) to enter a final decree closing the Chapter 11 Cases;

provided, however, that the foregoing is not intended to (1) expand the Bankruptcy Court's jurisdiction beyond that allowed by applicable law, (2) impair the rights of an Entity to (i) invoke the jurisdiction of a court, commission, or tribunal, including, without limitation, with respect to matters relating to a governmental unit's police and regulatory powers, and (ii) contest the invocation of any such jurisdiction; provided, however, that the invocation of such jurisdiction, if granted, shall not extend to the allowance or priority of Claims or the enforcement of any money judgment against a Debtor or a Reorganized Debtor, as the case may be, entered by such court, commission, or tribunal, and (3) impair the rights of an Entity to (i) seek the withdrawal of the reference in accordance with 28 U.S.C. § 157(d) and (ii) contest any request for the withdrawal of reference in accordance with 28 U.S.C. § 157(d).

ARTICLE XXII

MODIFICATION, REVOCATION, OR WITHDRAWAL OF THE PLAN

Modification of the Plan. The Debtors reserve the right, in accordance with the 22.1 Bankruptcy Code and the Bankruptcy Rules, to amend or modify the Plan, the Plan Supplement, or any exhibits to the Plan at any time prior to entry of the Confirmation Order, including, without limitation, to exclude one (1) or more Debtors from the Plan; provided, however, that any such amendments or modifications shall be subject to the consent of each of the Lender Steering Committee, the Creditors' Committee and the Producers' Committee which consents shall not be unreasonably withheld. Upon entry of the Confirmation Order, the Debtors may, upon order of the Bankruptcy Court, amend or modify the Plan, in accordance with section 1127(b) of the Bankruptcy Code, including, without limitation, to exclude one (1) or more Debtors from the Plan, or remedy any defect or omission or reconcile any inconsistency in the Plan in such manner as may be necessary to carry out the purpose and intent of the Plan; provided, however, that any such amendments or modifications shall be reasonably acceptable in form and substance to the Lender Steering Committee, the Creditors' Committee and the Producers' Committee, which consents shall not unreasonably be withheld. A holder of a Claim that has adopted the Plan shall be deemed to have accepted the Plan as modified if the proposed modification does not materially and adversely change the treatment of the Claim of such holder.

22.2 <u>Revocation or Withdrawal of the Plan</u>.

- (a) The Plan may be revoked or withdrawn prior to the Confirmation Date by the Debtors; provided, however, that the Lender Steering Committee and the Creditors' Committee consent.
- (b) If the Plan is revoked or withdrawn prior to the Confirmation Date, or if the Plan does not become effective for any reason whatsoever, the Plan shall be deemed null and void. In such event, nothing contained herein shall be deemed to constitute a waiver or release of any claims by the Debtors or any other Entity or to prejudice in any manner the rights of the Debtors or any other Entity in any further proceedings involving the Debtors.

ARTICLE XXIII

MISCELLANEOUS PROVISIONS

- 23.1 <u>Effectuating Documents and Further Transactions</u>. Each of the Debtors and the Reorganized Debtors is authorized to execute, deliver, file, or record such contracts, instruments, releases, indentures, and other agreements or documents and take such actions as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan and any securities issued pursuant to the Plan.
- Withholding and Reporting Requirements. In connection with the consummation of the Plan, the Debtors, the Reorganized Debtors, or the Disbursing Agent, as the case may be, shall comply with all withholding and reporting requirements imposed by any federal, state, local, or foreign taxing authority and all distributions hereunder shall be subject to any such withholding and reporting requirements. Notwithstanding the above, each holder of an Allowed Claim that is to receive a distribution under the Plan shall have the sole and exclusive responsibility for the satisfaction and payment of any tax obligations imposed by any governmental unit, including income, withholding and other tax obligations, on account of such distribution. Any party issuing any instrument or making any distribution under the Plan has the right, but not the obligation, to not make a distribution until such holder has made arrangements satisfactory to such issuing or disbursing party for payment of any such tax obligations.
- 23.3 Plan Supplement. The Exit Facility commitment letter, the Litigation Trust Agreement, the Management Incentive Plan, the New Holdco Bylaws, the New Holdco Certificate of Incorporation, the Second Lien Term Loan Facility term sheet, the Contributing Lender Assignment, the Warrant Agreement, the terms and conditions of the refinancing of the White Cliffs Credit Agreement, Schedules 1(A) and 1(B), the identity of the Persons who will serve as executive officers of New Holdco (if known by the date the Plan Supplement is filed), the identity of the Producer Representative, and any other appropriate documents, shall be contained in the Plan Supplement and filed with the Clerk of the Bankruptcy Court at least ten (10) days prior to the last day upon which holders of Claims may vote to accept or reject the Plan; provided, however, that the Debtors may amend (a) Schedules 1(A) and 1(B) through and including the Confirmation Date and (b) each of the other documents contained in the Plan Supplement, subject to Section 22.1 of the Plan, through and including the Effective Date in a manner consistent with the Plan and Disclosure Statement. Each of the documents contained in

the Plan Supplement as to form and substance shall be subject to the consent of each of the Lender Steering Committee and the Creditors' Committee, which consents shall not be unreasonably withheld. Upon its filing with the Bankruptcy Court, the Plan Supplement may be inspected in the office of the Clerk of the Bankruptcy Court during normal court hours. Holders of Claims or Equity Interests may obtain a copy of the Plan Supplement on the Debtors' website at www.kccllc.net/SemGroup.

- 23.4 <u>Payment of Statutory Fees.</u> All fees payable pursuant to section 1930 of title 28 of the United States Code shall be paid as and when due or otherwise pursuant to an agreement between the Reorganized Debtors and the United States Department of Justice, Office of the United States Trustee, until such time as a Chapter 11 Case for a Debtor shall be closed in accordance with the provisions of Section 23.14 of the Plan.
- 23.5 **Expedited Tax Determination.** The Reorganized Debtors may request an expedited determination of taxes under section 505(b) of the Bankruptcy Code for all returns filed for, or on behalf of, such Reorganized Debtors for all taxable periods through the Effective Date.
- 23.6 <u>Post-Confirmation Date Fees and Expenses</u>. From and after the Confirmation Date, the Reorganized Debtors and the Litigation Trust shall, in the ordinary course of business and without the necessity for any approval by the Bankruptcy Court, (a) retain professionals and (b) pay the reasonable fees and expenses (including reasonable professional fees and expenses) incurred by the Debtors, the Reorganized Debtors, or the Litigation Trust, as the case may be, related to implementation and consummation of or consistent with the provisions of the Plan.
- 23.7 **Substantial Consummation.** On the Effective Date, the Plan shall be deemed to be substantially consummated under sections 1101 and 1127(b) of the Bankruptcy Code.
- Severability. If, prior to the Confirmation Date, any term or provision of the Plan shall be held by the Bankruptcy Court to be invalid, void, or unenforceable, including, without limitation, the inclusion of one (1) or more Debtors in the Plan, the Bankruptcy Court shall, with the consent of the Debtors, the Lender Steering Committee, and the Creditors' Committee, have the power to alter and interpret such term or provision to make it valid or enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void, or unenforceable, and such term or provision shall then be applicable as altered or interpreted. Notwithstanding any such holding, alteration, or interpretation, the remainder of the terms and provisions of the Plan shall remain in full force and effect and shall in no way be affected, impaired, or invalidated by such holding, alteration, or interpretation. The Confirmation Order shall constitute a judicial determination and shall provide that each term and provision of the Plan, as it may have been altered or interpreted in accordance with the foregoing, is valid and enforceable pursuant to its terms.
- 23.9 <u>Governing Law.</u> Except to the extent that the Bankruptcy Code or other federal law is applicable, or to the extent that an exhibit hereto or document contained in the Plan Supplement provides otherwise, the rights, duties, and obligations arising under this Plan shall be governed by, and construed and enforced in accordance with, the Bankruptcy Code and, to the

extent not inconsistent therewith, the laws of the State of New York, without giving effect to principles of conflicts of laws.

- 23.10 **<u>Time.</u>** In computing any period of time prescribed or allowed by the Plan, unless otherwise set forth herein or determined by the Bankruptcy Court, the provisions of Bankruptcy Rule 9006 shall apply.
- 23.11 **Binding Effect.** The Plan shall be binding upon and inure to the benefit of the Debtors, the holders of Claims and Equity Interests, the Litigation Trust, and their respective successors and assigns, including, without limitation, the Reorganized Debtors.
- 23.12 **Exhibits/Schedules.** All exhibits and schedules to the Plan, including the Plan Supplement, are incorporated into and are a part of the Plan as if set forth in full herein.
- 23.13 **Notices.** All notices, requests, and demands to or upon the Debtors, the Debtors in Possession, or the Reorganized Debtors to be effective shall be in writing (including by facsimile transmission) and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made when actually delivered or, in the case of notice by facsimile transmission, when received and telephonically confirmed, addressed as follows:

SEMGROUP, L.P. 6120 S. Yale, Suite 700 Tulsa, Oklahoma 74136 Attention: Legal Department Telecopier: (918) 524-8276

With a copy to:

WEIL, GOTSHAL & MANGES LLP 200 Crescent Court, Suite 300 Dallas, Texas 75201 Attention: Martin A. Sosland, Esq. Telecopier: (214) 746-7777

- 23.14 Closing of the Chapter 11 Cases. The Reorganized Debtors shall, promptly upon the full administration of the Chapter 11 Cases, file with the Bankruptcy Court all documents required by Bankruptcy Rule 3022 and any applicable order of the Bankruptcy Court.
- 23.15 **Section Headings.** The section headings contained in this Plan are for reference purposes only and shall not affect in any way the meaning or interpretation of the Plan.
- 23.16 **Exemption from Registration.** Pursuant to section 1145 of the Bankruptcy Code, and except as provided in subsection (b) thereof, the issuance of the Plan Currency on account of, and in exchange for, the Claims against the Debtors shall be exempt from registration pursuant to section 5 of the Securities Act of 1933 and any other applicable nonbankruptcy law or regulation.

- 23.17 Exemption from Transfer Taxes. Pursuant to section 1146(c) of the Bankruptcy Code, the issuance, transfer, or exchange of Equity Interests or Plan Currency pursuant to the Plan, the creation of any mortgage, deed of trust, or other security interest, the making or assignment of any lease or sublease, or the making or delivery of any deed or other instrument of transfer under, in furtherance of, or in connection with the Plan, shall not be subject to any stamp, real estate transfer, mortgage recording, or other similar tax.
- 23.18 No Effect on Downstream Claims. No provision of this Plan (including, but not limited to, Section 1.133, 20.11 and 20.13) is intended to and no provision shall compromise, affect, discharge, or otherwise impact the Downstream Claims against any Downstream Purchasers (including, but not limited to, J. Aron & Company) and nothing contained herein shall release Producers' Downstream Claims against Downstream Purchasers (including, but not limited to, Downstream Claims against J. Aron & Company) for amounts due to Producers. Such Downstream Claims are not property of the Debtors or their estates and are not administered under the Plan, nor will the Downstream Claims constitute Litigation Trust Assets. Neither the Debtors, the Reorganized Debtors, the Prepetition Administrative Agent nor the Prepetition Lenders shall oppose or otherwise contest the Producers' prosecution of any Downstream Claims asserted as of September 21, 2009, including, but not limited to, Downstream Claims against J. Aron & Company, nor oppose efforts of First Purchaser Producers to seek remand or transfer of asserted Downstream Claims in litigation which has been or may hereinafter be transferred to the Bankruptcy Court. This Section 23.18 shall control in case of a conflict with any other provision in this Plan.

23.19 <u>Inconsistencies</u>. To the extent of any inconsistencies between the information contained in the Disclosure Statement and the terms and provisions of the Plan, the terms and provisions contained herein shall govern.

Dated: Wilmington, Delaware October 27, 2009

SEMCRUDE, L.P.

CHEMICAL PETROLEUM EXCHANGE, INCORPORATED

EAGLWING, L.P.

GRAYSON PIPELINE, L.L.C.

GREYHAWK GAS STORAGE COMPANY, L.L.C.

K.C. ASPHALT L.L.C.

SEMCANADA II, L.P.

SEMCANADA L.P.

SEMCRUDE PIPELINE, L.L.C.

SEMFUEL TRANSPORT LLC

SEMFUEL, L.P.

SEMGAS GATHERING LLC

SEMGAS STORAGE, L.L.C.

SEMGAS, L.P.

SEMGROUP ASIA, L.L.C.

SEMGROUP FINANCE CORP.

SEMGROUP, L.P.

SEMKAN, L.L.C.

SEMMANAGEMENT, L.L.C.

SEMMATERIALS VIETNAM, L.L.C.

SEMMATERIALS, L.P.

SEMOPERATING G.P., L.L.C.

SEMSTREAM, L.P.

SEMTRUCKING, L.P.

STEUBEN DEVELOPMENT COMPANY, L.L.C.

SEMCAP, L.L.C.

By:_/s/ Terrence Ronan_

Name: Terrence Ronan Title: Authorized Officer

Schedule 1.49¹

BP Oil Supply Company

CIMA Energy

CP Energy LLC

Calcasieu Refining Company

Central Crude Corporation

Chevron Texaco LP/Chevron Products Company

Cimmaron Transportation/Cimmaron Transmission Company/Cimmaron Gathering, LP

Coffeyville Resources/Coffeyville Resources Refining & Marketing, LLC

ConocoPhillips Company

Crude Marketing & Transportation

DCP NGL subsidiary of DCP Midstream

Delek Refining Inc

Husky Marketing and Supply Company

Interstate Petroleum/High Seirra Crude Oil and Marketing d/b/a Interstate Petroleum

Corporation

J. Aron & Company

Kelly McClaskey Oil Field Services

National Cooperative Refinery Association d/b/a NCRA

Oasis Trading and Marketing/Oasis Transportation and Marketing Corporation

Occidental Energy Marketing, Inc.

Plains Marketing LP

Plains Marketing GP, Inc.

Shell Trading US Company/Shell Oil Company

Sunoco Partners Marketing & Terminal L.P./Sunoco Logistics Partners, LP

Teppco Crude Oil LLC/Teppco Crude GP, LLC

Valero Energy

Valero Marketing and Supply Company

Ventura Refining and Transmission

¹ Including affiliates of the foregoing Entities except to the extent they are Prepetition Lenders or their affiliates (any other than Goldman Sachs & Co., J. Aron & Co., and their respective affiliates) who were such at 5:00 p.m. Eastern Daylight Time, on May 14, 2009.

Schedule 1

First Purchaser Producer Twenty-Day Claims (Listed by Debtor)

First Purchaser Producer Twenty-Day Claims (Listed By Debtor)

Debtor	Operator	1st Purchase 503(b)(9) Amount
EaglWing	ABRAXAS PETROLEUM CORPORATION	\$446,621.28
EaglWing	BRIGHTON ENERGY LLC	\$66,115.62
EaglWing	DAVIS PETROLEUM CORP	\$540,918.03
EaglWing	DINERO OPERATING CO	\$20,756.64
EaglWing EaglWing	GREAT PLAINS ENERGY INC	\$573,312.15
EaglWing EaglWing	KCS RESOURCES INC	\$23,353.76
~ ~	LANDMARK RESOURCES INC	\$1,049,669.48
EaglWing	LD DRILLING	\$1,455,846.29
EaglWing	MCCOY PETROLEUM CORPORATION	\$1,514,221.71
EaglWing	MULL DRILLING CO INC	\$3,332,368.44
EaglWing	MURFIN DRILLING COMPANY INC	\$8,092,255.58
EaglWing	PETROHAWK PROPERTIES LP	\$20,422.65
EaglWing	POGO PRODUCING COMPANY	\$253,196.76
EaglWing	PROLIFIC RESOURCES LLC	\$180,946.30
EaglWing	PURE PARTNERSHIP LLP	\$2,789.43
EaglWing	REIF OIL & GAS LLC	\$78,295.82
EaglWing	SAMSON CONTOUR ENERGY E&P LLC	\$277,505.16
EaglWing	SAMSON CONTOUR EXERCITED SAMSON LONE STAR LLC	\$5,690,774.45
EaglWing	SAMSON RESOURCES CO	\$5,338,379.05
EaglWing EaglWing	ST MARY LAND & EXPLORATION CO	\$7,195,925.73
EaglWing	STELLA ELIZABETH GRIMSLEY	\$38,512.03
EaglWing	VESS OIL CORP	\$7,638,554.83
EaglWing	VIKING RESOURCES INC	\$689,981.20
EaglWing	ZENERGY INC	\$1,780,598.25
Total EaglWing	24 Producers	\$46,301,320.66
A COURT THE CALL		
SemCrude	*CLAUD C HARP	\$39,853.81
SemCrude	*DAVID WIKEL	\$10,857.74
SemCrude	*FREDERICK F DRUMMOND	\$20,003.84
SemCrude	3 G OIL COMPANY	\$54,256.58
SemCrude	3M PRODUCTION, INC.	\$21,881.09
SemCrude	A C AILEY	\$10,617.08
0011101444		

First Purchaser Producer Twenty-Day Claims (Listed By Debtor)

Debtor	. Operator	1st Purchase 503(b)(9) Amount
SemCrude	AETHON I LP	\$87,838.96
SemCrude SemCrude	AINSWORTH OPERATING COMPANY	\$19,975.60
SemCrude SemCrude	AL CONTRERAS CONSULTANTS INC	\$12,272.57
SemCrude SemCrude	ALEXANDER RESOURCES INC	\$41,916.78
SemCrude SemCrude	ALLAN BAUGH	\$78,422.54
SemCrude	ALLAN BAUGH & RITA BAUGH,	\$9,362.08
SemCrude SemCrude	ALTEX ENERGY CORPORATION	\$2,320,674.73
SemCrude SemCrude	ALTMAN ENERGY INC	\$84,402.48
SemCrude SemCrude	AMERICAN ENERGIES CORP	\$19,992.08
SemCrude SemCrude	AMERICAN WEST	\$17,205.38
SemCrude SemCrude	ANSTINE - MUSGROVE OIL COMPANY	\$82,095.45
SemCrude	ANSTINE & MUSGROVE INC	\$278,925.42
SemCrude SemCrude	APACHE CORPORATION	\$48,785.43
SemCrude SemCrude	APOLLO ENERGIES INC	\$41,533.27
SemCrude SemCrude	ARDMORE PRODUCTION & EXPL CO	\$39,296.86
SemCrude SemCrude	ARGENT ENERGY INC	\$201,757.12
SemCrude	ARROW OIL & GAS INC	\$1,322,168.82
SemCrude	ARROWHEAD ENERGY INC	\$40,813.92
SemCrude	ATCHLEY RESOURCES INC	\$19,992.44
SemCrude	ATHAN INC	\$21,977.59
SemCrude	AVARD LLC	\$250,379.17
SemCrude	B & G PRODUCTION INC	\$43,510.92
SemCrude	BARNES OIL COMPANY	\$39,331.40
SemCrude	BARRIER RESOURCES	\$22,307.79
SemCrude	BAYS EXPLORATION INC	\$614,124.85
SemCrude	BC OPERATING INC	\$46,161.34
SemCrude	BEAR PETROLEUM INC	\$42,590.38
SemCrude	BEASLEY OIL COMPANY	\$67,447.37
SemCrude	BELL BROTHERS	\$23,100.01
SemCrude	BELL ENERGY	\$21,859.36
SemCrude	BELPORT OIL INC	\$19,076.10
SemCrude	BEREXCO INC	\$19,932.62

First Purchaser Producer Twenty-Day Claims (Listed By Debtor)

Debtor	Operator Control of the Control of t	1st Purchase 503(b)(9) Amount
SemCrude	BERRY PETROLEUM COMPANY	\$70,325.70
SemCrude	BIGGY'S PETROLEUM	\$20,182.01
SemCrude	BILL LAIRD	\$140,229.77
SemCrude	BILL WEEMS OIL INC	\$148,939.12
SemCrude SemCrude	BILLIE E EMERY	\$9,808.25
SemCrude SemCrude	BLACK DOT OIL INC	\$38,976.29
SemCrude SemCrude	BLACK GOLD OIL CO INC	\$23,536.71
SemCrude SemCrude	BLACKCAT OIL COMPANY	\$37,045.25
SemCrude SemCrude	BLACKSPOT OIL CO	\$19,439.52
SemCrude	BLAKE EXPLORATION, INC	\$180,862.09
SemCrude	BLUE DOLPHIN PRODUCTION LLC	\$627,728.91
SemCrude	BOB LOFTIS OIL & GAS	\$22,518.31
SemCrude	BOBBY HORN DBA HORN OIL COMPANY	\$11,710.10
SemCrude	BOBBY J DARNELL	\$51,489.71
SemCrude	BOBBY R PERKINS	\$19,809.12
SemCrude	BRADEN-DEEM INC	\$45,861.97
SemCrude	BRIDWELL OIL COMPANY	\$44,363.50
SemCrude	BRITT OIL COMPANY LLC	\$19,455.61
SemCrude	BRUCE OIL CO	\$19,420.93
SemCrude	BRUCE OIL COMPANY LLC	\$18,911.27
SemCrude	BURGUNDY OIL & GAS OF NM INC	\$90,424.81
SemCrude	BUTKIN OIL COMPANY LLC	\$529,823.52
SemCrude	C & A OIL COMPANY INC	\$41,566.56
SemCrude	C & W OIL COMPANY	\$97,800.33
SemCrude	CEDLC	\$19,188.96
SemCrude	C L & S OIL CO INC	\$21,233.07
SemCrude	CAL CREEK RESOURCES INC	\$25,172.48
SemCrude	CALVIN NOAH	\$122,977.70
SemCrude	CANOK INTERNATIONAL INC.	\$11,651.91
SemCrude	CARLSON OIL INC	\$25,029.12
SemCrude	CARMAC ENERGY CORP	\$290,363.57
SemCrude	CARROLL PRODUCTION CO	\$10,641.52

First Purchaser Producer Twenty-Day Claims (Listed By Debtor)

Debtor	Operator	1st Purchase 503(b)(9) Amount
SemCrude	CARTLEDGE & CARTLEDGE OIL CO	\$136,832.67
SemCrude SemCrude	CASE SALES COMPANY INC	\$40,366.88
SemCrude SemCrude	CASEY MUSGROVE OIL COMPANY	\$13,142.42
SemCrude	CASILLAS PETROLEUM CORP	\$81,288.46
SemCrude	CASTELLI EXPLORATION, INC.	\$137,355.01
SemCrude	CBM GAS COMPANY LLC	\$20,008.78
SemCrude	CEDAR RIDGE ENERGY INC	\$16,314.93
SemCrude SemCrude	CENPAT OPERATING LLC	\$84,178.35
SemCrude SemCrude	CENTER PRODUCTION CO LLC	\$230,417.96
SemCrude SemCrude	CENTRAL OPERATING INC	\$216,992.75
SemCrude SemCrude	CENTREX OPERATING COMPANY INC	\$69,430.77
SemCrude SemCrude	CHACO ENERGY COMPANY	\$18,995.36
SemCrude SemCrude	CHAMBERS. R.H., LLC	\$87,533.71
SemCrude SemCrude	CHAMP OPERATING COMPANY	\$18,824.94
SemCrude SemCrude	CHAP OPERATING CO., INC.	\$21,786.56
SemCrude SemCrude	CHAPARRAL ENERGY INC	\$69,713.66
SemCrude	CHAPMAN OIL COMPANY	\$41,212.30
SemCrude SemCrude	CHARTER OAK PRODUCTION CO LLC	\$22,068.01
SemCrude SemCrude	CHIZUM OIL LLC	\$139,835.24
SemCrude	CHOLLA PRODUCTION LLC	\$44,005.99
SemCrude SemCrude	CIRCLE C OILFIELD SUPPLY INC	\$23,731.39
SemCrude SemCrude	CIRCLE K OIL AND GAS INC	\$31,693.74
SemCrude SemCrude	CLARK EXPLORATION COMPANY	\$37,598.19
SemCrude SemCrude	CLIPPER ENERGY LLC	\$824,684.42
SemCrude SemCrude	CLYDE R JACOBS	\$19,690.32
SemCrude SemCrude	CMX INC	\$254,988.18
SemCrude SemCrude	COG OIL & GAS LP	\$1,035,545.53
SemCrude SemCrude	COLEMAN OIL	\$18,885.61
-	COLORADO INTERSTATE GAS CO	\$145,067.81
SemCrude SemCrude	COLPITT BROTHERS OIL	\$53,591.13
·	COLUMBUS OIL COMPANY	\$19,600.47
SemCrude	COMBS PRODUCTION LLC	\$11,636.94
SemCrude	COMPS LYONACTION FFC	,,

-2000 BBD War - 1 224 BBD BBD - 245 BBD	Operator	Ist Purchase 503(b)(9)/Amount
Debtor	Operator COMPASS DRILLING INC	\$39,832.10
SemCrude	CONSOLIDATED INVESTMENTS	\$63,384.88
SemCrude SemCrude	COOK OIL COMPANY	\$535,107.14
SemCrude SemCrude	COOKE COUNTY PRODUCTION CO INC	\$91,031.48
SemCrude SemCrude	COURTS COMPANY	\$17,676.64
SemCrude SemCrude	CRAWLEY PETROLEUM CORPORATION	\$216,952.47
SemCrude SemCrude	CREST RESOURCES INC	\$43,762.58
SemCrude SemCrude	CROSE OIL COMPANY	\$59,916.21
SemCrude SemCrude	CRYSTAL RIVER OIL & GAS LLC	\$21,281.90
SemCrude SemCrude	CUDD OPERATING CORP	\$81,175.59
SemCrude SemCrude	CULBREATH OIL & GAS INC	\$43,808.67
SemCrude SemCrude	CWIC,INC.	\$20,594.50
SemCrude SemCrude	CYCLONE PETROLEUM INC	\$103,752.82
SemCrude	D & H OIL INC	\$23,772.29
SemCrude	D C ENERGY INC	\$293,835.09
SemCrude	D E EXPLORATION INC	\$708,932.96
SemCrude	D R LAUCK OIL CO INC	\$323,589.39
SemCrude	D W EVANS OIL	\$43,227.67
SemCrude	DALE RADFORD	\$58,637.97
SemCrude	DALE WILLIAMS	\$19,079.37
SemCrude SemCrude	DAN L MCGONIGAL	\$19,806.27
SemCrude	DAN WALLACE INC	\$201,977.56
SemCrude	DANIEL S TRACHTENBERG	\$22,583.47
SemCrude	DARRELL'S DRILLING INC	\$20,927.28
SemCrude	DAVE HOLKE CONSULTING INC	\$21,715.03
SemCrude	DAVID K MOORE	\$60,269.17
SemCrude	DAVID R COAST	\$59,111.67
SemCrude	DAYSTAR PETROLEUM INC	\$614,739.57
SemCrude	DEAN E EYLER DBA DEAN E EYLER	\$40,759.91
SemCrude SemCrude	DEMO OIL COMPANY	\$8,979.55
SemCrude	DEVON RESOURCES CORPORATION	\$19,400.10
SemCrude SemCrude	DIAMOND CROSS OIL CO LLC	\$10,007.28
Semenue		

SemCrude DIVERSIFIED OPERATING CORP \$62,109.08 SemCrude DK OPERATING \$174,517.39 SemCrude D-MIL PRODUCTION INC \$19,583.76 SemCrude DOME ENGINEERING \$18,733.05 SemCrude DON DENNEY \$21,310.28 SemCrude DONALD F LESTER \$40,197.80 SemCrude DONRAY PETROLEUM LLC \$51,011.70 SemCrude DOUBLE EAGLE EXPLORATION INC \$224,908.36 SemCrude DOUBLE M PETROPROPERTIES, INC. \$44,287.10 SemCrude DOUBLE M PETROPROPERTIES, INC. \$19,082.11 SemCrude DREW WILSON \$19,082.11 SemCrude DREYER OIL \$18,001.84 SemCrude DREYER OIL \$18,001.84 SemCrude DREYER OIL \$18,001.84 SemCrude DRILLERS & PRODUCERS INC \$787,367.81 SemCrude DUNCAP OIL COMPANY \$61,483.75 SemCrude DUNCAP OIL COMPANY \$310,056.56 SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL A TERHUNE JR		Operator	Ist Purchase 503(b)(9) Amount
SemCrude DK OPERATING \$174,517.39 SemCrude D-MIL PRODUCTION INC \$19,583.76 SemCrude DOME ENGINEERING \$18,733.05 SemCrude DON DENNEY \$21,310.28 SemCrude DONALD F LESTER \$40,197.80 SemCrude DONRAY PETROLEUM LLC \$51,011.70 SemCrude DONBLE M PETROPROPERTIES, INC \$204,908.36 SemCrude DOUBLE M PETROPROPERTIES, INC \$44,287.10 SemCrude DOUBLE M PETROPROPERTIES, INC \$19,082.11 SemCrude DREW WILSON \$18,001.84 SemCrude DREVER OIL \$10,082.11 SemCrude DUNCAN OIL PROPERTIES INC \$60,248.04 SemCrude DUNLAP OIL COMPANY \$787,367.81 SemCrude DUNLAP OIL COMPANY \$310,056.56 SemCrude DUNNE EQUITIES OPERATING INC \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$37,417.02 SemCrude EARL NEATHERLIN INC \$10,292.23 SemCrude EARL NEATHERLIN INC \$10,292.23 SemCrude EAR	200000000000000000000000000000000000000		\$62,109.08
SemCrude D-MIL PRODUCTION INC \$19,583.76 SemCrude DOME ENGINEERING \$18,733.05 SemCrude DON DENNEY \$21,310.28 SemCrude DON ALD F LESTER \$40,197.80 SemCrude DONRAY PETROLEUM LLC \$51,011.70 SemCrude DOUBLE EAGLE EXPLORATION INC \$204,908.36 SemCrude DOUBLE M PETROPROPERTIES, INC. \$44,287.10 SemCrude DREW WILSON \$19,082.11 SemCrude DREYER OIL \$18,001.84 SemCrude DREYER OIL \$18,001.84 SemCrude DRILLERS & PRODUCERS INC \$787,367.81 SemCrude DUNCAN OIL PROPERTIES INC \$787,367.81 SemCrude DUNCAN OIL PROPERTIES INC \$787,367.81 SemCrude DUNNE EQUITIES OPERATING INC \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$57,417.02 SemCrude EARL A TERHUNE JR \$21,126.22 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE NEATHERLIN INC \$10,292.33 SemCrude EA	•		\$174,517.39
SemCrude DOME ENGINEERING \$18,733.05 SemCrude DON DENNEY \$21,310.28 SemCrude DONALD F LESTER \$40,197.80 SemCrude DONRAY PETROLEUM LLC \$51,011.70 SemCrude DOUBLE EAGLE EXPLORATION INC \$204,908.36 SemCrude DOUBLE M PETROPROPERTIES, INC. \$44,287.10 SemCrude DREW WILSON \$19,082.11 SemCrude DREYER OIL \$60,248.04 SemCrude DRILLERS & PRODUCERS INC \$60,248.04 SemCrude DUNCAN OIL PROPERTIES INC \$787,367.81 SemCrude DUNCAN OIL PROPERTIES INC \$787,367.81 SemCrude DUNLAP OIL COMPANY \$310,056.56 SemCrude DUNIAP OIL COMPANY \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$57,417.02 SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARL NEATHERLIN INC \$10,292.23 SemCrude EARL SBORO ENERGIES CORPORATION \$118,294.89 SemCrude	-	· ·	\$19,583.76
SemCrude DON DENNEY \$21,310.28 SemCrude DONALD F LESTER \$40,197.80 SemCrude DONRAY PETROLEUM LLC \$51,011.70 SemCrude DOUBLE EAGLE EXPLORATION INC \$204,908.36 SemCrude DOUBLE M PETROPROPERTIES, INC. \$44,287.10 SemCrude DREW WILSON \$19,082.11 SemCrude DREYER OIL \$18,001.84 SemCrude DREYER OIL \$60,248.04 SemCrude DUNCAN OIL PROPERTIES INC \$787,367.81 SemCrude DUNCAN OIL PROPERTIES INC \$787,367.81 SemCrude DUNLAP OIL COMPANY \$61,483.75 SemCrude DUNNE EQUITIES OPERATING INC \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$57,417.02 SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EARTH SOUND INC	_	***************************************	\$18,733.05
SemCrude DONALD F LESTER \$40,197.80 SemCrude DONRAY PETROLEUM LLC \$51,011.70 SemCrude DOUBLE EAGLE EXPLORATION INC \$204,908.36 SemCrude DOUBLE M PETROPROPERTIES, INC. \$44,287.10 SemCrude DREW WILSON \$19,082.11 SemCrude DREYER OIL \$60,248.04 SemCrude DRILLERS & PRODUCERS INC \$787,367.81 SemCrude DUNLAP OIL PROPERTIES INC \$787,367.81 SemCrude DUNLAP OIL COMPANY \$61,483.75 SemCrude DUNNE EQUITIES OPERATING INC \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$77,417.02 SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLE-BOING ENERGIES CORPORATION \$192,948.91 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EARTH SOUND INC \$10,4656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude	=		\$21,310.28
SemCrude DONRAY PETROLEUM LLC \$51,011.70 SemCrude DOUBLE EAGLE EXPLORATION INC \$204,908.30 SemCrude DOUBLE M PETROPROPERTIES, INC. \$44,287.10 SemCrude DREW WILSON \$19,082.11 SemCrude DREYER OIL \$18,001.84 SemCrude DRILLERS & PRODUCERS INC \$60,248.04 SemCrude DUNCAN OIL PROPERTIES INC \$787,367.81 SemCrude DUNLAP OIL COMPANY \$61,483.75 SemCrude DUNLAP OIL COMPANY \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$37,417.02 SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLSBORO ENERGIES CORPORATION \$138,650.68 SemCrude EARLSBORO ENERGIES CORPORATION \$138,650.68 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrud	-		\$40,197.80
SemCrude DOUBLE EAGLE EXPLORATION INC \$204,908.36 SemCrude DOUBLE M PETROPROPERTIES, INC. \$44,287.10 SemCrude DREW WILSON \$19,082.11 SemCrude DREYER OIL \$18,001.84 SemCrude DRILLERS & PRODUCERS INC \$60,248.04 SemCrude DUNCAN OIL PROPERTIES INC \$787,367.81 SemCrude DUNLAP OIL COMPANY \$61,483.75 SemCrude DUNNE EQUITIES OPERATING INC \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$9,774.00 SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EDWARD A EMERY \$71,390.44 SemCrude EORATING COMPANY \$21,100.79 SemCrude EMER KARST OIL CO \$9,975.29 SemCrude EMERSOURCE PETROL	-		\$51,011.70
SemCrude DOUBLE M PETROPROPERTIES, INC. \$44,287.10 SemCrude DREW WILSON \$19,082.11 SemCrude DREYER OIL \$18,001.84 SemCrude DRILLERS & PRODUCERS INC \$60,248.04 SemCrude DUNCAN OIL PROPERTIES INC \$787,367.81 SemCrude DUNLAP OIL COMPANY \$61,483.75 SemCrude DUNNE EQUITIES OPERATING INC \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$57,417.02 SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$21,100.79 SemCrude EMER KARST OIL CO \$9,979.29 SemCrude EMERSOURCE PETROLEUM INC \$9,979.29 SemCrude ENERSOURCE PET			\$204,908.36
SemCrude DREW WILSON \$19,082.11 SemCrude DREYER OIL \$18,001.84 SemCrude DRILLERS & PRODUCERS INC \$60,248.04 SemCrude DUNCAN OIL PROPERTIES INC \$787,367.81 SemCrude DUNLAP OIL COMPANY \$61,483.75 SemCrude DUNNE EQUITIES OPERATING INC \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$57,417.02 SemCrude EARL A TERHUNE IR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLSBORO ENERGIES CORPORATION \$192,948.91 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude EDWARD A EMERY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE P	-		\$44,287.10
SemCrude DREYER OIL \$18,001.84 SemCrude DRILLERS & PRODUCERS INC \$60,248.04 SemCrude DUNCAN OIL PROPERTIES INC \$787,367.81 SemCrude DUNLAP OIL COMPANY \$61,483.75 SemCrude DUNNE EQUITIES OPERATING INC \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$57,417.02 SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLSBORO ENERGIES CORPORATION \$192,948.91 SemCrude EARTH SOUND INC \$138,650.64 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude EDWARD A EMERY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERGEX INC \$11,527.86 SemCrude ENERSOURCE P			\$19,082.11
SemCrude DRILLERS & PRODUCERS INC \$60,248.04 SemCrude DUNCAN OIL PROPERTIES INC \$787,367.81 SemCrude DUNLAP OIL COMPANY \$61,483.75 SemCrude DUNNE EQUITIES OPERATING INC \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$57,417.02 SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLSBORG ENERGIES CORPORATION \$192,948.91 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude EGRET OPERATING COMPANY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERSOURCE PETROLEUM INC \$11,527.86 SemCrude ERDOL INC. \$41,827.94 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude<			\$18,001.84
SemCrude DUNCAN OIL PROPERTIES INC \$787,367.81 SemCrude DUNLAP OIL COMPANY \$61,483.75 SemCrude DUNNE EQUITIES OPERATING INC \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$57,417.02 SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLSBORO ENERGIES CORPORATION \$192,948.91 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude EOWARD A EMERY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude ELMER KARST OIL CO \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERGEX INC \$11,527.86 SemCrude ENERSOURCE PETROLEUM INC \$11,527.86 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESCHER CORPORAT		,_	\$60,248.04
SemCrude DUNLAP OIL COMPANY \$61,483.75 SemCrude DUNNE EQUITIES OPERATING INC \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$57,417.02 SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLSBORO ENERGIES CORPORATION \$192,948.91 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude EGRET OPERATING COMPANY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$11,527.86 SemCrude ERDOL INC. \$41,827.94 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESKRIDGE PRODUCTION COMPANY \$9,176.97			•
SemCrude DUNNE EQUITIES OPERATING INC \$310,056.56 SemCrude EAGLE CREEK CORPORATION \$57,417.02 SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLSBORO ENERGIES CORPORATION \$192,948.91 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude EDWARD A EMERY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$11,527.86 SemCrude ERDOL INC. \$41,827.94 SemCrude ESCHER CORPORATION \$8,937.40 SemCrude ESKRIDGE PRODUCTION COMPANY \$8,937.40			\$61,483.75
SemCrude EAGLE CREEK CORPORATION \$57,417.02 SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLSBORO ENERGIES CORPORATION \$192,948.91 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude EDWARD A EMERY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$11,527.86 SemCrude ERDOL INC. \$41,827.94 SemCrude ESCHER CORPORATION \$8,937.40 SemCrude ESKRIDGE PRODUCTION COMPANY \$9,176.97	•		·
SemCrude EARL A TERHUNE JR \$9,774.00 SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLSBORO ENERGIES CORPORATION \$138,650.68 SemCrude EARTH SOUND INC \$104,656.64 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude EGRET OPERATING COMPANY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$11,527.86 SemCrude ERDOL INC. \$41,827.94 SemCrude ESCHER CORPORATION \$8,937.40 SemCrude ESKRIDGE PRODUCTION COMPANY \$9,176.97	- ·		
SemCrude EARL NEATHERLIN INC \$21,126.22 SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLSBORO ENERGIES CORPORATION \$192,948.91 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude EGRET OPERATING COMPANY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$11,527.86 SemCrude ERDOL INC. \$11,527.86 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESCHER CORPORATION COMPANY \$8,937.40 SemCrude ESKRIDGE PRODUCTION COMPANY \$9,176.97			·
SemCrude EARLE-LE OIL CO \$10,292.23 SemCrude EARLSBORO ENERGIES CORPORATION \$192,948.91 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude EGRET OPERATING COMPANY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$11,527.86 SemCrude ERDOL INC. \$41,827.94 SemCrude ESCHER CORPORATION \$8,937.40 SemCrude ESKRIDGE PRODUCTION COMPANY \$9,176.97	=		•
SemCrude EARLSBORO ENERGIES CORPORATION \$192,948.91 SemCrude EARTH SOUND INC \$138,650.68 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude EGRET OPERATING COMPANY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$11,527.86 SemCrude ERDOL INC. \$11,527.86 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESKRIDGE PRODUCTION COMPANY \$9,974.09	-		•
SemCrude EARTH SOUND INC \$138,650.68 SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude EGRET OPERATING COMPANY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$19,943.35 SemCrude ERDOL INC. \$11,527.86 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESKRIDGE PRODUCTION COMPANY \$9,976.97		EARLSBORO ENERGIES CORPORATION	
SemCrude EDINGER ENGINEERING INC \$104,656.64 SemCrude EDWARD A EMERY \$71,390.04 SemCrude EGRET OPERATING COMPANY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$11,527.86 SemCrude ERDOL INC. \$11,527.86 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESKRIDGE PRODUCTION COMPANY \$9,977.40	•		
SemCrude EDWARD A EMERY \$71,390.04 SemCrude EGRET OPERATING COMPANY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$11,527.86 SemCrude ERDOL INC. \$11,527.86 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESKRIDGE PRODUCTION COMPANY \$9,976.97		EDINGER ENGINEERING INC	•
SemCrude EGRET OPERATING COMPANY \$21,100.79 SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$19,943.35 SemCrude ERDOL INC. \$11,527.86 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESKRIDGE PRODUCTION COMPANY \$8,937.40 SemCrude ESKRIDGE PRODUCTION COMPANY \$9,176.97	- "	EDWARD A EMERY	•
SemCrude ELMER KARST OIL CO \$20,135.56 SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$19,943.35 SemCrude ERDOL INC. \$11,527.86 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESKRIDGE PRODUCTION COMPANY \$8,937.40 SemCrude ESKRIDGE PRODUCTION COMPANY \$9,176.97		EGRET OPERATING COMPANY	· · · · · · · · · · · · · · · · · · ·
SemCrude EMERY WELL SERVICE & ED EMERY \$9,979.29 SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$19,943.35 SemCrude ERDOL INC. \$11,527.86 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESKRIDGE PRODUCTION COMPANY \$8,937.40 SemCrude ESKRIDGE PRODUCTION COMPANY \$9,176.97			•
SemCrude ENERGEX INC \$40,708.23 SemCrude ENERSOURCE PETROLEUM INC \$19,943.35 SemCrude ERDOL INC. \$11,527.86 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESKRIDGE PRODUCTION COMPANY \$8,937.40 SemCrude ESKRIDGE PRODUCTION COMPANY \$9,176.97	-		· ·
SemCrude ENERSOURCE PETROLEUM INC \$19,943.35 SemCrude ERDOL INC. \$11,527.86 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESKRIDGE PRODUCTION COMPANY \$8,937.40			•
SemCrude ERDOL INC. \$11,527.86 SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESKRIDGE PRODUCTION COMPANY \$8,937.40			•
SemCrude ESCHER CORPORATION \$41,827.94 SemCrude ESKRIDGE PRODUCTION COMPANY \$8,937.40 \$9,176.97	=		•
SemCrude ESKRIDGE PRODUCTION COMPANY \$8,937.40	*		•
\$9 176 97			•
OVIIIVIWAY	SemCrude	EUGENE SELLMEYER & HELEN L	\$9,176.97

CARDINARIO CONTRARADO DE CARDA	Operator	Purchase 503(b)(9) Amount
Debtor	EXOK INC	\$44,877.53
SemCrude SemCrude	F & J OIL COMPANY LLC	\$59,678.49
	F L LIVINGSTON	\$20,586.08
SemCrude SemCrude	F.H.A. INVESTMENTS	\$112,385.59
SemCrude SemCrude	FAGADAU ENERGY CORP	\$45,346.65
•	FAIRFIELD OIL & GAS CORP	\$193,998.18
SemCrude SemCrude	FAITH PETROLEUM	\$20,011.99
-	FALCON OIL PROPERTIES	\$19,498.21
SemCrude	FIRST NATIONAL OIL INC	\$78,436.36
SemCrude	FLEETWOOD OIL PROPERTIES	\$19,197.30
SemCrude	FOSSIL CREEK ENERGY CORPORATION	\$21,892.93
SemCrude SemCrude	FRANK & ETHEL KRAMER FAMILY LTD PRINSHP	\$118,060.30
	FULLER PRODUCTION INC	\$21,966.12
SemCrude	G & C PETROLEUM INC	\$18,442.49
SemCrude SemCrude	G & S PRODUCTION	\$20,445.03
SemCrude SemCrude	GABEL OIL COMPANY	\$20,623.85
SemCrude SemCrude	GABS OIL COMPANY	\$21,031.08
SemCrude SemCrude	GAI-JIN TRADING CO LLC	\$90,046.81
SemCrude SemCrude	GARRY DAVIS	\$171,278.96
SemCrude	GEMINI OIL COMPANY	\$275,230.40
SemCrude SemCrude	GENERAL PETROLEUM COMPANY	\$106,567.34
SemCrude SemCrude	GEO AMERICAN RESOURCES INC	\$21,950.47
SemCrude SemCrude	GEORGE M MYERS	\$9,843.39
SemCrude SemCrude	GEORGE W DENNISON SR	\$9,796.32
_	GLB EXPLORATION INC	\$19,407.48
SemCrude	GLENDELL GASKINS	\$19,551.14
SemCrude	GLIMP OIL COMPANY, L.L.C.	\$136,355.86
SemCrude	GMX RESOURCES INC	\$587,014.91
SemCrude	GOULD OIL COMPANY	\$38,680.23
SemCrude	GOOLD OIL COM ANY GP II ENERGY INC	\$357,031.27
SemCrude	GRA-EX, LLC	\$77,927.08
SemCrude	GRAYHORSE OPERATING INC	\$301,991.32
SemCrude	ONA I HONDE OF EXAMINO THE	ŕ

Debtor	Operator Operator	st Purchase 503(b)(9) Amount
SemCrude	GRAYSON OPERATING LLC	\$19,452.35
SemCrude	GRIFFITH PETROLEUM SERVICES INC	\$21,873.38
SemCrude	GROUND DEVELOPMENT COMPANY	\$126,706.27
SemCrude	GUINN COMPANY	\$22,587.89
SemCrude	HINC	\$39,723.44
SemCrude	H&G PETROLEUM CORPORATION	\$20,550.85
SemCrude	HACKER BROS. WELL SERVICE	\$21,625.75
SemCrude	HARRY A SPRING	\$215,499.14
SemCrude	HARVARD PETROLEUM COMPANY, LLC	\$21,962.40
SemCrude	HAWKINS OIL, L.L.C.	\$336,707.54
SemCrude	HAYDEN OPERATING	\$36,686.66
SemCrude	HAZELWOOD PROD & EXPLORATION CO LLC	\$88,424.76
SemCrude	HAZLEWOOD OIL & GAS CO INC	\$21,822.71
SemCrude	HEADINGTON OIL COMPANY LLC	\$3,411,574.73
SemCrude	HELM ENERGY COMPANY	\$17,812.91
SemCrude	HERMAN L LOEB	\$835,113.82
SemCrude	HESSE PETROLEUM COMPANY LLC	\$40,877.79
SemCrude	HEUSTON OIL LLC	\$9,148.43
SemCrude	HLH REALTY INC	\$20,468.64
SemCrude	HOLLAND OIL & GAS	\$20,495.12
SemCrude	HOLLEY OPERATING, L.L.C.	\$21,842.99
SemCrude	HOPE OIL COMPANY	\$107,287.50
SemCrude	HOPPER OIL COMPANY INC	\$81,717.25
SemCrude	HOYT OIL & GAS LLC	\$10,960.80
SemCrude	HUNT ENGINEERING LLC	\$18,254.3
SemCrude	HUNTINGTON ENERGY LLC	\$628,992.19
SemCrude	HUNTON OIL & GAS CORPORATION	\$22,130.0
SemCrude	HUTCHINSON OIL COMPANY	\$20,470.8
SemCrude	ILLINOIS EXPLORATION LLC	\$38,757.9
SemCrude	INGERSOLL OIL INC	\$2,280.2
SemCrude	INGERSOLL ROSINBAUM OIL INC	\$4,351.7
SemCrude	INTERGY PRODUCTION LLC	\$21,219.4

Debtor	Operator 1s	t Purchase 503(b)(9) Amount
SemCrude	IPACT	\$16,968.50
SemCrude	J & D INVESTMENTS	\$81,602.06
SemCrude	J B HERRMANN OIL & GAS	\$19,738.12
SemCrude	J D PITTMAN	\$26,985.96
SemCrude	J D PRODUCTION COMPANY	\$21,526.52
SemCrude	J H BOLAND	\$19,614.90
SemCrude	J&J SERVICES COMPANY	\$20,253.17
SemCrude	JACK CORMAN	\$20,053.56
SemCrude	JACK EXPLORATION INC	\$163,860.17
SemCrude	JACK L HADDOCK	\$19,544.39
SemCrude	JACK THOMPSON	\$19,080.65
SemCrude	JACKIE THOMPSON	\$19,451.01
SemCrude	JACKSON BROTHERS LLC	\$108,028.02
SemCrude	JAMES ALLEN MCGAUGHEY	\$17,764.54 \$15,032,54
SemCrude	JAMES D BULLARD	\$15,032.54
SemCrude	JAMES H. AND ESTER M. BURT	\$3,229.63 \$41,550.55
SemCrude	JAMES W OWENS	\$41,530.55 \$19,800.13
SemCrude	JANUARY INVESTMENTS L.L.C.	\$283,886.55
SemCrude	JASON OIL COMPANY LLC	\$283,880.33 \$299,701.47
SemCrude	JAYHAWK ENERGY INC	\$29,701.47
SemCrude	JEC OPERATING LLC	\$20,582.88
SemCrude	JEFFRIES PUMPING SERVICE INC	\$20,362.66 \$39,949.04
SemCrude	JIM AND JULIE MARTIN	\$99,796.26
SemCrude	JIMMY R CHATHAM	\$1,271,411.43
SemCrude	JMA ENERGY COMPANY LLC	\$20,339.09
SemCrude	JOE B CLIFTON	\$20,339.03 \$19,913.35
SemCrude	JOE B CLIFTON EXPLORATION INC	\$19,481.94
SemCrude	JOE BOB KELLEY AND/OR VICTOR SLONE JR	\$38,465.14
SemCrude	JOE E SHARBER	\$23,419.1
SemCrude	JOHN D PITMAN	\$19,717.4
SemCrude	JOKERS WILD ENERGY LLC	\$84,034.2
SemCrude	JORDAN & JORDAN OIL CO	φυπ,υυπ.ω.

Debtor	Operator 1981	st Purchase 503(b)(9) Amount
SemCrude	JOSIE OIL & GAS INC	\$19,940.82
SemCrude	K C RESOURCES INC	\$22,406.71
SemCrude	K&S OIL COMPANY	\$22,019.30
SemCrude	K. E. SNYDER COMPANY LLC	\$29,986.67
SemCrude	KAHAN AND ASSOCIATES INC	\$935,317.75
SemCrude	KECHI ENERGY LLC	\$21,851.06
SemCrude	KEITH F WALKER OIL & GAS COMPANY LLC	\$313,945.12
SemCrude	KENNETH LOWE	\$41,424.74
SemCrude	KENNETH RAY FIREY	\$9,569.07
SemCrude	KERNS PETROLEUM INC	\$26,302.54
SemCrude	KERR McGEE OIL & GAS ONSHORE LP	\$3,866,259.00
SemCrude	KIMRACO OIL INC	\$20,332.15
SemCrude	KINGERY DRILLING CO INC	\$489,436.07
SemCrude	KLABZUBA OIL & GAS INC	\$231,398.07
SemCrude	KOBY OIL CO LLC	\$105,499.83
SemCrude	KRUMME OIL COMPANY	\$137,279.66
SemCrude	L D HALE	\$42,537.77
SemCrude	LE JONES OPERATING INC	\$46,365.05
SemCrude	L W WILLIAMS JR	\$98,019.36
SemCrude	L WAYNE NEAL AND STELLA L NEAL	\$21,333.06
SemCrude	L WAYNE RAINEY	\$22,133.96
SemCrude	L.C.S. PRODUCTION COMPANY	\$1,729,956.98
SemCrude	LACHENMAYR OIL CO	\$164,305.23
SemCrude	LACKEY & ASSOCIATES	\$21,103.98
SemCrude	LAMAMCO DRILLING COMPANY	\$22,791.70
SemCrude	LANCE LEASING	\$21,357.84
SemCrude	LANCE RUFFEL OIL & GAS CORP	\$1,191,428.51
SemCrude	LANDERS & MUSGROVE OIL CO INC	\$19,235.41
SemCrude	LANDERS AND MUSGROVE OIL	\$19,209.66
SemCrude	LAPETCO INC	\$11,025.88
SemCrude	LARRON OIL CORPORATION	\$39,023.85
SemCrude	LARRY JAMES	\$22,975.52

America - Compunity (Congressing (1999)	Operator (1st Purchase 503(b)(9) Amount
Debtor	Operator LAWRENCE JAMES	\$22,468.38
SemCrude	LEDON TURPIN PRODUCTION	\$76,746.78
SemCrude	LEONARD VANN	\$17,561.99
SemCrude	LESTER DEAN IKENBERRY	\$12,335.90
SemCrude	LEWBET INC	\$19,109.44
SemCrude	LIBERTY OPERATING INC.	\$17,857.13
SemCrude	LINDELL HOWARD AND JACK TAYLOR	\$86,926.27
SemCrude	LINDENMUTH & ASSOCIATES, INC	\$39,993.88
SemCrude	LINK OIL COMPANY	\$41,901.78
SemCrude	LINK OIL COMPANT LINN ENERGY HOLDINGS	\$74,770.45
SemCrude	LITTLE BEAR RESOURCES LLC	\$158,492.95
SemCrude		\$21,500.14
SemCrude	LLOYD RICHARDSON	\$22,084.10
SemCrude	LOBAR OIL CO INC	\$20,802.55
SemCrude	LOBO EXPLORATION INC	\$19,597.07
SemCrude	LONG DRILLING CO	\$21,792.92
SemCrude	LORENTZ OIL & GAS L L C	\$20,235.59
SemCrude	LOWRY EXPLORATION INC	\$71,211.43
SemCrude	LUKE OIL COMPANY	\$52,320.56
SemCrude	LUNAR INVESTMENTS INC	\$21,292.82
SemCrude	M CRAIG DEISENROTH	\$148,483.44
SemCrude	M E BOOTH CORPORATION	\$19,677.48
SemCrude	M E KLEIN & ASSOCIATES INC	\$42,898.07
SemCrude	M&M ENERGY	\$23,028.13
SemCrude	MA & MC OIL & GAS LLC	\$39,110.00
SemCrude	MACKEY CONSULTING & LEASING	\$6,252.18
SemCrude	MAJESTIC OIL & GAS	\$44,485.78
SemCrude	MAP PRODUCTION COMPANY INC	\$21,610.51
SemCrude	MARCON OPERATING COMPANY INC	\$18,873.05
SemCrude	MARK E DOBRINSKI	\$17,832.53
SemCrude	MARK HAYNES	\$17,032.33 \$16,994.25
SemCrude	MARK HOLLOWAY INC	\$10,394.23 \$19,391.26
SemCrude	MARLIN OIL CORPORATION	# 19,091.20

munecongrame Leng Flaggaria	Operator Services	1st Purchase 503(b)(9) Amount
Debtor	MAYABB OIL COMPANY	\$48,166.98
SemCrude	MCCOY PETROLEUM CORPORATION	\$1,563,687.91
SemCrude	MCGINNESS OIL CO OF KS	\$20,555.57
SemCrude	MCM ENERGY INC.	\$20,321.14
SemCrude	MEADOWBROOK OIL CORP OF OKLAHOMA	\$19,498.52
SemCrude	MESA EXPLORATION COMPANY INC	\$109,959.01
SemCrude	MICKEY L PRUITT	\$78,032.15
SemCrude	MICKEY PRUITT	\$20,241.72
SemCrude	MID CONTINENT ENERGY CORP	\$194,861.92
SemCrude	MIDCO EXPLORATION	\$20,174.50
SemCrude	MID-CONTINENT ENERGY OPERATING CO	\$46,972.58
SemCrude	MID-CONTINENT EXERGY OF EXCURNING GO	\$85,660.65
SemCrude	MMG OIL COMPANY	\$983.62
SemCrude	MOLITOR OIL	\$77,466.85
SemCrude	MOLITOR OIL MONROE NATURAL GAS INC	\$20,041.51
SemCrude	MONTGOMERY EXPLORATION COMPANY	\$19,334.19
SemCrude	MORTON PRODUCTION INC	\$80,395.15
SemCrude	MSN OPERATING	\$20,821.94
SemCrude	MSN OFERATING M-TEX RESOURCES INC	\$22,839.54
SemCrude	MURFIN DRILLING COMPANY INC	\$72,959.98
SemCrude	MURFIN DRILLING COMPANY INC.	\$19,608.60
SemCrude	MUSTANG FUEL CORPORATION	\$207,662.80
SemCrude	N F PRODUCERS	\$19,445.03
SemCrude	NACOGDOCHES OIL & GAS LLC	\$99,054.68
SemCrude	NACOGDOCHES OIL & GAIO BLO NADEL & GUSSMAN ENERGY LLC	\$104,051.25
SemCrude	NADEL & GUSSMAN PERMIAN LLC	\$136,519.39
SemCrude	NATIONAL PIONEER OIL COMPANY LLC	\$19,571.59
SemCrude	NEW DOMINION LLC	\$3,789,223.65
SemCrude	NEWCASTLE RESOURCES, LLC	\$19,902.12
SemCrude	NEWELL OIL & GAS	\$101,589.85
SemCrude	NEWELL OIL & GAS NFR ENERGY LLC	\$67,787.34
SemCrude	NO WHERE OIL COMPANY	\$62,242.62
SemCrude	NO MHEKE OIL COMPANI	,

	Operator.	st Purchase 503(b)(9) Amount
Debtor SemCrude	NOBLE	\$285,063.00
SemCrude SemCrude	NOBLE ENERGY INC	\$872,160.17
-	NORTHBROOK OIL & GAS CO	\$71,943.75
SemCrude	NOVO EXPLORATION CORPORATION	\$40,732.70
SemCrude	NOVY OIL & GAS INC	\$166,543.00
SemCrude	NYTEX ENERGY LLC	\$42,143.54
SemCrude	O HILL ENTERPRISES, INC.	\$17,062.43
SemCrude	OAK HILLS ENERGY INC	\$22,240.09
SemCrude	OIL COMPANY OF AMERICA, INC.	\$104,019.27
SemCrude	OKLAHOMA ENERGY LLC	\$65,392.64
SemCrude	OKLAHOMA LIQUID RESOURCES INC	\$20,166.45
SemCrude	OKLAHOMA OIL & GAS MANAGEMENT INC	\$376,468.15
SemCrude	OSBORN HEIRS COMPANY	\$94,839.74
SemCrude	OVERLAND ENERGY INC	\$18,562.33
SemCrude	OXCO LLC	\$20,443.33
SemCrude	P & H CONSTRUCTION INC	\$21,068.48
SemCrude	PADA PETROLEUM INC.	\$22,069.40
SemCrude	PAMAVETTA OIL COMPANY	\$9,976.82
SemCrude	PANAVETTA OIL CONTANT PARAGON INDUSTRIES INC	\$43,926.90
SemCrude	PATRIOT PETROLEUM CO LLC	\$39,146.59
SemCrude	PATRIOT PETROLEOM CO LLC PATRON ENERGY LLC	\$22,310.15
SemCrude	PATRON ENERGY LDC PATTERSON OIL COMPANY INC	\$78,168.04
SemCrude	PAUL VAN ORSDOL	\$20,608.11
SemCrude	• • •	\$20,134.88
SemCrude	PEAK ENERGY LLC	\$217,825.87
SemCrude	PEDESTAL OIL COMPANY INC	\$21,524.77
SemCrude	PEEWE OIL CO.	\$17,289.93
SemCrude	PEMEX INC	\$211,634.90
SemCrude	PERKINS ENERGY COMPANY	\$945,581.10
SemCrude	PETCO PETROLEUM CORP	\$36,821.61
SemCrude	PETRO MAC CORPORATION	\$183,986.72
SemCrude	PETRO WARRIOR LLC	\$44,280.73
SemCrude	PETROLEUM ENERGY CORPORATION	\$ 1.320 VVV

consecution of the constitution of the constit	Operator Operator	st Purchase 503(b)(9) Amount
Debtor	PETROLEUM PROPERTY SERVICES	\$19,213.48
SemCrude	PHILLIPS EXPLORATION COMPANY LC	\$159,490.47
SemCrude	PHOENIX OIL & GAS INC	\$878,600.14
SemCrude	PHOENIX OIL & GAS INC PHOENIX PETROCORP INC	\$575,885.71
SemCrude	PICKRELL DRILLING CO	\$369,795.86
SemCrude	PIERCE PRODUCTION CO	\$164,010.65
SemCrude		\$23,149.00
SemCrude	PIONEER NATURAL	\$57,276.44
SemCrude	PIONEER OIL COMPANY INC PLATTE VALLEY OIL COMPANY INC	\$179,845.16
SemCrude		\$19,570.39
SemCrude	PLOTNER OIL COMPANY	\$19,277.63
SemCrude	PONTOTOC PRODUCTION CO INC	\$428,265.59
SemCrude	POPP OPERATING INC	\$22,300.75
SemCrude	PREMIER ENERGY LLC	\$21,136.98
SemCrude	PREMIER PETROLEUM INC.	\$1,939,056.57
SemCrude	PRIMA EXPLORATION INC	\$352,747.00
SemCrude	PRODUCERS CRUDE	\$128,012.83
SemCrude	PROTEGE ENERGY LLC	\$56,832.28
SemCrude	PSA INC	\$139,202.33
SemCrude	PULLERS OIL CO LLC	\$93,335.07
SemCrude	Q T OIL LTD CO	\$203,775.96
SemCrude	QUINQUE OPERATING COMPANY	\$20,074.65
SemCrude	R D FINNESY OIL COMPANY	\$21,842.57
SemCrude	R J SPERRY CO	\$220,889.17
SemCrude	R L INVESTMENT LLC	\$18,901.21
SemCrude	RADELL W KOEHN	\$58,410.39
SemCrude	RAINBOW OILWELL SERVICE INC	\$61,609.86
SemCrude	RA-JAC INC	\$817,597.46
SemCrude	RAMA OPERATING CO INC	\$81,999.83
SemCrude	RANDALL E JAMES	\$19,811.71
SemCrude	RANDALL H BAXTER	\$84,032.75
SemCrude	RANDON PRODUCTION COMPANY LLC	\$236,104.27
SemCrude	RANDY AND JAN WALKER OIL COMPANY LLC	4200,100

Debtor	Operator	lst Purchase 503(b)(9) Amount
SemCrude	RANGE PROD	\$232,423.07
SemCrude	RANKEN ENERGY CORPORATION	\$149,869.78
SemCrude	RAYDON EXPLORATION INC	\$621,036.38
SemCrude	RAYMOND H JONES FAMILY TRUST	\$48,571.59
SemCrude	RC HOLDINGS LLC	\$40,507.36
SemCrude	RDT PROPERTIES INC	\$70,139.87
SemCrude	RED FORK PRODUCTION LLC	\$40,879.44
SemCrude	RED OAK ENERGY INC	\$60,764.46
SemCrude	REDDY OIL & GAS CORP	\$98,471.96
SemCrude	REDDY OIL & GAS CORPORATION	\$3,790.80
SemCrude	REDLAND RESOURCES INC.	\$43,027.16
SemCrude	REGINALD D FISHER	\$40,153.12
SemCrude	REID WILLIAMS OIL CO	\$19,492.04
SemCrude	REID WILLIAMS OPERATING INC	\$22,454.54
SemCrude	REIF OIL & GAS LLC	\$38,402.27
SemCrude	REIN RESOURCES	\$80,141.10
SemCrude	REPUBLIC RESOURCES CORP	\$19,320.66
SemCrude	RESEARCH OIL COMPANY INC	\$19,423.58
SemCrude	RESERVE MANAGEMENT INC	\$39,570.72
SemCrude	RESOURCE OPERATIONS INC	\$216,719.61
SemCrude	RICHARD L SMITH	\$9,745.69
SemCrude	RITCHIE EXPLORATION INC	\$143,014.90
SemCrude	RKK PRODUCTION COMPANY	\$39,578.46
SemCrude	RKR EXPLORATION INC	\$39,538.84
SemCrude	ROBERSON OIL COMPANY INC	\$46,209.81
SemCrude	ROBERT W PASSMORE	\$85,143.14
SemCrude	ROBINOWITZ OIL COMPANY	\$193,026.15
SemCrude	ROBINSON ENERGY & EXPL INC	\$127,454.47
SemCrude	ROBINSON OIL COMPANY	\$41,886.97
SemCrude	RODGER E SPARKMAN	\$20,609.15
SemCrude	ROGER KLINGBERG	\$43,464.88
SemCrude	ROGERS COUNTY COAL SEAM LLC	\$18,470.84

	Oberator 2	ist Purchase 503(b)(9) Amount
Debtor SemCrude	RON GRIGGS	\$22,699.70
SemCrude SemCrude	ROSEWOOD RESOURCES INC	\$78,491.41
SemCrude SemCrude	ROSS HOENER INC.	\$20,174.90
SemCrude SemCrude	ROUGEOT OIL & GAS CORP	\$133,432.47
SemCrude SemCrude	ROX EXPLORATION INC	\$44,328.30
SemCrude SemCrude	ROXANNA OIL CO	\$79,097.31
SemCrude SemCrude	RUSK ENERGY LTD	\$23,761.09
SemCrude SemCrude	RUSSELL D MELOY	\$41,881.86
SemCrude SemCrude	RUTLEDGE - NEWSOM OIL CO	\$21,457.11
SemCrude SemCrude	S J MURPHY OIL & GAS INC	\$21,564.51
SemCrude SemCrude	SAJAC ENERGY COMPANY L.L.C.	\$21,491.66
SemCrude SemCrude	SALLEE OIL CORPORATION	\$98,924.65
SemCrude SemCrude	SAMSON RESOURCES CO	\$200,423.86
SemCrude SemCrude	SAND POINT PRODUCTION INC	\$41,537.48
SemCrude SemCrude	SAND RESOURCES INC	\$42,245.42
SemCrude SemCrude	SANDRIDGE ENERGY INC	\$888,813.58
SemCrude SemCrude	SANDRIDGE TERTIARY LLC	\$23,601.43
SemCrude	SANDSTONE RESOURCES INC	\$18,223.26
SemCrude SemCrude	SCHANKIE WELL SERVICE	\$216,834.09
SemCrude	SCOTT-BANES-SCOTT PARTNERSHIP	\$9,675.01
SemCrude	SEEKER LLC	\$40,188.60
SemCrude	SELCO DRILLING INC	\$18,663.42
SemCrude	SEMINOLE TANK SERVICE	\$49,043.57
SemCrude	SHIDELER ENTERPRISES INC	\$19,879.66
SemCrude	SHIELDS OPERATING INC.	\$20,108.74
SemCrude	SHORT AND SHORT LLC	\$77,911.38
SemCrude	SHORT OIL COMPANY	\$142,865.15
SemCrude	SISLER BROTHERS	\$23,352.52
SemCrude SemCrude	SLAWSON EXPLORATION CO INC	\$450,966.63
SemCrude SemCrude	SNEDCO INC.	\$21,038.08
SemCrude	SNYDER PARTNERS	\$208,766.06
SemCrude	SOONER TREND EXPLORATION INC	\$19,678.30
Schickade		

	Operator	1st Purchase 503(b)(9) Amount
Debtor	SOUTH BURBANK PETROLEUM CORP	\$781,299.58
SemCrude	SOUTHERN PRODUCTION	\$14,239.08
SemCrude	SOUTHERN RESOURCES INC	\$258,027.60
SemCrude SemCrude	SOUTHLAKE EXPLORATION INC	\$64,857.30
	SOUTHWEST GAS SYSTEMS, INC.	\$19,728.97
SemCrude SemCrude	SPALDING ENERGY, INC.	\$21,459.27
-	SPEARS OIL COMPANY	\$11,154.82
SemCrude	SPECIAL ENERGY CORPORATION	\$84,860.46
SemCrude	SPESS OIL COMPANY INC	\$128,229.41
SemCrude	SPRING OPERATING COMPANY	\$23,689.68
SemCrude	ST MARY LAND & EXPLORATION CO	\$110,972.04
SemCrude	ST MARY LAND & EXPLORATION COMPANY	\$21,143.49
SemCrude SemCrude	STAGHORN ENERGY LLC	\$149,345.56
	STAMPER OPERATING CO INC	\$21,319.00
SemCrude SemCrude	STAR PRODUCTION INC	\$1,036,578.47
SemCrude SemCrude	STEPHENS & JOHNSON OPERATING	\$893,509.83
SemCrude SemCrude	STEPHENS EXPLORATION INC	\$21,014.55
SemCrude SemCrude	STEVE W MEYER	\$18,742.76
SemCrude SemCrude	STEVENSON OPERATING LLC	\$22,101.63
SemCrude SemCrude	STIDHAM BOB & ALICE	\$20,818.15
SemCrude SemCrude	SUBSURFACE OPERATIONS	\$20,417.07
SemCrude SemCrude	SUBSURFACE PRODUCTION INC	\$20,165.83
•	SUHAIRD	\$61,803.76
SemCrude	SULLIVAN AND COMPANY LLC	\$925,126.34
SemCrude	SUMMIT OPERATING	\$22,234.62
SemCrude	SUNDOWN ENERGY INC	\$22,944.09
SemCrude	SYNERGY OIL & GAS LP	\$50,247.78
SemCrude	T & C MFG & OPERATING, INC.	\$21,325.29
SemCrude	T&LOILINC	\$20,106.64
SemCrude	T M ENTERPRISES INC	\$19,360.50
SemCrude	TECHNICAL ENERGY SERVICES	\$39,250.48
SemCrude	TED W MCGONIGAL	\$9,874.00
SemCrude	TED M MCGOLIGAT	•••

THE RESTRICTION OF THE PROPERTY OF	Operator,	lst Purchase 503(b)(9) Amount
Debtor	Operator TEMPEST	\$23,025.13
SemCrude	TE-RAY RESOURCES INC	\$80,788.95
SemCrude	TGT PETROLEUM CORP	\$351,048.52
SemCrude	THE BALE CORPORATION	\$25,502.58
SemCrude	THE GLOCO LLC	\$174,534.51
SemCrude	THOMAS D ROBBINS	\$30,771.33
SemCrude	THOMAS D RODDING THOMSEN BROTHERS OIL CO.	\$24,292.57
SemCrude	THOROUGHBRED ASSOCIATES LLC	\$423,680.65
SemCrude	THREE SANDS INVESTMENTS INC	\$20,291.75
SemCrude	TILLEY OIL & GAS INC	\$44,277.03
SemCrude	TIM JOHNSON	\$19,840.97
SemCrude	TIM TUCKER	\$41,960.30
SemCrude	TITAN ENERGY INC	\$1,113,276.54
SemCrude	TITAN RESOURCES LIMITED	\$64,188.59
SemCrude	TKO INVESTMENT COMPANY	\$19,424.92
SemCrude	TOKLAN OIL & GAS CORPORATION	\$62,988.79
SemCrude	TRACKER MINERAL	\$42,889.79
SemCrude	TRACY KENNEDY	\$9,912.75
SemCrude	TRENCHCO INC	\$19,674.15
SemCrude	TRIAD ENERGY INC	\$243,376.48
SemCrude	TRIAD ENERGY INC TRIPOWER RESOURCES INC	\$445,321.41
SemCrude	TROY JACO INC	\$97,801.81
SemCrude	TWILA MAE WILSON ESTATE	\$11,422.16
SemCrude	UMBERHAM OIL COMPANY INC	\$31,279.60
SemCrude	UNIT PETROLEUM COMPANY	\$19,392.42
SemCrude	UNIT PETROLEUM COMPANY LLC UNITED PRODUCTION COMPANY LLC	\$169,826.62
SemCrude		\$143,569.21
SemCrude	UPLANDS RESOURCES INC UPTON RESOURCES USA INC	\$899,967.68
SemCrude		\$18,531.00
SemCrude	VALIENT OIL & INVESTMENT CO.	\$20,353.60
SemCrude	VALIOHN ENERGY CORPORATION	\$41,844.78
SemCrude	VALLEY PETROLEUM COMPANY	\$20,287.31
SemCrude	VAMCO OPERATIONS	way y gar y 1 cm h

AMASS STOREST OF BUILDING		Lst Purchase 503(b)(9) Amount
i) Ebebtoπ	是自然的。由于全国的自然的自然的特别,是使用的自然是一个的自然的。	\$18,925.07
SemCrude	VAN OIL	\$257,780.49
SemCrude	VAUGHN GOOD OIL COMPANY LLC	\$476,008.46
SemCrude	VEENKER RESOURCES INC	\$77,699.68
SemCrude	VENIS EXPLORATION COMPANY	\$29,753.15
SemCrude	VERNON BROWN	\$61,027.86
SemCrude	VIATHON HYDROCARBONS, INC.	\$18,992.81
SemCrude	VICKIE R CHATHAM	\$37,991.52
SemCrude	VICTOR W PRYOR JR	\$17,428.10
SemCrude	VIKE OIL INC	\$83,618.45
SemCrude	VIKING RESOURCES INC	\$24,676.79
SemCrude	VIN INC	\$1,340,213.77
SemCrude	VINCENT OIL CORPORATION	\$127,480.78
SemCrude	VJI NATURAL RESOURCES	\$188,345.20
SemCrude	W D SHORT	\$19,425.13
SemCrude	W H PRICE OIL PROPERTIES	\$17,318.26
SemCrude	W J M OIL INCORPORATED	\$94,088.97
SemCrude	W M MONTGOMERY	\$18,230.16
SemCrude	W M PETROLEUM	\$19,750.34
SemCrude	W ROBERT WILSON	\$217,379.58
SemCrude	WABASH ENERGY CORP	\$57,956.15
SemCrude	WADI EXPLORATION AND PRODUCTION LLC	\$80,873.79
SemCrude	WALCHER OIL LC	\$21,350.77
SemCrude	WALSH & WATTS, INC.	\$602,185.94
SemCrude	WALSH PRODUCTION INC	\$54,658.97
SemCrude	WALTERS OIL COMPANY	\$57,033.03
SemCrude	WALTMAN FAMILY LTD PRTNRSHP	\$37,033.03 \$44,248.24
SemCrude	WARD PETROLEUM CORPORATION	·
SemCrude	WAYNE D STOKES	\$18,040.42
SemCrude	WEINKAUF PETROLEUM INC	\$387,123.08
SemCrude	WELLCO ENERGY INC	\$51,966.02
SemCrude	WELLSTAR CORPORATION	\$195,826.41
SemCrude	WENTWORTH OPERATING COMPANY	\$19,158.48

	Öperator	lst Purchase 503(b)(9) Amount
Debtor	WEST TEXAS ROYALTIES INC	\$57,730.14
SemCrude	WEST TEXAS ROTALTILLS INC. WFD OIL CORP	\$20,463.81
SemCrude	WHEELER OIL COMPANY	\$20,292.51
SemCrude	WHITDALE INC	\$20,669.62
SemCrude	WHIT DALE INC WHITE EXPLORATION INC	\$681,405.44
SemCrude	WHITE OPERATING CO	\$665,742.86
SemCrude	WHITE OFERATING CO WHITE PINE PETROLEUM CORP	\$78,545.43
SemCrude	WHITE FINE PETROLLOM CORR WHITEHEAD PRODUCTION COMPANY, INC	\$61,095.75
SemCrude	WHITEHEAD FRODUCTION COMPANY, INCOMPANY, INC	\$411,288.18
SemCrude	WILDHORSE OIL & GAS CORP	\$122,151.94
SemCrude	WILLIAM A SNEED	\$10,330.26
SemCrude	WILLIAM A SNEED WILLIAM H ZUHONE	\$29,744.24
SemCrude	WILLIAM M PARK	\$52,754.69
SemCrude	WILLIAM IM FARK WILLIAM R EARNHARDT COMPANY	\$100,415.63
SemCrude	WILLIAMS PROD	\$3,966,394.79
SemCrude	WILLIAMS PRODUCTION ROCKY MOUNTAIN CO	\$23,990.87
SemCrude	WILLIFORD ENERGY CO	\$82,723.17
SemCrude	WILLIFORD ENERGY CO WILTON PETROLEUM INC	\$37,934.26
SemCrude	WYNN-CROSBY 1998 LTD	\$68,956.27
SemCrude	XANADU EXPLORATION COMPANY	\$130,984.65
SemCrude	XTO ENERGY INC	\$25,287.74
SemCrude	ZENERGY INC	\$19,145.69
SemCrude	ZENITH DRILLING COMPANY	\$39,408.05
SemCrude	ZEPHYR OPERATING CO LLC	\$109,168.63
SemCrude	606 Producers	\$88,953,124.93
SemCrude Total	000 I tourners	
9 C	ACCT OF NANCY LOUISE GILL MONFORD	\$35,283.69
SemGas	ALTEX ENERGY CORPORATION	\$193,197.26
SemGas	ALTEX RESOURCES INC	\$883,680.91
SemGas	AMERICAN WARRIOR INC	\$6,917.01
SemGas	APOLLO ENERGIES INC	\$2,040.19
SemGas	ARIES RESOURCES LLC	\$14,114.87
SemGas	ARIES RESOURCES LLC	

Debtor	Operator	1st Purchase 503(b)(9) Amount
SemGas	ARROW OIL & GAS INC	\$7,845.28
SemGas SemGas	ATOKA OPERATING INC	\$168,765.17
SemGas SemGas	BARON EXPLORATION COMPANY	\$28,356.65
SemGas SemGas	BECKER OIL CORPORATION	\$166,658.78
SemGas	BEREN CORPORATION	\$1,974.17
SemGas	BKM PRODUCTION	\$538.70
SemGas SemGas	BLACK OIL EXPLORATION &PRODUCTION II LTD	\$351,510.62
SemGas SemGas	BLACKWELL BMC LP	\$495,413.18
-	BRECK OPERATING CORP	\$34,122.27
SemGas	CARMEN SCHMITT INC	\$2,299.61
SemGas	CHESAPEAKE EXP. LTD PARTNERSHIP	\$572,612.91
SemGas	CIMAREX ENERGY CO	\$49,522.04
SemGas	CMX INC	\$7,726.63
SemGas	DANNY MAYO	\$15,689.13
SemGas SemGas	DAYSTAR PETROLEUM INC	\$3,161.86
	DEUTSCH OIL COMPANY	\$5,459.36
SemGas	DOOLAREE CORPORATION	\$45,335.75
SemGas SemGas	EAGLE OIL & GAS COMPANY	\$7,609.74
	EDMISTON OIL CO INC	\$19,407.32
SemGas	ELIAS THOMAS	\$5.11
SemGas	ENMARK GAS GATHERING LP	\$8,030.45
SemGas	F G HOLL COMPANY LLC	\$107,832.63
SemGas	FELDERHOFF PRODUCTION COMPANY	\$20,095.92
SemGas	FRED J NUSS	\$4,681.32
SemGas	HARVEY CLARK	\$780.35
SemGas	HEAPE EXPLORATION INC	\$205,039.19
SemGas	HEP OIL COMPANY	\$25,275.10
SemGas		\$348.33
SemGas	HERMAN L LOEB	\$9,869.04
SemGas	HUTCHINSON OIL CO	\$7,642.87
SemGas	INDIAN OIL CO INC	\$8,530.47
SemGas	JETTA OPERATING COMPANY INC	\$4,990.64
SemGas	JOE PARR	ψ 1 ₉ 270.01

Debtor	Operator	Lst Purchase 503(b)(9). Amount
SemGas	JOHN O FARMER INC	\$2,955.82
SemGas	JRC PRODUCTION, LLC	\$580.70
SemGas	KEPCO INC	\$202.29
SemGas	KNIGHTON OIL COMPANY INC	\$8,665.04
SemGas	L D DRILLING INC	\$6,632.24
SemGas	LUMEN MIDSTREAM PARTNERSHIP LLC	\$60,095.30
SemGas	MCCOY PETROLEUM CORPORATION	\$477,407.35
SemGas	MID CONTINENT ENERGY CORP	\$3,783.48
SemGas	MID-CONTINENT ENERGY CORP	\$7,742.32
SemGas	MTM PETROLEUM INC	\$1,951.67
SemGas	NORTHSTAR INVESTMENTS, INC.	\$6,504.26
SemGas	OIL PRODUCERS OF KANSAS INC	\$2,249.91
SemGas	OKT PETROLEUM COMPANY INC	\$43,382.11
SemGas	PERRY LYNN LARSON	\$18,842.19
SemGas	PETROCORP INC	\$17,495.07
SemGas	PETROLEUM TECHNOLOGIES INC	\$7,629.26
SemGas	PINTAIL PETROLEUM LTD	\$15,069.41
SemGas	PRATT WELL SERVICE INC	\$4,638.62
SemGas	PRECISION OPERATING INC	\$1,616.69
SemGas	RAMA OPERATING CO INC	\$10,025.90
SemGas	REDWING GAS SYSTEMS, INC	\$3,645.10
SemGas	RITCHIE EXPLORATION INC	\$1,023.39
SemGas	ROBERT B ROWLAND	\$5,572.68
SemGas	RONALD PRATER TRUST	\$3,766.41
SemGas	SAND SPRINGS OIL & GAS CO	\$10,021.27
SemGas	SANDRIDGE ENERGY INC	\$41,437.50
SemGas	SIROKY OIL MANAGEMENT	\$10,970.45
SemGas	SPECIAL ENERGY CORPORATION	\$1,894,528.56
SemGas	STARLIGHT RIDGE COMPANY	\$752.11
SemGas	STEVE JONES	\$2,130.44
SemGas	THE QUINTIN LITTLE COMPANY	\$47,423.91
SemGas	TRANSPACIFIC OIL CORP.	\$5,226.27

TOTAL	708 Producers	\$141,721,082.79
SemGas Total	78 Producers	\$6,466,637.20
SemGas	ZENITH DRILLING COMPANY	\$5,807.60
SemGas	YOUNGER ENERGY COMPANY	\$65,539.10
SemGas	WW OIL & GAS LLC	\$6,953.83
SemGas	WHITE & ELLIS DRILLING INC	\$7,494.37
SemGas	WESTAR ENERGY CORP.	\$1,499.10
SemGas	VJI NATURAL RESOURCE INC	\$3,720.19
SemGas	VINCENT OIL CORPORATION	\$47,699.26
SemGas	TROY JACO INC	\$81,591.55
Debtor	Operator	1st Purchase 503(b)(9) Amount

Schedule 2

Secured First Purchaser Producer Claims (Listed By Debtor)

2000 11 S (2) P 2000 1				rst Purchase Amount			
Debtor	Operator	Kansas			Texas		Total
EaglWing EaglWing	*L D DRILLING	\$20,586.23	\$0.00	\$0.00	\$0.00	\$0.00	\$20,586.23
EaglWing	BRIGHTON ENERGY LLC	\$0.00	\$0.00	\$90,396.2 6	\$66,435.01	\$0.00	\$156,831.27
EaglWing	DAVIS PETROLEUM CORP	\$1,314,862.43	\$0.00	\$0.00	\$0.00	\$0.00	\$1,314,862.43
EaglWing	DINERO OPERATING CO	\$0.00	\$0.00	\$0.00	\$114,021.52	\$0.00	\$114,021.52
EaglWing	GREAT PLAINS ENERGY INC	\$1,597,219.98	\$0.00	\$0.00	\$0.00	\$0.00	\$1,597,219.98
EaglWing	HESS OIL COMPANY	\$41.61	\$0.00	\$0.00	\$0.00	\$0.00	\$41.61
EaglWing	KCS RESOURCES INC	\$0.00	\$0.00	\$0.00	\$46,043.11	\$0.00	\$46,043.11
EaglWing	LANDMARK RESOURCES INC	\$3,532,550.86	\$0.00	\$0.00	\$0.00	\$0.00	\$3,532,550.86
EaglWing EaglWing	LD DRILLING	\$4,200,073.60	\$0.00	\$0.00	\$0.00	\$0.00	\$4,200,073.60
EaglWing EaglWing	MCCOY OIL & GAS LLC,	\$2,404.15	\$0.00	\$0.00	\$0.00	\$0.00	\$2,404.15
Eagiwing	MCCOY PETROLEUM CORPORATION	\$5,279,915.87	\$0.00	\$198,446.85	\$0.00	\$0.00	\$5,478,362.72
EaglWing	MEDALLION PETROLEUM INC	\$0.00	\$0.00	\$22,448.09	\$0.00	\$0.00	\$22,448.09
EaglWing	MULL DRILLING CO INC	\$2,427,874.50	\$0.00	\$0.00	\$0.00	\$0.00	\$2,427,874.50
EaglWing	MURFIN DRILLING COMPANY INC	\$26,520,568.81	\$0.00	\$0.00	\$0.00	\$0.00	\$26,520,568.81
EaglWing	PETROHAWK PROPERTIES LP	\$0.00	00.02	\$41,517.87	\$0.00	\$0.00	\$41,517.87
EaglWing		\$0.00	\$0.00	\$0.00	\$710,674.65	\$0.00	\$710,674.65
EaglWing	POGO PRODUCING COMPANY	\$537,295.09	\$0.00	\$0.00	\$0.00	\$0.00	\$537,295.09
EaglWing	PROLIFIC RESOURCES LLC	\$0.00	\$0.00	\$0.00	\$2,789.43	\$0.00	\$2,789.43
EaglWing	PURE PARTNERSHIP LLP	\$272,335.87	\$0.00	\$0.00	\$0.00	\$0.00	\$272,335.87
EaglWing	REIF OIL & GAS LLC	\$0.00	\$0.00	\$4,270.75	\$0.00	\$0.00	\$4,270.75
EaglWing	RICKS EXPLORATION INC	\$0.00	\$0.00	\$0.00	\$468,095.32	\$0.00	\$468,095.32
EaglWing	SAMSON CONTOUR ENERGY E&P LLC	\$0.00	\$0.00	\$0.00	\$17,492,674.90	\$0.00	\$17,492,674.90
EaglWing	SAMSON LONE STAR LLC		\$5,539,975.15	\$7,189,918.45	\$0.00	\$1,634,114.70	\$14,535,959.89
EaglWing	SAMSON RESOURCES CO	\$171,951.59 \$173,160.47	\$0.00	\$0.00	\$0.00	\$0.00	\$173,160.47
EaglWing	STELLA ELIZABETH GRIMSLEY		\$0.00	\$0.00	\$0.00	\$0.00	\$22,238,319.70
EaglWing	VESS OIL CORP	\$22,238,319.70	\$0.00	\$0.00	\$0.00	\$0.00	\$1,775,475.74
EaglWing	VIKING RESOURCES INC	\$1,775,475.74	\$0.00 \$0.00	\$2,021,974.21	\$0.00	\$0.00	\$2,083,117.54
EaglWing	ZENERGY INC	\$61,143.33	\$0.00	\$40,982.19	\$0.00	\$0.00	\$40,982.19
EaglWing	ZINKE & TRUMBO LIMITED	\$0.00 \$70,125,779.83	\$5,539,975.15	\$9,609,954.67	\$18,900,733.94	\$1,634,114.70	\$105,810,558.29
EaglWing Total	28 Producers	\$70,125,779.85	33,337,773-13	35,007,554.07	310,200,700131	0.1,00 1,1	, ,
	4.50 1.500 0.01 0.0	\$0.00	\$0.00	\$21,563.01	\$0.00	\$0.00	\$21,563.01
SemCrude	*BERTSCHINGER OIL CO	\$0.00	\$0.00	\$87,299.70	\$0.00	\$0.00	\$87,299.70
SemCrude	*CLAUD C HARP		\$0.00	\$28,563.40	\$0.00	\$0.00	\$28,563.40
SemCrude	*DAVID WIKEL	\$0.00	\$0.00	\$47,210.21	\$0.00	\$0.00	\$47,210.21
SemCrude	*FREDERICK F DRUMMOND	\$0.00		\$21,755.27	\$0.00	\$0.00	\$21,755.27
SemCrude	*KEITH LOVELL	\$0.00	\$0.00	\$303.81	\$0.00	\$0.00	\$303.81
SemCrude	*MIDWEST OIL COMPANY	\$0.00	\$0.00		\$0.00	\$0.00	\$631.01
SemCrude	*QUANTUM OPERATING COMPANY	\$631.01	\$0.00	\$0.00	\$0.00	\$0.00	\$0.29
SemCrudo	*RANDALL POGUE	\$0.00	\$0.00	\$0.29	\$0.00	\$0.00	\$52.59
SemCrude	3 B C CO INC	\$0.00	\$0.00	\$52.59		\$0.00	\$92,184.69
SemCrude	3 G OIL COMPANY	\$92,184.69	\$0.00	\$0.00	\$0.00		\$26.31
SemCrude	3-D OIL & GAS	\$0.00	\$0.00	\$0.00	\$26.31	\$0.00	\$31,748.40
SemCrude	3M PRODUCTION, INC.	\$0.08	\$0.00	\$31,748.40	\$0.00	\$0.00	\$1,746.40 \$185.05
SemCrude	4-D PETROLEUM CONSULTANTS	\$185.05	\$0.00	\$0.00	\$0.00	\$0.00	\$1,258.72
SemCrude	4S OIL	\$0.00	\$0.00	\$0.00	\$1,258.72	\$0.00	
SemCrude	56 OIL & GAS LTD	\$0.00	\$0.00	\$545.80	\$0.00	\$0.00	\$545.80
SemCrude	A & T PRODUCTION COMPANY	\$0.00	\$0.00	\$82.08	\$0.00	\$0.00	\$82.08
SemCrude	A A S OIL COMPANY INC	\$1,104.88	\$0.00	\$0.00	\$0.00	\$0.00	\$1,104.88
SemCrude	A C AILEY	\$0.00	\$0.00	\$33,100.09	\$0.00	\$0.00	\$33,100.09
SemCrude	A E I CORP - OKLAHOMA	\$0.00	\$0.00	\$545.39	\$0.00	\$0.00	\$545.39
Demonde	A STATE OF THE PARTY OF THE PAR						

asosonasioninassassas			Fi	rst Purchase Amounts i	n Active States		
Debtor	Operator		New Mexico		Texas		Total
SemCrude	A L MARTIN & BETTY MARTIN	\$0.00	\$0.00	\$3,087.11	\$0.00	\$0.00	\$3,087.11
SemCrude	A L RAINWATER	\$0.00	\$0.00	\$9,467.43	\$0.00	\$0.00	\$9,467.43 \$199.04
SemCrude	A W SWIFT & CO	\$0.00	\$0.00	\$199.04	\$0.00	\$0.00	
SemCrude	A. G. CLAUSEN	\$0.00	\$0.00	\$0.00	\$111.56	\$0.00	\$111.56
SemCrude	AARON OIL CO	\$0.00	\$0.00	\$17,305.39	\$0.00	\$0.00	\$17,305.39
SemCrude SemCrude	ABC OIL & GAS	\$0.00	\$0.00	\$3,977.90	\$0.00	\$0.00	\$3,977.90
SemCrude	ABERNATHY AUTO, INC.	\$0.00	\$0.00	\$0.00	\$7,957.54	\$0.00	\$7,957.54 \$166.55
SemCrude	ABILENE AMERICAN ENERGY INC.	\$0.00	\$0.00	\$0.00	\$166.55	\$0.00	•
SemCrude	ABNG INC	\$0.00	\$0.00	\$21,511.51	\$0.00	\$0.00	\$21,511.51
SemCrude	ACCESS OPERATING COMPANY INC	\$0.00	\$0.00	\$203.32	\$0.00	\$0.00	\$203.32
SemCrude SemCrude	ACCESS PETROLEUM CORPORATION	\$0.00	\$0.00	\$10,246.04	\$0.00	\$0.00	\$10,246.04
SemCrude SemCrude	AEON ENERGY	\$0.00	\$0.00	\$10,426.35	\$0.00	\$0.00	\$10,426.35
SemCrude SemCrude	AETHON I LP	\$0.00	\$0.00	\$231,536.55	\$0.00	\$0.00	\$231,536.55
SemCrude SemCrude	AGAPE INC	\$0.00	\$0.00	\$7,782.64	\$0.00	\$0.00	\$7,782.64
SemCrude SemCrude	AINSWORTH OPERATING COMPANY	\$108,858.23	\$0.00	\$0.00	\$0.00	\$0.00	\$108,858.23
SemCrude SemCrude	AL CONTRERAS CONSULTANTS INC	\$0.00	\$0.00	\$17,615.73	\$0.00	\$0.00	\$17,615.73
SemCrude SemCrude	ALAN J VONFELDT	\$80.66	\$0.00	\$0.00	\$0.00	\$0.00	\$80.66
SemCrude	ALAN L LAMB	\$0.00	\$0.00	\$19,971.87	\$0.00	\$0.00	\$19,971.87
SemCrude SemCrude	ALBANY WELL SERVICE	\$0.00	\$0.00	\$0.00	\$146.46	\$0.00	\$146.46
SemCrude	ALEXANDER ENERGY CORP	\$0.00	\$0.00	\$139.08	\$0.00	\$0.00	\$139.08 \$93,947.17
SemCrude	ALEXANDER RESOURCES INC	\$0.00	\$0.00	\$0.00	\$93,947.17	\$0.00	
SemCrude	AL-LA-LYN OIL, INC.	\$263.81	\$0.00	\$0.00	\$0.00	\$0.00	\$263.81
SemCrude	ALLAN BAUGH	\$0.00	\$0.00	\$193,957.22	\$0.00	00.02	\$193,957.22
SemCrude	ALLAN BAUGH & RITA BAUGH,	\$0.00	\$0.00	\$18,015.04	\$0.00	\$0.00	\$18,015.04
SemCrude	ALTEX ENERGY CORPORATION	\$0.00	\$0.00	\$8,888,581.31	\$0.00	\$0.00	\$8,888,581.31
SemCrude SemCrude	ALTEX RESOURCES INC	\$0.00	\$0.00	\$28.25	\$0.00	\$0.00	\$28.25 \$349,992.30
SemCrude	ALTMAN ENERGY INC	\$0.00	\$0.00	\$349,992.30	\$0.00	\$0.00	\$349,992.30 \$30.76
SemCrude	AMBRIT ENERGY CORP	\$0.00	\$0.00	\$30.76	\$0.00	\$0.00	\$30.70 \$338.85
SemCrude	AMEREX CORPORATION	\$0.00	\$0.00	\$338.85	\$0.00	\$0.00	\$22,048.42
SemCrude SemCrude	AMERICAN ENERGIES CORP	\$22,048.42	\$0.00	\$0.00	\$0.00	\$0.00	\$22,048.42 \$5,741.16
SemCrude	AMERICAN WARRIOR INC	\$5,741.16	\$0.00	\$0.00	\$0.00	\$0.00	\$3,741.10 \$46,447.99
SemCrude	AMERICAN WEST	\$0.00	\$0.00	\$0.00	\$46,447.99	\$0.00	\$40,447.59 \$107.66
SemCrude	AMERICANA ENERGY INC	\$0.00	\$0.00	\$107.66	\$0.00	\$0.00	\$0.06
SemCrude	AMER-TEX ENERGY INC	\$0.00	\$0.00	\$0.06	\$0.00	\$0.00	\$5.40
SemCrude	ANA LENA CORP	\$0.00	\$0.00	\$5.40	\$0.00	\$0.00	\$27,326.94
SemCrude	ANADARKO MINERALS INC	\$0.00	\$0.00	\$27,326.94	\$0.00	\$0.00	\$27,320.94 \$5.31
SemCrude	ANADARKO PETROLEUM CORPORATION	\$0.00	\$0.00	\$5.31	\$0.00	\$0.00	
SemCrude	ANITA LOUISE LAIRD	\$0.00	\$0.00	\$21,066.89	\$0.00	\$0.00	\$21,066.89 \$62.92
SemCrude	ANJET PETROLEUM	\$0.00	\$0.00	\$62.92	\$0.00	\$0.00	\$62.92 \$31,420.84
SemCrude	ANSON OIL COMPANY	\$0.00	\$0.00	\$31,420.84	\$0.00	\$0.00	\$31,420.84 \$315,480.42
SemCrude	ANSTINE - MUSGROVE OIL COMPANY	\$315,480.42	\$0.00	\$0.00	\$0.00	\$0.00	\$315,460,42 \$668,909.20
SemCrude	ANSTINE & MUSGROVE INC	\$668,909.20	\$0.00	\$0.00	\$0.00	\$0.00	\$668,909.20 \$167,603.08
SemCrude	APACHE CORPORATION	\$0.00	\$167,603.08	\$0.00	\$0.00	\$0.00	\$18,391.09
SemCrude SemCrude	APEC INC	\$0.00	\$0.00	\$18,391.09	\$0.00	\$0.00	
SemCrude SemCrude	APOLLO ENERGIES INC	\$83,763.72	\$0.00	\$19.56	\$0.00	\$0.00	\$83,783.28
SemCrude SemCrude	ARBEN OIL INC	\$0.00	\$0.00	\$0.00	\$38,079.98	\$0.00	\$38,079.98 \$12,217.27
SemCrude SemCrude	ARBUCKLE ENTERPRISES INC	\$0.00	\$0.00	\$12,217.27	\$0.00	\$0.00	\$12,217.27 \$182.73
SemCrude SemCrude	ARC ENERGY, INC.	\$0.08	\$0.00	\$0.00	\$182.73	\$0.00	\$182.73 \$64,568.93
SemCrude SemCrude	ARDMORE PRODUCTION & EXPL CO	\$0.00	\$0.00	\$64,568.93	\$0.00	\$0.00	304,206.73
Seliforage	· ************************************						

				nar-soner v estosomburgus v	in Active States		
24 BANG - 24 BANG BA			ew Mexico	Oklahoma		Wyoming	Ilitali di
Debtor	Operator Supplement Su	25424360	ew Mexico \$0.00	\$0.00	\$0.00	\$0.00	\$474,584.52
SemCrude	ARGENT ENERGY INC	\$474,584.52 \$0.00	\$0.00	\$6.64	\$0.00	\$0.00	\$6.64
SemCrude	ARKOMA RESOURCES	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$10,328.37
SemCrude	ARMADILLA OIL & GAS LLC	\$10,328.37 \$0.00	\$0.00	\$0.00	\$3,350,389.32	\$0.00	\$3,350,389.32
SemCrude	ARROW OIL & GAS INC	\$0.00	\$0.00	\$10,038.44	\$0.00	\$0.00	\$10,038.44
SemCrude	ARROW OPERATING CO INC	\$0.00	\$0.00	\$90,678.70	\$0.00	\$0.00	\$90,678.70
SemCrude	ARROWHEAD ENERGY INC	\$0.00	\$0.00	\$167,048.95	\$0.00	\$0.00	\$167,048.95
SemCrude	ATCHLEY RESOURCES INC	\$0.00	\$0.00	\$20,756.13	\$0.00	\$0.00	\$20,756.13
SemCrude	ATHABASCA ENERGY CORPORATION	\$0.00 \$0.00	\$0.00	\$83,762.71	\$0.00	\$0.00	\$83,762.71
SemCrude	ATHAN INC	\$0.00	\$0.00	\$1.39	\$0.00	\$0.00	\$1.39
SemCrude	AULT-FOSTER PRODUCTION CO.	\$0.00	\$0.00	\$562,820.59	\$0.00	\$0.00	\$562,820.59
SemCrude	AVARDILC	\$0.00	\$0.00	\$0.00	\$350.27	\$0.00	\$350.27
SemCrude	AZTEX PETROLEUM	\$0.00	\$0.00	\$13.36	\$0.00	\$0.00	\$13.36
SemCrude	AZURE ENERGY LTD	\$8,210.15	\$0.00	\$0.00	\$0.00	\$0.00	\$8,210.15
SemCrude	B & B OIL CO	\$0.00	\$0.00	\$21,016.86	\$86,133.18	\$0.00	\$107,150.04
SemCrude	B & G PRODUCTION INC	\$0.00	\$0.00	\$14,255.32	\$0.00	\$0.00	\$14,255.32
SemCrude	B P CHESSER	\$0.00	\$0.00	\$0.00	\$23,230.01	\$0.00	\$23,230.01
SemCrude	B P OPERATING COMPANY	\$0.00	\$0.00	\$0.00	\$182.69	\$0.00	\$182.69
SemCrude	B.O. HAWKINS	\$0.00	\$0.00	\$471,784.84	\$0.00	\$0.00	\$471,784.84
SemCrude	BARNES OIL COMPANY	\$0.00	\$0.00	\$112,682.43	\$0.00	\$0.00	\$112,682.43
SemCrude	BARRIER RESOURCES	\$0.00	\$0.00	\$21,465.99	\$0.00	\$0.00	\$21,465.99
SemCrude	BARTA ENTERPRISES	\$107,566.01	\$0.00	\$0.00	\$0.00	\$0,00	\$107,566.01
SemCrude	BARTON EXPLORATION, INC.	\$0.00	\$0.00	\$1,877,959.36	\$0.00	\$0.00	\$1,877,959.36
SemCrude	BAYS EXPLORATION INC	\$0.00	\$0.00	\$86,354.85	\$0.00	\$0.00	\$86,354.85
SemCrude	BC OPERATING INC	\$19,460.82	\$0.00	\$0.00	\$0.00	\$0.00	\$19,460.82
SemCrude	B-C STEEL LLC	\$0.00	\$0.00	\$0.00	\$231.55	\$0.00	\$231.55
SemCrude	BEAIRD ENERGY CO.	\$45,384.75	\$0.00	\$0.00	\$0.00	\$0.00	\$45,384.75
SemCrude	BEAR PETROLEUM INC	\$0.00	\$0.00	\$233,543.42	\$0.00	\$0.00	\$233,543.42
SemCrude	BEASLEY OIL COMPANY	\$0.00	\$0.00	\$44,473.08	\$0.00	\$0.00	\$44,473.08
SemCrude	BECK RESOURCES INC	\$0.00	\$0.00	\$17.64	\$0.00	\$0.00	\$17.64
SemCrude	BELL & KINLEY CO.	\$44,314.75	\$0.00	\$0.00	\$0.00	\$0.00	\$44,314.75
SemCrude	BELL BROTHERS	\$0.00	\$0.00	\$0.00	\$80,642.64	\$0.00	\$80,642.64
SemCrude	BELL ENERGY	\$0.00	\$0.00	\$19,076.10	\$0.00	\$0.00	\$19,076.10
SemCrude	BELPORT OIL INC	\$108.59	\$0.00	\$0.00	\$0.00	\$0.00	\$108.59
SemCrude	BERENERGY CORPORATION BERENTZ DRILLING CO INC	\$18,815.06	\$0.00	\$0.00	\$0.00	\$0.00	\$18,815.06
SemCrude		\$0.00	\$0.00	\$94,344.10	\$0.00	\$0.00	\$94,344.10
SemCrude	BEREXCO INC BERTSCHINGER OIL COMPANY	\$0.00	\$0.00	\$22,137.96	\$0.00	\$0.00	\$22,137.96
SemCrude		\$0.00	\$0.00	\$0.00	\$1,277.35	\$0.00	\$1,277.35 \$553.08
SemCrude	BEX INC BEZ RESOURCES LLC	\$0.00	\$0.00	\$553.08	\$0.00	\$0.00	
SemCrude		\$0.00	\$0.00	\$0.00	\$1,598.12	\$0.00	\$1,598.12
SemCrude	BIG TIO INC. BIGGY'S PETROLEUM	\$22,077.90	\$0.00	00.00	\$0.00	\$0.00	\$22,077.90
SemCrude		\$0.00	\$0.00	\$16.28	\$0.00	\$0.00	\$16.28 \$345,086.40
SemCrude	BILL ENNIS BILL LAIRD	\$0.00	\$0.00	\$345,086.40	\$0.00	\$0.00	\$343,086.40 \$2,569.38
SemCrude	BILL MIOHNSON PRODICO INC	\$0.00	\$0.00	\$2,569.38	\$0.00	\$0.00	\$2,569.38 \$324,620.10
SemCrude	BILL WEEMS OIL INC	\$0.00	\$0.00	\$324,620.10	\$0.00	\$0.00	\$324,620.10
SemCrude	BILLIE E EMERY	\$0.00	\$0.00	\$9,808.25	\$0.00	\$0.00	\$30,580.09
SemCrude	BILLY JACK SHARBER OPERATING LLC	\$0.00	\$0.00	\$30,580.09	\$0.00	\$0.00	\$176,673.72
SemCrude	BLACK DOT OIL INC	\$0.00	\$0.00	\$176,673.72	\$0.00	\$0.00	\$176,673.72 \$26,673.50
SemCrude	BLACK GOLD OIL INC	\$0.00	\$0.00	\$26,673.50	\$0.00	\$0.00	320,013.30
SemCrude	DUACK GODD OFF OO 1140						

			sasyonian namaka	ISSUEDICALESCA MATERIA	s in Active States		
		Kansas			Texas	Wyoming	Total
Debtor	Operator Property of the Control of	\$0.00	\$0.00	\$1,391.78	\$0.00	\$0.00	\$1,391.78
SemCrude	BLACK RIVER OIL & GAS LLC	\$0.00	\$0.00	\$297,854.11	\$0.00	\$0.00	\$297,854.11
SemCrude	BLACKCAT OIL COMPANY	\$46.924.59	\$0.00	\$0.00	\$0.00	\$0.00	\$46,924.59
SemCrude	BLACKSPOT OIL CO	\$0.00	\$0.00	\$3,333.80	\$0.00	\$0.00	\$3,333.80
SemCrude	BLAIR OIL COMPANY	\$491,720.89	\$0.00	\$0.00	\$0.00	\$0.00	\$491,720.89
SemCrude	BLAKE EXPLORATION, INC	\$158.43	\$0.00	\$0.00	\$0.00	\$0.00	\$158.43
SemCrude	BLAKE PRODUCTION CO INC	\$0.00	\$0.00	\$1,271,706.88	\$65,197.56	\$0.00	\$1,336,904.44
SemCrude	BLUE DOLPHIN PRODUCTION LLC	\$0.00	\$0.00	\$0.00	\$23,590.79	\$0.00	\$23,590.79
SemCrude	BLUE RIDGE GROUP	\$0.00	\$0.00	\$0.00	\$14,959.29	\$0.00	\$14,959.29
SemCrude	BLUE RIDGE RESOURCES, LLC	\$0.00	\$0.00	\$1,266.89	\$0.00	\$0.00	\$1,266.89
SemCrude	BLUEBONNET OIL CORP	\$0.00	\$0.00	\$0.35	\$0.00	\$0.00	\$0.35
SemCrude	BLUEHAWK PROPERTIES INC	\$177.16	\$0.00	\$0.00	\$0.00	\$0.00	\$177.16
SemCrude	BLUERIDGE PETROLEUM CORP.	\$40,543,44	\$0.00	\$0.00	\$0.00	\$0.00	\$40,543.44
SemCrude	BLUESTEM INC	\$0.00	\$0.00	\$126.86	\$0.00	\$0.00	\$126.86
SemCrude	BOB & SON OIL CO.	\$0.00	\$0.00	\$65,165.82	\$0.00	\$0.00	\$65,165.82
SemCrude	BOB BUNCH	\$0.00	\$0.00	\$0.00	\$40,557.41	\$0.00	\$40,557.41
SemCrude	BOB G ISBELL OPERATING	\$0.00	\$0.00	\$91,426.63	\$0.00	\$0.00	\$91,426.63
SemCrude	BOB LOFTIS OIL & GAS	\$0.00	\$0.00	\$0.00	\$5,454.07	\$0.00	\$5,454.07
SemCrude	BOBBY D MORRISON	\$0.00	\$0.00	\$37,542.08	\$0.00	\$0.00	\$37,542.08
SemCrude	BOBBY HORN DBA HORN OIL COMPANY	\$0.00	\$0.00	\$232,572.74	\$0.00	\$0.00	\$232,572.74
SemCrude	BOBBY J DARNELL	\$0.00	\$0.00	\$0.00	\$32.14	\$0.00	\$32.14
SemCrude	BOBBY J WEST	\$0.00	\$0.00	\$40,069.56	\$0.00	\$0.00	\$40,069.56
SemCrude	BOBBY R PERKINS	\$0.00	\$0.00	\$30,702.72	\$0.00	\$0.00	\$30,702.72
SemCrude	BOBWHITE PRODUCTION CO INC	\$0.00	\$0.00	\$1.26	\$0.00	\$0.00	\$1.26
SemCrude	BO-MC RESOURCES CORPORATION	\$0.00	\$0.00	\$42,628.21	\$0.00	\$0.00	\$42,628.21
SemCrude	BOONE OPERATING INC	\$0.00	\$0.00	\$0.00	\$8,098.46	\$0.00	\$8,098.46
SemCrude	BOO-ROW PIPE & SUPPLY, INC.	\$25,009.31	\$0.00	\$0.00	\$0.00	\$0.00	\$25,009.31
SemCrude	BOWMAN OIL COMPANY	\$0.00	\$0.00	\$175.97	\$0.00	\$0.00	\$175.97
SemCrude	BOYCE PRODUCING CORP BOYD & McWILLIAMS ENERGY GROUP	\$0.00	\$0.00	\$0.00	\$11,840.18	\$0.00	\$11,840.18
SemCrude	BRAD COOK OIL PROPERTIES	\$0.00	\$0.00	\$127.29	\$0.00	\$0,00	\$127.29
SemCrude		\$0.00	\$0.00	\$66,865.29	\$0.00	\$0.00	\$66,865.29
SemCrude	BRADEN-DEEM INC BRADLEY & BRADLEY OIL CO.	\$25,569.26	\$0.00	\$0.00	\$0.00	\$0.00	\$25,569.26
SemCrude	BRADLET & BRADLET OIL CO. BRAMMER SUPPLY INC.	\$0.00	\$0.00	\$11.88	\$0.00	\$0.00	\$11.88
SemCrude	BRAYO NATURAL RESOURCES INC	\$0.00	\$0.00	\$4,286.67	\$0.00	\$0.00	\$4,286.67
SemCrude	BRECK OPERATING COMPANY	\$0.00	\$0.00	\$10.89	\$0.00	\$0.00	\$10.89
SemCrude	BRENT BAKER OIL & GAS INC	\$0.00	\$0.00	\$2,307.59	\$0.00	\$0.00	\$2,307.59
SemCrude	BRIDWELL OIL COMPANY	\$0.00	\$0.00	\$0.00	\$247,118.54	\$0.00	\$247,118.54
SemCrude	BRISCOE OIL OPERATING CO INC	\$0.00	\$0.00	\$1.16	\$0.00	\$0.00	\$1.16
SemCrude	BRISTOL RESOURCES CORPORATION	\$0.00	\$0.00	\$3.92	\$0.00	\$0.00	\$3.92
SemCrude		\$0.00	\$0.00	\$61,228.36	\$0.00	\$0.00	\$61,228.36
SemCrude	BRITT OIL COMPANY LLC BRITTANY ENERGY LLC	\$0.00	\$0.00	\$317,966.90	\$0.00	\$0.00	\$317,966.90
SemCrude		\$0.00	\$0.00	\$379.90	\$0.00	\$0.00	\$379.90
SemCrude	BROCON INC BROWN AND BORELLI INC	\$0.00	\$0.00	\$350.10	\$0.00	\$0.00	\$350.10
SemCrude	BRUCE OIL CO	\$19,420,93	\$0.00	00.00	\$0.00	\$0.00	\$19,420.93
SemCrude	BRUCE OIL COMPANY LLC	\$18,911.27	\$0.00	\$0.00	\$0.00	\$0.00	\$18,911.27
SemCrude	BRYAN D CAIN	\$0.00	\$0.00	\$1,752.08	\$0.00	\$0.00	\$1,752.08
SemCrude	BRYANT AND SON	\$0.00	\$0.00	\$9.35	\$0.00	\$0.00	\$9.35
SemCrude	BRYANT LYN PRODUCTION	\$0.00	\$0.00	\$61,563.92	\$0.00	\$0.00	\$61,563.92
SemCrude	BRYCE F HAYES OIL & GAS	\$207.42	\$0.00	\$0.00	\$0.00	\$0.00	\$207.42
SemCrude	DRICE FINIES OIL & ONS						

Marine and Speniare Appleador				First Purchase Amount	s in Active States		
	Operator	Kansas	New Mexico				Total
Debtor SemCrude	BSO INCORPORATED	\$0.00	\$0.00	\$8.26	\$0.00	\$0.00	\$8.26
SemCrude SemCrude	BUD RICHARDSON II	\$0.00	\$0.00	\$78.79	\$0.00	\$0.00	\$78.79
SemCrude	BURGUNDY OIL & GAS OF NM INC	\$0.00	\$234,364.02	\$0.00	\$0.00	\$0.00	\$234,364.02
SemCrude	BURKETT DRILLING CO.	\$0.00	\$0.00	\$0.00	\$343.56	\$0.00	\$343.56
SemCrude	BURTON OIL PROPERTIES	\$0.00	\$0.00	\$19,339.23	\$0.00	\$0.00	\$19,339.23
SemCrude	BUTKIN OIL COMPANY LLC	\$0.00	\$0.00	\$1,427,636.11	\$0.00	\$0.00	\$1,427,636.11
SemCrude	BUXTON ENERGY LLC	\$0.00	\$0.00	\$864.89	\$0.00	\$0.00	\$864.89
SemCrude	BVD INC	\$0.00	\$0.00	\$6,808.96	\$0.00	\$0.00	\$6,808.96
SemCrude	C & A OIL COMPANY INC	\$0.00	\$0.00	\$78,636.68	\$0.00	\$0.00	\$78,636.68
SemCrude	C & B OIL COMPANY	\$0.00	\$0.00	\$24,820.66	\$0.00	\$0.00	\$24,820.66
SemCrude	C & L OIL & GAS CORPORATION	\$0.00	\$0.00	\$24,973.07	\$0.00	\$0.00	\$24,973.07
SemCrude	C & W OIL COMPANY	\$0.00	\$0.00	\$282,120.81	\$0.00	\$0.00	\$282,120.81
SemCrude	C D OPERATING CO INC	\$0.00	\$0.00	\$555.55	\$0.00	\$0.00	\$555.55
SemCrude	CEDLC	\$0.00	\$0.00	\$19,188.96	\$0.00	\$0.00	\$19,188.96
SemCrude	C H TODD INC	\$43.68	\$0.00	\$0.00	\$0.00	\$0.00	\$43.68
SemCrude	CJINVESTMENT	\$0.00	\$0.00	\$230.30	\$0.00	\$0.00	\$230.30
SemCrude	C L & S OIL CO INC	\$0.00	\$0.00	\$21,494.76	\$0.00	\$0.00	\$21,494.76
SemCrude	C&L OIL AND GAS CORPORATION	\$0.00	\$0.00	\$345.00	\$0.00	\$0.00	\$345.00
SemCrude	C&M OIL COMPANY INC	\$0.00	\$0.00	\$2,774.24	\$0.00	\$0.00	\$2,774.24
SemCrude	CAL CREEK RESOURCES INC	\$0.00	\$0.00	\$25,172.48	\$0.00	\$0.00	\$25,172.48
SemCrude	CALUMET OIL COMPANY	\$0.00	\$0.00	\$257.16	\$0.00	\$0.00	\$257.16
SemCrude	CALVIN NOAH	\$280,319.47	\$0.00	\$0.00	\$0.00	\$0.00	\$280,319.47 \$1,498.52
SemCrude	CANADIAN RIVER RESOURCES LLC	\$0.00	\$0.00	\$1,498.52	\$0.00	00.02	
SemCrude	CANOK INTERNATIONAL INC.	\$0.00	\$0.00	\$25,705.53	\$0.00	\$0.00	\$25,705.53 \$3,351.06
SemCrude	CANOK INTERNATIONAL INC	\$0.00	\$0.00	\$3,351.06	\$0.00	\$0.00	\$1,545.96
SemCrude	CANTRELL ENERGY CORPORATION	\$0.00	\$0.00	\$1,545.96	\$0.00	\$0.00 \$0.00	\$595.63
SemCrude	CAPSTONE OIL & GAS	\$0.00	\$0.00	\$595.63	\$0.00	\$0.00	\$1,015.05
SemCrude	CARL E GUNGOLL EXPLORATION INC	\$0.00	\$0.00	\$1,015.05	\$0.00	\$0.00 \$0.00	\$685,967.36
SemCrude	CARMAC ENERGY CORP	\$0.00	\$0.00	\$685,967.36	\$0.00	\$0.00	\$19.291.12
SemCrude	CARPENTER OIL COMPANY	\$0.00	\$0.00	\$19,291.12	\$0.00	\$0.00	\$43.10
SemCrude	CARROLL L BREWER	\$0.00	\$0.00	\$43.10	\$0.00	\$0.00	\$31,097,24
SemCrude	CARROLL PRODUCTION CO	\$0.00	\$0.00	\$31,097.24	\$0.00	\$0.00	\$383,027.36
SemCrude	CARTLEDGE & CARTLEDGE OIL CO	\$0.00	\$0.00	\$383,027.36	\$0.00 \$0.00	\$0.00	\$87,480.27
SemCrude	CASE SALES COMPANY INC	\$0.00	\$0.00	\$87,480.27	\$0.00 \$0,00	\$0.00	\$35,788.20
SemCrude	CASEY MUSGROVE OIL COMPANY	\$0.00	\$0.00	\$35,788.20		\$0.00	\$378,945.51
SemCrude	CASILLAS PETROLEUM CORP	\$0.00	\$0.00	\$378,945.51	\$0.00 \$0.00	\$0.00	\$113.21
SemCrude	CASONDRA J SANFORD	\$0.00	\$0.00	\$113.21	\$0.00	\$0.00	\$319,000.91
SemCrudo	CASTELLI EXPLORATION, INC.	\$319,000.91	\$0.00	\$0.00		\$0.00 \$0.00	\$47,965.27
SemCrude	CBM GAS COMPANY LLC	\$0.00	\$0.00	\$47,965.27	\$0.00	\$0.00	\$1,849.97
SemCrude	CD OPERATING COMPANY INC	\$0.00	\$0.00	\$1,849.97	\$0.00 \$0.00	\$0.00	\$16.57
SemCrude	CDS OIL COMPANY	\$0.00	\$0.00	\$16.57		\$0.00	\$1,159.12
SemCrude	CEDAR OIL COMPANY	\$0.00	\$0.00	\$0.00	\$1,159.12	\$0.00	\$79,951.90
SemCrude	CEDAR RIDGE ENERGY INC	\$0.00	\$0.00	\$0.00	\$79,951.90 \$0.00	\$0.00	\$608,827.24
SemCrude	CENPAT OPERATING LLC	\$0.00	\$0.00	\$608,827.24	\$0.00	\$0.00	\$611,742.32
SemCrude	CENTER PRODUCTION CO LLC	\$0.00	\$0.00	\$611,742.32 \$0.00	\$0.00 \$20,261.84	\$0.00	\$20,261.84
SemCrude	CENTRAL BASIN OIL INVESTMENTS INC	\$0.00	\$0.00		\$20,261.84	\$0.00	\$614,038.28
SemCrude	CENTRAL OPERATING INC	\$614,038.28	\$0.00	\$0.00	\$0.00	\$0.00	\$304,941.24
SemCrude	CENTREX OPERATING COMPANY INC	\$0.00	\$0.00	\$304,941.24	\$0.00	\$0.00	\$18.87
SemCrude	CFC OIL, INC.	\$0.00	\$0.00	\$18.87	J9.VV	\$0.00	± ~

		The second secon	o-santa supungi prosessi sas 1988		n Active States		
				Oldanoma	Texas	Avoming	Total
Debror	Operator	Kansas	\$0.00	\$18,995.36	\$0.00	\$0.00	\$19,968.75
SemCrude	CHACO ENERGY COMPANY	\$973.39	\$0.00	\$0.00	\$214,328.21	\$0.00	\$214,328.21
SemCrude	CHAMBERS. R.H., LLC	\$0.00 \$0.00	\$0.00	\$0.00	\$55,806.16	\$0.00	\$55,806.16
SemCrude	CHAMP OPERATING COMPANY	• • •	\$0.00	\$0.70	\$0.00	\$0.00	\$0.70
SemCrude	CHANEY & COBURN INC	\$0.00	\$0.00	\$0.00	\$50,043.78	\$0.00	\$50,043.78
SemCrude	CHAP OPERATING CO., INC.	\$0.00 \$0.00	\$0.00	\$197,871.90	\$0.00	\$0.00	\$197,871.90
SemCrude	CHAPARRAL ENERGY INC	• • • • • • • • • • • • • • • • • • • •	\$0.00	\$146,928.06	\$0.00	\$0.00	\$146,928.06
SemCrude	CHAPMAN OIL COMPANY	\$0.00	\$0.00	\$2,209.08	\$0.00	\$0.00	\$2,209.08
SemCrude	CHARLES ANSON	\$0.00	\$0.00	\$957.92	\$0.00	\$0.00	\$957.92
SemCrude	CHARLES H COLPITT DRILLING CO	\$0.00	\$0.00	\$21,392.00	\$0.00	\$0.00	\$21,392.00
SemCrude	CHARLIE HARGUS	\$0.00	\$0.00	\$44,095.73	\$0.00	\$0.00	\$44,095.73
SemCrude	CHARTER OAK PRODUCTION CO LLC	\$0.00	\$0.00	\$1,304.17	\$0.00	\$0.00	\$1,304.17
SemCrude	CHER OIL COMPANY LTD	\$0.00	\$0.00	\$48,936.55	\$0.00	\$0.00	\$48,936.55
SemCrude	CHEROKEE MINERALS INC	\$0.00	\$0.00	\$27,038.36	\$0.00	\$0.00	\$27,038.36
SemCrude	CHEROKEE ROYALTIES & MINERALS INC	\$0.00	\$0.00	\$35,781.06	\$0.00	\$0.00	\$35,781.06
SemCrude	CHESAPEAKE OPERATING INC	\$0.00	\$0.00	\$96,131.20	\$0.00	\$0.00	\$96,131.20
SemCrude	CHESTER HORTON	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$20,419.60
SemCrude	CHEYENNE OIL PROPERTIES	\$20,419.60	\$0.00	\$394,319.22	\$0.00	\$0.00	\$394,319.22
SemCrude	CHIZUM OIL LLC	\$0.00	\$0.00	\$139,378.50	\$0.00	\$0.00	\$139,378.50
SemCrude	CHOLLA PRODUCTION LLC	\$0.00	\$0.00	\$33,830.53	\$0.00	\$0.00	\$33,830.53
SemCrude	CIMARRON RIVER OPERATING CO	\$0.00	\$0.00	\$31,993.17	\$0.00	\$0.00	\$31,993.17
SemCrude	CIMMARON TRANS	\$0.00 \$0.00	\$0.00	\$89,085.83	\$0.00	\$0.00	\$89,085.83
SemCrude	CIRCLE C OILFIELD SUPPLY INC	-	\$0.00	\$0.00	\$0.00	\$0.00	\$18,371.09
SemCrude	CIRCLE G PETROLEUM	\$18,371.09	\$0.00	\$44,303.23	\$0.00	\$0.00	\$44,303.23
SemCrude	CIRCLE K OIL AND GAS INC	\$0.00 \$0.00	\$0.00	\$4,485.59	\$0.00	\$0.00	\$4,485.59
SemCrude	CIRRUS PRODUCTION CO	\$0.00 \$0.00	\$0.00	\$421.07	\$0.00	\$0.00	\$421.07
SemCrude	CK RESOURCES	\$0.00	\$0.00	\$468.89	\$0.00	\$0.00	\$468.89
SemCrude	CLARENCE E JAMES	\$0.00 \$74,962.51	\$0.00	\$0.00	\$0.00	\$0.00	\$74,962.51
SemCrude	CLARK EXPLORATION COMPANY	\$74,962.31	\$0.00	\$9.16	\$0.00	\$0.00	\$9.16
SemCrude	CLARKS WELL SERVICE	\$0.00	\$0.00	\$0.00	\$18,196.03	\$0.00	\$18,196.03
SemCrude	CLAVO OIL LLC	\$0.00 \$24.02	\$0.00	\$0.00	00.02	\$0.00	\$24.02
SemCrude	CLAY MOORE	\$0.00	\$0.00	\$0.00	\$767.90	\$0.00	\$767.90
SemCrude	CLAYTONVILLE VENTURES	\$2,546,136.81	\$0.00	\$0.00	\$0.00	\$0.00	\$2,546,136.81
SemCrude	CLIPPER ENERGY LLC	\$2,340,130.81	\$0.00	\$61,938.31	\$0.00	\$0.00	\$61,938.31
SemCrude	CLYDE R JACOBS	\$703,827.33	\$0.00	\$62,072.05	\$0.00	\$0.00	\$765,899.38
SemCrude	CMX INC	\$703,827.33	\$0.00	\$20,184.03	\$19,357.62	\$0.00	\$39,541.65
SemCrude	COASTAL PLAINS ENERGY INC	\$0.00	\$2,904,705.17	\$0.00	\$0.00	\$0.00	\$2,904,705.17
SemCrude	COG OIL & GAS LP	\$0.00	\$0.00	\$354.84	\$0.00	\$0.00	\$354.84
SemCrude	COGBURN & SIMMONS WELL SERVICE	\$0.00	\$0.00	\$225.66	\$0.00	\$0.00	\$225.66
SemCrude	COLLIER PETROLEUM CORPORATION	\$0.00	\$0.00	\$0.00	\$3,694.10	\$0.00	\$3,694.10
SemCrude	COLO EXPLORATION COMPANY INC	\$26,099.90	\$0.00	\$0.00	\$0.00	\$0.00	\$26,099.90
SemCrude	COLORADO INTERSTATE GAS CO	\$20,099.90	\$0.00	\$112,584.15	\$0.00	\$0.00	\$112,584.15
SemCrude	COLPITT BROTHERS OIL	\$0.00 \$0.00	\$0.00	\$127,796.24	\$0.00	\$0.00	\$127,796.24
SemCrude	COLUMBUS OIL COMPANY	\$0.00	\$0.00	\$19,984.49	\$0.00	\$0.00	\$19,984.49
SemCrude	COMANCHE PETROLEUM	\$0.00	\$0.00	\$0.34	\$0.00	\$0.00	\$0.34
SemCrude	COMANCHE RESOURCES CO	\$0.00	\$0.00	\$1,576.73	\$0.00	00.02	\$1,576.73
SemCrude	COMBINED RESOURCES CORPORATION	\$0.00	\$0.00	\$13,078.20	\$0.00	\$0.00	\$13,078.20
SemCrude	COMBS PRODUCTION LLC	\$0.00	\$0.00	\$224,749.42	\$0.00	\$0.00	\$224,749.42
SemCrude	COMPASS DRILLING INC	\$0.00	\$0.00	\$0.00	\$4,781.02	00.02	\$4,781.02
SemCrude	CONCHO OILFIELD SERV & OPER CO	30.00	Ψν.συ		•		

			on was multiples of people was the	SPACE THE PROPERTY OF THE PARTY	in Active States		
			ew Mexico	Ser ureinsezen	Texas W	voming	a Lotal
Debtor	Operator Company of the Company of t	7,000,000	\$0.00	\$273.61	\$0.00	\$0.00	\$273.61
SemCrude	CONDELL POLLARD	\$0.00 \$0.00	\$0.00	\$19,563.78	\$0.00	\$0.00	\$19,563.78
SemCrude	CONNER PRODUCTION CO LLC	\$0.00 \$0.00	\$0.00	\$0.00	\$1,231.28	\$0.00	\$1,231.28
SemCrude	CONNER-MARRS COMPANY	\$0.00	\$0.00	\$0.16	\$0.00	\$0.00	\$0.16
SemCrude	CONSOLIDATED EXPLORATION	\$0.00	\$0.00	\$289,036.72	\$0.00	\$0.00	\$289,036.72
SemCrude	CONSOLIDATED INVESTMENTS	\$0.00	\$0.00	\$1,821,441.29	\$0.00	\$0.00	\$1,821,441.29
SemCrude	COOK OIL COMPANY	\$0,00	\$0.00	\$168,453.85	\$0.00	\$0.00	\$168,453.85
SemCrude	COOKE COUNTY PRODUCTION CO INC	\$0.00 \$2,457.09	\$0.00	\$0.00	\$0.00	\$0.00	\$2,457.09
SemCrude	COOPER SERVICE COMPANY	\$43,735.27	\$0.00	\$0.00	\$0.00	\$0.00	\$43,735.27
SemCrude	CORAL COAST PETROLEUM	\$43,733.27 \$0.00	\$0.00	\$117.22	\$0.00	\$0.00	\$117.22
SemCrude	CORNELIUS OIL INC	\$0.00 \$0.00	\$0.00	\$18.07	\$0.00	\$0.00	\$18.07
SemCrude	CORNER ENERGY LLC	\$0.00	\$0.00	\$600.57	\$0.00	\$0.00	\$600.57
SemCrude	CORTEZ OPERATING COMPANY	\$3.418.90	\$0.00	\$0.00	\$0.00	\$0.00	\$3,418.90
SemCrude	COUNTRYSIDE OIL, LLC	\$3,418.90 \$0.00	\$0.00	\$35,427.07	\$0.00	\$0.00	\$35,427.07
SemCrude	COURTS COMPANY	\$0.00	\$0.00	\$24,069.46	\$0.00	\$0.00	\$24,069.46
SemCrude	COVINGTON OIL COMPANY INC	\$0.00	\$0.00	\$1,819.46	\$0.00	\$0.00	\$1,819.46
SemCrude	COWBOY OIL & GAS INC	\$1,479.10	\$0.00	\$0.00	\$0.00	\$0.00	\$1,479.10
SemCrude	CRAIG OIL COMPANY	\$0.00	\$0.00	\$617,163.12	\$0.00	\$0.00	\$617,163.12
SemCrude	CRAWLEY PETROLEUM CORPORATION	\$0.00	\$0.00	\$8,988.51	\$0.00	\$0.00	\$8,988.51
SemCrude	CRB RESOURCES INC	\$19,330.31	\$0.00	\$0.00	\$0.00	\$0.00	\$19,330.31
SemCrude	CREDO PETROLEUM CORPORATION	\$0.00	\$0,00	\$0.00	\$919.74	\$0.00	\$919.74
SemCrude	CREEK ENERGY INC	\$0.00	\$0.00	\$89,239.26	\$0.00	\$0.00	\$89,239.26
SemCrude	CREST RESOURCES INC	\$0.00	\$0.00	\$117,878.30	\$0.00	\$0.00	\$117,878.30
SemCrude	CROSE OIL COMPANY	\$0.00	\$0.00	\$19,233.56	\$0.00	\$0.00	\$19,233.56
SemCrude	CROSS PROPERTIES INC	\$0.00	\$0.00	\$19,710.03	\$0.00	\$0.00	\$19,710.03
SemCrude	CROW CREEK ENERGY LLC	\$753,365.6 1	\$0.00	\$24,267.18	\$0.00	\$0.00	\$777,632.79
SemCrude	CROWN ENERGY COMPANY	\$0.00	\$0.00	\$28,027.26	\$0.00	\$0.00	\$28,027.26
SemCrude	CRUSADER ENERGY CORPORATION	\$0.00	\$0.00	\$84,476.22	\$28,134.96	\$0.00	\$112,611.18
SemCrude	CRYSTAL RIVER OIL & GAS LLC	\$22,376.22	\$0.00	\$183,761.64	\$0.00	\$0.00	\$206,137.86
SemCrude	CUDD OPERATING CORP	\$0.00	\$0.00	\$38,896.39	\$0.00	\$0.00	\$38,896.39
SemCrude	CUESTA PETROLEUM, INC.	\$0.00	\$0.00	\$43,808.67	\$0.00	\$0.00	\$43,808.67
SemCrude	CULBREATH OIL & GAS INC	\$0.00	\$0.00	\$2,376.16	\$0.00	\$0.00	\$2,376.16
SemCrude	CUMMINGS OIL AND CATTLE	\$19,830.29	\$0.00	\$0.00	\$0.00	\$0.00	\$19,830.29
SemCrude	CURRY OIL	\$0.00	\$0.00	\$61,108.74	\$0.00	\$0.00	\$61,108.74
SemCrude	CURZON OPERATING CO. LTD.	\$0.00	\$0.00	\$0.00	\$42,735.64	\$0.00	\$42,735.64
SemCrude	CWIC,INC. CYCLONE PETROLEUM INC	\$309,264.98	\$0.00	\$0.00	\$0.00	\$0.00	\$309,264.98 \$51.36
SemCrude	CYCLONE PETROLEOM INC.	\$0.00	\$0.00	\$51.36	\$0.00	\$0.00	\$31.30 \$135,689.17
SemCrude		\$0.00	\$0.00	\$135,689.17	\$0.00	\$0.00	
SemCrude	D & H OIL INC D & J OIL COMPANY INC	\$0.00	\$0.00	\$21.82	\$0.00	\$0.00	\$21.82
SemCrude		\$0.00	\$0.00	\$0.00	\$9,282.08	\$0.00	\$9,282.08
SemCrude	D & L OPERATING CO D & R ENTERPRISES	\$0.00	\$0.00	\$0.00	\$23,925.28	\$0.00	\$23,925.28 \$6,974.04
SemCrude	D& SOIL COMPANY	\$6,974.04	\$0.00	\$0.00	\$0.00	\$0.00	\$18.08
SemCrude	D & T SWABBING & WELL SERVICE INC	\$0.00	\$0.00	\$18.08	\$0.00	\$0.00	\$753,000.93
SemCrude	D C ENERGY INC	\$0.00	\$0.00	\$645,982.79	\$107,018.14	\$0.00	\$1,671,314.15
SemCrude	D E EXPLORATION INC	\$1,671,314.15	\$0.00	\$0.00	\$0.00	\$0.00	\$35,492.33
SemCrude	D L EXPLORATION INC	\$0.00	\$0.00	\$0.00	\$35,492.33	\$0.00	\$35,492.33 \$21,858.54
SemCrude	D O WALKER INC	\$0.00	\$0.00	\$21,858.54	\$0.00	\$0.00	\$922,234.22
SemCrude	D R LAUCK OIL CO INC	\$922,234.22	\$0.00	\$0.00	\$0.00	\$0.00	\$68,181.40
SemCrude	D W EVANS OIL	\$68,181.40	\$0.00	\$0.00	\$0.00	\$0.00	300,101.40
SemCrude	D M EAVISO AIT						

				irst Parchase Amoun	is in Active States		
	Constitution of the consti	Kansas	law Meyico	Oklahoma	Texas	Wyoming	Total
Debton	DEPARTMENT TO THE PROPERTY OF	\$0.00	\$0.00	\$0.00	\$19,097.79	\$0.00	\$19,097.79
SemCrude	D W P PRODUCTION	\$0.00	\$0.00	\$0.00	\$774.34	\$0.00	\$774.34
SemCrude	D.K.D.	\$0.00	\$0.00	\$23,724.76	\$0.00	\$0.00	\$23,724.76
SemCrude	D.S.J. INC	\$0.00	\$0.00	\$27.51	\$0.00	\$0.00	\$27.51
SemCrude	DAB AND ASSOCIATES INC	\$0.00	\$0.00	\$0.00	\$10,840.13	\$0.00	\$10,840.13
SemCrude	DALE P. SIMPSON	\$0.00	\$0.00	\$179,822.24	\$0.00	\$0.00	\$179,822.24
SemCrude	DALE RADFORD	\$0.00	\$0.00	\$6,280.34	\$0.00	\$0.00	\$6,280.34
SemCrude	DALE RADFORD & HERBERT RADFORD, J/Ts	\$0.00	\$0.00	\$77,739.68	\$0.00	\$0.00	\$77,739.68
SemCrude	DALE WILLIAMS	\$0.00	\$0,00	\$35,938.96	\$0.00	\$0.00	\$35,938.96
SemCrude	DAN L MCGONIGAL	\$0.00	\$0.00	\$528,829.05	\$0.00	\$0.00	\$528,829.05
SemCrude	DAN WALLACE INC	\$0.00	\$0.00	\$843.48	\$0.00	\$0.00	\$843.48
SemCrude	DANCO ENTERPRISES INC	\$0.00	\$0.00	\$0.00	\$5,959.37	\$0.00	\$5,959.37
SemCrude	DANIEL EXPLORATIONS INC.	\$0.00	\$0.00	\$35,222.07	\$0.00	\$0.00	\$35,222.07
SemCrude	DANIEL S TRACHTENBERG	\$0.00	\$0.00	\$8.69	\$0.00	\$0.00	\$8.69
SemCrude	DANNY D DENHAM	\$0.00	\$0.00	\$38.94	\$0.00	\$0.00	\$38.94
SemCrude	DARLING OIL CORPORATION	\$0.00	\$0.00	\$22,175.17	\$0.00	\$0.00	\$22,175.17
SemCrude	DARRELL'S DRILLING INC	\$0.00	\$0.00	\$722.30	\$0.00	\$0.00	\$722.30
SemCrude	DATA ENERGY INC.	\$0.00	\$0.00	\$27,646.10	\$0.00	\$0.00	\$27,646.10
SemCrude	DAUCO PETROLEUM CORPORATION	\$0.00	\$0.00	\$65,802.07	\$0.00	\$0.00	\$65,802.07
SemCrude	DAVE HOLKE CONSULTING INC	\$0.00	\$0.00	\$9,642.84	\$0.00	\$0.00	\$9,642.84
SemCrude	DAVID B WILLIAMS	\$0.00	\$0.00	\$32,567.28	\$0.00	\$0.00	\$32,567.28
SemCrude	DAVID D KIRK	\$0.00 \$0.00	\$29,955.97	\$0.00	\$0.00	\$0.00	\$29,955.97
SemCrude	DAVID H ARRINGTON OIL & GAS IN	\$0.00	\$0.00	\$299,974.15	\$0.00	\$0.00	\$299,974.15
SemCrude	DAVID K MOORE	\$0.00	\$0.00	\$127,991.39	\$0.00	\$0.00	\$127,991.39
SemCrude	DAVID R COAST	\$86,056.69	\$0.00	\$632.85	\$0.00	00.08	\$86,689.54
SemCrude	DAWSON MARKWELL EXPLORATION CO	\$0.00	\$0.00	\$14,250.96	\$0.00	\$0.00	\$14,250.96
SemCrude	DAXIN ENERGY CORPORATION	\$1,772,082.08	\$0.00	\$0.00	\$0.00	\$0.00	\$1,772,082.08
SemCrude	DAYSTAR PETROLEUM INC	\$0.00	\$0.00	\$24.90	\$0.00	\$0.00	\$24,90
SemCrude	DEAN E EYLER	\$0.00	\$0.00	\$51,516.48	\$0.00	\$0.00	\$51,516.48
SemCrude	DEAN E EYLER DBA DEAN E EYLER	\$0.00	\$0.00	\$25,165.57	\$0.00	\$0.00	\$25,165.57
SemCrude	DEE JAY EXPLORATION COMPANY INC	\$0.00	\$0.00	\$26,669.62	\$0.00	\$0.00	\$26,669.62
SemCrude	DEGGE OILFIELD SERVICE	\$0.00	\$0.00	\$0.00	\$45.67	\$0.00	\$45.67
SemCrude	DEL PETROLEUM	\$0.00	\$0.00	\$13,330.02	\$0.00	\$0.00	\$13,330.02
SemCrude	DELTA D OPERATIONS INC	\$0.00	\$0.00	\$2,358.65	\$0.00	\$0.00	\$2,358.65
SemCrude	DEMCO OIL & GAS COMPANY	\$0.00	\$0.00	\$8,979.55	\$0.00	\$0.00	\$8,979.55
SemCrude	DEMO OIL COMPANY	\$0.00	\$0.00	\$0.70	\$0.00	\$0.00	\$0.70
SemCrude	DENNETT'S INC	\$0.00	\$0.00	\$0.00	\$52.24	\$0.00	\$52.24
SemCrude	DERYL JAMES OIL COMPANY	\$80,100.08	\$0.00	\$0.00	\$0.00	\$0.00	\$80,100.08
SemCrude	DEUTSCH OIL COMPANY	\$0.00	\$0.00	\$86,325.83	\$0.00	\$0.00	\$86,325.83
SemCrude	DEVON RESOURCES CORPORATION	\$0.00	\$0.00	\$0.00	\$120.66	\$0.00	\$120.66
SemCrude	DEWEY WILSON	\$0.00	\$0.00	\$41,072.45	\$0.00	\$0.00	\$41,072.45
SemCrude	DIAMOND CROSS OIL CO LLC	\$0.00	\$0.00	\$3.05	\$0.00	\$0.00	\$3.05
SemCrude	DIGGER OIL & GAS CO INC	\$128,133.27	\$0.00	\$0.00	\$0.00	\$0.00	\$128,133.27
SemCrude	DIVERSIFIED OPERATING CORP	\$146.02	\$0.00	\$0.00	\$0.00	\$0.00	\$146.02
SemCrude	DIXON ENERGY INC	\$140.02 \$353,868.89	\$0.00	\$0.00	\$0.00	\$0.00	\$353,868.89
SemCrude	DK OPERATING	\$0.00 \$0.00	\$0.00	\$0.00	\$346.38	\$0.00	\$346.38
SemCrude	DKMR OIL COMPANY INC	\$0.00 \$0.00	\$0.00	\$59,252.70	\$0.00	\$0.00	\$59,252.70
SemCrude	D-MIL PRODUCTION INC	\$21,661.62	\$0.00	\$0.00	\$0.00	\$0.00	\$21,661.62
SemCrude	DOCKING DEVELOPMENT LLC	\$21,001.02	\$0.00	\$60,163.64	\$0.00	\$0.00	\$60,163.64
SemCrude	DOME ENGINEERING	30.00	\$V.0V	44.7-210.			

			100000000000000000000000000000000000000	first Purchase Amounts in	Active States		
		Kansas	Variable Par	Oklahoma		Wyoming	Total
Debtor	Operator	\$0.00	\$0.00	\$28,319.16	\$0.00	\$0.00	\$28,319.16
SemCrude	DON DENNEY	\$0.00	\$0.00	\$18.34	\$0.00	\$0.00	\$18.34
SemCrude	DON E BROWN	\$34.38	\$0.00	\$0.00	\$0.00	\$0.00	\$34,38
SemCrude	DON G. FOX	\$0.00	\$0.00	\$19,306.57	\$0.00	\$0.00	\$19,306.57
SemCrude	DON SHARBER	\$0.00	\$0.00	\$19.63	\$0.00	\$0.00	\$19.63
SemCrude	DON YERBY	\$0.00	\$0.00	\$63,939.39	\$0.00	\$0.00	\$63,939.39
SemCrude	DONALD F LESTER	\$136,730.04	\$0.00	\$0,00	\$0.00	\$0.00	\$136,730.04
SemCrude	DONALD G JENKINS	\$25,639.34	\$0.00	\$0.00	\$0.00	\$0.00	\$25,639.34
SemCrude	DONALD I VONFELDT	\$0.00	\$0.00	\$89,728.98	\$0.00	\$0.00	\$89,728.98
SemCrude	DONRAY PETROLEUM LLC	\$552,268.57	\$0.00	\$0.00	\$0.00	\$0.00	\$552,268.57
SemCrude	DOUBLE EAGLE EXPLORATION INC	\$0.00	\$0.00	\$4,115.94	\$0.00	\$0.00	\$4,115.94
SemCrude	DOUBLE EAGLE PETROLEUM CORP	\$0.00 \$0.00	\$0.00	\$23,019.17	\$0.00	\$0.00	\$23,019.17
SemCrude	DOUBLE JOIL CO LLC	\$0.00 \$0.00	\$0.00	\$0.00	\$330,825.93	\$0.00	\$330,825.93
SemCrude	DOUBLE M PETROPROPERTIES, INC.	\$0.00	\$0.00	\$3,470.17	\$0.00	\$0.00	\$3,470.17
SemCrude	DOUG HEWITT	\$0.00	\$0.00	\$58.15	\$0.00	\$0.00	\$58.15
SemCrude	DOUG MEEHAN D/B/A HILLTOP OIL & GAS	\$0.00	\$0.00	\$5,109.59	\$0.00	\$0.00	\$5,109.59
SemCrude	DOUGLAS BALDRIDGE	\$0.00	\$0.00	\$1,203.73	\$0.00	\$0.00	\$1,203.73
SemCrude	DOUGLASS DIETZ & DALEY INC	\$0.00	\$0.00	\$0.00	\$37,954.81	\$0.00	\$37,954.81
SemCrude	DRAGOON OIL AND GAS COMPANY	\$89,945.01	\$0.00	\$0.00	\$0.00	\$0.00	\$89,945.01
SemCrude	DRAKE EXPLORATION LLC	\$0.00	\$0.00	\$83,381.49	\$0.00	\$0.00	\$83,381.49
SemCrude	DREW WILSON	\$59,610.82	\$0.00	\$0.00	\$0.00	\$0.00	\$59,610.82
SemCrude	DREYER OIL	\$227,637.03	\$0.00	\$0.00	\$0.00	\$0.00	\$227,637.03
SemCrude	DRILLERS & PRODUCERS INC	\$227,037.03	\$0.00	\$164.28	\$0.00	\$0.00	\$164.28
SemCrude	DRUMRIGHT WELL SERVICE INC	\$0.00	\$0.00	\$103,366.66	\$0.00	\$0.00	\$103,366.66
SemCrude	DUCK BROS DOZER & TRUCKING INC	\$0.00	\$0.00	\$0.00	\$562.80	\$0.00	\$562.80
SemCrude	DUNAGIN OPERATING	\$0.00	\$0.00	\$1,690,744,67	\$549,352.55	\$0.00	\$2,240,097.22
SemCrude	DUNCAN OIL PROPERTIES INC	\$0.00	\$0.00	\$274,716.23	\$0.00	\$0.00	\$274,716.23
SemCrude	DUNLAP OIL COMPANY	\$202,379.99	\$0.00	\$60,717.01	\$0.00	\$0.00	\$263,097.00
SemCrude	DUNNE EQUITIES OPERATING INC	\$0.00	\$0.00	\$25.63	\$0.00	\$0.00	\$25.63
SemCrude	DURANGO EXPLORATION	\$0.00	\$0.00	\$3,269.15	\$0.00	\$0.00	\$3,269.15
SemCrude	DWIGHT EPPERSON	\$0.00	\$0.00	\$2,036.33	\$0.00	\$0.00	\$2,036.33
SemCrude	DYNO ENERGY LLC	\$0.00	\$0.00	\$4,446.29	\$0.00	\$0.00	\$4,446.29
SemCrude	E & S RESOURCES LLC	\$21,268.28	\$0.00	\$8,928.93	\$0.00	\$0.00	\$30,197.21
SemCrude	E L THOMAS	\$0.00	\$0.00	\$23.38	\$0.00	\$0.00	\$23.38
SemCrude	E LYLE JOHNSON	\$38,146.19	\$0.00	\$0.00	\$0.00	\$0.00	\$38,146.19
SemCrude	E R OPERATING COMPANY	\$173,664.89	\$0.02	\$0.00	\$0.00	\$0.00	\$173,664.89
SemCrude	EAGLE CREEK CORPORATION	\$0.00	\$0.00	\$41.19	\$0.00	\$0.00	\$41.19
SemCrude	EAGLE RIVER ENERGY CORPORATION	\$0.00	\$0.00	\$19,369.58	\$0.00	\$0.00	\$19,369.58
SemCrude	EARL A TERHUNE JR	\$0.00	\$0.00	\$0.00	\$50,981.21	\$0.00	\$50,981.21
SemCrude	EARL NEATHERLIN INC	\$0.00	\$0.00	\$37.39	\$0.00	\$0.00	\$37.39
SemCrude	EARL OIL INC.	\$0.00	\$0.00	\$10,292.23	\$0.00	\$0.00	\$10,292.23
SemCrude	EARLE-LE OIL CO	\$0.00	\$0.00	\$495,630.57	\$0.00	\$0.00	\$495,630.57
SemCrude	EARLSBORO ENERGIES CORPORATION	\$0.00	\$0.00	\$0.00	\$137.34	\$0.00	\$137.34
SemCrude	ECHO PRODUCTION INC	\$0.00	\$0.00	\$1,350.33	\$0.00	\$0.00	\$1,350.33
SemCrude	ED PRIER INC	\$0.00	\$0.00	\$7.09	\$0.00	\$0.00	\$7.09
SemCrude	EDDIE'S SUBMERSIBLE SERVICE INC	\$0.00	\$0.00	\$254,544.07	\$0.00	\$0.00	\$254,544.07
SemCrude	EDINGER ENGINEERING INC	\$0.00	\$0.00	\$0.00	\$3,231.35	\$0.00	\$3,231.35
SemCrude	EDMONDSON OIL & GAS	\$0.00	\$0.00	\$113,043.93	\$0.00	\$0.00	\$113,043.93
SemCrude	EDWARD A EMERY	\$0.00	\$0,00	\$2,640.92	\$0.00	\$0.00	\$2,640.92
SemCrude	EDWARD C LAWSON PROPERTIES	30.00	+2102	• •			

				First Purchase Amoun	its in Active States		
Debroz	Operator	Kansas	New Mexico	Oklahoma	Texas	Wyoming	Lotal
The state of the s	EDWARD K WELLS	\$0.00	\$0.00	\$138. 5 9	\$0.00	\$0.00	\$138.59
SemCrude	EDWARD LEE BARTON	\$0.00	\$0.00	\$9,718.30	\$0.00	\$0.00	\$9,718.30
SemCrude	EGRET OPERATING COMPANY	\$0.00	\$0.00	\$133,082.43	\$0.00	\$0.00	\$133,082.43
SemCrude	ELCO OIL & GAS	\$0.00	\$0.00	\$0.00	\$60.50	\$0.00	\$60.50
SemCrude	ELLIS & WILLA MAE OVERMAN	\$0.00	\$0.00	\$3.51	\$0.00	\$0.00	\$3.51
SemCrude	ELLIS & WILLA MAE OVERSIAN ELLIS RESOURCES GROUP INC	\$0.00	\$0.00	\$24,939.78	\$0.00	\$0.00	\$24,939.78
SemCrude	ELMER G. & VIRGINIA HALSTEAD	\$0.00	\$0.00	\$157.91	\$0.00	\$0.00	\$157.91
SemCrude		\$634,091.91	\$0.00	\$0.00	\$0.00	\$0.00	\$634,091.91
SemCrude	ELMER KARST OIL CO	\$0.00	\$0.00	\$22,699.63	\$0.00	\$0.00	\$22,699.63
SemCrude	ELSON OIL COMPANY LLC	\$0.00	\$0.00	\$109.27	\$0.00	\$0.00	\$109.27
SemCrude	ELVIN EARL MINER	\$0.00	\$0.00	\$20,072.64	\$0.00	\$0.00	\$20,072.64
SemCrude	EMERY WELL SERVICE & ED EMERY	\$0.00	\$0.00	\$10.51	\$0.00	\$0.00	\$10.51
SemCrude	EMPIRE PETROLEUM	\$0.00	\$0.00	\$14,549.63	\$0.00	\$0.00	\$14,549.63
SemCrude	ENDEAVOR ENERGY RESOURCES LP	\$0.00	\$0.00	\$0.00	\$41,242.26	\$0.00	\$41,242.26
SemCnude	ENERGEX INC	\$0.00	\$0.00	\$125,203.55	\$0.00	\$0.00	\$125,203.55
SemCrude	ENERSOURCE PETROLEUM INC	\$0.00	\$0.00	\$23,522.49	\$0.00	\$0.00	\$23,522.49
SemCrude	ENEXCO INC	\$0.00	\$0.00	\$0.00	\$824.37	\$0.00	\$824.37
SemCrude	ENRICH OIL CORPORATION	\$93.28	\$0.00	\$0.00	\$0.00	\$0.00	\$93.28
SemCrude	ENTERPRISE, INC	\$0.00	\$0.00	\$17,791.30	\$0.00	\$0.00	\$17,791.30
SemCrude	ENTERPRISING ENTERPRISES	\$12,089.27	\$0.00	\$0.00	\$0.00	\$0.00	\$12,089.27
SemCrude	EQUINOX OIL COMPANY	\$0.00	\$0.00	\$11,527.86	\$0.00	\$0.00	\$11,527.86
SemCrude	ERDOL INC.	\$0.00	\$0.00	\$17.01	\$0.00	\$0.00	\$17.01
SemCrude	ERNST PETROLEUM COMPANY	\$0.00	\$0.00	\$94,464.02	\$0.00	\$0.00	\$94,464.02
SemCrude	ESCHER CORPORATION	\$0.00	\$0.00	\$41,630.98	\$0.00	\$0.00	\$41,630.98
SemCrude	ESKRIDGE PRODUCTION COMPANY	\$0.00	\$0.00	\$63.04	20.00	\$0.00	\$63.04
SemCrude	ESSEX ENERGY INC	\$0.00	\$0.00	\$17,403.89	\$0.00	\$0.00	\$17,403.89
SemCrude	EUGENE SELLMEYER & HELEN L	\$18,842.63	\$0.00	\$0.00	\$0.00	\$0.00	\$18,842.63
SemCrude	EVANS OIL INC	\$0.00	\$0.00	\$0.00	\$4,686.36	\$0.00	\$4,686.36
SemCrude	EVARADO OIL COMPANY	\$0.00	\$0.00	\$0.00	\$15,307.46	\$0.00	\$15,307.46
SemCrude	EXCO RESOURCES INC	\$0.00	\$0.00	\$156,081.89	\$0.00	\$0.00	\$156,081.89
SemCrude	EXOK INC	\$0.00	\$0.00		\$0.00	\$0.00	\$130.96
SemCrude	EXPLORATION ASSOCIATES II LLC	\$0.00	\$0.00	* * * * * * * * * * * * * * * * * * * *	\$2,840.66	\$0.00	\$2,840.66
SemCrude	EXZENA OIL CORP	\$0.00	\$0.00		\$0.00	\$0.00	\$193,531.05
SemCrude	P & J OIL COMPANY LLC	\$5,193.71	\$0.00		\$0.00	\$0.00	\$5,193.71
SemCrude	F G HOLL COMPANY LLC	\$0.00	\$0.00		\$63,044.48	\$0.00	\$63,044.48
SemCrude	F L LIVINGSTON	\$0.00	\$0.00		\$0.00	\$0.00	\$351,535.89
SemCrude	F.H.A. INVESTMENTS	\$0.00	\$0.00		\$0.00	\$0.00	\$51,610.44
SemCrude	FAGADAU ENERGY CORP	\$0,00	\$0.00		\$0.00	\$0.00	\$3,718.00
SemCrude	FAIRCHILD WELL SERVICE INC.	\$0.00	\$0.00	- •	\$0.00	\$0.00	\$462,704.80
SemCrude	FAIRFIELD OIL & GAS CORP	\$0.00	\$0.00	*	\$0.00	\$0.00	\$54,925.60
SemCrude	FAIRY ISAACS	\$106,506.84	\$0.00		\$0.00	\$0.00	\$106,506.84
SemCrude	FAITH PETROLEUM	\$0.00	\$0.00		\$0.00	\$0.00	\$58,719.02
SemCrude	FALCON OIL PROPERTIES	\$0.00	\$0.00	·	\$0.00	\$0.00	\$11,389.00
SemCrude	FERGUSON - HUNTER EXPLORATION INC	\$62,805.32	\$0.00	,	\$0.00	\$0.00	\$198,330.56
SemCrude	FIRST NATIONAL OIL INC	\$62,803.32 \$0.00	\$0.00		\$0.00	\$0.00	\$392.50
SemCrude	FLAT ROCK ENERGY COMPANY	\$0.00	\$0.00		\$0.00	\$0.00	\$147,438.78
SemCrude	FLEETWOOD OIL PROPERTIES	\$0.00	\$0.00		\$0.00	\$0.00	\$147.26
SemCrude	FLOYD ANDERSON	\$0.00	\$0.00		\$0.00	\$0.00	\$4,431.16
SemCrude	POREMAN ENTERPRISES INC	\$0.00	\$0.00	•	\$27,677.56	\$0.00	\$27,677.56
SemCrude	FOREST OIL CORPORATION	30.00	30.00	50.00			

			meson - seemenmes - 45-495	t Purchase Amounts i			
				Oklahoma	Texas W	yoming	Arrana and a series of the ser
Debtor	Operator and a second s	Kansas \$0.00	SO.00	\$62,890,67	\$0.00	\$0.00	\$62,890.67
SemCrude	FOSSIL CREEK ENERGY CORPORATION	\$0.00 \$0.00	\$0.00	\$0.00	\$20,519.14	\$0.00	\$20,519.14
SemCrude	FOSSIL OPERATING, INC.	\$0.00 \$0.00	\$0.00	\$0.00	\$159.01	\$0.00	\$159.01
SemCrude	FOUR L OIL		\$0.00	\$0.00	\$0.00	\$0.00	\$17,555.52
SemCrude	FOUR WAY OPERATING	\$17,555.52	\$0.00	\$536.25	\$0.00	\$0.00	\$536.25
SemCrude	FOX NORTHEASTERN OIL & GAS CORP	\$0.00	\$0.00	\$19,126.92	\$0.00	\$0.00	\$19,126.92
SemCrude	FOXBOROUGH ENERGY COMPANY LLC	\$0.00	\$0.00	\$0.00	\$259,667.62	\$0.00	\$259,667.62
SemCrude	FRANK & ETHEL KRAMER FAMILY LTD PRTNSHP	\$0.00	\$0.00	\$3,408.53	\$0.00	\$0.00	\$3,408.53
SemCrude	FRANKS BROS PRODUCTION	\$0.00	\$0.00	\$25,839.61	\$0.00	\$0.00	\$25,839.61
SemCrude	FREEMONT EXPLORATION INC	\$0.00	\$0.00	\$107.15	\$0.00	\$0.00	\$107.15
SemCrude	FREMONT ENERGY CORP	\$0.00	\$0.00	\$0.75	\$0.00	\$0.00	\$0.75
SemCrude	FREMONT EXPLORATION INC	\$0.00	\$0.00	\$50.68	00.02	\$0.00	\$50,68
SemCrude	FRICKEY INVESTMENT MANAGEMENT CO	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$19,998.73
SemCrude	FROELICH OIL	\$19,998.73	\$0.00	\$0.00	\$5,228.70	\$0.00	\$5,228.70
SemCrude	FRONTIER OPERATING CORPORATION	\$0.00 \$0.00	\$0.00	\$113,785.10	\$0.00	\$0.00	\$113,785.10
SemCrude	FULLER PRODUCTION INC	• • • • • • • • • • • • • • • • • • • •	\$0.00	\$0.00	\$1,571.07	\$0.00	\$1,571.07
SemCrude	FURRY PRODUCING COMPANY	\$0.00 \$0.00	\$0.00	\$100,604.71	\$0.00	\$0.00	\$100,604.71
SemCrude	G & C PETROLEUM INC	\$0.00	\$0.00	\$179,275.37	\$0.00	\$0.00	\$179,275.37
SemCrude	G & S PRODUCTION	\$0.00	\$0.00	\$21,036.57	\$0.00	\$0.00	\$21,036.57
SemCrude	G B F OIL COMPANY LLC	\$0.00 \$0.00	\$0.00	\$0.00	\$250.91	\$0,00	\$250.91
SemCrude	G B PATRICK	\$19,360.30	\$0.00	\$0.00	\$0.00	\$0.00	\$19,360.30
SemCrude	G3 OPERATING LLC	\$19,300.30 \$20,623.85	\$0.00	\$0.00	\$0.00	\$0.00	\$20,623.85
SemCrude	GABEL OIL COMPANY	\$20,023.63 \$92,182.98	\$0.00	\$0.00	\$0.00	\$0.00	\$92,182.98
SemCrude	GABS OIL COMPANY	\$92,182.98	\$0.00	\$247,736.69	\$0.00	\$0.00	\$247,736.69
SemCrude	GAI-JIN TRADING CO LLC	\$0.00	\$0.00	\$53,028.20	\$0.00	\$0.00	\$53,028.20
SemCrude	GAIL B GORE	\$0.00	\$0.00	\$1,364.35	\$0.00	\$0.00	\$1,364.35
SemCrude	GALA RESOURCES LTD.	\$16,744.57	\$0.00	\$0.00	\$0.00	\$0.00	\$16,744.57
SemCrude	GAMBLER RESOURCES	\$0.00	\$0.00	\$470,115.20	\$0.00	\$0.00	\$470,115.20
SemCrude	GARRY DAVIS	\$0.00	\$0.00	\$21,888.96	\$0.00	\$0.00	\$21,888.96
SemCrude	GARY KLUSMEYER & GAIL LYNN KLUSMEYER	\$0.00	\$0.00	\$1,367.98	\$0.00	\$0.00	\$1,367.98
SemCrude	GARY NOBLE SHIELDS	\$338.11	\$0.00	\$0.00	\$0.00	\$0.00	\$338.11
SemCrude	GARY R. SHAFFER	\$0.00	\$0.00	\$0.00	\$2,266.97	\$0.00	\$2,266.97
SemCrude	GAS CORPORATION OF AMERICA	\$0.00	\$0.00	\$820,409.94	\$0.00	\$0.00	\$820,409.94
SemCrude	GEMINI OIL COMPANY	\$1,057.94	\$0.00	\$0.00	\$0.00	\$0.00	\$1,057.94
SemCrude	GEN-DOR OIL COMPANY	\$0.00	\$0.00	\$146,486.93	\$0.00	\$0.00	\$146,486.93
SemCrude	GENERAL PETROLEUM COMPANY	\$0.00	\$0.00	\$13.97	\$0.00	\$0.00	\$13.97
SemCrude	GENESIS INC	\$0.00	\$0.00	\$12.98	\$0.00	\$0.00	\$12.98
SemCrude	GENIE OIL & GAS CORPORATION	\$0.00	\$0.00	\$2,745.30	\$0.00	\$0.00	\$2,745.30
SemCrude	GENO OIL AND GAS CORP	\$0.00	\$0.00	\$88,515.33	\$0.00	\$0.00	\$88,515.33
SemCrude	GEO AMERICAN RESOURCES INC	\$0.00	\$0.00	\$12,452.63	\$0.00	\$0.00	\$12,452.63
SemCrude	GEO EXPLORATION LLC	\$0.00	\$0.00	\$6,697.18	\$0.00	\$0.00	\$6,697.18
SemCrude	GEORGE H MASSAD	\$0.00	\$0.00	\$21,147.32	\$0.00	\$0,00	\$21,147.32
SemCrude	GEORGE H RUSSELL O&G INV INC	\$0.00	\$0.00	\$11,551.83	\$0.00	\$0.00	\$11,551.83
SemCrude	GEORGE M MYERS	\$0.00	\$0.00	\$112.10	\$0.00	\$0.00	\$112.10
SemCrude	GEORGE MASSAD OIL CO (ESTATE)	\$0.00	\$0.00	\$29,817.49	\$0.00	\$0.00	\$29,817.49
SemCrude	GEORGE W DENNISON SR	\$953.94	\$0.00	\$0.00	\$0.00	\$0.00	\$953.94
SemCrude	GEO-SAX PETROLEUM COMPANY	\$24,239.85	\$0.00	\$0.00	\$0.00	\$0.00	\$24,239.85
SemCrude	GERALD MACK ROBERTS	\$0.00	\$0.00	\$9.18	\$0.00	\$0.00	\$9.18
SemCrude	GEYER BROTHERS EQUIPMENT	\$9,008.44	\$0.00	\$0.00	00.00	\$0.00	\$9,008.44
SemCrude	GIANT HOLDING LLC	\$7,000.44	** * *				

Debtor Operator Kansas New Mexico Ottahoma Texas Wyomine	00 \$4,608.59 00 \$21,611.62 00 \$117,484.88 00 \$494,897.17 00 \$40,455.78 00 \$19,134.46 00 \$1,779,212.52 00 \$24,171.79 00 \$472,983.29 00 \$928,495.43 00 \$194,837.44 00 \$63,071.44 00 \$4,974.86 00 \$872,206.80 00 \$187,730.19 00 \$327.58 00 \$14,467.63
Debtor Operator	000 \$20,507.53 000 \$4,608.59 00 \$21,611.62 00 \$117,484.88 00 \$494,897.17 00 \$40,455.78 00 \$19,134.46 00 \$1,779,212.52 00 \$24,171.79 00 \$472,983.29 00 \$928,495.43 00 \$194,837.44 00 \$63,071.44 00 \$472,986.80 00 \$187,730.19 .00 \$327.58 .00 \$14,467.63
SemCrude GINTHER OIL OPERATIONS \$0.00 \$0.00 \$4,608.59 \$0.00 \$0.0	00 \$21,611.62 00 \$117,484.88 00 \$494,897.17 00 \$40,455.78 00 \$19,134.46 00 \$1,779,212.52 00 \$24,171.79 00 \$472,983.29 00 \$928,495.43 00 \$194,837.44 00 \$63,071.44 00 \$4,974.86 00 \$872,206.80 00 \$187,730.19 00 \$327.58 00 \$14,467.63
SemCrude GLACIER PETROLEUM CO. \$0.00 \$0.00 \$0.00 \$21,611.62 \$0.00 \$0.00 \$0.00 \$0.00 \$117,484.88 \$0.00 \$0.00 \$0.00 \$117,484.88 \$0.00 \$0.00 \$0.00 \$0.00 \$117,484.88 \$0.00 \$0	00 \$117,484.88 00 \$494,897.17 00 \$40,455.78 00 \$19,134.46 00 \$1,779,212.52 00 \$24,171.79 00 \$472,983.29 00 \$928,495.43 00 \$194,837.44 00 \$63,071.44 00 \$4,974.86 00 \$872,206.80 00 \$187,730.19 00 \$327.58 00 \$14,467.63
SemCrude GLB EXPLORATION INC \$0.00 \$0.00 \$117,484.88 \$0.00 \$0.00 SemCrude GLENDELL GASKINS \$0.00 \$0.00 \$494,897.17 \$0.00 \$0.00 SemCrude GLIMP OIL COMPANY, L.L.C. \$0.00 \$0.00 \$40,455.78 \$0.00 \$0.00 SemCrude GLM ENERGY INC \$0.00 <	00 \$494,897.17 00 \$40,455.78 00 \$19,134.46 00 \$1,779,212.52 00 \$24,171.79 00 \$472,983.29 00 \$928,495.43 00 \$194,837.44 00 \$63,071.44 00 \$4,974.86 00 \$872,206.80 00 \$187,730.19 00 \$327.58 00 \$14,467.63
SemCrude GLENDELL GASKINS \$0.00 \$0.00 \$494,897.17 \$0.00 \$0.00 \$0.00 \$494,897.17 \$0.00 \$0	00 \$44,455.78 00 \$19,134.46 00 \$1,779,212.52 00 \$24,171.79 00 \$472,983.29 00 \$928,495.43 00 \$194,837.44 00 \$63,071.44 00 \$4,974.86 00 \$872,206.80 00 \$187,730.19 00 \$327.58 00 \$14,467.63
SemCrude GLIMP OIL COMPANY, L.L.C. \$0.00 \$0.00 \$40,455.78 \$0.00	00 \$19,134.46 00 \$1,779,212.52 00 \$24,171.79 00 \$472,983.29 00 \$928,495.43 00 \$194,837.44 .00 \$63,071.44 .00 \$4,974.86 .00 \$872,206.80 .00 \$187,730.19 .00 \$327.58 .00 \$14,467.63
SemCrude GLM ENERGY INC \$3,00 \$0,00 \$0,00 \$0,00 \$0,00 \$0,00 \$0,00 \$0,00 \$0,00 \$0,00 \$1,701,995.46 \$0,00 \$0,00 \$1,701,995.46 \$0,00 \$0,00 \$1,701,995.46 \$0,00 \$0,00 \$1,701,995.46 \$0,00 \$0,00 \$1,701,995.46 \$0,00 \$0,00 \$1,701,995.46 \$0,00 \$0,00 \$1,701,995.46 \$0,00 \$0,00 \$24,171.79 \$0,00 \$0,00 \$24,171.79 \$0,00 \$0,00 \$24,171.79 \$0,00 <td>00 \$1,779,212.52 00 \$24,171.79 00 \$472,983.29 00 \$928,495.43 00 \$194,837.44 00 \$63,071.44 00 \$4,974.86 00 \$872,206.80 .00 \$187,730.19 .00 \$327.58 .00 \$14,467.63</td>	00 \$1,779,212.52 00 \$24,171.79 00 \$472,983.29 00 \$928,495.43 00 \$194,837.44 00 \$63,071.44 00 \$4,974.86 00 \$872,206.80 .00 \$187,730.19 .00 \$327.58 .00 \$14,467.63
SemCrude GLOBAL EXPLORATION LTD \$15,154.40 \$0.00 \$77,217.06 \$0.00 \$1,701,995.46 \$0.00 SemCrude GMX RESOURCES INC \$0.00 \$0.00 \$24,171.79 \$0.00 \$0.00 SemCrude GOULD OIL COMPANY \$0.00 \$0.00 \$0.00 \$472,983.29 \$0.00 SemCrude GP II ENERGY INC \$0.00 \$287,578.66 \$0.00 \$640,916.77 \$0.00 SemCrude GRA-EX, LLC \$194,837.44 \$0.00	00 \$24,171.79 00 \$472,983.29 00 \$928,495.43 00 \$194,837.44 00 \$63,071.44 00 \$4,974.86 00 \$872,206.80 00 \$187,730.19 00 \$327.58 00 \$14,467.63
SemCrude GMX RESOURCES INC \$0.00 \$0.00 \$24,171.79 \$0.00 \$0.00 SemCrude GOUKER & HARRIS \$0.00 \$0.00 \$0.00 \$472,983.29 \$0.00 SemCrude GOULD OIL COMPANY \$0.00 \$0.00 \$0.00 \$640,916.77 \$0.00 SemCrude GP II ENERGY INC \$194,837.44 \$0.00	00 \$472,983.29 00 \$928,495.43 00 \$194,837.44 00 \$63,071.44 00 \$4,974.86 00 \$872,206.80 00 \$187,730.19 00 \$327.58 00 \$14,467.63
SemCrude GOUKER & HARRIS 30.00 \$0.00 \$0.00 \$472,983.29 \$0.00 SemCrude GOULD OIL COMPANY \$0.00 \$287,578.66 \$0.00 \$640,916.77 \$0.00 SemCrude GP II ENERGY INC \$194,837.44 \$0.00	00 \$928,495.43 00 \$194,837.44 00 \$63,071.44 00 \$4,974.86 00 \$872,206.80 00 \$187,730.19 00 \$327.58 00 \$14,467.63
SemCrude GOULD OIL COMPANY 30.00 \$287,578.66 \$0.00 \$640,916.77 \$0.00 SemCrude GP II ENERGY INC \$194,837.44 \$0.00 \$0	00 \$194,837.44 00 \$63,071.44 00 \$4,974.86 00 \$872,206.80 00 \$187,730.19 00 \$327.58 00 \$14,467.63
SemCrude GP II ENERGY INC \$0.00 <td>00 \$63,071.44 00 \$4,974.86 00 \$872,206.80 00 \$187,730.19 00 \$327.58 00 \$14,467.63</td>	00 \$63,071.44 00 \$4,974.86 00 \$872,206.80 00 \$187,730.19 00 \$327.58 00 \$14,467.63
SemCrude GRA-EX, LLC \$194,87,144 \$0.00 \$0.00 \$63,071.44 \$0.00 SemCrude GRAWARD OPERATING INC \$0.00 \$0.00 \$4,974.86 \$0.00<	.00 \$4,974.86 .00 \$872,206.80 .00 \$187,730.19 .00 \$327.58 .00 \$14,467.63
SemCrude GRAWARD OPERATING INC \$0.00 \$0.00 \$4,974.86 \$0.00 \$0.00 \$0.00 \$4,974.86 \$0.00	.00 \$872,206.80 .00 \$187,730.19 .00 \$327.58 .00 \$14,467.63
SemCrude GRAY PRODUCTION \$0.00 \$0.00 \$872,206.80 \$0.00 \$150.00	.00 \$187,730.19 .00 \$327.58 .00 \$14,467.63
SemCrude GRAYHORSE OPERATING INC 50.00 \$0.00 \$187,730.19 \$0.00 \$187,730.19 \$0.00	.00 \$327.58 .00 \$14,467.63
SemCrude GRAYSON OPERATING LLC \$0.00 \$0.00 \$327.58 \$1 SemCrude GREEN, HUGH \$0.00 \$0.00 \$14,467.63 \$0.00 \$1 SemCrude GREENLEE OIL AND GAS LLC \$0.00 \$0.00 \$14,467.63 \$0.00 <	.00 \$14,467.63
SemCrude GREEN, HUGH \$0.00 \$0.00 \$14,467.63 \$0.00 <td></td>	
SemCrude GREENLEE OIL AND GAS LLC 30.00 30.00 \$0.00	
SemCrude GREG HALL OIL & GAS LLC \$0.00 \$15,322.84 \$0.00 \$.00 \$8.52
SamCrade GREG HALL OIL & GAS LLC	.00 \$15,322.84
SemCrude	.00 \$39,695.31
Som Cride GREG SPARKS 50.00 30.00 30.00 30.00	.00 \$20,943.63
SemCrude GRIFFIN PETROLEUM CO \$0.00 \$0.00 \$1.137.11 \$0.00 \$.00 \$31,132.11
SemCrude GRIFFIN RESOURCES 50.00 30.00 31,752.71 St. 100 S	.00 \$179.23
SemCrude GRIFFIN WELL SERVICE INC 50.00 \$0.00 \$0.00 \$.00 \$24,617.72
SemCrude GRIFFITH PETROLEUM SERVICES INC 50.00 50.00 50.00 50.00 \$2.44.176.27 \$0.00 \$.00 \$244,176.27
SamCrude GROUND DEVELOPMENT COMPANY \$0.00	.00 \$23,869.88
Sem Crude GLINN COMPANY 50.00 35.00 \$25,000.00	00 \$20,693.10
SemCrude GUY HAILE \$20,093.10	1.00 \$20,026.74
SemCrude H & M PETROLEUM CORP \$20,020.74	379,131.62
SemCrude HINC \$79,131.02	38,368.52
SemCrude H&G PETROLEUM CORPORATION 50.00 \$0.00 \$0.00 \$0.00 \$0.00),00 \$1,206.82
SemCrude H. H. BLAIR OIL COMPANY INC. 51,200.82).00 \$46,261.14
SemCrude HACKER BROS. WELL SERVICE	0.00 \$1,496.90
SemCrude HALCYON NATURAL GAS LLC 30.00 9.00 \$416.53 \$0.00 \$).00 \$616.53
SemCrude HALL EXPLORATION INC 50.00 \$35.80 \$6.00).00 \$25.89
SomCrade HALLMARK RESOURCES INC).00 \$601.93
Som Cruda HAMMER OIL COMPANY),00 \$27,865.90
SemCrude HAN OPERATING INC 50.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00),00 \$86.69
SemCrude HARDING & SHELTON INC \$0.00 \$0.00	0.00 \$1,861.98
SomCride HARGUS OIL CO	0.00 \$2,546.05
SemCrude HAROLD CUNDIFF \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	0.00 \$9,592.49
SemCrude HAROLD WESLEY LANGLEY	0.00 \$80.28
SamOrida HARRINGTON OIL & GAS \$0.00 \$0.00	0.00 \$626,477.06
SamCride HARRY A SPRING \$626,477.06 \$0.00	0.00 \$10,287.05
SemCrude HARTER ENERGY INC \$0.00 \$0.00 \$10,207.00	0.00 \$21,962.40
SamOnda HARVARD PETROLEUM COMPANY, LLC \$0.00 \$21,962.40 \$0.00	0.00 \$3,683.32
SamCrude HAVENS OIL COMPANY \$0.00 \$0.00 \$0.00	0.00 \$751,924.91
SemCrude HAWKINS OIL, L.L.C. \$751,924.91 \$0.00 \$0.00 \$0.00 \$0.00	0.00 \$113,675.93
SemCrude HAYDEN OPERATING \$113,675.93 \$0.00 \$0.00 \$0.00	

				· re-consistence · · · · · · · · · · · · · · · · · · ·	in Active States	ricum via va Se vlemba vilokis	
					Control of the second s	Yvoming	Total
Debtor	Operator		Yew Mexico \$0.00	Oldahoma \$202,556.06	\$0.00	\$0.00	\$202,556.06
SemCrude	HAZELWOOD PROD & EXPLORATION CO LLC	\$0.00	*****	\$64,347.41	\$0.00	\$0.00	\$64,347.41
SemCrude	HAZLEWOOD OIL & GAS CO INC	\$0.00	\$0.00	\$0.00	\$500.38	\$0.00	\$500.38
SemCrude	H-B OIL CO.	\$0.00	\$0.00	\$2,251.51	\$0.00	\$0.00	\$2,251.51
SemCrude	HEART TO HEART	\$0.00	\$0.00	\$341.86	\$0.00	\$0.00	\$341.86
SemCrude	HEARTLAND EXPLORATION INC	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$38,257.69
SemCrude	HELM ENERGY COMPANY	\$38,257.69	\$0.00	\$29,265.68	\$0.00	\$0.00	\$29,265.68
SemCrude	HERITAGE PETROLEUM INC.	\$0.00	\$0.00	\$29,203.00	\$0.00	\$0.00	\$2,274,596.31
SemCrude	HERMAN L LOEB	\$2,274,596.31	\$0.00		\$0.00	\$0.00	\$10,424.88
SemCrude	HERMAN RAY	\$0.00	\$0.00	\$10,424.88 \$0.00	\$0.00	\$0.00	\$108,599.20
SemCrude	HESSE PETROLEUM COMPANY LLC	\$108,599.20	\$0.00	\$30,173,42	\$0.00	\$0.00	\$30,173.42
SemCrude	HEUSTON OIL LLC	\$0.00	\$0.00	\$30,173.42	\$0.00	\$0.00	\$20,107.83
SemCrude	HIGH BLUFF OPERATING, LLC	\$20,107.83	\$0.00	\$9,320.36	\$0.00	\$0.00	\$9,320.36
SemCrude	HILL OIL COMPANY INC	\$0.00	\$0.00	\$9,320.36 \$386.64	\$0.00	\$0.00	\$386.64
SemCrude	HINKLE ENGINEERING INC	\$0.00	\$0.00	-	\$0.00	\$0.00	\$38,711.70
SemCrude	HLH REALTY INC	\$0.00	\$0.00	\$38,711.70	\$0.00	\$0,00	\$13,915.20
SemCrude	HOCO INC	\$0.00	\$0.00	\$13,915.20	\$0.00	\$0.00	\$161,364.25
SemCrude	HODGDEN OPERATING CO INC	\$0.00	\$0.00	\$161,364.25	\$63,810.39	\$0.00	\$63,810.39
SemCrude	HOLLAND OIL & GAS	\$0.00	\$0.00	\$0.00 \$0.00	\$669.46	\$0.00	\$669.46
SemCrude	HOLLEY OIL COMPANY LLC	\$0.00	\$0.00		\$49,390.74	\$0.00	\$49,390.74
SemCrude	HOLLEY OPERATING, L.L.C.	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$406.48
SemCrude	HONEY OIL COMPANY INC	\$406.48	\$0.00	00.00	\$0.00	\$0.00	\$233,519.68
SemCrude	HOPE OIL COMPANY	\$0.00	\$0.00	\$233,519.68	\$0.00	\$0.00	\$162.97
SemCrude	HOPPE OIL & GAS COMPANY INC	\$162.97	\$0.00	\$0.00	\$0.00 \$0.00	\$0.00	\$232,676.46
SemCrude	HOPPER OIL COMPANY INC	\$232,676.46	\$0.00	\$0.00	\$0.00	\$0.00	\$142.51
SemCrude	HORIZON NATURAL GAS COMPANY	\$0.00	\$0.00	\$142.51	\$0.00 \$0.00	\$0.00	\$70,904.12
SemCrude	HOWARD & TAYLOR OIL CO	\$0.00	\$0.00	\$70,904.12	\$48,058.93	\$0.00	\$48,058.93
SemCrude	HOWARD STOVALL	\$0.00	\$0.00	\$0.00	\$48,038.93 \$0.00	\$0.00	\$23,734.44
SemCrude	HOYT OIL & GAS LLC	\$0.00	00.02	\$23,734.44	\$0.00 \$0.00	\$0.00	\$18,653.79
SemCrude	HUGHES GAS SYSTEMS INC	\$0.00	\$0.00	\$18,653.79	\$0.00	\$0.00	\$1,834.07
SemCrude	HULL ENTERPRISES INC	\$0.00	\$0.00	\$1,834.07	\$0.00 \$0.00	\$0.00	\$34,786.59
SemCrude	HUNT ENGINEERING LLC	\$0.00	\$0.00	\$34,786.59		\$0.00	\$2,217,621.23
SemCrude	HUNTINGTON ENERGY LLC	\$0.00	\$0.00	\$95,575.74	\$2,122,045.49 \$0.00	\$0.00	\$172,124.68
SemCrude SemCrude	HUNTON OIL & GAS CORPORATION	\$0.00	\$0.00	\$172,124.68	\$0.00	\$0.00 \$0.00	\$104,547.47
SemCrude SemCrude	HUTCHINSON OIL COMPANY	\$104,547.47	\$0.00	\$0.00	*	\$0.00	\$18.77
SemCrude SemCrude	1 & F OIL & FARM	\$18.77	\$0.00	\$0.00	\$0.00	\$0.00	\$38.53
SemCrude SemCrude	I D OIL COMPANY INC	\$0.00	\$0.00	\$38.53	\$0.00	\$0.00	\$19,666.42
- · · · · ·	IC-CO	\$0.00	\$0.00	\$19,666.42	\$0.00		\$6,100.34
SemCrude	ICE WELL SERVICES INC	\$0.00	\$0.00	\$0.00	\$6,100.34	\$0.00	\$1,192.88
SemCrude	ICON ENERGY	\$0.00	\$0.00	\$1,192.88	\$0.00	\$0.00	\$92,526.75
SemCrude	ILLINOIS EXPLORATION LLC	\$92,526.75	\$0.00	\$0.00	\$0.00	\$0.00	\$39,507.96
SemCrude	I-MAC PETROLEUM SERVICE INC	\$0.00	\$0.00	\$39,507.96	\$0.00	\$0.00	\$39,307.90 \$24.64
SemCrude	INDIAN OIL COMPANY	\$0.00	\$0.00	\$24.64	\$0.00	\$0.00	\$24.04 \$2,280.25
SemCrude	INGERSOLL OIL INC	\$0.00	\$0.00	\$2,280.25	\$0.00	\$0.00	\$2,280.23
SemCrude	INGERSOLL OIL INC	\$0.00	\$0,00	\$7,861.19	\$0.00	\$0.00	
SemCrude	INTERGY PRODUCTION LLC	\$0.00	\$0.00	\$26,046.44	\$0.00	\$0.00	\$26,046.44
SemCrude	INVESTMENT EQUIPMENT LLC	\$0.00	\$0.00	\$71,853.91	\$0.00	\$0.00	\$71,853.91 \$528.93
SemCrude	IOWA STEEL & IRON WORKS	\$0.00	\$0.00	\$528.93	\$0.00	\$0.00	
SemCrude		\$0.00	\$0.00	\$0.00	\$76,474.36	\$0.00	\$76,474.36
SemCrude	IPACT IRON POST OIL & GAS INC	\$0.00	\$0.00	\$25,755.45	\$0.00	\$0.00	\$25,755.45
SemCrude	MON LOST OFF & OVER THE	-					

				First Purchase Amount	s in Active States		
			New Mexico	Oklahoma	Texas	Wyoming	- Cotal
Debtor	Operatorial Inc.	\$0.00	\$0.00	\$22,379.93	\$0.00	\$0.00	\$22,379.93
SemCrude	J&D INVESTMENTS	\$207,484.79	\$0.00	\$0.00	\$0.00	\$0.00	\$207,484.79
SemCrude	J&HOIL	\$20,623.52	\$0.00	\$0.00	\$0.00	\$0.00	\$20,623.52
SemCrude	J & M OPERATING COMPANY	\$0.00	\$0.00	\$25,271.94	\$0.00	\$0.00	\$25,271.94
SemCrude		\$0.00	\$0.00	\$8.00	\$2,372.68	\$0.00	\$2,372.68
SemCrude	J&R PRODUCING CO. J&S OF ENID INC.	\$0,00	\$0.00	\$2,591.35	\$0.00	\$0.00	\$2,591.35
SemCrude	J A MARRS OIL COMPANY INC	\$0.00	\$0.00	\$40,038.07	\$0.00	\$0.00	\$40,038.07
SemCrude		\$0.00	\$0.00	\$0.00	\$57,688.52	\$0.00	\$57,688.52
SemCrude	J B HERRMANN OIL & GAS	\$0.00	\$0.00	00.02	\$19,240.81	\$0.00	\$19,240.81
SemCrude	J C DANIELS ENERGY	\$545.48	\$0.00	00.02	\$0.00	\$0.00	\$545.48
SemCrude	J CARTER OIL	\$0.00	\$0.00	\$142,003.78	\$0.00	\$0.00	\$142,003.78
SemCrude	J D PITTMAN	\$43,187.21	\$0.00	\$0.00	\$0.00	\$0.00	\$43,187.21
SemCrude	J D PRODUCTION COMPANY	\$0.00	\$0.00	\$0.00	\$259.04	\$0.00	\$259.04
SemCrude	I G OIL & GAS	\$0.00	\$0.00	\$122,753.39	\$0.00	\$0.00	\$122,753.39
SemCrude	J H BOLAND	\$0.00	\$0.00	\$2,274.44	\$0.00	\$0.00	\$2,274.44
SemCrude	J HERMAN ROARK	\$0.00	\$0.00	\$98.15	00.02	\$0.00	\$98.15
SemCrude	J L THOMAS ENGINEERING INC	\$0.00	\$0.00	\$33.37	\$0.00	\$0.00	\$33.37
SemCrude	JL WATKINS	\$0.00	\$0.00	\$578.10	\$0.00	\$0.00	\$578.10
SemCrude	J S SMITH	\$0.00	\$0.00	\$20,890.21	\$0.00	\$0.00	\$20,890.21
SemCrude	J W MEADORS TRUST	\$0.00	\$0.00	\$0.00	\$53,717.16	\$0.00	\$53,717.16
SemCrude	J&J SERVICES COMPANY	\$0.00	\$0.00	\$76.09	\$0.00	\$0.00	\$76.09
SemCrude	1&M WELL SERVICING & TRUCKING INC	\$0.00	\$0.00	\$0.00	\$529.44	\$0.00	\$529.44
SemCrude	J. V. RHYNE	\$0.00	\$0.00	\$0.00	\$21,768.13	\$0.00	\$21,768.13
SemCrude	J.T. RANCHES, L.L.C.	\$0.00	\$0.00	\$40,632.43	\$0.00	\$0.00	\$40,632.43
SemCrude	JACK B SELLERS ESTATE	\$0.00	\$0.00	\$3,811.00	\$0.00	\$0.00	\$3,811.00
SemCrude	JACK CHILES ESTATE	\$0.00	\$0.00	\$40,965.68	\$0.00	\$0.00	\$40,965.68
SemCrude	JACK CORMAN	\$18,552.05	\$0.00	\$513,492.42	\$0.00	\$0.00	\$532,044.47
SemCrude	JACK EXPLORATION INC	\$78,585.19	\$0.00	\$0.00	\$0.00	\$0.00	\$78,585.19
SemCrude	JACK L HADDOCK		\$0.00	\$284.09	\$0.00	\$0.00	\$284.09
SemCrude	JACK L MURPHY	\$0.00	\$0.00	\$0.00	\$7,672.96	\$0.00	\$7,672.96
SemCrude	JACK NEAL ENGINEERING	\$0.00	\$0.00	\$5.04	\$0.00	\$0.00	\$5.04
SemCnude	JACK TERRY	\$0.00	\$0.00	\$46,753.71	\$0.00	\$0.00	\$46,753.71
SemCrude	JACK THOMPSON	\$0.00		\$38,604.39	\$0.00	\$0.00	\$38,604.39
SemCrude	JACKIE THOMPSON	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$346,386.02
SemCrude	JACKSON BROTHERS LLC	\$346,386.02	\$0.00	\$105.91	\$0.00	\$0.00	\$105.91
SemCrude	JACKSON OILFIELD SERVICE INC	\$0.00	\$0.00	\$0.00	\$303.20	\$0.00	\$303.20
SemCrude	JACO OIL & GAS	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$20,003.16
SemCrude	JAED PRODUCTION COMPANY INC	\$20,003.16	\$0.00	\$673.68	\$0.00	\$0.00	\$673.68
SemCrude	JAMES A TAYLOR	\$0.00	\$0.00		\$17,764.54	\$0.00	\$17,764.54
SemCrude	JAMES ALLEN MCGAUGHEY	\$0.00	\$0.00		\$17,704.54	\$0.00	\$6,097.66
SemCrude	JAMES C KELLEY OIL AND GAS	\$0.00	\$0.00		\$0.00	\$0.00	\$15,032.54
SemCrude	JAMES D BULLARD	\$0.00	\$0.00		\$0.00	\$0.00	\$10,004.46
SemCrude	JAMES E WARREN	\$10,004.46	\$0.00		\$0.00	\$0.00	\$19,332.17
SemCrude	JAMES H & CYNTHIA J TICKEL	\$0.00	\$0.00			\$0.00	\$3,229.63
SemCrude SemCrude	JAMES H. AND ESTER M. BURT	\$0.00	\$0.00		\$0.00	\$0.00	\$3,788.92
SemCrude SemCrude	JAMES N HARDWICK	\$0.00	\$0.00		\$0.00	\$0.00 \$0.00	\$58,270.09
	JAMES Q MAGUIRE DBA JAMES Q	\$0.00	\$0.00		\$0.00	\$0.00	\$80,235.14
SemCrude SemCrude	JAMES W OWENS	\$0.00	\$0.00		\$0.00		\$23.79
SemCrude	JANES W OWERS JANET S. HARGRAVE	\$0.00	\$0.00		\$0.00	00.00	\$82,626.12
SemCrude	JANUARY INVESTMENTS L.L.C.	\$0.00	\$0.00	\$82,626.12	\$0.00	\$0.00	302,020.12
SemCrude	MINTORIX INTROVERSED BIBLO.						

			20.000	live Purchase Amain	is in Active States		
		Капѕаѕ	New Mexico	Oldahoma	Texas	Wyoming	Total
Debtor	Operator	\$747,830.10	\$0.00	\$0.00	\$0.00	\$0.00	\$747,830.10
SemCrude	JASON OIL COMPANY LLC	\$0.00	\$0.00	\$825.56	\$0.00	\$0.00	\$825,56
SemCrude	JAY PETROLEUM INC	\$0.00	\$0.00	\$84.55	\$0.00	\$0.00	\$84.55
SemCrude	JAY R STURMAN	\$0.00	\$0.00	\$21,352.19	\$0.00	\$0.00	\$21,352.19
SemCrude	JEC OPERATING LLC	\$0.00	\$0.00	\$40,549.59	\$0.00	\$0.00	\$40,549.59
SemCrude	JEFFREY RIXLEBEN	\$0.00	\$0.00	\$268,169.65	\$0.00	\$0.00	\$268,169.65
SemCrude	JEFFRIES PUMPING SERVICE INC	\$6,006.53	\$0.00	\$0.00	\$0.00	\$0.00	\$6,006.53
SemCrude	JENKINS & BROYLES PRODUCTION	\$19,742.76	\$0.00	\$0.00	\$0.00	\$0.00	\$19,742.76
SemCrude	JENKINS AND SONS PRODUCTION INC	\$0.00	\$0.00	\$325.88	\$0.00	\$0.00	\$325.88
SemCrude	JENKINS PRODUCTION COMPANY	\$0.00	\$0.00	\$0.00	\$1,608.29	\$0.00	\$1,608.29
SemCrude	JERICHO OIL & GAS COMPANY	\$0.00	\$0.00	\$2,194.21	\$0.00	\$0.00	\$2,194.21
SemCrude	JERICO RESOURCES JERRY SANNER OIL PROPERTIES	\$0.00	\$0.00	\$31,299.64	\$0.00	\$0.00	\$31,299.64
SemCrude	JERRY SANNER OIL FROFER TIES JERRY SCOTT DRILLING COMPANY	\$0.00	\$0.00	\$4,776.68	\$0.00	\$0.00	\$4,776.68
SemCrude		\$0.00	\$0.00	\$263.33	\$0.00	\$0.00	\$263.33
SemCrude	JERRY W LINE SR	\$0.00	\$0.00	\$7.70	\$0.00	\$0.00	\$7.70
SemCrude	JESS WEBB	\$0.00	\$0.00	\$10,766.71	\$0.00	\$0.00	\$10,766.71
SemCrude	JESSE H WYRICK JEWELL CONSULTING INC.	\$0.00	\$0.00	\$24,672.88	\$0.00	\$0.00	\$24,672.88
SemCrude		\$0.00	\$0.00	\$153,453.24	\$0.00	\$0.00	\$153,453.24
SemCrude	IIM AND JULIE MARTIN IIM CHAPMAN OIL & GAS	\$0.00	\$0.00	\$95.13	\$0.00	\$0.00	\$95.13
SemCrude		\$0.00	\$0.00	\$193.08	\$0.00	\$0.00	\$193.08
SemCrude	IIM COX OIL INC JIMMY R CHATHAM	\$0,00	\$0.00	\$256,108.48	\$0.00	\$0.00	\$256,108.48
SemCrude		\$20,259.29	\$0.00	\$0.00	\$0.00	\$0.00	\$20,259.29
SemCrude	JIMMY RAY WOLFE	\$0.00	\$0.00	\$0.00	\$5,702.47	\$0.00	\$5,702.47
SemCrude	JJH PROD. CO. JMA ENERGY COMPANY LLC	\$0.00	\$0.00	\$3,233,073.66	\$0.00	\$0.00	\$3,233,073.66
SemCrude	JMA ENERGY COMPANY LLC JOANNE CLEARY	\$0.00	\$0.00	\$289.26	\$0.00	\$0.00	\$289.26
SemCrude	JOANNE CLEAR I JOE B CLIFTON	\$0.00	\$0.00	\$74,137.16	\$0.00	\$0.00	\$74,137.16
SemCrude	JOE B CLIFTON EXPLORATION INC	\$0.00	\$0.00	\$146,825.36	\$0.00	\$0.00	\$146,825.36
SemCrude	JOE BOB KELLEY AND/OR VICTOR SLONE JR	\$0.00	\$0.00	\$41,497.64	\$0.00	\$0.00	\$41,497.64
SemCrude	JOE BRAKE	\$209.01	\$0.00	\$0.00	\$0.00	\$0.00	\$209.01
SemCrude	JOE E SHARBER	\$0.00	\$0.00	\$53,708.53	00.00	\$0.00	\$53,708.53
SemCrude	JOE ROBERSON	\$0.00	\$0.00	\$25,397.51	\$0.00	\$0.00	\$25,397.51
SemCrude	JOHN ALMACK	\$0.00	\$0.00	\$63.82	\$0.00	\$0.00	\$63.82
SemCrude	JOHN D PITMAN	\$102,962.64	\$0.00	\$0.00	\$0.00	\$0.00	\$102,962.64
SemCrude	JOHN G GRAVES OIL PROPERTIES INC	\$0.00	\$0.00	\$21.86	\$0.00	\$0.00	\$21.86
SemCrude	JOHN G MALONEY	\$0.00	\$0.00	\$2,346.07	\$0.00	\$0.00	\$2,346.07
SemCrude	JOHN O MADORET	\$0.00	\$89,126.44	\$0.00	\$0.00	\$0.00	\$89,126.44
SemCrude	JOHN M WOODY D/B/A DOODLEBUG	\$8,648.21	\$0.00	\$0.00	\$0.00	\$0.00	\$8,648.21
SemCrude	JOHN MASSAD	\$0.00	\$0.00	\$71.62	\$0.00	\$0.00	\$71.62
SemCrude	JOHN R EVANS	\$17,434.82	\$0.00	\$0.00	\$0.00	\$0.00	\$17,434.82
SemCrude	JOHN R WARDEN INC	\$0.00	\$0.00	\$23,481.92	\$0.00	\$0.00	\$23,481.92
SemCrude	IOKERS WILD ENERGY LLC	\$0.00	\$0.00	\$71,586.51	\$0.00	\$0.00	\$71,586.51
SemCrude	JOLEN OPERATING COMPANY	\$0.00	\$0.00	\$97,959.76	\$0.00	\$0.00	\$97,959.76
SemCrude	JONES & BUCK DEVELOPMENT	\$71,333.74	\$0.00	\$0.00	\$0.00	\$0.00	\$71,333.74
SemCrude SemCrude	IORDAN & JORDAN OIL CO	\$0.00	\$0,00	\$207,223.63	\$0.00	00.02	\$207,223.63 \$12,643.92
SemCrude SemCrude	JORDAN & JORDAN GID GO JORDAN DEVELOPMENT COMPANY LLC	\$0.00	\$0.00	\$12,643.92	\$0.00	\$0.00	\$12,643.92 \$101,399.18
SemCrude SemCrude	JOSIE OIL & GAS INC	\$0.00	\$0.00	\$101,399.18	\$0.00	\$0.00	\$101,399.18
SemCrude SemCrude	JRHOP INC	\$0.00	\$0.00	\$34,718.23	\$0.00	\$0.00	\$34,718.23 \$42.95
SemCrude SemCrude	JWS	\$0.00	\$0.00	\$0.00	\$42.95	00.00	\$4,716.37
SemCrude	K & W VANPELT COMPANY	\$0.00	\$0.00	\$0.00	\$4,716.37	\$0.00	.p4, / 1♥.J {
Scholage	50 mm 51 4 1 75 17 mm 4 m 2 1 1 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2						

		10,77246		(C)(T)(B)(B)(B)(C)(B)(B)(B)(B)(B)(B)(B)(B)(B)(B)(B)(B)(B)	n Active States		
				Account to the Contract of the	Texas	4411624631424634-4-2	Total
Debtor	Operator	Hitte virilitation of the control of the cont	ew Mexico \$0.00	\$33,264.19	\$0.00	\$0.00	\$33,264.19
SemCrude	K C RESOURCES INC	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$20,068.98
SemCrude	K G OIL COMPANY	\$20,068.98	\$0.00	\$0.00	\$0.00	\$0.00	\$1,524.03
SemCnide	K LANCE SNYDER	\$1,524.03	\$0.00	\$528.07	\$0.00	\$0.00	\$528.07
SemCrude	K OIL COMPANY	\$0.00	\$0.00	\$0.00	\$43,384.23	\$0.00	\$43,384.23
SemCrude	K&D OIL & GAS COMPANY	\$0.00	\$0.00	\$41,202.40	\$0.00	\$0.00	\$41,202.40
SemCrude	K&S OIL COMPANY	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$73,392.27
SemCrude	K. E. SNYDER COMPANY LLC	\$73,392.27	\$0.00	\$1,003,854.97	\$0.00	\$0.00	\$2,766,186.28
SemCrude	KAHAN AND ASSOCIATES INC	\$1,762,331.31	\$0.00	\$19,546.30	\$0.00	\$0.00	\$19,546.30
SemCrude	KAL ENERGY INC	\$0.00 \$0.00	\$0.00	\$71.69	\$0.00	\$0.00	\$71.69
SemCrude	KATO OPERATING COMPANY	\$0.00 \$1,473.86	\$0.00	\$0.00	\$0.00	\$0.00	\$1,473.86
SemCrude	KC OIL	\$1,473.80	\$0.00	\$76,337.99	\$0.00	\$0.00	\$76,337.99
SemCrude	KECHI ENERGY LLC	\$0.00	\$0.00	\$113.57	\$0.00	\$0.00	\$113.57
SemCrude	KEENER OIL & GAS COMPANY	\$0.00	\$0.00	\$1,317,509.55	\$0.00	\$0.00	\$1,317,509.55
SemCrude	KEITH F WALKER OIL & GAS COMPANY LLC	\$0.00	\$0.00	\$246.64	\$0.00	\$0.00	\$246.64
SemCrude	KELMAR OIL COMPANY	\$0.00	\$0.00	\$28,697.36	\$0.00	\$0.00	\$28,697.36
SemCrude	KENNETH D & SHIRLEY A KLINGER	\$0.00 \$0.00	\$0.00	\$51,678.79	\$0.00	\$0.00	\$51,678.79
SemCrude	KENNETH LOWE	\$0.00	\$0.00	\$9,569.07	\$0.00	\$0.00	\$9,569.07
SemCrude	KENNETH RAY FIREY	\$0.00	\$0.00	\$21,214.19	\$0.00	\$0.00	\$21,214.19
SemCrude	KENNETH STALNAKER	\$0.00	\$0.00	\$17,591.84	\$0.00	\$0.00	\$17,591.84
SemCrude	KENT ALLEN	\$0.00	\$0.00	\$20,835.12	\$0.00	\$0.00	\$20,835.12
SemCrude	KEPCO INC	\$0.00 \$308.52	\$0.00	\$0.00	\$0.00	\$0.00	\$308.52
SemCrude	KERA OIL	\$308.32	\$0.00	\$0.00	\$97,817.63	\$0.00	\$97,817.63
SemCrude	KERN CO	\$0.00	\$26,302.54	\$0.00	\$0.00	\$0.00	\$26,302.54
SemCrude	KERNS PETROLEUM INC	\$0.00	\$0.00	\$0.57	\$0.00	\$0.00	\$0.57
SemCrude	KERR-MCGEE CORPORATION	\$0.00	\$0.00	\$3,375.00	\$0.00	\$0.00	\$3,375.00
SemCrude	KEY ENERGY CORPORATION	\$12,142.54	\$0.00	\$0.00	\$0.00	\$0.00	\$12,142.54
SemCrude	KEY GAS CORP	\$0.00	\$0.00	\$285.06	\$0.00	\$0.00	\$285.06
SemCrude	KEY OPERATING COMPANY	\$0.00	\$0.00	\$2,348.45	\$0.00	\$0.00	\$2,348.45
SemCrude	KIBTEC OIL & GAS INC.	\$0.00	\$0.00	\$223.19	\$0.00	\$0.00	\$223.19
SemCrude	KIM ALBRITTON DBA KIM RAY OIL & GAS	\$0.00	\$0,00	\$481.31	\$0.00	\$0.00	\$481.31
SemCrude	KIM RAY OIL & GAS	\$0.00	\$0.00	\$54,233.74	\$0.00	\$0.00	\$54,233.74
SemCrude	KIMRACO OIL INC	\$0.00	\$0.00	\$1,413,918.93	\$0.00	\$0.00	\$1,413,918.93
SemCrude	KINGERY DRILLING CO INC	\$0.00	\$0.00	\$30.43	\$0.00	\$0.00	\$30.43
SemCrude	KINGFISHER RESOURCES INC.	\$0.00	\$0.00	\$1,305.40	\$0.00	\$0.00	\$1,305.40
SemCrude	KIRKPATRICK OIL & GAS LLC	\$0.00	\$0.00	\$1,406.06	\$0.00	\$0.00	\$1,406.06
SemCrude	KIRKPATRICK OIL COMPANY	\$0.00	\$0.00	\$21,816.01	\$0.00	\$0.00	\$21,816.01
SemCrude	KIRRIE OIL COMPANY LLC	\$881.16	\$0.00	\$0.00	\$0.00	\$0.00	\$881.16
SemCrude	KIZZAR WELL SERVICING INC	\$395,211.52	\$0.00	\$0.00	\$0.00	\$0.00	\$395,211.52
SemCrude	KLM EXPLORATION CO	\$0.00	\$0.00	\$237.97	\$0.00	\$0.00	\$237.97
SemCrude	KLO	\$0.00	\$0.00	\$290,458.55	\$0.00	\$0.00	\$290,458.55
SemCrude	KOBY OIL CO LLC	\$0.00	\$0.00	\$4.58	\$0.00	\$0.00	\$4.58
SemCrude	KOEHN PUMPING LLC	\$0.00	\$0.00	\$9.73	\$0.00	\$0.00	\$9.73
SemCrude	KROTZER OIL COMPANY	\$0.00	\$0.00	\$0.83	\$0.00	\$0.00	\$0.83
SemCrude	KRS OIL COMPANY	\$0.00	\$0.00	\$305,954.69	\$0.00	\$0.00	\$305,954.69
SemCrude	KRUMME OIL COMPANY	\$0.00	\$0.00	\$405.32	\$0.00	\$0.00	\$405.32
SemCrude	KWB INC	\$0.00	\$0.00	\$22,027.31	\$0.00	\$0.00	\$22,027.31
SemCrude	KWB OIL PROPERTY MANAGEMENT	\$71,877.21	\$0.00	\$0.00	\$0.00	\$0.00	\$71,877.21
SemCrude	L D HALE	\$0.00	\$0.00	\$188,633.79	\$0.00	\$0.00	\$188,633.79
SemCrude	L E JONES OPERATING INC	\$0.00	*****	• •			

Depti Name						in Warry States		公司的 的問題
Semicropies	(1907年)(第1980年)(1937年)				Oklahoma	Texas	Wyoming	Total
SenfCride LESTILAM STATE	Debtor		DESTRUCTION OF THE PROPERTY OF					\$2,894.54
SemCrude L.G.MARTIN ARID JOLAS AMERIN \$10.00 \$30.00 \$35.00 \$30.00 \$35.		L E SPLAWN				\$0.00	\$0.00	
SenfCrude PRINTT OIL PROPERTIES \$0.00 \$0.00 \$245,304.14 \$0.00 \$0.00 \$245,304.16 \$266.00 \$240,205.25 \$260.00 \$0.00 \$0.00 \$24,210.32 \$260.00 \$0.00 \$24,210.32 \$260.00 \$0.00 \$24,210.32 \$260.00 \$24,210.32 \$260.00 \$24,210.32 \$260.00 \$24,210.32 \$260.00 \$24,210.32 \$260.00 \$24,210.32 \$260.00 \$24,210.32 \$260.00 \$24,210.32 \$260.00 \$24,210.32 \$260.00 \$24,210.32 \$260.00 \$260.00 \$24,210.32 \$260.00 \$26	SemCrude				- •	\$0.00	\$0.00	\$3.10
SemCrude	SemCrude			•		\$0.00	\$0.00	
SemCrude LWAYNE NEAL AND STELLAL NEAL \$0.00 \$0.00 \$0.00 \$22,133.96 \$0.00 \$22,133.96 \$0.00	SemCrude	L W WILLIAMS JR		• • • •	•		\$0.00	\$24,210.32
SemCrude	SemCrude						\$0.00	\$22,133.96
SemCrade Let Oil PROPERTIES NC S0.00 S0.00 S0.00 S5.004,42.05 S0.00 S5.004,42.05 SemCrade Let CAMPANA INC S0.00	SemCrude						\$0.00	\$61,467.18
SemCrade LCS_PRODUCTION COMPANY \$0.00 \$0.00 \$915.63 \$0.00 \$915.65 \$0.00 \$0.0	SemCrude	L&J OIL PROPERTIES INC					\$0.00	\$5,202,442.05
SemCrude LACHEMANA INC CO	SemCrude						\$0.00	\$915.63
SemCrude	SemCrude		***				\$0.00	\$508,856.63
SemCrude	SemCrude			•			\$0.00	\$47,053.67
SemCrude	SemCrude			•			\$0.00	\$8.93
SemCrude LADY BUG OIL COMPANY S0.00 \$30.00 \$33.959.86 \$30.00 \$30.00 \$33.959.86 \$30.00 \$30.00 \$31.37584 \$50.00 \$30.00 \$31.37584 \$50.00 \$30.00	SemCrude		* *	·			\$0.00	\$2,059.77
SemCrude LAMANCO BRILLING COMPANY \$0.00 \$0.00 \$21,357.84 \$0.00 \$21,357.84 \$0.00 \$21,357.84 \$0.00 \$21,357.84 \$0.00 \$21,357.84 \$0.00 \$20.	SemCrude	LADY BUG OIL COMPANY				***		\$38,969.86
SemCrude	SemCrude	LAMAMCO DRILLING COMPANY	•				00.02	\$21,357.84
SemCrude	SemCrude		• • • • • • • • • • • • • • • • • • • •					\$4,077,669.96
SemCrude	SemCrude	LANCE RUFFEL OIL & GAS CORP						
SemCrude LANDERS AND MUSGROVE OIL \$500.58 \$50.00 \$32,855.90 \$0.00 \$32,855.90 \$0.00 \$32,855.90 \$0.00 \$32,855.90 \$0.00 \$0.00 \$5.51.37 \$0.00 \$0.00 \$5.51	SemCrude							
SemCrude	SemCrude	LANDERS AND MUSGROVE OIL			•			\$32,855.90
SemCrude	SemCrude	LANDMARK ENERGY LLC						\$6,551.37
SemCrude LANDMARK RESOURCES INC \$0.00 \$0.00 \$26,955.27 \$0.00 \$0.00 \$25,955.27 \$0.00 \$0.00 \$25,955.27 \$0.00 \$0.00 \$25,955.27 \$0.00 \$0.00 \$33,042.54 \$0.00 \$0.00 \$33,042.54 \$0.00 \$0.00 \$33,042.54 \$0.00 \$0.00 \$33,042.54 \$0.00 \$0.00 \$33,042.54 \$0.00 \$0.00 \$39,98.79 \$0.00 \$0.00 \$39,98.79 \$0.00 \$0.00 \$39,98.79 \$0.00 \$0.00 \$39,98.79 \$0.00 \$0.00 \$39,98.79 \$0.00	SemCrude							\$10,105.39
SemCrude LAPETCO INC S0.00 \$0.00 \$33,042.54 \$0.00 \$0.00 \$39,908.79 \$0.00 \$0.00 \$39,908.79 \$0.00 \$0.00 \$39,908.79 \$0.00 \$0.00 \$39,908.79 \$0.00 \$0.00 \$39,908.79 \$0.00 \$0.00 \$35,908.79 \$0.00 \$0.00 \$35,908.79 \$0.00 \$0.00 \$35,908.79 \$0.00 \$0.00 \$35,908.79 \$0.00 \$0.00 \$35,908.79 \$0.00 \$0.00 \$0.00 \$35,88.55 \$0.00	SemCrude	LANDMARK RESOURCES INC	·					\$26,955.27
SemCrude	SemCrude		The state of the s		•			\$3,042.54
SemCrude LARIO OIL AND GAS CO \$0.00 \$0.00 \$358.85 \$0.00 \$0.00 \$838.85 \$80.00 \$0.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$154.61 \$80.00 \$150.00	SemCrude	LARIAT ENERGY & SUPPLY CO			,			\$39,908.79
SemCrude LARKIN OIL COMPANY \$0.00 \$0.00 \$154.61 \$0.00 \$0.00 \$1.6273.16 \$0.00 \$	SemCrude	LARIO OIL AND GAS CO			•			\$858.85
SemCrude	SemCrude			-			· ·	\$154.61
SemCrude LARRON OIL CORPORATION S873.38 S0.00 S0.00 S0.00 S0.00 S873.38 SemCrude LARRY D HALE S0.00 S0	SemCrude	LARKIN OIL COMPANY					·	\$116,273.16
SemCrude LARRY JAMES S0.00 S0.00 S59,150.67 S0.00 S0.00 S59,150.67 SemCrude LARSON EXPLORATION COMPANY S0.00 S0.00 S21.47 S0.00 S0.00 S21.38 SemCrude LATIGO DRILLING INC S0.00 S0.00 S0.00 S4,987.97 S0.00 S0.00 S4,987.97 SemCrude LATIGO OIL & GAS INC S0.00 S0.00 S0.00 S63,944.33 S0.00 S0	SemCrude	LARRON OIL CORPORATION				·		\$873.38
SemCrude LARRY JAMES S0.00 S0.00 S21.47 S0.00 S0.00 S0.00 S21.47 S0.00 S0.00 S0.00 S21.47 S0.00 S0.00 S0.00 S0.00 S21.47 S0.00 S0.00 S0.00 S21.47 S0.00 S0.00 S0.00 S0.00 S21.47 S0.00 S0.00 S0.00 S21.47 S0.00 S0.00 S0.00 S21.47 S0.00	SemCrude	LARRY D HALE					-	\$59,150.67
SemCrude LARSON EXPLORATION COMPANY \$0.00 \$0.00 \$81.38 \$0.00 \$0.00 \$81.38 \$0.00 \$0.00 \$0.00 \$81.38 \$0.00 \$0.00 \$0.00 \$81.38 \$0.00	SemCrude	LARRY JAMES						\$21.47
SemCrude LATIGO DRILLING INC \$0.00 \$0.00 \$4,987.97 \$0.00 \$0.00 \$4,987.97 \$0.00 \$0.00 \$4,987.97 \$0.00 \$0.00 \$63,944.33 \$0.00 \$0.00 \$63,944.33 \$0.00 \$0.00 \$63,944.33 \$0.00 \$0.00 \$63,944.33 \$0.00 \$0.00 \$63,944.33 \$0.00		LARSON EXPLORATION COMPANY	- · · · · · · · · · · · · · · · · · · ·					\$81.38
SemCrude	SemCrude							\$4,987.97
SemCrude LAWRENCE JAMES \$0.00	SemCrude	LATIGO OIL & GAS INC						\$63,944.33
SemCrude LCB RESOURCES INC \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$3,234.34 SemCrude LEBEN OIL COMPANIES LLC \$3,234.34 \$0.00 \$0.00 \$244,785.01 \$0.00 \$0.00 \$244,785.01 SemCrude LEDON TURPIN PRODUCTION \$0.00 \$0.00 \$244,785.01 \$0.00 \$0.00 \$244,785.01 SemCrude LEED ENERGY LLC \$0.00 \$0.00 \$0.00 \$28,518.15 \$0.00 \$0.00 \$28,518.15 SemCrude LEEJAN OIL COMPANY \$0.00 \$0.00 \$0.00 \$31,118.97 \$0.00 \$31,118.97 SemCrude LEEMAN ENERGY CORP \$0.00 \$0.00 \$23,462.91 \$0.00 \$0.00 \$20,00 \$0.00 \$31,118.97 \$0.00 \$31,118.97 \$0.00 \$32,3462.91 \$0.00 \$0.00 \$31,118.97 \$0.00 \$32,3462.91 \$0.00 \$0.00 \$23,462.91 \$0.00 \$0.00 \$23,462.91 \$0.00 \$0.00 \$23,462.91 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$		LAWRENCE JAMES						\$90.57
SemCrude LEBEN OIL COMPANIES LLC \$3,234.34 \$0.00 \$0.00 \$244,785.01 \$0.00 \$244,785.01 SemCrude LEDON TURPIN PRODUCTION \$0.00 \$0.00 \$224,785.01 \$0.00 \$0.00 \$28,518.15 SemCrude LEED ENERGY LLC \$0.00 \$0.00 \$0.00 \$31,118.97 \$0.00 \$31,118.97 SemCrude LEEJAN OIL COMPANY \$0.00 \$0.00 \$0.00 \$31,118.97 \$0.00 \$23,462.91 SemCrude LEEMAN ENERGY CORP \$0.00 \$0.00 \$23,462.91 \$0.00 \$0.00 \$20,000 \$0.00 \$20,000 \$22,751.91 \$20,000 \$22,751.91 \$20,000 \$22,751.91 \$20,000 \$20,000 \$22,751.91 \$20,000 \$20,000 \$	SemCrude			* * * * * * * * * * * * * * * * * * * *			· ·	\$3,234,34
SemCrude LEDON TURPIN PRODUCTION \$0.00 \$0.00 \$28,518.15 \$0.00 \$0.00 \$28,518.15 \$0.00 \$0.00 \$28,518.15 \$0.00 \$0.00 \$31,118.97 \$0.00 \$31,118.97 \$0.00 \$31,118.97 \$0.00 \$31,118.97 \$0.00 \$31,118.97 \$0.00 \$31,118.97 \$0.00 \$31,118.97 \$0.00 \$31,118.97 \$0.00		LEBEN OIL COMPANIES LLC						
SemCrude LEED ENERGY LLC \$0.00 \$0.00 \$0.00 \$1,118.97 SemCrude LEEJAN OIL COMPANY \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$23,462.91 \$0.00 \$23,462.91 \$0.00 \$0.00 \$23,462.91 \$0.00 \$0.00 \$23,462.91 \$0.00		LEDON TURPIN PRODUCTION				•	•	
SemCrude LEEJAN OIL COMPANY \$0.00 \$0.00 \$23,462.91 \$0.00 \$0.00 \$23,462.91 SemCrude LEEMAN ENERGY CORP \$0.00 \$0.00 \$0.00 \$1,400.09 \$0.00 \$1,400.09 SemCrude LEFFEL'S INC \$0.00 \$0.00 \$20,375.15 \$0.00		LEED ENERGY LLC	* *		•			
SemCrude LEEMAN ENERGY CORP \$6.00 \$23,402.91 \$0.00 \$1,400.09 SemCrude LEFFEL'S INC \$0.00 \$0.00 \$0.00 \$1,400.09 \$0.00 \$20,375.15 SemCrude LEMLEY OIL CORPORATION \$0.00 \$0.00 \$22,751.91 \$0.00 \$0.00 \$22,751.91 SemCrude LEONARD VANN \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$223.05 SemCrude LEROY HERRMAN \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$12,335.90 SemCrude LESTER DEAN IKENBERRY \$0.00 \$0.00 \$0.00 \$0.00 \$10,120.09 SemCrude LESTER TAJCHMAN D/B/A TAJCHMAN \$10,120.09 \$0.00 \$99,867.58 \$0.00 \$0.00 \$99,867.58 SemCrude LEWBET INC \$0.00 \$0.00 \$73.98 \$0.00 \$0.00 \$13,45.19 SemCrude LEWIS OIL CORPORATION \$0.00 \$0.00 \$21,345.19 \$0.00 \$21,345.19 SemCrude LIBERTY OPERATING INC. \$0.00	•	LEEJAN OIL COMPANY						•
SemCrude LEFFEL'S INC \$0.00 \$0.00 \$0.00 \$0.00 \$20,375.15 SemCrude LEMLEY OIL CORPORATION \$0.00 \$0.00 \$22,751.91 \$0.00 \$0.00 \$22,751.91 SemCrude LEONARD VANN \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$223.05 SemCrude LEROY HERRMAN \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$12,335.90 SemCrude LESTER DEAN IKENBERRY \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$10,120.09 SemCrude LESTER TAJCHMAN D/B/A TAJCHMAN \$10,120.09 \$0.00 \$99,867.58 \$0.00 \$0.00 \$99,867.58 SemCrude LEWBET INC \$0.00 \$0.00 \$73.98 \$0.00 \$0.00 \$73.98 SemCrude LEWIS OIL CORPORATION \$0.00 \$0.00 \$21,345.19 \$0.00 \$0.00 \$21,345.19 SemCrude LIBERTY OPERATING INC. \$0.00 \$0.00 \$229,220.93 \$0.00 \$0.00 \$289,220.93 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>•</td>								•
SemCrude LEMLEY OIL CORPORATION \$0.00 \$0.00 \$22,751.91 \$0.00 \$22,751.91 SemCrude LEONARD VANN \$0.00 \$0.00 \$0.00 \$0.00 \$223.05 SemCrude LEROY HERRMAN \$0.00 \$0.00 \$10.00 \$0.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.120.09 \$10.00 \$10	- · · · ·	LEFFEL'S INC	•					,
SemCrude LEONARD VANN \$0.00 \$223.05 SemCrude LEROY HERRMAN \$223.05 \$0.00 \$0.00 \$0.00 \$0.00 \$12,335.90 SemCrude LESTER DEAN IKENBERRY \$0.00 \$0.00 \$10,120.09 \$0.00 \$0.00 \$0.00 \$10,120.09 SemCrude LESTER TAJCHMAN D/B/A TAJCHMAN \$10,120.09 \$0.00 \$99,867.58 \$0.00 \$0.00 \$99,867.58 SemCrude LEWBET INC \$0.00 \$0.00 \$73.98 \$0.00 \$0.00 \$73.98 SemCrude LEWIS OIL CORPORATION \$0.00 \$0.00 \$21,345.19 \$0.00 \$0.00 \$21,345.19 SemCrude LIBERTY OPERATING INC. \$0.00 \$0.00 \$289,220.93 \$0.00 \$0.00 \$289,220.93		LEMLEY OIL CORPORATION						,
SemCrude LEROY HERRMAN \$223.05 \$0.00 \$0.00 \$0.00 \$12,335.90 SemCrude LESTER DEAN IKENBERRY \$0.00 \$0.00 \$10,120.09 \$0.00 \$0.00 \$0.00 \$10,120.09 SemCrude LESTER TAJCHMAN D/B/A TAJCHMAN \$10,120.09 \$0.00 \$99,867.58 \$0.00 \$0.00 \$99,867.58 SemCrude LEWBET INC \$0.00 \$0.00 \$73.98 \$0.00 \$0.00 \$73.98 SemCrude LEWIS OIL CORPORATION \$0.00 \$0.00 \$21,345.19 \$0.00 \$0.00 \$21,345.19 SemCrude LIBERTY OPERATING INC. \$0.00 \$0.00 \$289,220.93 \$0.00 \$0.00 \$289,220.93								
SemCrude LESTER DEAN IKENBERRY \$0.00 \$0.00 \$10,120.09 SemCrude LESTER TAJCHMAN D/B/A TAJCHMAN \$10,120.09 \$0.00 \$0.00 \$0.00 \$0.00 \$99,867.58 SemCrude LEWBET INC \$0.00 \$0.00 \$99,867.58 \$0.00 \$0.00 \$73.98 SemCrude LEWIS OIL CORPORATION \$0.00 \$0.00 \$21,345.19 \$0.00 \$0.00 \$21,345.19 SemCrude LIBERTY OPERATING INC. \$0.00 \$0.00 \$289,220.93 \$0.00 \$0.00 \$289,220.93		LEROY HERRMAN				*		
SemCrude LESTER TAJCHMAN D/B/A TAJCHMAN \$10,120.09 \$0.00 \$99,867.58 SemCrude LEWBET INC \$0.00 \$99,867.58 \$0.00 \$99,867.58 SemCrude LEWIS OIL CORPORATION \$0.00 \$0.00 \$13,345.19 \$0.00 \$0.00 \$21,345.19 SemCrude LIBERTY OPERATING INC. \$0.00 \$0.00 \$289,220.93 \$0.00 \$0.00 \$289,220.93		LESTER DEAN IKENBERRY						
SemCrude LEWIS OIL CORPORATION \$0.00 \$0.00 \$73.98 \$0.00 \$0.00 \$73.98 SemCrude LEWIS OIL CORPORATION \$0.00 \$0.00 \$21,345.19 \$0.00 \$0.00 \$21,345.19 SemCrude LIBERTY OPERATING INC. \$0.00 \$0.00 \$289,220.93 \$0.00 \$0.00 \$289,220.93		LESTER TAJCHMAN D/B/A TAJCHMAN						
SemCrude LEWIS OIL CORPORATION \$0.00 \$0.00 \$21,345.19 \$0.00 \$21,345.19 SemCrude LIBERTY OPERATING INC. \$0.00 \$0.00 \$22,345.19 \$0.00 \$0.00 \$289,220.93				* * * *	,			
SemCrude LIBERTY OPERATING INC. \$0.00 \$0.00 \$289,220.93 \$0.00 \$289,220.93				*				
\$0.00			-		,			
	SemCrude	LINDELL HOWARD AND JACK TAYLOR	\$0.00	\$0.00	\$289,220.93	30.00	₩.JU	\$\$\$\$\$ ₃ \$\$\$\$\$\$\$\$

			anamaki i a manamani ya masa w	First Purchase Amou	REPLANTAGE STATE		
			New Mexico		Texas	Wyoming	
Debtor	Operator 1888 1888 1888 1888 1888 1888 1888 18	Kansas \$0.00	\$0.00	\$5,202.40	\$0.00	\$0.00	\$5,202.40
SemCrude	LINDENMUTH & ASSOCIATES INC	\$0.00	\$0.00	\$41,948.17	\$0.00	\$0.00	\$41,948.17
SemCrude	LINDENMUTH & ASSOCIATES, INC	\$105,627.42	\$0.00	\$0.00	\$0.00	\$0.00	\$105,627.42
SemCrude	LINK OIL COMPANY	\$0.00	\$0.00	\$0.00	\$189,984.78	\$0.00	\$189,984.78
SemCrude	LINN ENERGY HOLDINGS	\$0.00	\$0.00	\$515,739.02	\$0.00	\$0.00	\$515,739.02
SemCrude	LITTLE BEAR RESOURCES LLC	\$0.00	\$0.00	\$13,737.45	\$0.00	\$0.00	\$13,737.45
SemCrude	LITZENBERGER EXPLORATION INC.	\$0.00	\$0.00	\$22,748.21	\$0.00	\$0.00	\$22,748.21
SemCrude	LLOYD RICHARDSON	\$0.00	\$0.00	\$120,743.45	\$0.00	\$0.00	\$120,743.45
SemCrude	LOBAR OIL CO INC	\$0.00	\$0.00	\$40,831.05	\$0.00	\$0.00	\$40,831.05
SemCrude	LOBO EXPLORATION INC	\$101.79	\$0.00	\$0.00	\$0.00	\$0.00	\$101.79
SemCrude	LOCO PRODUCTION	\$0.00	\$0.00	\$20,349.59	\$0.00	\$0.00	\$20,349.59
SemCrude	LONE OAK PLANTATION INC	\$3,466.61	\$0.00	\$0.00	\$0.00	\$0.00	\$3,466.61
SemCrude	LONE STAR OIL CO.	\$39,263.61	\$0.00	\$0.00	\$0.00	\$0.00	\$39,263.61
SemCrude	LONG DRILLING CO	\$23,833.79	\$0.00	\$0.00	\$0.00	\$0.00	\$23,833.79
SemCrude	LOREN L. HAYDEN LORENTZ OIL & GAS L L C	\$0.00	\$0.00	\$41,005.29	\$0.00	\$0.00	\$41,005.29
SemCrude	LOWRY EXPLORATION INC	\$0.00	\$0.00	\$93,503.95	\$0.00	\$0.00	\$93,503.95
SemCrude	LUCKY ACE PETROLEUM, LLC	\$0.00	\$0.00	\$24,790.18	\$0.00	\$0.00	\$24,790.18
SemCrude	LUKE OIL COMPANY	\$0.00	\$0.00	\$169,710.91	\$0.00	\$0.00	\$169,710.91
SemCrude	LUNAR INVESTMENTS INC	\$0.00	\$0.00	\$165,442.21	\$0.00	\$0.00	\$165,442.2I
SemCrude	M AND V RESOURCES INC	\$0.00	\$0.00	\$25,275.15	\$0.00	\$0.00	\$25,275.15
SemCrude	M CRAIG DEISENROTH	\$0.00	\$0.00	\$99,249.70	\$0.00	\$0.00	\$99,249.70 \$318,091.24
SemCrude SemCrude	M E BOOTH CORPORATION	\$0.00	\$0.00	\$318,091.24	\$0.00	\$0.00	\$123,016.18
SemCrude SemCrude	M E KLEIN & ASSOCIATES INC	\$0.00	\$0.00	\$123,016.18	\$0.00	\$0.00	\$123,010.18
SemCrude SemCrude	M M & M RESOURCES	\$0.00	\$0.00	\$358.82	\$0.00	\$0.00	\$41,469.34
SemCrude	M M ENERGY INC	\$17,048.22	\$0.00	\$24,421.12	\$0.00	\$0.00	\$70,282.28
SemCrude	M&M ENERGY	\$0.00	\$0.00	\$70,282.28	\$0.00	\$0.00 \$0.00	\$29,271.70
SemCrude	MA & MC OIL & GAS LLC	\$29,271.70	\$0.00	\$0.00	\$0.00	\$0.00	\$57,744.56
SemCrude	MACKELLAR INC	\$0.00	\$0.00	\$57,744.56	\$0.00 \$0.00	\$0.00	\$98,066.54
SemCrude	MACKEY CONSULTING & LEASING	\$0.00	\$0.00	\$98,066.54	\$0.00	\$0.00	\$6,489.34
SemCrude	MACMART INC	\$0.00	\$0.00	\$6,489.34	\$0.00	\$0.00	\$6,087.65
SemCrude	MAGNUM ENERGY INC.	\$0.00	\$0.00	\$6,087.65	\$0.00		\$47.64
SemCrude	MAIER RESOURCES INC	\$0.00	\$0.00	\$47.64	\$0.00 \$22,563.52		\$22,563.52
SemCrude	MAJESTIC OIL & GAS	\$0.00	\$0.00	\$0.00 \$2.23	\$0.00		\$2.23
SemCrude	MALY & MALY OIL	\$0.00	\$0.00	\$2.23 \$45.60	\$0.00		\$45.60
SemCrude	MANTEX INC	\$0.00	\$0.00	\$0.00	\$173,783.57		\$173.783.57
SemCrude	MAP PRODUCTION COMPANY INC	\$0.00	\$0.00 \$0.00	\$20,485.38	\$0.00		\$20,485.38
SemCrude	MARATHON OIL COMPANY	\$0.00	\$0.00	\$26,249.30	\$0,00		\$26,249.30
SemCrude	MAR-BAR ENTERPRISES INC	\$0.00	\$0.00	\$0.00	\$30,841.56		\$30,841.56
SemCrude	MARCON OPERATING COMPANY INC	\$0.00	\$0.00	\$1,434.48	\$0.00		\$1,434.48
SemCrude	MARJO OPERATING CO INC	\$0.00	\$0.00		\$0.00		\$33.96
SemCrude	MARK CONNELL	\$33.96	\$0.00		\$0.00		\$36,976.48
SemCrude	MARK E DOBRINSKI	\$36,976.48 \$0.00	\$0.00		\$0.00		\$186.81
SemCrude	MARK HARGUS - KAMCO HOT OIL	\$0.00	\$0.00		\$0.00		\$323,795.81
SemCrude	MARK HAYNES	\$0.00	\$0.00		\$0.00		\$53,473.61
SemCrade	MARK HOLLOWAY INC	\$0.00 \$39,133.94	\$0.00	•	\$0.00		\$42,962.20
SemCrude	MARLIN OIL CORPORATION	\$39,133.94 \$6,185.21	\$0.00	•	\$0.00		\$6,185.21
SemCrude	MARSHALL OIL COMPANY	\$0.00	\$0.00		\$0.00		\$79,562.15
SemCrude	MARTIN OIL & GAS	\$0.00	\$0.00	- ,	\$0.00		\$2.73
SemCrude	MARTIN OIL PROPERTIES	30.00	00.00	-			

		a a markara samagaara waasii		mang many a company mang man	in Active States		
			vew Mexico	Oklahoma	Texas Wi	oming	Total
Debtor	Operator	Kansas \$0.00	SO.00	\$9,323.72	\$0.00	\$0.00	\$9,323.72
SemCrude	MARY ELLEN HOLMES	*****	\$0.00	\$760.93	\$0.00	\$0.00	\$760.93
SemCrude	MATTEA OIL & GAS LLC	\$0.00	\$0.00	\$323.19	\$0.00	\$0.00	\$323.19
SemCrude	MAURICE WESSEL	\$0.00 \$0.00	\$0.00	\$40,595.34	\$0.00	\$0.00	\$40,595.34
SemCrude	MAXIM ENERGY CORPORATION	\$0.00	\$0.00	\$178,270.18	\$0.00	\$0.00	\$178,270.18
SemCrude	MAYABB OIL COMPANY	\$0.00 \$43.41	\$0.00	\$0.00	\$0.00	\$0.00	\$43.41
SemCrude	McCANN DRILLING, INC.		\$0.00	\$1,652.73	\$0.00	\$0.00	\$4,862,337.81
SemCrude	MCCOY PETROLEUM CORPORATION	\$4,860,685.08	\$0.00	\$0.00	\$0.00	\$0.00	\$31,203.42
SemCrude	MCELVAIN OIL & GAS PROPERTIES	\$31,203.42 \$109,943.18	\$0.00	\$0.00	\$0.00	\$0.00	\$109,943.18
SemCrude	MCGINNESS OIL CO OF KS	\$109,943.18	\$0.00	\$141.61	\$0.00	\$0.00	\$141.61
SemCrude	MCGOWEN RESOURCES CO. INC.	\$0.00 \$0.00	\$0.00	\$23,825.63	\$0.00	\$0.00	\$23,825.63
SemCrude	MCM ENERGY INC.	\$1,153.00	\$0.00	\$0.00	\$0.00	\$0.00	\$1,153.00
SemCrude	MCNEISH OIL OPERATIONS	\$1,155.00	\$0.00	\$12.08	\$0.00	\$0.00	\$12.08
SemCrude	MEADE ENERGY CORPORATION	\$0.00	\$0.00	\$55,023.99	\$0.00	\$0.00	\$55,023.99
SemCrude	MEADOWBROOK OIL CORP OF OKLAHOMA	\$45.446.89	\$0.00	\$0.00	\$0.00	\$0.00	\$45,446.89
SemCrude	MEM PARTNERSHIP LP	\$45,440.89	\$0.00	\$107.01	\$0.00	\$0.00	\$107.01
SemCrude	MENDELL PETROLEUM CORP	\$0.00	\$0.00	\$40,954.00	\$11,485.00	\$0.00	\$52,439.00
SemCrude	MER RESOURCES INC.	\$0.00	\$0.00	\$213.50	\$0.00	\$0.00	\$213.50
SemCrude	MERCER PETROLEUM MANAGEMENT	\$0.00	\$0.00	\$240,757.54	\$0.00	\$0.00	\$240,757.54
SemCrude	MESA EXPLORATION COMPANY INC	\$0.00	\$0.00	\$0.00	\$72.34	\$0.00	\$72.34
SemCrude	MESQUITE LAND & MINERALS CO	\$1,156.83	\$0.00	\$0.00	\$0.00	\$0.00	\$1,156.83
SemCrude	MESSENGER PETROLEUM INC	\$21,873.19	\$0.00	\$0.00	\$0.00	\$0.00	\$21,873.19
SemCrude	MEWBOURNE OIL COMPANY	\$0.00	\$0.00	\$213,163.54	\$0.00	\$0.00	\$213,163.54
SemCrude	MICKEY L PRUITT	\$0.00	\$0.00	\$66,247.97	\$0.00	\$0.00	\$66,247.97
SemCrude	MICKEY PRUITT	\$621,907.76	\$0.00	\$0.00	\$0.00	\$0.00	\$621,907.76
SemCrude	MID CONTINENT ENERGY CORP	\$376.68	\$0.00	\$0.00	\$0.00	\$0.00	\$376.68
SemCrude	MID CONTINENT ENERGY OP CO	\$59,336.97	\$0.00	\$0.00	\$0.00	\$0.00	\$59,336.97
SemCrude	MIDCO EXPLORATION	\$0.00	\$130,765.52	\$0.00	\$0.00	\$0.00	\$130,765.52
SemCrude	MID-CONTINENT ENERGY OPERATING CO	\$0.00	\$0.00	\$243,813.46	\$0.00	\$0.00	\$243,813.46
SemCrude	MIDNIGHT HOUR LLC	\$54,668.46	\$0.00	\$0.00	\$0.00	\$0.00	\$54,668.46
SemCrude	MIDWEST CASING PULLING	\$0.00	\$0.00	\$106.71	\$0.00	\$0.00	\$106.71
SemCrude	MIDWEST ENTERPRISES INC	\$0.00	\$0.00	\$20,104.20	\$0.00	\$0.00	\$20,104.20
SemCrude	MIDWESTERN EXPLORATION CO	\$364.43	\$0.00	\$0.00	\$0.00	\$0.00	\$364.43
SemCrude	MIKE KELSO OIL COMPANY	\$58.65	\$0.00	\$0.00	\$0.00	\$0.00	\$58.65
SemCrude	MILLENIUM OIL & GAS INC	\$0.00	\$0.00	\$2,467.55	\$0.00	\$0.00	\$2,467.55
SemCrude	MILLER MCGEE OPERATING INC	\$0.00	\$0.00	\$437.33	\$0.00	\$0.00	\$437.33
SemCrude	MINIHAN OIL & GAS CO	\$0.00	\$0.00	\$0.00	\$51,933.79	\$0.00	\$51,933.79
SemCrude	MIOCENE OIL & GAS, LTD.	\$0.00	\$0.00	\$0.00	\$3,943.90	\$0.00	\$3,943.90
SemCrude	MMG OIL COMPANY	\$0.00	\$0.00	\$0.00	\$18,040.84	\$0.00	\$18,040.84
SemCrude	MOLE HOLE PRODUCTION	\$186,506.60	\$0.00	\$0.00	\$0.00	\$0.00	\$186,506.60
SemCrude	MOLITOR OIL	\$0.00	\$0.00	\$10,134.95	\$0.00	\$0.00	\$10,134.95
SemCrude	MONA BONESS	\$0.00	\$0.00	\$78,239.92	\$0.00	\$0.00	\$78,239.92
SemCrude	MONROE NATURAL GAS INC	\$0.00	\$0.00	\$72,320.37	\$0.00	\$0.00	\$72,320.37
SemCrude	MONTGOMERY EXPLORATION COMPANY	\$0.00	\$0.00	\$25.69	\$0.00	\$0.00	\$25.69
SemCrude	MOORE & MOORE ENERGY INC MOORE PETROLEUM INVESTMENTS CORP	\$0.00	\$0.00	\$26,501.88	\$0,00	\$0.00	\$26,501.88
SemCrude		\$0.00	\$0.00	\$0.00	\$152.22	\$0.00	\$152.22
SemCrude	MORGAN OIL	. \$0.00	\$0.00	\$87,071.72	\$0.00	\$0.00	\$87,071.72
SemCrude	MORRIS CHILES MORRIS E STEWART OIL CO	\$0.00	\$0.00	\$3,103.23	\$0.00	\$0.00	\$3,103.23
SemCrude	MORRIS E STEWART OIL CO MORTON PRODUCTION INC	\$0.00	\$0.00	\$325,033.36	\$0.00	\$0.00	\$325,033.36
SemCrude	MOKTOM EKODOCHOM INC						

	N. V. Dierrei	nganethanga angelengengan (- a - a - a - a - a - a - a - a - a -	K	of Durchage Amounts	in Active States		
		Kansas		Oklahoma	Texas	Wyoming	Total
Debtor	Operator	So.oo	\$0.00	\$0.00	\$64,039.80	\$0.00	\$64,039.80
SemCrude	MSN OPERATING	\$0.00	\$0.00	\$0.00	\$227,370.38	\$0.00	\$227,370.38
SemCrude	M-TEX RESOURCES INC	\$0.00	\$0.00	\$102.37	\$0.00	\$0.00	\$102.37
SemCrude	MURAS ENERGY INC	• • • • • • • • • • • • • • • • • • • •	\$0.00	\$25.12	\$0.00	\$0.00	\$25.12
SemCrude	MURCHISON OIL & GAS INC	\$0,00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.01
SemCrude	MURFIN DRILLING COMPANY INC	\$0.01	\$0.00	\$0.00	\$0.00	\$0.00	\$19,608.60
SemCrude	MURFIN DRILLING COMPANY INC.	\$19,608.60	\$0.00	\$370,007.44	\$0.00	\$0.00	\$370,007.44
SemCrude	MUSTANG FUEL CORPORATION	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$72,033.89
SemCrude	N F PRODUCERS	\$72,033.89	\$0.00	\$0.00	\$0.00	\$0.00	\$316,193.90
SemCrude	NACOGDOCHES OIL & GAS LLC	\$316,193.90	\$0.00	\$277,626.84	\$0.00	\$0.00	\$317,581.72
SemCrude	NADEL & GUSSMAN ENERGY LLC	\$39,954.88		\$277,020.84	\$15,327.36	\$0.00	\$318,095.87
SemCrude	NADEL AND GUSSMAN PERMIAN LLC	\$0.00	\$302,768.51 \$0.00	\$617.26	\$0.00	\$0.00	\$617.26
SemCrude	NANCY S ESLICK	\$0.00	-	\$0.00	\$0.00	\$0.00	\$17,461.22
SemCrude	NASH OIL & GAS INC	\$17,461.22	\$0.00	\$20,485.02	\$0.00	\$0.00	\$20,485.02
SemCrude	NATIONAL PIONEER OIL COMPANY LLC	\$0.00	\$0.00	\$438.65	\$0.00	\$0.00	\$438.65
SemCrude	NATIVE GAS COMPANY LLC	\$0.00	\$0.00	\$627.27	\$0.00	\$0.00	\$627.27
SemCrude	NATURAL GAS COMPRESSION CORP	\$0.00	\$0.00	\$853.86	\$0.00	\$0.00	\$853.86
SemCrude	NELSON EXPLORATION CORP	00.00	\$0.00	\$0.00	\$183,448.42	\$0.00	\$183,448.42
SemCrude	NETA R GOLDSMITH	\$0.00	\$0.00		\$0.00	\$0.00	\$8,037,337.92
SemCrude	NEW DOMINION LLC	\$0.00	\$0.00	\$8,037,337.92 \$0.00	\$2,602.40	\$0.00	\$2,602.40
SemCrude	NEW MEXICO WESTBROOK OIL CORP	\$0.00	\$0.00	\$0.00	\$104,777.91	\$0.00	\$104,777.91
SemCrude	NEWCASTLE RESOURCES, LLC	\$0.00	\$0.00	\$311,522.85	\$0.00	\$0.00	\$311,522.85
SemCrude	NEWELL OIL & GAS	\$0.00	\$0.00	\$311,322.83	\$163,042.37	\$0.00	\$163,042.37
SemCrade	NFR ENERGY LLC	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$40,354.56
SemCrude	NICHOL'S WELL PULLING	\$40,354.56	\$0.00	\$0.00 \$0.00	\$22,879.61	\$0.00	\$22,879.61
SemCrude	NICK OIL	\$0.00	\$0.00	\$264.85	\$0.00	\$0.00	\$264.85
SemCrude	NIELSON & ASSOCIATES INC	\$0.00	\$0.00	\$204.63 \$157,031.27	\$0.00	\$0.00	\$157,031.27
SemCrude	NO WHERE OIL COMPANY	\$0.00	\$0.00	\$137,031.27	\$690,491.43	\$0.00	\$690,491.43
SemCrude	NOBLE	\$0.00	00.02	\$9,159.53	\$2,036,650.71	\$0.00	\$2,069,682.44
SemCrude	NOBLE ENERGY INC	\$23,872.20	\$0.00	\$9,139.33 \$78.71	\$0.00	\$0.00	\$78.71
SemCrude	NOBLE RESOURCES INC.	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$14.70
SemCrude	NORSTAR PETROLEUM INC.	\$14.70	\$0.00	\$141,895.51	\$0.00	\$0.00	\$141,895.51
SemCrude	NORTHBROOK OIL & GAS CO	\$0.00	\$0.00	\$1,119.69	\$0.00	\$0.00	\$1,119.69
SemCrude	NORTHPORT PRODUCTION COMPANY	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$19,412.39
SemCrude	NORTHSTAR INVESTMENTS, INC.	\$19,412.39	\$0.00	\$59.24	\$0.00	\$0.00	\$59.24
SemCrude	NORTHWEST OIL COMPANY	\$0.00	\$0.00	\$0.00	\$105,605.45	\$0.00	\$105,605.45
SemCrude	NOVO EXPLORATION CORPORATION	\$0.00	\$0.00	\$178,515.56	\$0.00	\$0.00	\$297,488.48
SemCrude	NOVY OIL & GAS INC	\$118,972.92	\$0.00	\$59,197.87	\$0.00	\$0.00	\$59,197.87
SemCrude	NUB'S WELL SERVICING, INC	\$0.00	\$0.00	\$39,197.87 \$81,179.52	\$0.00	\$0.00	\$81,179.52
SemCrude	NU-HOPE OIL COMPANY	\$0.00	\$0.00	\$0.00	\$11,561.56	\$0.00	\$11,561.56
SemCrude	NUMBER ONE OIL & GAS	\$0.00	\$0.00	\$25,413.02	\$0.00	\$0.00	\$25,413.02
SemCrude	NWM OPERATING INC	\$0.00	\$0.00	\$25,413.02	\$0.00	\$0.00	\$12,700.97
SemCrude	NWM PROPERTIES INC	\$0.00	\$0.00	•	\$0.00	\$0.00	\$174,938.53
SemCrude	NYTEX ENERGY LLC	\$0.00	\$0.00	\$174,938.53 \$0.00	\$18,591.83	\$0.00	\$18,591.83
SemCrude	O HILL ENTERPRISES, INC.	\$0.00	\$0.00	\$0.00	\$1,966.85	\$0.00	\$1,966.85
SemCrude	O M BELL	\$0.00	\$0.00		\$0.00	\$0.00	\$45,817.08
SemCrude	OAK HILLS ENERGY INC	\$0.00	\$0.00	\$45,817.08	\$0.00	\$0.00	\$17,473.59
SemCrude	OCHILTREE OIL CO INC	\$0.00	\$0.00	\$17,473.59	\$0.00	\$0.00	\$58,812.67
SemCrude	OCONEE OIL & GAS LLC	\$0.00	\$0.00	\$58,812.67 \$307.60	\$0.00	\$0.00	\$307.60
SemCrude	OCTAGON RESOURCES INC	00.00	\$0.00	\$307.0U	\$0.00	*****	

			anowa gura a William Residentifica		. neverval cue la company	IN NOVEMBER OF S	
	Operator			Oklahoma	Texas	Wyoming	Total
Debtor			New Mexico \$0.00	\$0.00	\$286,31	\$0.00	\$286.31
SemCrude	OHLHAUSEN OIL COMPANY	\$0.00 \$0.00	\$0.00	\$0.75	\$0.00	\$0.00	\$0.75
SemCrude	OIL CENTER OPERATING INC	•	\$0.00	\$0.00	\$0.00	\$0.00	\$356,474.31
SemCrude	OIL COMPANY OF AMERICA, INC.	\$356,474.31	\$0.00	\$10,991.47	\$0.00	\$0.00	\$51,971.61
SemCrude	OIL PRODUCERS OF KANSAS INC	\$40,980.14	\$0.00	\$0.00	\$0.00	\$0.00	\$828.98
SemCrude	OIL PROPERTY MANAGEMENT INC.	\$828.98 \$0.00	\$0.00	\$7,230.62	\$0.00	\$0.00	\$7,230.62
SemCrude	OIL VENTURES LTD.	\$0.00	\$0.00	\$0.38	\$0.00	\$0.00	\$0.38
SemCrude	OKIE CRUDE COMPANY	\$0.00 \$0.00	\$0.00	\$152,487.44	\$0.00	\$0.00	\$152,487.44
SemCrude	OKLAHOMA ENERGY LLC	\$0.00 \$0.00	\$0.00	\$42,303.86	\$0.00	\$0.00	\$42,303.86
SemCrude	OKLAHOMA LIQUID RESOURCES INC	\$0.00	\$0.00	\$978,575.40	\$0.00	\$0.00	\$978,575.40
SemCrude	OKLAHOMA OIL & GAS MANAGEMENT INC	\$0.00	\$0.00	\$29,838.43	\$0.00	\$0.00	\$29,838.43
SemCrude	OKLAHOMA SUB-SURFACE PUMP & SUPPLY CO	\$0.00 \$0.00	\$0.00	\$273.88	\$0.00	\$0.00	\$273.88
SemCrude	OKT PETROLEUM COMPANY INC	\$0.00	\$0.00	\$2,111.14	\$0.00	\$0.00	\$2,111.14
SemCrude	ON LINE OIL INC	\$0.00	\$0.00	\$21,343.45	\$0.00	\$0.00	\$21,343.45
SemCrude	O'NEAL DRILLING COMPANY	\$0.00	\$0.00	\$596.19	\$0.00	\$0.00	\$596.19
SemCrude	OPEN DOOR HOLDINGS INC	\$0.00	\$0.00	\$5,373.08	\$0.00	\$0.00	\$5,373.08
SemCrude	OPEN DOOR PRODUCTIONS INC	\$249,767.44	\$0.00	\$0.00	\$0.00	\$0.00	\$249,767.44
SemCrude	OSBORN HEIRS COMPANY	\$0.00	\$0.00	\$15.56	\$0.00	\$0.00	\$15.56
SemCrude	OUACHITA EXPLORATION INC	\$41,799.56	\$0.00	\$0.00	\$0.00	\$0.00	\$41,799.56
SemCrude	OVERLAND ENERGY INC	\$0.00	\$0.00	\$40,384.02	\$0.00	\$0.00	\$40,384.02
SemCrude	OXCOLLC	\$0.00	\$0.00	\$181.61	\$0.00	\$0.00	\$181.61
SemCrude	OXY USA INC	\$0.00	\$0.00	\$30,882.68	\$0.00	\$0.00	\$30,882.68
SemCrude	P&H CONSTRUCTION INC	\$0.00	\$0.00	\$79.92	\$0.00	\$0.00	\$79.92
SemCrude	PACE EXPLORATION LLC	\$109,939,17	\$0.00	\$0.00	\$0.00	\$0.00	\$109,939.17
SemCrude	PADA PETROLEUM INC.	\$0,00	\$0.00	\$439.23	\$0.00	\$0.00	\$439.23
SemCrude	PALM & COOK OIL & GAS COMPANY	\$1,844.14	\$0.00	\$0.00	\$0.00	\$0.00	\$1,844.14
SemCrude	PALOMINO PETROLEUM INC	\$0.00	\$0.00	\$75.69	\$0.00	\$0.00	\$75.69
SemCrude	PALUCA PETROLEUM INC.	\$0.00	\$0.00	\$10,730.46	\$0.00	\$0.00	\$10,730.46
SemCrude	PAMAVETTA OIL COMPANY	\$0.00	\$0.00	\$51,568.62	\$0.00	\$0.00	\$51,568.62
SemCrude	PAPAL ENTERPRISES INC	\$0.00	\$0.00	\$0.00	\$12,714.27	\$0.00	\$12,714.27
SemCrude	PAR 3	\$0.00	\$0.00	\$115,656.85	\$0.00	\$0.00	\$115,656.85
SemCrude	PARAGON INDUSTRIES INC	\$0.00	\$0.00	\$103.02	\$0.00	\$0.00	\$103.02
SemCrude	PARDEE PRODUCTION CO	\$0.00	\$0.00	\$38,222.29	\$0.00	\$0.00	\$38,222.29
SemCrude	PARKER BROTHERS OIL COMPANY	\$2,340.80	\$0.00	\$0.00	\$0.00	\$0.00	\$2,340.80
SemCrude	PARRISH OIL COMPANY LLC PARSONS ENGINEERING CORP	\$0.00	\$0.00	\$5,248.59	\$0.00	\$0.00	\$5,248.59
SemCrude		\$0.00	\$0.00	\$97,726.75	\$0.00	\$0.00	\$97,726.75
SemCrude	PATRIOT PETROLEUM CO LLC	\$0.00	\$0.00	\$124,219.88	\$0.00	\$0.00	\$124,219.88
SemCrude	PATRON ENERGY LLC PATTERSON OIL COMPANY INC	\$0.00	\$0.00	\$280,647.24	\$0.00	\$0.00	\$280,647.24
SemCrude	PAUL BOWMAN OIL COMPANY	\$1,060.32	\$0.00	\$0.00	\$0.00	\$0.00	\$1,060.32
SemCrude	= - 1 -	\$0.00	\$0.00	\$29,247.66	\$0.00	\$0.00	\$29,247.66
SemCrude	PAUL D ALBRIGHT PAUL E KLOBERDANZ	\$21,315.03	\$0.00	\$0.00	\$0.00	\$0.00	\$21,315.03
SemCrude		\$0.00	\$0.00	\$81,954.94	\$0.00	\$0.00	\$81,954.94
SemCrude	PAUL VAN ORSDOL PBT ENTERPRISES INC.	\$0.00	\$0.00	\$23,813.35	\$0.00	\$0.00	\$23,813.35
SemCrude	PEAK ENERGY LLC	\$72,643.73	\$0.00	\$0.00	\$0.00	\$0.00	\$72,643.73
SemCrude	PEAK ENERGY LLC PEDESTAL OIL COMPANY INC	\$0.00	\$0.00	\$560,691.04	\$0.00	\$0.00	\$560,691.04
SemCrude	PEEWE OIL CO.	\$66,929.53	\$0.00	\$0.00	\$0.00	\$0.00	\$66,929.53
SemCrude		\$0.00	\$0.00	\$17,289.93	\$0.00	\$0.00	\$17,289.93
SemCrude	PEMEX INC PENNER ENERGY, INC	\$0.00	\$0.00	\$82.65	\$0.00	\$0.00	\$82.65
SemCrude	PEPIN OPERATING COMPANY INC.	\$0.00	\$0.00	\$25.33	\$0.00	\$0.00	\$25.33
SemCrude	refin oferating contain the						

			Patrico Saurius II (Seus e	- 1000 AGUST - CHINADAS	in Active States		(22 <u>08</u> 100000000000000000000000000000000000
		Kansas		Oklahoma	Texas	vomine	Total
Debter	Operator Operator	Kansas S0 00	\$0.00	\$105.83	\$0.00	\$0.00	\$105,83
SemCrude	PERKINS BROS	\$0.00	\$0.00	\$693,594.74	\$0.00	\$0.00	\$693,594.74
SemCrude	PERKINS ENERGY COMPANY	\$0.00	\$0.00	\$2,313.86	\$0.00	\$0.00	\$2,313.86
SemCrude	PERKINS OIL COMPANY	\$0.00	\$0.00	\$8,603.87	\$0.00	\$0.00	\$8,603.87
SemCrude	PERRY ROULSTON	\$0.00	\$0.00	\$2,305,429.81	\$0.00	\$0.00	\$2,305,429.81
SemCrude	PETCO PETROLEUM CORP	\$0.00	\$0.00	\$67,484.17	\$0.00	\$0.00	\$67,484.17
SemCrude	PETRO MAC CORPORATION	\$0.00	\$0.00	\$215,507.84	\$0.00	\$0.00	\$215,507.84
SemCrude	PETRO WARRIOR LLC	\$0.00	\$0.00	\$3,619.58	\$0.00	\$0.00	\$3,619.58
SemCrude	PETRO-DAVIS	\$0.00	\$0.00	\$6,576.90	\$0.00	\$0.00	\$6,576.90
SemCrude	PETRO-DAVIS INC	\$0.00	\$0.00	\$30,810.74	\$0.00	\$0.00	\$30,810.74
SemCrude	PETRODUDES	\$0.00	\$50.79	\$0.00	\$0.00	\$0.00	\$50.79
SemCrude	PETROLEUM DEVELOPMENT CORP	\$12.66	\$0.00	\$0.00	\$0.00	\$0.00	\$12.66
SemCrude	PETROLEUM MANAGEMENT INC	\$20,979,75	\$0.00	\$0.00	\$0.00	\$0.00	\$20,979.75
SemCrude	PETROLEUM PROPERTY SERVICES	\$0.00	\$0.00	\$1,109.26	\$0.00	\$0.08	\$1,109.26
SemCrude	PETROQUEST OIL & GAS LLC	\$14,619.96	\$0.00	\$0.00	\$0.00	\$0.00	\$14,619.96
SemCrude	PETROSANTANDER USA INC	\$50,226.34	\$0.00	\$0.00	\$0.00	\$0.00	\$50,226.34
SemCrude	PHILLIP GROVES	\$245,701.27	\$0.00	\$0.00	\$0.00	\$0.00	\$245,701.27
SemCrude	PHILLIPS EXPLORATION COMPANY LC	\$0.00	\$0.00	\$1,813,772.90	\$0.00	\$0.00	\$1,813,772.90
SemCrude	PHOENIX OIL & GAS INC	\$0.00	\$0,00	\$1,784,419,44	\$0.00	\$0.00	\$1,784,419.44
SemCrude	PHOENIX PETROCORP INC	\$0.00	\$0.00	\$0.00	\$21,109.29	\$0.00	\$21,109.29
SemCrude	PICKETT OIL COMPANY	\$873,095.45	\$0.00	\$0.00	\$0.00	\$0.00	\$873,095.45
SemCrude	PICKRELL DRILLING CO	\$0.00	\$164,010.65	\$0.00	\$0.00	\$0.00	\$164,010.65
SemCrude	PIERCE PRODUCTION CO	\$0.00	\$0.00	\$9,276.76	\$0.00	\$0.00	\$9,276.76
SemCrude	PIKE OIL & GAS INC	\$0.00	\$0.00	\$41,890.90	\$0.00	\$0.00	\$41,890.90
SemCrude	PINNACLE CORPORATION PINNACLE OPERATING COMPANY LLC	\$0.00	\$0.00	\$638.97	\$0.00	\$0.00	\$638.97
SemCrude		\$2,785.40	\$0.00	\$0.00	\$0.00	\$0.00	\$2,785.40
SemCrude	PINTAIL PETROLEUM LTD	\$0.00	\$0.00	\$0.00	\$251.39	\$0.00	\$251.39
SemCrude	PIONEER EXPLORATION LTD PIONEER OIL COMPANY INC	\$95,981.65	\$0.00	\$0.00	\$0.00	\$0.00	\$95,981.65
SemCrude		\$0.00	\$0.00	\$3,436.83	\$0.00	\$0.00	\$3,436.83
SemCrude	PITCOCK INC PLATTE VALLEY OIL COMPANY INC	\$164,647.86	\$0.00	\$0.00	\$0.00	\$0.00	\$164,647.86
SemCrude	PLOTNER OIL COMPANY	\$0.00	\$0.00	\$44,798.11	\$0.00	\$0.00	\$44,798.11
SemCrude	PLUMMER ENERGY INC	\$0.00	\$0.00	\$1,858.60	\$0.00	\$0.00	\$1,858.60
SemCrude	PNG OPERATING COMPANY	\$0.00	\$0.00	\$4,323.46	\$0.00	\$0.00	\$4,323.46
SemCrude	PONTOTOC OPERATIONS INC	\$0.00	\$0.00	\$9.00	\$0.00	\$0.00	\$9.00
SemCrude	PONTOTOC PRODUCTION CO INC	\$0.00	\$0.00	\$23,650.05	\$0.00	\$0.00	\$23,650.05
SemCrude	POPP OPERATING INC	\$1,135,503.52	\$0.00	\$0.00	\$0.00	\$0.00	\$1,135,503.52
SemCrude	POTTER TRUCKING INC	\$29,305.02	\$0.00	\$0.00	\$0.00	\$0.00	\$29,305.02
SemCrude	PRAIRIE EXPLORATION	\$0.00	\$0.00	\$8,644.21	\$0,00	\$0.00	\$8,644.21
SemCrude	PRATT WELL SERVICE, INC.	\$42,364.44	\$0.00	\$0.00	\$0.00	\$0.00	\$42,364.44
SemCrude	PREMIER ENERGY LLC	\$0.00	00.00	\$67,066.83	\$0.00	\$0.00	\$67,066.83
SemCrude	PREMIER PETROLEUM INC.	\$23,727.39	\$0.00	\$0.00	\$0.00	\$0.00	\$23,727.39
SemCrude	PRICKLY PAIR PRODUCTION	\$0.00	\$0.08	\$0.00	\$2,350.83	\$0.00	\$2,350.83
SemCrude	PRODUCERS CRUDE	\$0.00	\$0.00	\$908,908.00	\$0.00	\$0.00	\$908,908.00
SemCrude	PRODUCEAS CRUDE PROSPEROUS OIL AND GAS INC	\$0.00	\$0.00	\$288.49	\$0.00	\$0.00	\$288.49 \$570 255 45
SemCrude	PROTEGE ENERGY LLC	\$42,013.81	\$0.00	\$168,259.74	\$368,081.90	\$0.00	\$578,355.45 \$3,572.05
SemCrude SemCrude	PROVEN RESERVES MANAGEMENT INC	\$0.00	\$0.00	\$3,572.05	\$0.00	00.02	\$3,572.05 \$33.71
	PRUCO	\$0.00	\$0.00	\$33.71	\$0.00	\$0.00	\$33.71 \$241,673.16
SemCrude SemCrude	PSA INC	\$0.00	00.02	\$241,673.16	\$0.00	\$0.00	\$396,950.40
SemCrude SemCrude	PULLERS OIL CO LLC	\$0.00	\$0.00	\$396,950.40	\$0.00	\$0.00	#370,7JV.40
Semerage	1 Opposite out on the						

				TO THE PROPERTY OF THE PROPERT	its in Active States	oganganakanasis): 17 filip	
		Interest Contractions		Oklahoma	Texas W	vomino	Total
Debtor	Operator		New Mexico \$0.00	\$19.817.82	\$0.00	\$0.00	\$19,817.82
SemCrude	PUMPING SERVICE INC	\$0.00	\$0.00 \$0.00	\$550,792.75	\$0.00	\$0.00	\$550,792.75
SemCrude	QT OIL LTD CO	\$0.00	\$0.00	\$6,146.55	\$0.00	\$0.00	\$6,146.55
SemCrude	QUAD J OIL & GAS LLC	\$0.00	\$0.00	\$0,140.00	\$0.00	\$0.00	\$224.11
SemCrude	QUALITY WELL SERVICE	\$224.11	\$0.00	\$0.00	\$12,070.62	\$0.00	\$12,070.62
SemCrude	QUEST OIL CORPORATION	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$478,696.64
SemCrude	QUINQUE OPERATING COMPANY	\$478,696.64	\$0.00	\$18,074.71	\$0.00	\$0.00	\$18,074.71
SemCrude	R & B TRUCKING	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$2,477.91
SemCrude	R & D OIL, LLC	\$2,477.91	\$0.00 \$0.00	\$0.00	\$0.00	\$0.00	\$72,670.58
SemCrude	R D FINNESY OIL COMPANY	\$72,670.58	\$0.00	\$0.00	\$0.00	\$0.00	\$1,005.89
SemCrude	R E FRENCH	\$1,005.89	\$0.00	\$99.38	\$0.00	\$0.00	\$99.38
SemCrude	R G BURK OIL PRODUCTION	\$0.00	\$0.00	\$80,426.61	\$0.00	\$0.00	\$80,426.61
SemCrude	R J SPERRY CO	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$186.71
SemCrude	R J WIXSON CO	\$186.71	\$0.00	\$0.00	\$0.00	\$0.00	\$653,791.44
SemCnide	R L INVESTMENT LLC	\$653,791.44	\$0.00	\$1,113.20	\$0.00	\$0.00	\$1,113.20
SemCrude	R L PERKINS	\$0.00 \$0.00	\$0.00	\$0.00	\$1,349.26	\$0.00	\$1,349.26
SemCrude	R M HILL OPERATING INC	\$0.00 \$0.00	\$0.00	\$1.04	\$0.00	\$0.00	\$1.04
SemCrude	R. COURTNEY WARREN INC.	\$0.00	\$0.00	\$494.34	\$0.00	\$0.00	\$494.34
SemCrude	RACHAEL EXPLORATION		\$0.00	\$0.00	\$0.00	\$0.00	\$39,338.42
SemCrude	RADELL W KOEHN	\$39,338.42	\$0.00	\$2,454.33	\$0.00	\$0.00	\$2,454.33
SemCrude	RAF OIL CO	\$0.00 \$0.00	\$0.00	\$300.87	\$0.00	\$0.00	\$300.87
SemCrude	RAFTER H OPRATING LLC	\$0.00 \$0.00	\$0.00	\$202,501.73	\$0.00	\$0,00	\$202,501.73
SemCrude	RAINBOW OILWELL SERVICE INC	\$0.00 \$0.00	\$0.00	\$34,509.32	\$0.00	\$0.00	\$34,509.32
SemCrude	RAINMAKER OIL AND GAS CORPORATION	\$0.00	\$0.00	\$267,419.98	\$0.00	\$0.00	\$267,419.98
SemCrude	RA-JAC INC	\$0.00	\$0.00	\$20,866.31	\$0.00	\$0.00	\$20,866.31
SemCrude	RALPH STONE	\$2,379,370.45	\$0.00	\$710.81	\$0.00	\$0.00	\$2,380,081.26
SemCrude	RAMA OPERATING CO INC	\$2,379,370.43	\$0.00	\$262,140.69	\$0.00	\$0.00	\$262,140.69
SemCrude	RANDALL E JAMES	\$0.00	\$0.00	\$39,382.25	\$0.00	\$0.00	\$39,382.25
SemCrude	RANDALL H BAXTER	\$300,816.06	\$0.00	\$0.00	\$0.00	\$0.00	\$300,816.06
SemCrude	RANDON PRODUCTION COMPANY LLC	\$0.00	\$0.00	\$612,616.68	\$0.00	\$0.00	\$612,616.68
SemCrude	RANDY AND IAN WALKER OIL COMPANY LLC	\$0.00	\$0.00	\$1,260,610.04	\$0.00	\$0.00	\$1,260,610.04
SemCrude	RANGE PROD	\$0.00	\$0.00	\$526,604.17	\$0.00	\$0.00	\$526,604.17
SemCrude	RANKEN ENERGY CORPORATION	\$0.00	\$0.00	\$22,306.96	\$0.00	\$0.00	\$22,306.96
SemCrude	RAW PRODUCTION LLC	\$543.57	\$0.00	\$0.00	\$0.00	\$0.00	\$543.57
SemCrude	RAY C FLEMING JR	\$523,269.07	\$0.00	\$1,454,985.24	\$19,795.89	\$0.00	\$1,998,050.20
SemCrude	RAYDON EXPLORATION INC	\$0.00	\$0.00	\$0.00	\$36,129.65	\$0.00	\$36,129.65
SemCrude	RAYMOND C TAYLOR, JR	\$0.00	\$0.00	\$130,395.13	\$0.00	\$0.00	\$130,395.13
SemCrude	RAYMOND H JONES FAMILY TRUST	\$0.00	\$0.00	\$159.70	\$0.00	\$0.00	\$159.70
SemCrude	RAYMOND L JAMES	\$0.00	\$0.00	\$9,457.74	\$0.00	\$0.00	\$9,457.74
SemCrude	RAY'S OIL & GAS LLC	\$0.00	\$0.00	\$59,635.35	\$0.00	\$0.00	\$59,635.35
SemCrude	RAY'S OIL COMPANY	\$0.00	\$0.00	\$69,209.88	\$0.00	\$0.00	\$69,209.88
SemCrude	RC HOLDINGS LLC	\$0.00	\$0.00	\$389.04	\$0.00	\$0.00	\$389.04
SemCrude	RDH ENTERPRISES INC	\$0.00	\$0.00	\$242,077.37	\$0.00	\$0.00	\$242,077.37
SemCrude	RDT PROPERTIES INC	\$0.00	\$0.00	\$0.00	\$207.56	\$0.00	\$207.56
SemCrude	REATA RESOURCES	\$0.00	\$0.00	\$125.15	\$0.00	\$0.00	\$125.15
SemCrude	REBACH PETROLEUM LLC	\$0.00	\$0.00	\$134.16	\$0.00	00.02	\$134.16
SemCrude	RED FORK OIL PRODUCTION INC	\$0.00	\$0.00	\$157,929.97	\$0.00	\$0.00	\$157,929.97
SemCrude	RED FORK PRODUCTION LLC	\$161,257.95	\$0.00	\$0.00	\$0.00	\$0.00	\$161,257.95
SemCrude	RED OAK ENERGY INC	\$0.00	\$0.00	\$1,780.86	\$0.00	\$0.00	\$1,780.86
SemCrude	RED ROCK RESOURCES LLC	40.00	*	•			

				First Purchase Amou	nts in Active States		
		Kansas	New Mexico	Oklahoma			Total
Debtor	Operator (Control of the Control of	\$0.00	\$0.00	\$319,368.55	\$0.00	\$0.00	\$319,368.55
SemCrude	REDDY OIL & GAS CORP	\$0.00	\$0.00	\$19,190.31	\$0.00	\$0.00	\$19,190.31
SemCrude	REDDY OIL & GAS CORPORATION	\$0.00	\$0.00	\$389.44	\$0.00	\$0.00	\$389.44
SemCrude	REDHEN OIL COMPANY INC.	\$64,480.44	\$0.00	\$0.00	\$0.00	\$0.00	\$64,480.44
SemCrude	REDLAND RESOURCES INC.	\$0.00	\$0.00	\$137,411.20	\$0.00	\$0.00	\$137,411.20
SemCrude	REED POWER TONGS INC	\$98,367.30	\$0.00	\$0.00	\$0.00	\$0.00	\$98,367.30
SemCrude	REGINALD D FISHER	\$0.00	\$0.00	\$60,568.84	\$0.00	\$0.00	\$60,568.84
SemCrude	REID WILLIAMS OIL CO	\$0.00	\$0.00	\$65,090.89	\$0.00	\$0.00	\$65,090.89
SemCrude	REID WILLIAMS OPERATING INC	\$117,323.64	\$0.00	\$0.00	\$0.00	\$0.00	\$117,323.64
SemCrude	REIF OIL & GAS LLC	\$91.93	\$0.00	\$0.00	\$0.00	\$0.00	\$91.93
SemCrude	REILLY OIL COMPANY INC	\$215,914.56	\$0.00	\$0.00	\$0.00	\$0.00	\$215,914.56
SemCrude	REIN RESOURCES	\$0.00	\$0.00	\$202.43	\$0.00	\$0.00	\$202.43
SemCrude	RENO RESOURCES	\$0.00	\$0.00	\$39,564.08	\$0.00	\$0.00	\$39,564.08
SemCrude	REPUBLIC RESOURCES CORP	\$0.00	\$0.00	\$21,102.09	\$0.00	\$0.00	\$21,102.09
SemCrude	RESEARCH OIL COMPANY INC	\$0.00	\$0.00	\$126,760.78	\$0.00	\$0.00	\$126,760.78
SemCrude	RESERVE MANAGEMENT INC	\$828,039.49	\$0.00	\$0.00	00.02	\$0.00	\$828,039.49
SemCrude	RESOURCE OPERATIONS INC	\$0.00	\$0.00	\$74,979.55	\$0.00	\$0.00	\$74,979 <i>.5</i> 5
SemCrude	RESOURCES OPERATING CO INC	\$0.00	\$0.00	\$64.01	\$0.00	\$0.00	\$64.01
SemCrude	REVELATION RESOURCES	\$0.00	\$0.00	\$11.14	\$0.00	\$0.00	\$11.14
SemCrude	RHOADES OIL COMPANY	\$0.00	\$0.00	\$46,004.21	\$0.00	\$0.00	\$46,004.21
SemCrude	RICHARD L SMITH	\$0.00	\$0.00	\$0.00	\$9,537.50	\$0.00	\$9,537.50
SemCrude	RICHARDSON RESOURCES INC	\$0.00	\$0.00	\$18.59	\$0.00	\$0.00	\$18.59
SemCrude	RICK MINER OIL COMPANY	\$0.00	\$0.00	\$1,057.87	\$0.00	\$0.00	\$1,057.87
SemCrude	RICKS EXPLORATION INC	\$0.00	\$0.00	\$327.54	\$0.00	\$0.00	\$327.54
SemCrude	RIO BLANCO OPERATING LLC	\$516,076.04	\$0.00	\$0.00	\$0.00	\$0.00	\$516,076.04
SemCrude	RITCHIE EXPLORATION INC	\$0.00	\$0.00	\$50,048.15	\$0.00	\$0.00	\$50,048.15
SemCrude	RKK PRODUCTION COMPANY	\$0.00	\$0.00	\$120,024.26	\$0.00	\$0.00	\$120,024.26
SemCrude	RKR EXPLORATION INC	\$0.00	\$0.00	\$911.64	\$0.00	\$0.00	\$911.64
SemCrude	ROBCO	\$0.00	\$0.00	\$139,856.45	\$0.00	\$0.00	\$139,856.45
SemCrude	ROBERSON OIL COMPANY INC	\$0.00	\$0.00		\$0.00	\$0.00	\$436.36
SemCrude	ROBERT L. COX	\$0.00	\$0.00		\$0.00	\$0.00	\$40,726.24
SemCrude	ROBERT MONTGOMERY	\$0.00	\$0.00		\$0.00	\$0.00	\$289,127.15
SemCrude	ROBERT W PASSMORE	\$592,420.04	\$0.00		\$0.00	\$0.00	\$592,420.04
SemCrude	ROBINOWITZ OIL COMPANY	\$394,700.01	\$0,00	\$0.00	\$0.00	\$0.00	\$394,700.01
SemCrude	ROBINSON ENERGY & EXPL INC	\$0.00	\$0.00	\$41,886.97	\$0.00	\$0.00	\$41,886.97
SemCrude	ROBINSON OIL COMPANY	\$0.00	\$0.00	\$3,408.51	\$0.00	\$0.00	\$3,408.51
SemCrude	ROBLYN OIL INC ROBSON ENERGY OKLAHOMA LLC	\$0.00	\$0.00	\$10,350.86	\$0.00	\$0.00	\$10,350.86
SemCrude	*****	\$0.00	\$0.00	\$3,025.06	\$0.00	\$0.00	\$3,025.06
SemCrude	ROCK OIL COMPANY	\$0.00	\$0.00	\$0.00	\$2,293.79	\$0.00	\$2,293.79
SemCrude	ROCK OPERATING	\$0.00	\$0.00	\$0.00	\$30,461.71	\$0.00	\$30,461.71
SemCrude	ROCKWOOD RESOURCES INC	\$0.00	\$0.00		\$0.00	\$0.00	\$239.52
SemCrude	ROD BRITT & JEFF PETERS	\$0.00	\$0.00	\$68,778.67	\$0.00	\$0.00	\$68,778.67
SemCrude	RODGER E SPARKMAN	00.00	\$0.00	\$52.35	\$0.00	\$0.00	\$52.35
SemCrude	ROGER JOHN WOLFGRAM	\$0.00	\$0.00	\$0.00	\$135,678.62	\$0.00	\$135,678.62
SemCrude	ROGER KLINGBERG	\$0.00	00.02		\$0.00	\$0.00	\$226.16
SemCrude	ROGER W HALL	\$0.00	\$0.00		\$0.00	\$0.00	\$37,549.42
SemCrude	ROGERS COUNTY COAL SEAM LLC	\$0.00	\$0.00	-	\$0.00	\$0.00	\$102,901.18
SemCrude	RON GRIGGS	\$0.00	\$0.00		\$0.00	\$0.00	\$138.58
SemCrude	RONALD D. BLUBAUGH	\$247,162.43	\$0.00		\$0.00	\$0.00	\$247,162.43
SemCrude	ROSEWOOD RESOURCES INC	<i>\$22.77,402.</i> 10	- ****				

and the second of the second of the				First Purchase Amount	s in Active States		
	Operator:	Kansas	New Mexico	Oklaboma			Total
Debtor SemCrude	ROSS ENGINEERING INC	\$0.00	\$0.00	\$0.00	\$45.10	\$0,00	\$45.10
SemCrude	ROSS HOENER INC.	\$239,136.18	\$0.00	\$0.00	\$0.00	\$0.00	\$239,136.18
SemCrude SemCrude	ROUGEOT OIL & GAS CORP	\$0.00	\$0.00	\$287,728.12	\$0.00	\$0.00	\$287,728.12
SemCrude SemCrude	ROX EXPLORATION INC	\$0.00	\$0.00	\$182,310.52	\$0.00	\$0.00	\$182,310.52
SemCrude	ROXANNA OIL CO	\$0.00	\$0.00	\$151,759.23	\$0.00	\$0.00	\$151,759.23
SemCrude	ROY BAGLEY JR	\$0.00	\$0.00	\$0.00	\$2.08	\$0.00	\$2.08
SemCrude	ROY WEBB	\$0.00	\$0.00	\$60,976.85	\$0.00	\$0.00	\$60,976.85
SemCrude	ROYCE WILLIAMS	\$0.00	\$0.00	\$85.50	\$0.00	\$0.00	\$85.50
SemCrude	RUH OIL COMPANY	\$0.00	\$0.00	\$177.68	\$0.00	\$0.00	\$177.68
SemCrude	RUMLEY TRUST	\$0.00	\$0.00	\$119.33	\$0.00	\$0.00	\$119.33
SemCrude	RUSK ENERGY LTD	\$0.00	\$0.00	\$0.00	\$23,761.09	\$0.00	\$23,761.09
SemCrude	RUSSELL D MELOY	\$0.00	\$0.00	\$113,437.75	\$0.00	\$0.00	\$113,437.75
SemCrude SemCrude	RUTLEDGE - NEWSOM OIL CO	\$0.00	\$0.00	\$0.00	\$151,671.30	\$0.00	\$151,671.30
SemCrude SemCrude	S & S OIL & GAS CORP.	\$0.00	\$0.00	\$2,921.86	\$0.00	\$0.00	\$2,921.86
SemCrude SemCrude	S & S OIL COMPANY INC	\$0.00	\$0.00	\$32,768.15	\$0.00	\$0.00	\$32,768.15
SemCrude SemCrude	S B STREET OPERATING INC	\$0.00	\$0.00	\$0.00	\$4,470.78	\$0.00	\$4,470.78
SemCrude SemCrude	S C SMITH OIL COMPANY	\$0.00	\$0.00	\$9,650.82	\$0.00	\$0.00	\$9,650.82
SemCrude	S I MURPHY OIL & GAS INC	\$0.00	\$0.00	\$119,740.55	\$0.00	\$0.00	\$119,740.55
SemCrude	S/H PETROLEUM CORPORATION	\$0.00	\$0.00	\$1,208.12	\$0.00	\$0.00	\$1,208.12
SemCrade	SABO PRODUCTION CO	\$0.00	\$0.00	\$1.40	\$0.00	\$0.00	\$1.40
SemCrude	SAJAC ENERGY COMPANY L.L.C.	\$0.00	\$0.00	\$66,043.73	\$0.00	\$0.00	\$66,043.73
SemCrude	SALLEE OIL CORPORATION	\$0.00	\$0.00		\$0.00	\$0.00	\$281,433.02
SemCrude	SAMSON RESOURCES CO	\$0.00	\$0.00	\$353,792.96	\$0.00	\$0.00	\$353,792.96
SemCrude	SAND POINT LLC	\$0.00	\$0.00		\$0.00	\$0.00	\$18.69
SemCrude	SAND POINT PRODUCTION INC	\$0.00	\$0.00	\$140,931.48	\$0.00	\$0.00	\$140,931.48
SemCrude	SAND RESOURCES INC	\$0.00	00.02	\$42,245.42	\$0.00	\$0.00	\$42,245.42
SemCrude	SANDRIDGE ENERGY INC	\$0.00	00.02		\$0.00	\$0.00	\$2,120,091.48
SemCrude	SANDRIDGE TERTIARY LLC	\$0.00	00.02		\$92,990.48	\$0.00	\$92,990.48
SemCrude	SANDSTONE RESOURCES INC	\$0.00	00.02		\$0.00	\$0.00	\$40,406.05
SemCrude	SANOOK INC	\$0.00	\$0.00	\$86.00	\$0.00	\$0.00	\$86.00
SemCrude	SARGECO INC	\$0.00	\$0.00	\$0.00	\$147.40	\$0.00	\$147.40
SemCrude	SASAKWA OPERATING LLC	\$0.00	\$0.00		\$0.00	\$0.00	\$41,767.07
SemCrude	SCAYERS OIL AND GAS INC	\$0.00	\$0.00		\$0.00	\$0.00	\$3,881.37
SemCrude	SCHANKIE WELL SERVICE	\$653,490.74	\$0.00		\$0.00	\$0.00	\$653,490.74
SemCrude	SCOTT A BROGNA	\$0.00	\$0.00		\$0.00	\$0.00	\$28.64
SemCrude	SCOTT-BANES-SCOTT PARTNERSHIP	\$0.00	\$0.00		\$0.00	\$0.00	\$36,715.58
SemCrude	SEAGULL OPERATING CO	\$0.00	\$0.00	\$28,367.11	\$0.00	\$0.00	\$28,367.11
SemCrude	SEED GROUP	\$4,631.34	\$0.00		\$0.00	\$0.00	\$4,631.34
SemCrude	SEEKER LLC	\$160,880.80	\$0.00		\$0.00	\$0.00	\$160,880.80
SemCrude	SELCO DRILLING INC	\$57,666.09	\$0.00		\$0.00	\$0.00	\$57,666.09
SemCrude	SELCO OIL & GAS INC	\$0.00	\$0.00		\$0.00	\$0.00	\$65,864.09
SemCrude	SELL DRILLING COPMANY	\$0.00	\$0.00	\$40,216.17	\$0,00	\$0.00	\$40,216.17
SemCrude	SELLERS OIL COMPANY	\$0.00	\$0.00	•	\$0.00	\$0.00	\$49,386.24
SemCrude	SENAX INC	\$0.00	\$0.00		\$0.00	\$0.00	\$71.20
SemCrude	SEVEN WELLS OIL & GAS EXPL.	\$0.00			\$1,360.11	\$0.00	\$1,360.11
SemCrude	SEYMOUR OIL CO	\$90.14	\$0.00		\$0.00	\$0.00	\$90.14
SemCrude	SGO PETROLEUM INC	\$0.00	\$0.00		\$0.00	\$0.00	\$13.71
SemCrude	SHERMAN OIL & GAS CO	\$0.00	\$0.00		\$0.00	\$0.00	\$31.68
SemCrude	SHIDELER ENTERPRISES INC	\$0.00	\$0.00	\$44,097.30	\$0.00	\$0.00	\$44,097.30
5511151464	**************************************						

smooth in consequence in the				First Purchase Amou	ifs in Active States		
Debtor	Operator	Kansas	New Mexico		Texas	Wyoming	
SemCrude	SHIELDS OPERATING INC.	\$0.00	\$0.00	\$59,722.12	\$0.00	\$0.00	\$59,722.12
SemCrude	SHIRLEY A. KLINGER	\$0.00	\$0.00	\$17,371.85	\$0.00	\$0.00	\$17,371.85
SemCrude	SHIRLEY A. WARD	\$22.76	\$0.00	\$0.00	\$0.00	\$0.00	\$22.76
SemCrude	SHOFNER OIL CO INC	\$124.98	\$0.00	\$0.00	\$0.00	\$0.00	\$124.98
SemCrude	SHORT AND SHORT LLC	\$0.00	\$0.00	\$198,785.29	\$0.00	\$0.00	\$198,785.29
SemCrude	SHORT OIL COMPANY	\$0.00	\$0.00	\$495,737.62	\$0.00	\$0.00	\$495,737.62
SemCrude	SIEGRIST INVESTMENT CO.	\$0.00	\$0.00	\$233.10	\$0.00	\$0.00	\$233.10
SemCrude	SINGER OIL COMPANY	\$0.00	\$0.00	\$308.61	\$0.00	\$0.00	\$308.61
SemCrude	SISLER BROTHERS	\$0.00	\$0.00	\$147,253.24	\$0.00	\$0.00	\$147,253.24
SemCrude SemCrude	SLAWSON EXPLORATION CO INC	\$39,650.78	\$0.00	\$508,955.16	\$0.00	\$0.00	\$548,605.94
SemCrude	SMITH OIL COMPANY	\$0.00	\$0.00	\$63.60	\$0.00	\$0.00	\$63.60
SemCrude	SNEDCO INC.	\$42,136.54	\$0,00	\$0.00	\$0.00	\$0.00	\$42,136.54
SemCrude	SNYDER PARTNERS	\$0.00	\$0.00	\$397,214.79	\$0.00	\$0.00	\$397,214.79
SemCrude SemCrude	SOJOURNER DRILLING CORPORATION	\$0.00	\$0.00	\$0.00	\$3,619.67	\$0.00	\$3,619.67
	SONNY'S TANK TRUCKS INC	\$0.00	\$0.00	\$6,419.14	\$0.00	\$0.00	\$6,419.14
SemCrude	SOONER EXPLORATION & DEVELOPMENT CO INC	\$0.00	\$0.00	\$21,813.79	\$0.00	\$0.00	\$21,813.79
SemCrude	SOONER OIL FIELD SERVICE INC	\$0.00	\$0.00	\$741.32	\$0.00	\$0.00	\$741.32
SemCrude	SOONER OIL FIELD SERVICES INC.	\$0.00	\$0.00	\$94.11	\$0.00	\$0.00	\$94.11
SemCrude	SOONER PIPE & SUPPLY OIL CORP	\$0.00	\$0.00	\$8,177.20	\$0.00	\$0.00	\$8,177.20
SemCrude	SOONER TREND EXPLORATION INC	\$0.00	\$0.00	\$25,438.46	00.00	\$0.00	\$25,438.46
SemCrude	SOUTH BURBANK PETROLEUM CORP	\$0.00	\$0,00	\$846,929.19	\$0.00	\$0.00	\$846,929.19
SemCrude	SOUTH BORDAIN FETROEDOM GOLD	\$0.00	\$0.00	\$300.15	\$0.00	\$0.00	\$300.15
SemCrude	SOUTH FOTT INC	\$0.00	\$0.00	\$122.89	\$0.00	\$0.00	\$122.89
SemCrude	SOUTHERN PRODUCTION	\$0.00	\$0.00	\$37,373.50	\$0.00	\$0.00	\$37,373.50
SemCrude	SOUTHERN PRODUCTION SOUTHERN RESOURCES INC	\$0.00	\$0.00	\$510,841.23	\$0.00	\$0.00	\$510,841.23
SemCrude	SOUTHERN RESOURCES INC SOUTHLAKE EXPLORATION INC	\$0.00	\$0.00	\$0.00	\$197,328.98	\$0.00	\$197,328.98
SemCrude	SOUTHEAR OIL & GAS	\$0.00	\$0.00	\$30,321.18	\$0.00	\$0.00	\$30,321.18
SemCrude	SOUTHWEST ENERGY COMPANY	\$0.00	\$0.00	\$508.32	\$0.00	\$0.00	\$508.32
SemCrude	SOUTHWEST GAS SYSTEMS, INC.	\$0.00	\$0.00	\$0.00	\$19,728.97	\$0.00	\$19,728.97
SemCrude	SOUTHWEST PETROLEUM CORPORATION	\$0.00	\$0.00	\$90.47	\$0.00	\$0.00	\$90.47
SemCrude	SOUTHWIND EXPORATION LLC	\$4,623.48	\$0.00	\$0.00	\$0.00	\$0.00	\$4,623.48
SemCrude	SPALDING ENERGY, INC.	\$0.00	\$0.00	\$0.00	\$109,292.15	\$0.00	\$109,292.15
SemCrude	SPARES ENERGY INC	\$0.00	\$9.00	\$379.17	\$0.00	\$0.00	\$379.17
SemCrude	SPEARS OIL COMPANY	\$0.00	\$0.00	\$20,936.07	\$0.00	\$0.00	\$20,936.07
SemCrude	SPECIAL ENERGY CORPORATION	\$0.00	\$0.00	\$1,852,915.16	\$0.00	\$0.00	\$1,852,915.16
SemCrude	SPELLER OIL CORPORATION	\$0.00	\$0.00	\$108.80	\$0.00	\$0.00	\$108.80
SemCrude	SPESS OIL COMPANY INC	\$64,331.36	\$0.00	\$183,747.15	\$0.00	\$0.00	\$248,078.51
SemCrude		\$39,056.68	\$0.00	\$0.00	\$0.00	\$0.00	\$39,056.68
SemCrude	SPINES EXPLORATION INC. SPRING OPERATING COMPANY	\$0.00	\$0.00	\$467,331.66	\$0.00	\$0.00	\$467,331.66
SemCrude	SPRING OPERATING COMPANY SPRING VALLEY OIL & GAS INC	\$0.00	\$0.00	\$21,961.70	\$0.00	\$0.00	\$21,961.70
SemCrude	* ·	\$0.00	\$0.00	\$101.37	\$0,00	\$0.00	\$101.37
SemCrude	SSB PRODUCTION L C ST MARY LAND & EXPLORATION CO *	\$0.00	\$0.00	\$58.18	\$370,674.86	\$0.00	\$370,733.04
SemCrude	ST MARY LAND & EXPLORATION CO ST MARY LAND & EXPLORATION COMPANY	\$0.00	\$0.00	\$67,948.79	\$0.00	\$0.00	\$67,948.79
SemCrude		\$0.00	\$0.00	\$1,145.45	\$0.00	\$0.00	\$1,145.45
SemCrude	STACY OPERATING LLC	\$0.00	\$0.00	\$447,970.27	\$0.00	\$0.00	\$447,970.27
SemCrude	STAGHORN ENERGY LLC	\$0.00	\$0.00	\$0.00	\$28,383.84	\$0.00	\$28,383.84
SemCrude	STAMPER OPERATING CO INC	\$0.00	\$0.00	\$0.00	\$3,182,221.73	00.02	\$3,182,221.73
SemCrude	STAR PRODUCTION INC	\$825.89	\$0.00	\$0.00	\$0.00	\$0.00	\$825.89
SemCrude	STARR F SCHLOBOHM	\$1,808.33	\$0.00	\$0.00	\$0.00	\$0.00	\$1,808.33
SemCrude	STELBAR OIL CORPORATION INC	¥1,000.33	45.00				

		- ONE ANY CONTROL OF THE CONTROL OF	esse essentiality	rst Purchase Amounts	ii ArtiveStales		
			w Mexico	Oklahoma	Texas	Wyoming	Total
Debtor	Operator (in the control of the cont	\$61.54	\$0.00	\$0.00	\$0.00	\$0.00	\$61.54
SemCrude	STEPCO INC	\$0.00	\$0.00	\$2,148,721.32	\$0.00	\$0.00	\$2,148,721.32
SemCrude	STEPHENS & JOHNSON OPERATING	\$0.00	\$0.00	\$62,925.09	\$0.00	\$0.00	\$62,925.09
SemCrude	STEPHENS EXPLORATION INC	\$0.00	\$0.00	\$30,913.00	\$0.00	\$0.00	\$30,913.00
SemCrude	STESCO OPERATING COMPANY	\$108.96	\$0.00	\$0.00	\$0.00	\$0.00	\$108.96
SemCrude	STEVE JONES	\$0.00	\$0.00	\$51,835.93	\$0.00	\$0.00	\$51,835.93
SemCrude	STEVE W MEYER	\$24,368.50	\$0.00	\$0.00	\$0.00	\$0.00	\$24,368.50
SemCrude	STEVENSON OPERATING LLC	\$0.00	\$0.00	\$30,570.37	\$0.00	\$0.00	\$30,570.37
SemCrude	STIDHAM_BOB & ALICE	\$0.00	\$0.00	\$0.00	\$168.72	\$0.00	\$168.72
SemCrude	STINGER OIL & GAS INC	\$0.00	\$0.00	\$14.77	\$0.00	00.00	\$14.77
SemCrude	STORM EXPLORATION CO.	\$0.00	\$0.00	\$11,926.92	\$0.00	\$0.00	\$11,926.92
SemCrude	STRONGARM WELL SERVICE INC	\$0.00	\$0.00	\$0.00	\$34,724.29	\$0.00	\$34,724.29
SemCrude	STROUBE ENERGY CORPORATION	\$5,476.88	\$0.00	\$0.00	\$0.00	\$0.00	\$5,476.88
SemCrude	STUBBS & FLEMING	\$0.00	\$0.00	\$29,908.83	\$0.00	\$0.00	\$29,908.83
SemCrude	SUBSURFACE MINERALS GROUP LLC	\$0.00	\$0.00	\$0.00	\$81,492.76	\$0.00	\$81,492.76
SemCrude	SUBSURFACE OPERATIONS	\$0.00	\$0.00	\$0.00	\$127,950.41	\$0.00	\$127,950.41
SemCrude	SUBSURFACE PRODUCTION INC	\$0.00	\$0.00	\$107,413.01	\$0.00	\$0.00	\$107,413.01
SemCrude	SUHAIRD	\$0.00	\$0.00	\$2,509,066.14	\$0.00	\$0.00	\$2,509,066.14
SemCrude	SULLIVAN AND COMPANY LLC	\$0.00	\$0.00	\$2.73	\$0.00	\$0.00	\$2.73
SemCrude	SUMMIT OIL COMPANY	\$0.00	\$0.00	\$45,671.50	\$0.00	\$0.00	\$45,671.50
SemCrude	SUMMIT OPERATING	\$0.00	\$0.00	\$0.00	\$2,698.27	\$0.00	\$2,698.27
SemCrude	SUNDANCE CONSULTING CORPORATION	\$0.00	\$0.00	\$43,377.87	00.02	\$0.00	\$43,377 <i>.</i> 87
SemCrude	SUNDOWN ENERGY INC	\$0.00	\$0.00	\$0.00	\$4,585.56	\$0.00	\$4,585.56
SemCrude	SUNRAY OIL CO	\$0.00	\$0.00	\$19,215.49	\$0.00	\$0.00	\$19,215.49
SemCrude	SUNSET ENERGY LLC	\$0.00	\$0.00	\$2,743.78	\$0.00	\$0.00	\$2,743.78
SemCrude	SUNTERRA ENERGY CORP	\$0.00	\$0.00	\$0.00	\$27,621.52	\$0.00	\$27,621.52
SemCrude	SUOCO OIL CORPORATION	\$0.00	\$0.00	\$103.37	\$0.00	\$0.00	\$103.37
SemCrude	SUPERIOR OIL & GAS LLC SYNERGY OIL & GAS LP	\$0.00	00.02	\$0.00	\$74,941.66	\$0.00	\$74,941.66
SemCrude	T & C MFG & OPERATING, INC.	\$44,398.67	\$0.00	\$0.00	\$0.00	\$0.00	\$44,398.67
SemCrude	T&LOILINC	\$0.00	\$0.00	\$40,286.36	\$0.00	\$0.00	\$40,286.36
SemCrude	T D ENERGY COMPANY INC	\$0.00	\$0.00	\$18.30	\$0.00	\$0.00	\$18.30
SemCrude	TILCOILINC	\$0.00	\$0.00	\$21,462.81	\$0.00	\$0.00	\$21,462.81
SemCrude	T M ENTERPRISES INC	\$39,317.05	\$0.00	\$0.00	\$0.00	\$0.00	\$39,317.05
SemCrude	TACK OPERATING LLC	\$0.00	\$0.00	\$419.28	\$0.00	\$0.00	\$419.28
SemCrude	TALLGRASS PETROLEUM CORP	\$0.00	\$0.00	\$0.01	\$0.00	\$0.00	\$0.01
SemCrude	TBK ENERGY INC	\$0.00	\$0.00	\$2,492.07	\$0.00	\$0.00	\$2,492.07
SemCrude	TECHNICAL ENERGY SERVICES	\$0.00	\$0.00	\$102,615.21	\$0.00	\$0.00	\$102,615.21
SemCrude	TED BRANHAM	\$0.00	\$0.00	\$9,130.06	\$0.00	\$0.00	\$9,130.06
SemCrude	TED W MCGONIGAL	\$0.00	\$0.00	\$45,325.42	\$0.00	\$0.00	\$45,325.42
SemCrude	TEE CEE'S INC	\$0.00	\$0.00	\$20,720.88	\$0,00	\$0.00	\$20,720.88
SemCrude	TELESIS OPERATING CO INC.	\$0.00	\$0.00	\$0.00	\$27,428.06	\$0.00	\$27,428.06
SemCrude	TEMPEST	\$0.00	\$0.00	\$0.00	\$163,269.34	\$0.00	\$163,269.34
SemCrude SemCrude	TEPE OIL AND GAS	\$22.27	\$0.00	\$0.00	\$0.00	\$0.00	\$22.27 \$11,830,96
SemCrude SemCrude	TEPPCO	\$0.00	00.02	\$11,830.96	\$0.00	\$0.00	\$11,830.96 \$257,774.50
SemCrude SemCrude	TE-RAY RESOURCES INC	\$0.00	\$0.00	\$257,774.50	\$0.00	\$0.00	
SemCrude SemCrude	TERRA PRODUCTION COMPANY	\$0.00	\$0.00	\$4,235.08	\$0.00	\$0.00	\$4,235.08 \$91.92
SemCrude SemCrude	TERRY L SMITH	\$0.00	\$0.00	\$91.92	\$0.00	\$0.00	\$91.92 \$37.37
SemCrude SemCrude	TEXACO EXPLORATION & PRODUCTION	\$0.00	\$0.00	\$37.37	\$0.00	\$0.00	\$196.09
SemCrude SemCrude	TEXACO INC	\$0.08	\$0.00	\$196.09	\$0.00	\$0.00	\$170.03
Demoinac	10.0100 11.0						

					te in define Xintee		
			New Mexico	Oklahoma	Tay Same	Wyoming	Total
Debtor	Operator	Kansas S0.00	\$0.00	\$41,026.14	\$0.00	\$0.00	\$41,026.14
SemCrude	TEX-OK ENERGY LIMITED PARTNERSHIP	\$0.00	\$0.00	\$243.01	\$0.02	\$0.00	\$243.01
SemCrude	TEXOMA OIL & GAS CORP	\$0.00	\$0.00	\$0.00	\$282.19	\$0.00	\$282.19
SemCrude	TEXOMA OIL TOOLS, INC.	\$933,754.84	\$0.00	\$0.00	00.02	\$0.00	\$933,754.84
SemCrude	TGT PETROLEUM CORP	\$933,734.84 \$0.00	\$0.00	\$84,229.49	\$0.00	\$0.00	\$84,229.49
SemCrude	THE BALE CORPORATION	\$0.00	\$0.00	\$81.96	\$0.00	\$0.00	\$81.96
SemCrude	THE DAUBE COMPANY	\$0.00	\$0.00	\$3,958.68	\$0.00	\$0.00	\$3,958.68
SemCrude	THE DEHART COMPANY A PARTNERSHIP	\$0.00	\$0.00	\$63,681.17	\$0.00	\$0.00	\$63,681.17
SemCrude	THE GLENN SUPPLY CO. INC.	\$0.00	\$0.00	\$546,390.26	\$0.00	\$0.00	\$546,390.26
SemCrude	THE GLOCO LLC	\$0.00	\$0.00	\$23,816,58	\$0.00	\$0.00	\$23,816.58
SemCrude	THE QUINTON LITTLE CO INC	\$8,154.87	\$0.00	\$0.00	\$0.00	\$0.00	\$8,154.87
SemCrude	THE TREES OIL COMPANY	\$0.00	\$0.00	\$86,127.51	\$0.00	\$0.00	\$86,127.51
SemCrude	THOMAS D ROBBINS	\$0.00	\$0.00	\$372.73	\$0.00	\$0.00	\$372.73
SemCrude	THOMAS EXPLORATION CO	\$20,761.66	\$0.00	\$0.00	\$0.00	\$0.00	\$20,761.66
SemCrude	THOMAS GARNER AND TRENTON L TEAGER	\$0.00	\$0.00	\$9.62	\$0.00	\$0.00	\$9.62
SemCrude	THOMAS J JONES III	\$81,939.26	\$0.00	\$0.00	\$0.00	\$0.00	\$81,939.26
SemCrude	THOMSEN BROTHERS OIL CO.	\$1,734,52	\$0.00	\$0.00	\$0.00	\$0.00	\$1,734.52
SemCrude	THORNTON DRILLING COMPANY	\$504,847.62	\$0.00	\$0.00	\$0.00	\$0.00	\$504,847.62
SemCrude	THOROUGHBRED ASSOCIATES LLC	\$0,00	\$0.00	\$3,362.06	\$0.00	\$0.00	\$3,362.06
SemCrude	THR OIL COMPANY	\$0.00	\$0.00	\$22,737.23	\$0.00	\$0.00	\$22,737.23
SemCrude	THREE SANDS INVESTMENTS INC	\$82,668.23	\$0.00	\$0.00	\$0.00	\$0.00	\$82,668.23
SemCrude	THREE-D RESOURCES INC	\$66.50	\$0.00	\$0.00	\$0.00	\$0.00	\$66.50
SemCrude	THRESHOLD DEVELOPMENT COMPANY	\$0.00	\$0.00	\$97.49	\$0.00	\$0.00	\$97.49
SemCrude	TIDEWATER OPERATING CO INC	\$59.37	\$0.00	\$0.00	\$0.00	\$0.00	\$59.37
SemCrude	TIGER OIL AND GAS LLC	\$0.00	\$0.00	\$80,092.08	\$0.00	\$0.00	\$80,092.08
SemCrude	TILLEY OIL & GAS INC	\$0.00	\$0.00	\$19,937.64	\$0.00	\$0.00	\$19,937.64
SemCrude	TIM JOHNSON	\$0.00	\$0.00	\$63,932.14	\$0.00	\$0.00	\$63,932.14
SemCrude	TIM TUCKER	\$2,768.71	\$0.00	\$0.00	\$0.00	\$0.00	\$2,768.71
SemCrude	TINA A WATSON TITAN ENERGY INC	\$0.00	\$0.00	00.02	\$2,707,629.53	\$0.00	\$2,707,629.53
SemCrude		\$18,438.71	\$0.00	\$239,046.00	\$0.00	\$0.00	\$257,484.71
SemCrude	TITAN RESOURCES LIMITED TKL OIL PROPERTIES, INC.	\$0.00	\$0.00	\$29,550.17	\$0.00	\$0.00	\$29,550.17
SemCrude	TKO INVESTMENT COMPANY	\$19,424.92	\$0.00	\$0.00	\$0.00	\$0.00	\$19,424.92
SemCrude	TLC OIL INC	\$0.00	\$0.00	\$102,026.29	\$0.00	\$0.00	\$102,026.29
SemCrude	TLS OIL AND GAS INC	\$0.00	\$0.00	\$994.70	\$0.00	\$0.00	\$994.70
SemCrude	TOKLAN OIL & GAS CORPORATION	\$0.00	\$0.00	\$171,821.96	\$0.00	\$0.00	\$171,821.96
SemCrude	TOLLETT OIL	\$0.00	\$0,00	\$0.00	\$12,794.75	\$0.00	\$12,794.75
SemCrude	TON BROWN CONSTRUCTION	\$8,323.19	\$0.00	\$0.00	\$0.00	\$0.00	\$8,323.19
SemCrude	TOMO OIL & GAS INC	\$0.00	\$0.00	\$0.00	\$96.02	\$0.00	\$96.02
SemCrude	TOMCO OIL INC	\$539.23	\$0.00	\$0.00	\$0.00	\$0.00	\$539.23
SemCrude	TOP O TEXAS PRODUCTION CO INC	\$0.00	\$0.00	\$0.00	\$674.08	\$0.00	\$674.08
SemCrude	TOPSAIL SECONDARY RECOVERY	\$0.00	\$0.00	\$2,156.67	\$0.00	\$0.00	\$2,156.67
SemCrude	TRACE OIL COMPANY	\$0.00	\$0.00	\$12,062.48	\$0.00	\$0.00	\$12,062.48
SemCrude	TRACE OIL COMPANY TRACKER MINERAL	\$0.00	\$0.00	\$0.00	\$64,505.63	\$0.00	\$64,505.63
SemCrude	TRACK KENNEDY	\$0.00	\$0.00	\$19,371.84	\$0.00	\$0.00	\$19,371.84
SemCrude	TRANSCOASTAL OPERATING LLC	\$0.00	\$0.00	\$273.58	\$0.00	\$0.00	\$273.58
SemCrude	TRANSCOASTAL OF BRATING LEC	\$0.00	\$0.00	\$73.83	\$0.00	\$0.00	\$73.83
SemCrude	TRENCHCO INC	\$0.00	\$0.00	\$87,860.99	\$0.00	\$0.00	\$87,860.99
SemCrude	TREY RESOURCES INC	\$0.00	\$0.00	\$103,061.80	\$0.00	\$0.00	\$103,061.80
SemCrude	TRIAD ENERGY INC	\$0.00	\$0.00	\$731,400.79	\$0.00	\$0.00	\$731,400.79
SemCrude	LIGHT PARKOT HAC						

				First Purchase Amount	s in Active States		
		Kansas	New Mexico	Oklahoma			Total
Debtor	Operator TRI-EX PETROLEUM LP	\$81,425.99	\$0.00	\$18,715.22	\$0.00	\$0.00	\$100,141.21
SemCrude	TRILOGY OPERATING, INC.	\$0.00	\$307,645.10	\$0.00	\$0.00	\$0.00	\$307,645.10
SemCrude		\$0.00	\$0.00	\$3.49	\$0.00	\$0.00	\$3.49
SemCrude	TRINEHA INC TRIPLEDEE DRILLING CO	\$0.00	\$0.00	\$24,041.33	\$0.00	\$0.00	\$24,041.33
SemCrude	TRIPOWER RESOURCES INC	\$0.00	\$0.00	\$1,693,270.39	\$0.00	\$0.00	\$1,693,270.39
SemCrude	TRIPOWER RESOURCES INC TRIVESTCO ENERGY COMPANY	\$0.00	\$0.00	\$100.18	\$0.00	\$0.00	\$100.18
SemCrude		\$277,275,06	\$0.00	\$0.00	\$0.00	\$0.00	\$277,275.06
SemCrude	TROY JACO INC	\$0.00	\$0.00	\$1,769.79	\$0.00	\$0.00	\$1,769.79
SemCrude	TRUMAN E WEAVER DBA TRYNOR OIL AND GAS INC.	\$0.00	\$0.00	\$5,138.49	\$0.00	\$0.00	\$5,138.49
SemCrude	TSBT OIL & GAS PROPERTIES	\$0.00	\$0.00	\$390.60	\$0.00	\$0.00	\$390.60
SemCrude	TULSA MINERALS INC	\$0.00	\$0.00	\$22.15	\$0.00	\$0.00	\$22.15
SemCrude	TULUM CORPORATION	\$0.00	\$0.00	\$11,346.01	\$0.00	\$0.00	\$11,346.01
SemCrude		\$0.00	\$0.00	\$0.00	\$135.72	\$0.00	\$135.72
SemCrude	TURTLE BACK CORPORATION	\$0.00	\$0.00	\$8,622.43	\$0.00	\$0.00	\$8,622.43
SemCrude	TUSCANY OIL & GAS LLC TWENTY/TWENTY OIL & GAS INC	\$0.00	\$0.00	\$416.93	\$0.00	\$0.00	\$416.93
SemCrude		\$0.00	\$0.00	\$21,919.83	\$0.00	\$0.00	\$21,919.83
SemCrude	TWILA MAE WILSON ESTATE TWIN OAKS CONSTRUCTION COMPANY	\$0.00	\$0.00	\$0.00	\$23,425.14	\$0.00	\$23,425.14
SemCrude		\$23,543.78	\$0.00	\$0.00	\$0.00	\$0.00	\$23,543.78
SemCrude	TY HAZEN OIL UMBERHAM OIL COMPANY INC	\$0.00	\$0.00	\$105,750.14	\$0.00	\$0.00	\$105,750.14
SemCrude	OWREKHWIM OFF CONTENTS INC	\$0.00	\$0.00	\$38,931.37	\$0.00	\$0.00	\$38,931.37
SemCrude	UNIT PETROLEUM COMPANY UNITED ENERGY & EXPLORATION INC	\$0.00	\$0.00	\$13,916.43	\$0.00	\$0.00	\$13,916.43
SemCrude	UNITED OIL CORPORATION	\$0.00	\$0.00	\$48,685.60	\$0.00	\$0.00	\$48,685.60
SemCrude	UNITED OIL CORPORATION UNITED PRODUCTION COMPANY LLC	\$0.00	\$0.00	\$587,951.29	\$0.00	\$0.00	\$587,951.29
SemCrude	UPLANDS RESOURCES INC	\$0.00	\$0.00	\$176,241.92	\$0.00	\$0.00	\$176,241.92
SemCrude	UPTEX OPERATING LP	\$0.00	\$0.00	\$0.00	\$277,971.93	\$0.00	\$277,971.93
SemCrude	V C MCCLINTOCK	\$0.00	\$0.00	\$1,041.55	\$0.00	\$0.00	\$1,041.55
SemCrude	VAIL ENERGY	\$10,807.94	\$0.00	\$0.00	\$0.00	\$0,00	\$10,807.94
SemCrude	VALIENT OIL & INVESTMENT CO.	\$0.00	\$0.00	\$37,129.84	\$0.00	\$0.00	\$37,129.84
SemCrude	VALIENT OIL & INVESTMENT OF	\$0.00	\$0.00	\$20,587.98	\$0.00	\$0.00	\$20,587.98
SemCrude	VALLEY PETROLEUM COMPANY	\$0.00	\$0.00	\$0.00	\$149,149.25	\$0.00	\$149,149.25
SemCrude	VAMCO OPERATIONS	\$127,158.69	\$0.00	\$0.00	\$0.00	\$0.00	\$127,158.69
SemCrude	VAN OIL	\$19,914.21	\$0.00	\$0.00	\$0.00	\$0.00	\$19,914.21
SemCrude	VANTAGE PETROLEUM INC	\$15,953.94	\$0.00	\$0.00	\$0.00	\$0.00	\$15,953.94
SemCrude	VASSAUR DRILLING CO	\$0.00	\$0.00	\$241.08	\$0.00	\$0.00	\$241.08
SemCrude	VAUGHAN PRODUCTION COMPANY	\$0.00	\$0.00	\$117.55	00.02	\$0.00	\$117.55
SemCrude	VAUGHN GOOD OIL COMPANY LLC	\$0.00	\$0.00	\$568,978.21	\$0.00	\$0.00	\$568,978.21
SemCrude	VECTOR EXPLORATION INC	\$0.00	\$0.00	\$7.40	\$0.00	\$0.00	\$7.40
SemCrude	VEENKER RESOURCES INC	\$635,424.38	\$0.00	\$1,201,518.48	\$0.00	\$0.00	\$1,836,942.86
SemCrude	VEGA ENERGY COMPANY	\$0.00	\$0.00	\$3,500.03	\$0.00	\$0.00	\$3,500.03
SemCrude	VELDON ZOLLINGER INC	\$0.00	\$0.00	\$41.36	\$0.00	\$0.00	\$41.36
SemCrude	VELMA ENERGY LLC	\$0.00	\$0.00	\$34,847.33	\$0.00	\$0.00	\$34,847.33
SemCrude	VELMA ENERGY LLC VENIS EXPLORATION COMPANY	\$0.00			\$0.00	\$0.00	\$161,867.12
SemCrude	VERNA M AKIN	\$0.00			\$0.00	\$0.00	\$52.57
SemCrude	VERNON BROWN	\$0.00			\$0.00	\$0.00	\$61,157.14
SemCrude	VESS OIL CORP	\$84,239.74		\$0.00	\$0.00	\$0.00	\$84,239.74
SemCrude	VESS OIL CORP VIATHON HYDROCARBONS, INC.	\$0.00		\$0.00	\$103,073.43	\$0.00	\$103,073.43
SemCrude	VICKIE R CHATHAM	\$0.00		\$19,701.56	\$0.00	\$0.00	\$19,701.56
SemCrude	VICTOR W PRYOR JR	\$0.00		\$57,646.62	\$0.00	\$0.00	\$57,646.62
SemCrude	VIKE OIL INC	\$0.00		\$37,869.33	\$0.00	\$0.00	\$37,869.33
SemCrude	AINT OUR TIME						

				SOLOMINIS IS CONTROL OF THE SOLOMINIS OF T				
Septimus VILE OIL INC.								
SemCrude VIER OIL INC. \$132.5694.72 \$10.00 \$0.00 \$0.00 \$0.00 \$32.5694.72	Debtor		13.41123.1136.1137.11	A CONTRACTOR OF THE PARTY OF TH			\$0.00	
SemCrude	SemCrude				· ·	\$0.00	\$0.00	\$325,694.72
SeniCrude						\$0.00	\$0.00	\$25,078.71
SentCrude						\$681.37	\$0.00	
SemCrade			•				\$0.00	
SemCnude					\$0.00	\$0.00	00.00	
SemCrude VIRIGIL RAT IMATABE \$10.00 \$0.00 \$77,891.79 \$0.00 \$0.00 \$77,891.79 \$0.00 \$0.00 \$77,891.79 \$0.00 \$0.00 \$77,891.79 \$0.00 \$0.00 \$77,891.79 \$0.00 \$			· ·	• • •	\$957.61	\$0.00	•	-
SemCrude					\$77,891.79	\$0.00		
SemiCrude VII NATURAL RESOURCES : EMM PROJECT \$2,365.93 \$0.00 \$0.0		VISTA ENERGY INC			\$0.00	\$0.00		
SemCrude				\$0.00	\$0.00	\$0.00		
SemCrude VRC OIL & ORS O'PERATING CO \$782,246.25 \$0.00 \$0.00 \$0.00 \$0.00 \$782,246.25 \$0.00 \$0.00 \$0.00 \$0.00 \$782,246.25 \$0.00 \$0.00 \$0.00 \$0.00 \$782,246.25 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$782,246.25 \$0.00			-		\$0.00	\$0.00		
SemCrude W D SHORT			- /		\$36,269.19	\$0.00		*
SemCrude				·		\$0.00	·	
SemCrude W H PRICE S0.00 S0.00 S2,239.91 S0.00 S0.00 S2,239.91 S0.00 S0.00 S2,239.91 S0.00 S0.00 S3,096.24 SemCrude W H PRICE S0.00 S0.00 S0.00 S145,396.06 S0.00 S0.00 S0.00 S145,396.06 S0.00 S0.00 S0.00 S145,396.06 S0.00 S0.0				\$0.00	\$919.10	\$0.00		
Semicrade				\$0.00	\$2,239.91			,
SemCrude W M OIL INCORPORATED \$0.00 \$0.00 \$145,959.95 \$0.00 \$0.00 \$236,159.59 \$30.00 \$30.00 \$326,159.59 \$30.00 \$30.00 \$326,159.59 \$30.00 \$30.00 \$326,159.59 \$30.00 \$30.00 \$326,159.59 \$30.00 \$30.00 \$326,159.59 \$30.00 \$30.00 \$326,179.59 \$30.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$30.00 \$320,77 \$300.00 \$300			• •	\$0.00	\$53,096.24			,
SemCrude			• • • • •	\$0.00	\$145,396.06	* * * * * * * * * * * * * * * * * * * *		
SemCrude W M PETROLEUM \$18,868.77 \$0.00 \$0.00 \$50.77 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$52,07.7 \$0.00 \$0.00 \$0.00 \$0.00 \$52,07.7 \$0.00			\$0.00	\$0.00	\$246,159.59		*	
SemCrude		** *** *** * * * * * * * * * * * * * *	\$18,868.97	\$0.00	\$0.00			
SemCrude				\$0.00	\$520.77			
SemCrude WADSRT WILSON \$50.00 \$			\$0.00	\$0.00	\$11,540.81			
SemCrude WABASH ENERGY CORP \$889,936,97 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$234,401.25 \$0.00 \$234,401.25 \$0.00 \$234,401.25 \$0.00 \$234,401.25 \$0.00 \$234,401.25 \$0.00 \$234,401.25 \$0.00 \$234,401.25 \$0.00 \$234,401.25 \$0.00 \$234,401.25 \$0.00 \$234,401.25 \$0.00 \$0			\$0.00	\$0.00				•
SemCrude WAD EXPLORATION AND PRODUCTION LLC \$0.00 \$0.0			\$589,936.97	\$0.00	\$0.00	•	· ·	
SemCrude WALCHER OIL CC \$281,040.23 \$0.00 \$0.00 \$30.00 \$50.00 \$57,751.01 \$50.00 \$50.00 \$50.00 \$51.4,008.57 \$50.00 \$50.00 \$51.4,008.57 \$50.00 \$50.00 \$51.4,008.57 \$50.00 \$50.00 \$51.4,008.57 \$50.00 \$577.67 \$50.00 \$50.00 \$577.67 \$50.00 \$50.00 \$577.67 \$50.00 \$50.00	-	WADASII ENERGI GOLU WADI EYPI ORATION AND PRODUCTION LLC	\$0.00	\$0.00				
SemCrude WALLER RESOURCES INC \$0.00 \$0.00 \$31,71.01 \$0.00 \$45,224.70 \$0.00 \$45,224.70 \$0.00 \$45,224.70 \$0.00 \$45,224.70 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$15.40.85.7 \$0.00 \$0.00			\$281,040.23	\$0.00	-	* *		
SemCrude WALSH & WATTS, INC. \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$154,008.57 \$0.00 \$0.00 \$131,106.27 \$0.00 \$0.00 \$14,411.69 \$0.00			\$0.00	\$0.00				•
SemCrude WALSMITH DRILLING COMPANY \$0.00 \$0.00 \$17.01 \$0.00 \$0.00 \$485.18 \$0.00 \$0.00 \$485.18 \$0.00 \$0.00 \$485.18 \$0.00 \$0.00 \$154,008.57 \$0.00			\$0.00					,
SemCrude WALTER GRIGGS \$0.00 \$0.00 \$154,008.57 \$0.00 \$0.00 \$114,008.57 \$0.00 \$0.00 \$114,008.57 \$0.00 \$0.00 \$114,008.57 \$0.00 \$0.00 \$114,008.57 \$0.00 \$0.00 \$114,008.57 \$0.00 \$0.00 \$114,008.57 \$0.00 \$0.00 \$114,008.57 \$0.00 \$0.00 \$114,008.57 \$0.00 \$0.00 \$114,008.57 \$0.00 \$0.00 \$114,008.57 \$0.00 \$0.00 \$114,008.57 \$0.00 \$			\$0.00			-		
SemCrude WALTERS OIL COMPANY \$0.00 \$0.00 \$131,106.27 \$0.00 \$0.00 \$0.00 \$131,106.27 \$0.00			\$0.00		•	*	-	
SemCrude WALTMAN FAMILY LTD PRINRSHP \$0.00 \$1.00 \$131,100.27 \$0.00 \$0.00 \$185,153.15 \$0.00 \$0.00 \$185,153.15 \$0.00 \$0.00 \$14,411.69 \$0.00 \$14,411.69 \$0.00 \$14,411.69 \$0.00 \$14,411.69 \$0.00 \$14,411.69 \$0.00 \$14,411.69 \$0.00 \$14,411.69 \$0.00 \$14,411.69 \$0.00 \$14,411.69 \$0.00 \$14,411.69 \$0.00 \$14,411.69 \$0.00 \$16,1067.44 \$0.00 \$0.00 \$16,1067.44 \$0.00 \$0.00 \$16,1067.44 \$0.00 \$0.00 \$16,1067.44 \$0.00 \$0.00 \$0.00 \$16,1067.44 \$0.00 \$0.00 \$0.00 \$0.00 \$16,1067.44 \$0.00			\$0.00			• • • • •		
SemCrude WARD PETROLEUM CORPORATION \$0.00 \$0.00 \$14,411.69 \$0.00 \$0.00 \$14,411.69 \$0.00 \$0.00 \$14,411.69 \$0.00 \$0.00 \$577.67 \$0.00 \$0.		WALTMAN FAMILY LTD PRTNRSHP	\$0.00	• • • •	·			
SemCrude WARPATH ENERGY INC \$0.00 \$0.00 \$0.00 \$577.67 \$0.00 \$577.67 SemCrude WATSON OIL & GAS, INC. \$0.00 \$0.00 \$161,067.44 \$0.00 \$0.00 \$577.67 SemCrude WAYNE D STOKES \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$53.16 SemCrude WAYNE D STOKES \$53.16 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$50.0								
SemCrude WATSON OIL & GAS, INC. \$0.00 \$0.00 \$0.00 \$161,067.44 \$0.00 \$0.00 \$161,067.44 SemCrude WAYNE D STOKES \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$53.16 SemCrude WAYNE DOBBS INC \$53.16 \$0.00 \$0.00 \$0.00 \$0.00 \$553.16 SemCrude WAYNE E WALCHER \$569.11 \$0.00 \$0.00 \$0.00 \$0.00 \$569.11 SemCrude WAYNE THOMAS OIL & GAS \$0.00 \$0.00 \$12,470.22 \$0.00 \$0.00 \$97.34 SemCrude WEAVER ENERGY CORPORATION \$0.00 \$0.00 \$97.34 \$0.00 \$0.00 \$97.34 SemCrude WEINKAUF PETROLEUM INC \$1,238.90 \$0.00 \$1,224,309.80 \$0.00 \$0.00 \$1,225,548.70 SemCrude WEILDON CORPORATION \$0.00 \$0.00 \$0.00 \$35.60 \$0.00 \$0.00 \$35.60 SemCrude WELLCO ENERGY INC \$0.00 \$0.00 \$97,337.79 \$0.00 </td <td></td> <td></td> <td>\$0.00</td> <td></td> <td></td> <td></td> <td></td> <td>• • • • • •</td>			\$0.00					• • • • • •
SemCrude WAYNE D STOKES \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$53.16 SemCrude WAYNE DOBBS INC \$53.16 \$0.00 \$0.00 \$0.00 \$0.00 \$569.11 SemCrude WAYNE E WALCHER \$569.11 \$0.00 \$0.00 \$0.00 \$0.00 \$569.11 SemCrude WAYNE THOMAS OIL & GAS \$0.00 \$0.00 \$12,470.22 \$0.00 \$0.00 \$97.34 \$0.00 \$0.00 \$97.34 SemCrude WEAVER ENERGY CORPORATION \$0.00 \$0.00 \$97.34 \$0.00 \$0.00 \$97.34 SemCrude WEINKAUF PETROLEUM INC \$1,238.90 \$0.00 \$1,224,309.80 \$0.00 \$0.00 \$1,225,548.70 SemCrude WELDON CORPORATION \$0.00 \$0.00 \$0.00 \$31.558 \$0.00 \$4,315.58 SemCrude WELDON DELANEY \$0.00 \$0.00 \$35.60 \$0.00 \$0.00 \$97,337.79 SemCrude WELLCO ENERGY INC \$0.00 \$0.00 \$0.00 \$0.00 <td< td=""><td>*</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>	*							
SemCrude WAYNE DOBBS INC \$53.16 \$0.00 \$0.00 \$0.00 \$0.00 \$569.11 SemCrude WAYNE E WALCHER \$569.11 \$0.00 \$0.00 \$0.00 \$0.00 \$569.11 SemCrude WAYNE THOMAS OIL & GAS \$0.00 \$0.00 \$12,470.22 \$0.00 \$0.00 \$97.34 \$0.00 \$0.00 \$97.34 SemCrude WEAVER ENERGY CORPORATION \$0.00 \$0.00 \$97.34 \$0.00 \$0.00 \$97.34 SemCrude WEINKAUF PETROLEUM INC \$1,238.90 \$0.00 \$1,224,309.80 \$0.00 \$0.00 \$1,225,548.70 SemCrude WELDON CORPORATION \$0.00 \$0.00 \$0.00 \$4,315.58 \$0.00 \$35.60 \$0.00 \$35.60 \$0.00 \$35.60 \$0.00 \$35.60 \$0.00 \$37.37.79 \$0.00 \$0.00 \$97.337.79 \$0.00 \$0.00 \$97.337.79 \$0.00 \$0.00 \$585,786.05 \$0.00 \$0.00 \$0.00 \$585,786.05 \$0.00 \$0.00 \$0.00 \$585,786.05 <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>•</td></td<>								•
SemCrude WAYNE E WALCHER \$569.11 \$0.00 \$0.00 \$0.00 \$0.00 \$12,470.22 SemCrude WAYNE THOMAS OIL & GAS \$0.00 <td></td> <td></td> <td></td> <td></td> <td></td> <td>=</td> <td></td> <td></td>						=		
SemCrude WAYNE THOMAS OIL & GAS \$0.00 \$0.00 \$12,470.22 \$0.00 \$0.00 \$97.34 \$0.00 \$0.00 \$97.34 \$0.00 \$0.00 \$97.34 \$0.00						*		
SemCrude WEAVER ENERGY CORPORATION \$0.00 \$0.00 \$97.34 \$0.00 \$0.00 \$1,225,548.70 SemCrude WEINKAUF PETROLEUM INC \$1,238.90 \$0.00 \$1,224,309.80 \$0.00 \$0.00 \$1,225,548.70 SemCrude WELDON CORPORATION \$0.00 \$0.00 \$0.00 \$4,315.58 \$0.00 \$35.60 SemCrude WELDON DELANEY \$0.00 \$0.00 \$97,337.79 \$0.00 \$0.00 \$97,337.79 SemCrude WELLCO ENERGY INC \$0.00 \$0.00 \$97,337.79 \$0.00 \$0.00 \$97,337.79 SemCrude WELLSTAR CORPORATION \$585,786.05 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$585,786.05 SemCrude WENTWORTH OPERATING COMPANY \$0.00 \$0.00 \$679.31 \$0.00 \$0.00 \$679.31 SemCrude WESLEY J KRUMME \$0.00 \$0.00 \$0.00 \$92,367.56 \$0.00 \$92,367.56 SemCrude WEST TEXAS ROYALTIES INC \$0.00 \$0.00 \$63,078.57 \$0.00			* *					
SemCrude WEINKAUF PETROLEUM INC \$1,238,90 \$0.00 \$1,224,309,80 \$0.00 \$4,315.58 SemCrude WELDON CORPORATION \$0.00 \$0.00 \$35.60 \$0.00 \$35.60 SemCrude WELDON DELANEY \$0.00 \$0.00 \$97,337.79 \$0.00 \$97,337.79 SemCrude WELLCO ENERGY INC \$0.00 \$0.00 \$97,337.79 \$0.00 \$0.00 \$585,786.05 SemCrude WELLSTAR CORPORATION \$585,786.05 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$41,694.67 SemCrude WENTWORTH OPERATING COMPANY \$0.00 \$0.00 \$679.31 \$0.00 \$0.00 \$679.31 SemCrude WESLEY J KRUMME \$0.00 \$0.00 \$0.00 \$92,367.56 \$0.00 \$92,367.56 SemCrude WEST TEXAS ROYALTIES INC \$0.00 \$0.00 \$555.99 \$0.00 \$0.00 \$63,078.57 SemCrude WESTSTAR OIL & GAS INC. \$0.00 \$63,078.57 \$0.00 \$63,078.57		WEAVER ENERGY CORPORATION						
SemCrude WELDON CORPORATION \$0.00<								
SemCrude WELDON DELANEY \$0.00 \$0.00 \$13.00 \$0.00 \$97,337.79 SemCrude WELLCO ENERGY INC \$0.00 \$0.00 \$97,337.79 \$0.00 \$0.00 \$585,786.05 SemCrude WELLSTAR CORPORATION \$585,786.05 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$41,694.67 SemCrude WENTWORTH OPERATING COMPANY \$0.00 \$0.00 \$679.31 \$0.00 \$0.00 \$679.31 SemCrude WESLEY J KRUMME \$0.00 \$0.00 \$0.00 \$92,367.56 \$0.00 \$92,367.56 SemCrude WEST TEXAS ROYALTIES INC \$0.00 \$0.00 \$555.99 \$0.00 \$0.00 \$63,078.57 SemCrude WESTSTAR OIL & GAS INC. \$0.00 \$0.00 \$63,078.57 \$0.00 \$0.00 \$63,078.57							-	
SemCrude WELLCO ENERGY INC \$0.00 \$0.00 \$97,357.79 \$0.00 \$585,786.05 SemCrude WELLSTAR CORPORATION \$585,786.05 \$0.00 \$0.00 \$0.00 \$0.00 \$41,694.67 SemCrude WENTWORTH OPERATING COMPANY \$0.00 \$0.00 \$679.31 \$0.00 \$0.00 \$679.31 SemCrude WESLEY J KRUMME \$0.00 \$0.00 \$0.00 \$92,367.56 \$0.00 \$92,367.56 SemCrude WEST TEXAS ROYALTIES INC \$0.00 \$0.00 \$555.99 \$0.00 \$0.00 \$555.99 SemCrude WESTSTAR OIL & GAS INC. \$0.00 \$0.00 \$63,078.57 \$0.00 \$0.00 \$63,078.57		WELDON DELANEY						·
SemCrude WELLSTAR CORPORATION \$585,786.05 \$0.00 \$0.00 \$0.00 \$41,694.67 SemCrude WENTWORTH OPERATING COMPANY \$0.00 \$0.00 \$41,694.67 \$0.00 \$0.00 \$679.31 SemCrude WESLEY J KRUMME \$0.00 \$0.00 \$679.31 \$0.00 \$92,367.56 \$0.00 \$92,367.56 SemCrude WEST TEXAS ROYALTIES INC \$0.00 \$0.00 \$555.99 \$0.00 \$555.99 SemCrude WESTSTAR OIL & GAS INC. \$0.00 \$0.00 \$63,078.57 \$0.00 \$0.00 \$63,078.57	-	WELLCO ENERGY INC					*	
SemCrude WENTWORTH OPERATING COMPANY \$0.00 \$0.00 \$679.31 \$0.00 \$0.00 \$679.31 SemCrude WESLEY J KRUMME \$0.00 \$0.00 \$0.00 \$92,367.56 \$0.00 \$92,367.56 SemCrude WEST TEXAS ROYALTIES INC \$0.00 \$0.00 \$555.99 \$0.00 \$0.00 \$555.99 SemCrude WESTSTAR OIL & GAS INC. \$0.00 \$0.00 \$63,078.57 \$0.00 \$0.00 \$63,078.57		WELLSTAR CORPORATION	•	•				
SemCrude WESLEY J KRUMME \$0.00 <td></td> <td>WENTWORTH OPERATING COMPANY</td> <td></td> <td></td> <td>•</td> <td></td> <td></td> <td></td>		WENTWORTH OPERATING COMPANY			•			
SemCrude WEST TEXAS ROYALTIES INC \$0.00		WESLEY J KRUMME	· · · · · · · · · · · · · · · · · · ·					
SemCrude WESTSTAR OIL & GAS INC. \$0.00 \$0.00 \$333.99 \$0.00 \$63,078.57			· · · · · · · · · · · · · · · · · · ·			,		
\$6.00 \$1181 \$65.076.37 \$0.00 \$0.00								•
	•	WFD OIL CORP	\$0.00	\$0.00	303,076.37	φυ.υυ	\$0.00	**-

	337956267FE65267			irst Purchase Amnu	nts in Active States		
	Operator and sufficient and sufficie	The state of the s	New Mexico	Oklahoma	Texas	Wyoming	Potal Potal
Debtor	Operator WHEELER OIL COMPANY	\$42,026.10	\$0.00	\$0.00	\$0.00	\$0.00	\$42,026.10
SemCrude	WHITDALE INC	\$0.00	\$0.00	\$69,345.40	\$0.00	\$0.00	\$69,345.40
SemCrude	WHITE & ELLIS DRILLING CO	\$29,597.74	\$0.00	\$0.00	\$0.00	\$0.00	\$29,597.74
SemCrude	WHITE EXPLORATION INC	\$1,752,092.22	\$0.00	\$0,00	\$0.00	\$0.00	\$1,752,092.22
SemCrude	WHITE OPERATING CO	\$0.00	\$0.00	\$2,298,226.00	\$0.00	\$0.00	\$2,298,226.00
SemCrude	····	\$197,141.38	\$0.00	\$0.00	\$0.00	\$0.00	\$197,141.38
SemCrude	WHITE PINE PETROLEUM CORP	\$0.00	\$0.00	\$0.00	\$1,858.27	\$0.00	\$1,858.27
SemCrude	WHITE WING PETROLEUM INC.	\$0.00	\$0.00	\$206,732.61	\$0.00	\$0.00	\$206,732.61
SemCrude	WHITEHEAD PRODUCTION COMPANY, INC	\$845.03	\$0.00	\$0.00	\$0.00	\$0.00	\$845.03
SemCrude	WHITETAIL CRUDE INC WHITING OIL AND GAS CORPORATION	\$0.00	\$0.00	\$278,608.09	\$0.00	\$0.00	\$278,608.09
SemCrude	WIBLE LIMITED LIABILITY COMPANY	\$0.00	\$0.00	\$3.97	\$0.00	\$0.00	\$3.97
SemCrude	WILBUR TROY WILSON ESTATE	\$0.00	\$0.00	\$11.823.43	\$0.00	\$0.00	\$11,823.43
SemCrude		\$0.00	\$0.00	\$4,847.62	\$0.00	\$0.00	\$4,847.62
SemCrude	WILD GOOSE OIL LLC	\$6,014.12	\$0.00	\$429,002.59	\$0.00	\$0.00	\$435,016.71
SemCrude	WILDHORSE OIL & GAS CORP	\$0.00	\$0.00	\$20,726.83	\$0.00	\$0.00	\$20,726.83
SemCrude	WILLIAM A SNEED	\$0.00	\$0.00	\$0.00	\$24,286.83	\$0.00	\$24,286.83
SemCrude	WILLIAM B LONG	\$0.00	\$0.00	\$190.07	\$0.00	\$0.00	\$190.07
SemCrude	WILLIAM GLENN COOK	\$0.00	\$0.00	\$37,026.21	\$0.00	\$0.00	\$37,026.21
SemCrude	WILLIAM H ZUHONE	\$0.00	\$0.00	\$119,717.28	\$0,00	\$0.00	\$119,717.28
SemCrude	WILLIAM M PARK WILLIAM R EARNHARDT COMPANY	\$0.00	\$0.00	\$361,938.94	\$0.00	\$0.00	\$361,938.94
SemCrude	WILLIAMS INSPECTION & POLYLINE CO	\$0.00	\$0.00	\$1,243.00	\$0.00	\$0.00	\$1,243.00
SemCrude	WILLIE & SANDRA ELLITHORP	\$0.00	\$0.00	\$1,388.92	\$0.00	\$0.00	\$1,388.92
SemCrude	WILLIFORD ENERGY CO	\$0.00	\$0.00	\$187,812.14	\$0.00	\$0.00	\$187,812.14
SemCrude	WILLIFORD ENERGY CO WIL-R-CO OIL AND GAS COMPANY	\$0.00	\$0.00	\$218.81	\$0.00	\$0.00	\$218.81
SemCrude	WILTON PETROLEUM INC	\$65,024.14	\$0.00	\$0.00	\$0.00	\$0.00	\$65,024.14
SemCrude	WILTON PETROLEON INC WIND RIVER RESOURCES INC	\$0.00	\$0.00	\$7,361.60	\$0.00	\$0.00	\$7,361.60
SemCrude	WIND RIVER RESOURCES INC WITHROW OIL COMPANY INC	\$0.00	\$0.00	\$178.14	\$0.00	\$0.00	\$178.14
SemCrude	WOOLLEY OIL COMPANY	\$0.00	\$0.00	\$1,335.47	\$0.00	\$0.00	\$1,335.47
SemCrude	WOOLSEY PETROLEUM CORPORATION	\$329.99	\$0.00	\$0.00	\$0.00	\$0.00	\$329.99
SemCrude	WYNN-CROSBY 1998 LTD	\$0.00	\$138,074.64	\$0.00	\$0.00	\$0.00	\$138,074.64
SemCrude	XANADU EXPLORATION COMPANY	\$0.00	\$0.00	\$381,409.96	\$0.00	\$0.00	\$381,409.96
SemCrude	XANADO EXPLORATION COMI ANY I	\$0.00	\$0.00	\$0.00	\$94.89	\$0.00	\$94.89
SemCrude		\$0.00	\$0.00	\$0.00	\$551.14	\$0.00	\$551.14
SemCrude	XOPHI ENERGY, INC XTO ENERGY INC	\$0.00	\$99,914.38	\$3,341.56	\$0.00	\$0.00	\$103,255.94
SemCrude	YALE OIL ASSOCIATION INC	\$0.00	\$0.00	\$422.74	\$0.00	\$0.00	\$422.74
SemCrude	ZENERGY INC	\$19,462.71	\$0.00	\$844.75	\$87,481.15	\$0.00	\$107,788.61
SemCrude	ZENITH DRILLING COMPANY	\$198,273.39	\$0.00	\$0.00	\$0.00	\$0.00	\$198,273.39
SemCrude	ZEPHYR OPERATING CO LLC	\$0.00	\$0.00	\$263,066.52	\$0.00	\$0.00	\$263,066.52
SemCrude SemCrude Total	1478 Producers	\$57,490,551.30	\$4,982,044.93	\$124,875,957.08	\$30,685,326.41	\$0.00	\$218,033,879.72
Semerade 10tai	14/0 1 10000013	40.7,10.1,10.	- 7 .				
SemGas	ACCT OF NANCY LOUISE GILL MONFORD	\$0.00	\$0.00	\$0.00	\$106,592.48	\$0.00	\$106,592.48
SemGas SemGas	ALTEX ENERGY CORPORATION	\$0.00	\$0.00	\$541,897.52	\$0.00	\$0.00	\$541,897.52
SemGas SemGas	ALTEX RESOURCES INC	\$0.00	\$0.00	\$2,490,886.57	\$0.00	\$0.00	\$2,490,886.57
SemGas SemGas	AMERICAN WARRIOR INC	\$11,923.24	\$0.00	\$0.00	\$0.00	\$0.00	\$11,923.24
SemGas SemGas	APOLLO ENERGIES INC	\$4,800.69	\$0.00	\$0.00	\$0.00	\$0.00	\$4,800.69
	ARIES RESOURCES LLC	\$0.00	\$0.00	\$0.00	\$36,997.39	\$0.00	\$36,997.39
SemGas	ARROW OIL & GAS INC	\$0.00	\$0.00	\$0.00	\$19,861.29	\$0.00	\$19,861.29
SemGas SemGos	ATOKA OPERATING INC	\$0.00	\$0.00	\$0.00	\$290,913.61	\$0.00	\$290,913.61
SemGas	VIOVV OF EVVIING INC						

SemGas	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$11,019.65 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	in Active States W S68,827.43 \$0.00 \$0.00 \$0.00 \$2,520.57 \$846,436.70 \$1,138,365.00 \$79,590.88 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41 \$0.00	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$68,827.43 \$68,827.43 \$429,268.38 \$3,720.07 \$2,520.57 \$846,436.70 \$1,138,365.00 \$79,590.88 \$11,019.65 \$84.90 \$6,417.88 \$16.08 \$2,902.55 \$1,405,948.00 \$21.39 \$113,082.37 \$18,822.26
Defitor SemGas	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$11,019.65 \$0.00 \$0.00 \$0.00 \$2,902.55 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$68,827.43 \$0.00 \$0.00 \$2,520.57 \$846,436.70 \$1,138,365.00 \$79,590.88 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$68,827.43 \$429,268.38 \$3,720.07 \$2,520.57 \$846,436.70 \$1,138,365.00 \$79,590.88 \$11,019.65 \$84.90 \$6,417.88 \$16.08 \$2,902.55 \$1,405,948.00 \$21.39 \$113,082.37
SemGas BARRON EXPLORATION COMPANT \$429,268.38 \$0.00 SemGas BECKER OIL CORPORATION \$3,720.07 \$0.00 SemGas BEREN CORPORATION \$0.00 \$0.00 SemGas BKM PRODUCTION \$0.00 \$0.00 SemGas BLACK OIL EXPLORATION &PRODUCTION II LTD \$0.00 \$0.00 SemGas BLACK WELL BMC LP \$0.00 \$0.00 SemGas BRECK OPERATING CORP \$0.00 \$0.00 SemGas BRITTANY ENERGY LLC \$0.00 \$0.00 SemGas CAMBRON INVESTMENTS \$84.90 \$0.00 SemGas CARMEN SCHMITT INC \$6,417.88 \$0.00 SemGas CARMEN SCHMITT INC \$16.08 \$0.00 SemGas CARMEN SCHMITT INC \$16.08 \$0.00 SemGas CBL RESOURCES, LLC \$0.00 \$0.00 SemGas CHE RESOURCES, LLC \$0.00 \$0.00 SemGas CHESAPEAKE EXP. LTD PARTNERSHIP \$0.00 \$0.00 SemGas CHESAPEAKE OPERATING INC \$0.00 <	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$11,019.65 \$0.00 \$0.00 \$0.00 \$2,902.55 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$0.00 \$0.00 \$2,520.57 \$846,436.70 \$1,138,365.00 \$79,590.88 \$0.00 \$0.00 \$0.00 \$0.00 \$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$3,720.07 \$2,520.57 \$846,436.70 \$1,138,365.00 \$79,590.88 \$11,019.65 \$84.90 \$6,417.88 \$16.08 \$2,902.55 \$1,405,948.00 \$21.39 \$113,082.37
SemGas BECKER OIL CORPORATION \$3,720.07 \$0.00 SemGas BEREN CORPORATION \$0.00 \$0.00 SemGas BKM PRODUCTION \$0.00 \$0.00 SemGas BLACK OIL EXPLORATION & PRODUCTION II LTD \$0.00 \$0.00 SemGas BLACK WELL BMC LP \$0.00 \$0.00 SemGas BRECK OPERATING CORP \$0.00 \$0.00 SemGas BRITTANY ENERGY LLC \$0.00 \$0.00 SemGas CAMBRON INVESTMENTS \$84.90 \$0.00 SemGas CARMEN SCHMITT INC \$6,417.88 \$0.00 SemGas CASTELLI EXPLORATION INC \$16.08 \$0.00 SemGas CBL RESOURCES, LLC \$0.00 \$0.00 SemGas CHESAPEAKE EXP. LTD PARTNERSHIP \$0.00 \$0.00 SemGas CHESAPEAKE OPERATING INC \$21.39 \$0.00 SemGas CIMAREX ENERGY CO \$0.00 \$0.00 SemGas DANNY MAYO \$18,822.26 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$11,019.65 \$0.00 \$0.00 \$2,902.55 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$0.00 \$2,520.57 \$846,436.70 \$1,138,365.00 \$79,590.88 \$0.00 \$0.00 \$0.00 \$0.00 \$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$2,520.57 \$846,436.70 \$1,138,365.00 \$79,590.88 \$11,019.65 \$84.90 \$6,417.88 \$16.08 \$2,902.55 \$1,405,948.00 \$21.39 \$113,082.37
SemGas BEREN CORPORATION \$0.00 \$0.00 SemGas BKM PRODUCTION \$0.00 \$0.00 SemGas BLACK OIL EXPLORATION & PRODUCTION II LTD \$0.00 \$0.00 SemGas BLACK WELL BMC LP \$0.00 \$0.00 SemGas BRECK OPERATING CORP \$0.00 \$0.00 SemGas BRITTANY ENERGY LLC \$0.00 \$0.00 SemGas CAMBRON INVESTMENTS \$84.90 \$0.00 SemGas CARMEN SCHMITT INC \$6,417.88 \$0.00 SemGas CASTELLI EXPLORATION INC \$16.08 \$0.00 SemGas CBL RESOURCES, LLC \$0.00 \$0.00 SemGas CHESAPEAKE EXP. LTD PARTNERSHIP \$0.00 \$0.00 SemGas CHESAPEAKE OPERATING INC \$21.39 \$0.00 SemGas CIMAREX ENERGY CO \$0.00 \$0.00 SemGas DANNY MAYO \$0.00 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$0.00 \$0.00 \$0.00 \$0.00 \$11,019.65 \$0.00 \$0.00 \$2,902.55 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$2,520.57 \$846,436.70 \$1,138,365.00 \$79,590.88 \$0.00 \$0.00 \$0.00 \$0.00 \$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$2,520.57 \$846,436.70 \$1,138,365.00 \$79,590.88 \$11,019.65 \$84.90 \$6,417.88 \$16.08 \$2,902.55 \$1,405,948.00 \$21.39 \$113,082.37
SemGas BKM PRODUCTION \$0.00 \$0.00 SemGas BLACK OIL EXPLORATION & PRODUCTION II LTD \$0.00 \$0.00 SemGas BLACK WELL BMC LP \$0.00 \$0.00 SemGas BRECK OPERATING CORP \$0.00 \$0.00 SemGas BRITTANY ENERGY LLC \$0.00 \$0.00 SemGas CAMBRON INVESTMENTS \$84.90 \$0.00 SemGas CARMEN SCHMITT INC \$6,417.88 \$0.00 SemGas CARMEN SCHMITT INC \$16.08 \$0.00 SemGas CASTELLI EXPLORATION INC \$16.08 \$0.00 SemGas CHESAPEAKE EXP. LTD PARTNERSHIP \$0.00 \$0.00 SemGas CHESAPEAKE OPERATING INC \$21.39 \$0.00 SemGas CIMAREX ENERGY CO \$0.00 \$0.00 SemGas DANNY MAYO \$18,822.26 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.0	\$0.00 \$0.00 \$0.00 \$11,019.65 \$0.00 \$0.00 \$2,902.55 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$846,436.70 \$1,138,365.00 \$79,590.88 \$0.00 \$0.00 \$0.00 \$0.00 \$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$1,138,365.00 \$79,590.88 \$11,019.65 \$84.90 \$6,417.88 \$16.08 \$2,902.55 \$1,405,948.00 \$21.39 \$113,082.37
SemGas BLACK OIL EAPLORATION & FRODOCTION IP TO \$0.00 \$0.00 SemGas BLACK WELL BMC LP \$0.00 \$0.00 SemGas BRECK OPERATING CORP \$0.00 \$0.00 SemGas BRITTANY ENERGY LLC \$0.00 \$0.00 SemGas CAMBRON INVESTMENTS \$84.90 \$0.00 SemGas CARMEN SCHMITT INC \$6,417.88 \$0.00 SemGas CARMEN SCHMITT INC \$16.08 \$0.00 SemGas CASTELLI EXPLORATION INC \$16.08 \$0.00 SemGas CBL RESOURCES, LLC \$0.00 \$0.00 SemGas CHESAPEAKE EXP. LTD PARTNERSHIP \$0.00 \$0.00 SemGas CHESAPEAKE OPERATING INC \$21.39 \$0.00 SemGas CIMAREX ENERGY CO \$0.00 \$0.00 SemGas DANNY MAYO \$18,822.26 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$	\$0.00 \$0.00 \$11,019.65 \$0.00 \$0.00 \$0.00 \$2,902.55 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$1,138,365.00 \$79,590.88 \$0.00 \$0.00 \$0.00 \$0.00 \$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$79,590.88 \$11,019.65 \$84.90 \$6,417.88 \$16.08 \$2,902.55 \$1,405,948.00 \$21.39 \$113,082.37
SemGas BLACK WELL BMC LP \$0.00 \$0.00 SemGas BRECK OPERATING CORP \$0.00 \$0.00 SemGas BRITTANY ENERGY LLC \$0.00 \$0.00 SemGas CAMBRON INVESTMENTS \$84.90 \$0.00 SemGas CARMEN SCHMITT INC \$6,417.88 \$0.00 SemGas CASTELLI EXPLORATION INC \$16.08 \$0.00 SemGas CBL RESOURCES, LLC \$0.00 \$0.00 SemGas CHESAPEAKE EXP. LTD PARTNERSHIP \$0.00 \$0.00 SemGas CHESAPEAKE OPERATING INC \$21.39 \$0.00 SemGas CIMAREX ENERGY CO \$0.00 \$0.00 SemGas CMX INC \$18,822.26 \$0.00 SemGas DANNY MAYO \$0.00 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$0.00 \$11,019.65 \$0.00 \$0.00 \$0.00 \$2,902.55 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$79,590.88 \$0.00 \$0.00 \$0.00 \$0.00 \$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$11,019.65 \$84.90 \$6,417.88 \$16.08 \$2,902.55 \$1,405,948.00 \$21.39 \$113,082.37
SemGas BRECK OPERATING CORP \$0.00 \$0.00 SemGas BRITTANY ENERGY LLC \$0.00 \$0.00 SemGas CAMBRON INVESTMENTS \$84.90 \$0.00 SemGas CARMEN SCHMITT INC \$6,417.88 \$0.00 SemGas CASTELLI EXPLORATION INC \$16.08 \$0.00 SemGas CBL RESOURCES, LLC \$0.00 \$0.00 SemGas CHESAPEAKE EXP. LTD PARTNERSHIP \$0.00 \$0.00 SemGas CHESAPEAKE OPERATING INC \$21.39 \$0.00 SemGas CIMAREX ENERGY CO \$0.00 \$0.00 SemGas CMX INC \$18,822.26 \$0.00 SemGas DANNY MAYO \$0.00 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$11,019.65 \$0.00 \$0.00 \$0.00 \$2,902.55 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$84.90 \$6,417.88 \$16.08 \$2,902.55 \$1,405,948.00 \$21.39 \$113,082.37
SemGas BRIT TANY ENERGY LLC \$84.90 \$0.00 SemGas CAMBRON INVESTMENTS \$6,417.88 \$0.00 SemGas CARMEN SCHMITT INC \$16.08 \$0.00 SemGas CASTELLI EXPLORATION INC \$16.08 \$0.00 SemGas CBL RESOURCES, LLC \$0.00 \$0.00 SemGas CHESAPEAKE EXP. LTD PARTNERSHIP \$0.00 \$0.00 SemGas CHESAPEAKE OPERATING INC \$21.39 \$0.00 SemGas CIMAREX ENERGY CO \$0.00 \$0.00 SemGas CMX INC \$18,822.26 \$0.00 SemGas DANNY MAYO \$0.00 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$0.00 \$0.00 \$0.00 \$2,902.55 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$0.00 \$0.00 \$0.00 \$0.00 \$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$6,417.88 \$16.08 \$2,902.55 \$1,405,948.00 \$21.39 \$113,082.37
SemGas CAMBRON INVESTMENTS \$6,417.88 \$0.00 SemGas CARMEN SCHMITT INC \$16.08 \$0.00 SemGas CASTELLI EXPLORATION INC \$16.08 \$0.00 SemGas CBL RESOURCES, LLC \$0.00 \$0.00 SemGas CHESAPEAKE EXP. LTD PARTNERSHIP \$0.00 \$0.00 SemGas CHESAPEAKE OPERATING INC \$21.39 \$0.00 SemGas CIMAREX ENERGY CO \$0.00 \$0.00 SemGas CMX INC \$18,822.26 \$0.00 SemGas DANNY MAYO \$0.00 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$0.00 \$0.00 \$2,902.55 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$0.00 \$0.00 \$0.00 \$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$16.08 \$2,902.55 \$1,405,948.00 \$21.39 \$113,082.37
SemGas CARMEN SCHMITTINC \$16.08 \$0.00 SemGas CASTELLI EXPLORATION INC \$0.00 \$0.00 SemGas CBL RESOURCES, LLC \$0.00 \$0.00 SemGas CHESAPEAKE EXP. LTD PARTNERSHIP \$0.00 \$0.00 SemGas CHESAPEAKE OPERATING INC \$21.39 \$0.00 SemGas CIMAREX ENERGY CO \$0.00 \$0.00 SemGas CMX INC \$18,822.26 \$0.00 SemGas DANNY MAYO \$0.00 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$25,140.12 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$0.00 \$2,902.55 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$0.00 \$0.00 \$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$2,902.55 \$1,405,948.00 \$21.39 \$113,082.37
SemGas CASTELLI EXPLORATION INC \$0.00 \$0.00 SemGas CBL RESOURCES, LLC \$0.00 \$0.00 SemGas CHESAPEAKE EXP. LTD PARTNERSHIP \$0.00 \$0.00 SemGas CHESAPEAKE OPERATING INC \$21.39 \$0.00 SemGas CIMAREX ENERGY CO \$0.00 \$0.00 SemGas CMX INC \$18,822.26 \$0.00 SemGas DANNY MAYO \$0.00 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$25,140.12 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$2,902.55 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$0.00 \$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00 \$0.00	\$1,405,948.00 \$21.39 \$113,082.37
SemGas CBL RESOURCES, LEC \$0.00 \$0.00 SemGas CHESAPEAKE EXP. LTD PARTNERSHIP \$0.00 \$0.00 SemGas CHESAPEAKE OPERATING INC \$0.00 \$0.00 SemGas CIMAREX ENERGY CO \$18,822.26 \$0.00 SemGas CMX INC \$18,822.26 \$0.00 SemGas DANNY MAYO \$0.00 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$25,140.12 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$1,405,948.00 \$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00	\$21.39 \$113,082.37
SemGas CHESAPEARE EXP. LTD PARTNERSHIT \$0.00 SemGas CHESAPEAKE OPERATING INC \$21.39 \$0.00 SemGas CIMAREX ENERGY CO \$0.00 \$0.00 SemGas CMX INC \$18,822.26 \$0.00 SemGas DANNY MAYO \$0.00 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$25,140.12 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$0.00 \$0.00 \$0.00 \$0.00 \$0.00	\$0.00 \$113,082.37 \$0.00 \$42,425.41	\$0.00 \$0.00 \$0.00	\$113,082.37
SemGas CHESAPEARE OPERATING INC \$0.00 \$0.00 SemGas CIMAREX ENERGY CO \$18,822.26 \$0.00 SemGas CMX INC \$0.00 \$0.00 SemGas DANNY MAYO \$0.00 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$25,140.12 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$0.00 \$0.00 \$0.00 \$0.00	\$113,082.37 \$0.00 \$42,425.41	\$0.00	
SemGas CIMAREX ENERGY CO \$18,822.26 \$0.00 SemGas CMX INC \$0.00 \$0.00 SemGas DANNY MAYO \$0.00 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$25,140.12 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$0.00 \$0.00 \$0.00	\$0.00 \$42,425.41		
SemGas CMX INC \$0.00 \$0.00 SemGas DANNY MAYO \$0.00 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$25,140.12 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$0.00 \$0.00	\$42,425.41	£0.00	\$10,044.20
SemGas DANNY MAYO \$7,290.59 \$0.00 SemGas DAYSTAR PETROLEUM INC \$7,290.59 \$0.00 SemGas DEUTSCH OIL COMPANY \$25,140.12 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$0.00	- '	20.00	\$42,425.41
SemGas DAYSTAR PETROLEGISTING \$0.00 SemGas DEUTSCH OIL COMPANY \$25,140.12 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00	•		\$0.00	\$7,290.59
SemGas DEUTSCHOIL COMPANY \$0.00 \$0.00 SemGas DOOLAREE CORPORATION \$0.00 \$0.00 \$0.00 \$0.00 \$0.00	30.00	\$0.00	\$0.00	\$25,140.12
SemGas DOOLAREE CORPORATION \$0.00 \$0.00	\$0.00	\$126,508.89	\$0.00	\$126,508.89
M) (R) MOUNT OF COMPANY	\$0.00	\$18,114.40	\$0.00	\$18,114,40
SemGas EAGLE OIL & GAS COMPANY	\$0.00	\$0.00	\$0.00	\$48,961.69
SemGas EDMISTON OIL CO INC \$48,961.69 \$0.00 \$0.00 \$0.00	\$0.00	\$44.79	\$0.00	\$44.79
SemGas ELIAS THOMAS	\$0.00	\$19,633.61	\$0.00	\$19,633.61
SemGas ENMARK GAS GAT FIERING ET	\$0.00	\$0.00	\$0.00	\$285,466.62
SemGas F G HOLL COMPANY LLC	\$0.00	\$49,046.39	\$0.00	\$49,046.39
SemGas FELDERHOFF PRODUCTION COMPANY \$0.00 \$0.00 \$0.00 \$0.00	\$0.00	\$0.00	\$0.00	\$11,442.43
SemGas FRED J NOSS	\$0.00	\$474.36	\$0.00	\$474.36
SemGas GRAYSON PETROLEON CONFANT	\$0.00	\$1,787.92	\$0.00	\$1,787.92
SemGas HARVEY CLARK	\$0.00	\$530,604.26	\$0.00	\$530,604.26
SemGas HEAPE EXPLORATION INC	\$0.00	\$56,100.68	\$0.00	\$56,100.68
SemGas HEP OIL COMPANY SO ON	\$0.00	\$0.00	\$0.00	\$844.07
SemGas HERMAN L LUEB	\$0.00	\$0.00	\$0.00	\$24,608.43
SemGas HUTCHINSON OIL CO	\$0.00	\$0.00	\$0,00	\$24,654.69
SemGas INDIAN OIL CO INC 324,034.09	\$0.00	\$16,774.57	\$0.00	\$16,774.57
SemGas JETTA OPERATING COMPANY INC	\$0.00	\$0.00	\$0.00	\$12,611.30
SemGas JUE PARR	\$0.00	\$0.00	\$0.00	\$8,072.28
SemGas JOHN O FARMER INC	\$0.00	\$0.00	\$0.00	\$6,624.89
SemGas JRC PRODUCTION, ELC	\$69,401.50	\$517.92	\$0.00	\$69,919.42
SemGas KEPCO INC	\$0.00	\$0.00	\$0.00	\$8,043.08
SemGas KEY GAS HOLDING LLC	\$0.00	\$0.00	\$0.00	\$20,157.66
SemGas KNIGHTUN OIL COMPANT INC	\$0.00	\$0.00	\$0.00	\$17,799.14
SemGas L D DRILLING INC	\$0.00	\$0.00	\$0.00	\$63,100.06
SemGas LUMEN MIDSTREAM PARTNERSHIP LLC	\$0.00	\$0.00	\$0.00	\$1,131,453.52
ScmGas MCCOY PETROLEUM CORPORATION \$1,131,453.52 50.00	\$0.00	\$0.00	\$0.00	\$7,000.39
SemGas MID CONTINENT ENERGY CORP \$7,000.39 \$0.00	\$0.00	\$0.00	\$0.00	\$13,904.90
SemGos MID-CONTINENT ENERGY CORP \$13,904.90 \$0.00	\$0.00	\$0.00	\$0.00	\$4,950.04
SemGas MTM PETROLEUM INC \$4,950,04 \$0.00	\$0.00	\$0.00	\$0.00	\$17,553.87
SemGas NORTHSTAR INVESTMENTS, INC. \$17,553.87 \$0.00	\$0.00	20,00	*****	•

				irst Purchase Amount			
Debton	Operator	Kansas				(1) (1) (1) (1)	Total
SemGas	OIL PRODUCERS OF KANSAS INC	\$5,634.31	\$0.00	\$0.00	\$0.00	\$0.00	\$5,634.31
SemGas	OKT PETROLEUM COMPANY INC	\$0.00	\$0.00	\$0.00	\$106,026.91	\$0.00	\$106,026.91
SemGas	PATRON ENERGY LLC	\$0.00	\$0.00	\$4,840.18	\$0.00	\$0.00	\$4,840.18
SemGas	PERRY LYNN LARSON	\$0.00	\$0.00	\$0.00	\$48,319.80	\$0.00	\$48,319.80
SemGas	PETROCORP INC	\$0.00	\$0.00	\$0.00	\$49,267.84	\$0.00 \$0.00	\$49,267.84 \$20,885.74
SemGas	PETROLEUM TECHNOLOGIES INC	\$20,885.74	\$0.00	\$0.00 \$0.00	\$0.00 \$0,00	\$0.00	\$36,826.32
SemGas	PINTAIL PETROLEUM LTD	\$36,826.32	\$0.00		\$0.00	\$0.00	\$12,633.53
SemGas	PRATT WELL SERVICE INC	\$12,633.53	\$0.00	\$0.00		\$0.00	\$7,433.17
SemGas	PRECISION OPERATING INC	\$0.00	\$0.00	\$0.00	\$7,433.17 \$0.00	\$0.00	\$20,793.40
SemGas	RAMA OPERATING CO INC	\$20,793.40	\$0.00	\$0.00	*		•
SemGas	REDWING GAS SYSTEMS, INC	\$3,827.35	\$0.00	\$0.00	\$0.00	\$0.00	\$3,827.35
SemGas	RENEGADE RESOURCES CORP.	\$1,474.74	00.02	\$0.00	\$0.00	\$0.00	\$1,474.74
SemGas	RITCHIE EXPLORATION INC	\$1,299.82	\$0.00	\$0.00	\$0.00	\$0.00	\$1,299.82
SemGas	ROBERT B ROWLAND	\$0.00	\$0.00	00.02	\$21,162.00	\$0.00	\$21,162.00
SemGas	RONALD PRATER TRUST	\$9,602.38	\$0.00	\$0.00	\$0.00	\$0.00	\$9,602.38
SemGas	SAND SPRINGS OIL & GAS CO	\$0.00	\$0.00	\$0.00	\$18,603.43	\$0.00	\$18,603.43
SemGas	SANDRIDGE ENERGY INC	\$0.00	\$0.00	\$118,654.66	\$0.00	\$0.00	\$118,654.66
SemGas	SIROKY OIL MANAGEMENT	\$25,181.38	\$0.00	\$0.00	\$0.00	\$0.00	\$25,181.38
SemGas	SPECIAL ENERGY CORPORATION	\$0.00	\$0.00	\$4,963,350.47	\$0.00	\$0.00	\$4,963,350.47
SemGas	STARLIGHT RIDGE COMPANY	\$0.00	\$0.00	\$0.00	\$3,765.49	\$0.00	\$3,765.49
SemGas	STEVE JONES	\$3,886.74	\$0.00	\$0.00	\$0.00	\$0.00	\$3,886.74
SemGas	THE OUINTIN LITTLE COMPANY	\$0.00	\$0.00	\$0.00	\$123,284.74	\$0.00	\$123,284.74
SemGas	TRANSPACIFIC OIL CORP.	\$13,146.29	00.02	\$0.00	\$0.00	\$0.00	\$13,146.29
SemGas	TROY JACO INC	\$0.00	\$0.00	\$0.00	\$192,340.82	\$0.00	\$192,340.82
SemGas SemGas	VINCENT OIL CORPORATION	\$50,084.22	\$0.00	\$0.00	\$0.00	\$0.00	\$50,084.22
SemGas SemGas	VII NATURAL RESOURCE INC	\$8,187.95	\$0.00	\$0.00	\$0.00	\$0.00	\$8,187.95
SemGas SemGas	WESTAR ENERGY CORP.	\$3,697.06	\$0.00	\$0.00	\$0.00	\$0.00	\$3,697.06
*	WHITE & ELLIS DRILLING INC	\$19,481.05	\$0.00	\$0.00	\$0.00	\$0.00	\$19,481.05
SemGas	WW OIL & GAS LLC	\$17,257.32	\$0.00	\$0.00	\$0.00	\$0.00	\$17,257.32
SemGas		\$0.00	\$0.00	\$0.00	\$160,402.57	\$0.00	\$160,402.57
SemGas	YOUNGER ENERGY COMPANY	\$14,859.01	\$0.00	\$0.00	\$0.00	\$0.00	\$14,859.01
SemGas	ZENITH DRILLING COMPANY 87 Producers	\$2,483,513.27	\$0.00	\$8,202,953.10	\$5,701,775.69	\$0.00	\$16,388,242.06
SemGas Total	0/ Flouuceis	OMITORIA E DIMIT	2,110				
TOTAL	1530 Producers	\$130,099,844.40	\$10,522,020.08	\$142,688,864.85	\$55,287,836.04	\$1,634,114.70	5340,232,680.07

Amounts are for goods sold to Debtors from June 1, 2008 to July 21, 2008 and include amounts for the First Purchaser Producer Twenty-Day Claims. To calculate the amount of each claimant's Secured First Purchaser Producer Claim, the amount paid under the Plan for that claimant's First Purchaser Producer Twenty-Day Claim, if any, must be subtracted from the amount listed herein. Moreover, certain of the Claims listed herein may include amounts held in suspense, which will be treated as General Unsecured Claims under the Plan, and will be subtracted from amounts listed herein.

Schedule 3

		I consequence to the consequence of the consequence	UNDERSTRUKTUREN SERVIK SER
		503(b)(9) Disputed Amount Propose	d Senlement Amount
Debtor	Claimant	\$1,271,797.53	\$0.00
EaglWing	CENTRAL CRUDE CORP.	\$133,037.10	\$0.00
EaglWing	GULFMARK ENERGY, INC.	\$24,875.25	\$16,417.67
EaglWing	QUESTAR EXPL & PROD CO	\$160,676.20	\$106,046.29
EaglWing	VALERO MARKETING AND SUPPLY CO.	\$23,786.59	\$15,699.15
EaglWing	ZENERGY INC 5 Claimants	\$1,614,172.67	\$138,163.11
Total EaglWing	2 Claimancs		
		\$67,339.21	\$44,443.88
SemCrude	3M PRODUCTION, INC.	\$25,938.70	\$17,119.54
SemCrude	APACHE CORPORATION	\$115,365.83	\$76,141.45
SemCrude	ATLAS PIPELINE MID-CONTINENT LLC	\$26,072.63	\$17,207.94
SemCrude	ATOKA MIDSTREAM LLC	\$20,228.30	\$13,350.68
SemCrude	B & R PRODUCTION	\$4,336,378.95	\$2,862,010.11
SemCrude	BARR	\$174,364.95	\$115,080.87
SemCrude	BARRY SPARKS OIL CO	\$40,223.34	\$26,547.40
SemCrude	BELPORT OIL INC	\$20,655.39	\$13,632.56
SemCrude	BEREXCO INC	\$18,614.08	\$12,285.29
SemCrude	BLACKCAT OIL COMPANY	\$21,523.69	\$14,205.64
SemCrude	BLUE BONNET ENERGY CORP	\$23,966.92	\$15,818.17
SemCrude	BRECK OPERATING COMPANY	\$68,186.22	\$45,002.90
SemCrude	C&J TRUCKS, INC	\$212,440.86	\$140,210.97
SemCrude	CEN TRANS CENTERPOINT ENERGY FIELD SERVICES INC	\$62,798.05	\$41,446.71
SemCrude	CENTERPOINT ENERGY FIELD SERVICES INC	\$14,964.79	\$9,876.76
SemCrude		\$10,961,910.77	\$8,850,000.00
SemCrude	CENTRAL CORP	\$8,086,884.89	\$5,337,344.03
SemCrude	CENTRAL KANSAS	\$1,353,618.80	\$893,388.41
SemCrude	CHESAPEAKE OPERATING INC	\$273,043.52	\$180,208.72
SemCrude	CHEVRON TEXACO	\$0.00	\$0.00
SemCrude	CHEVRON PRODUCTS INC	\$640,705.69	\$422,865.75
SemCrude	CIMA ENERGY	\$0.00	\$0.00
SemCrude	CIMMARON GATHERING LP	\$8,124,188.43	\$5,361,964.36
SemCrude	CMT	\$37,202.72	\$24,553.80
SemCrude	COFFEYVILLE RES		

	(Listed by b		
	Claimant	503(b)(9) Disputed Amount 1 Propo	sed Settlement Amount
Debtor SemCrude	COLUMBUS OIL COMPANY	\$20,600.20	\$13,596.13
SemCrude SemCrude	CONSOLIDATED PIPE & SUPPLY	\$164,607.12	\$108,640.70
	CROSS BAR ENERGY, LLC	\$365,763.78	\$241,404.09
SemCrude	DEKA EXPLORATION INC	\$126,430.86	\$83,444.37
SemCrude	DON E BROWN	\$31,886.93	\$21,045.37
SemCrude	DORADO OIL	\$401,063.90	\$264,702.17
SemCrude SemCrude	DYNEGY MSI	\$1,939,927.54	\$1,280,352.18
	EAGLE ROCK	\$3,889,824.47	\$2,567,284.15
SemCrude	EAGLE ROCK OPERATING LP	\$50,863.17	\$33,569.69
SemCrude	ELITE RESOURCES INC	\$21,754.89	\$14,358.23
SemCrude	ENCANA MARKETING (USA) INC	\$110,050.15	\$72,633.10
SemCrude SemCrude	ENERFIN RESOURCES I LIMITED PARTNERSHIP	\$191,656.96	\$126,493.59
SemCrude SemCrude	ENOGEX PRODUCTS CORPORATION	\$175,444.72	\$115,793.52
SemCrude SemCrude	ENTERPRISE FIELD SERVICES LLC	\$53,074.55	\$35,029.20
	ENVIROSAFE LOADING SYS.	\$230.76	\$152.30
SemCrude	FAGADAU ENERGY CORP	\$21,814.52	\$14,397.58
SemCrude	GAMMA RESOURCES INC	\$19,796.36	\$13,065.60
SemCrude	GREEN OPERATING COMPANY	\$19,822.27	\$13,082.70
SemCrude	GREG SPARKS	\$105,072.12	\$69,347.60
SemCrude	HEADINGTON OIL CO	\$141,990.46	\$93,713.70
SemCrude	HILAND PARTNERS	\$129,961.26	\$85,774.43
SemCrude	HUSKY MARKETING	\$1,818,416.00	\$1,200,154.56
SemCrude	HUTCHINSON OIL COMPANY	\$124,180.81	\$81,959.33
SemCrude	JIM SNYDER	\$20,602.46	\$13,597.62
SemCrude SemCrude	KLO	\$73,577.64	\$48,561.24
	KRUMME OIL COMPANY	\$68,225.30	\$45,028.70
SemCrude	LANDERS AND MUSGROVE OIL	\$41,911.61	\$27,661.66
SemCrude	LINN ENERGY HOLDINGS	\$513,855.66	\$339,144.74
SemCrude	LIONEL HARRIS OIL CO.	\$1,417.46	\$935.52
SemCrude	MACLASKEY	\$5,031,318.45	\$3,320,670.18
SemCrude	MARJO OPERATING CO INC	\$844,701.40	\$557,502.92
SemCrude	MARJO OPERATING CO INC MARJO OPERATING CO INC	\$40,246.79	\$26,562.88
SemCrude	MARIO OPERATING CO INC MARK L SHIDLER INC	\$23,244.46	\$15,341.34
SemCrude	MERCURIA	\$796,536.82	\$525,714.30
SemCrude	MIDSTREAM ENERGY SERVICES LLC	\$127,303.91	\$84,020.58
SemCrude	MITING LUCKIAL EMERGI OPKAICED DDC		

		SERVICE OF THE PROPERTY OF THE	
Debtor	Clamant	503(b)(9) Disputed Amount 1 Pr	oposed Settlement Amount
SemCrude	MORAN OIL ENTERPRISES	\$21,443.90	\$14,152.97
SemCrude	MUSTANG GAS PRODUCTS LLC	\$444,126.77	\$293,123.67
SemCrude	NAVAJO	\$489,589.67	\$323,129.18
SemCrude	NEWPARK ENVIRONMENTAL SVCS	\$29,404.86	\$19,407.21
SemCrude	NOBLE	\$17,818,576.79	\$11,760,260.68
SemCrude	PERRY L LARSON OPERATING CO	\$88,741.15	\$58,569.16
SemCrude	PETRO SC	\$492,139.26	\$324,811.91
SemCrude	PHOENIX PETROCORP INC	\$63,527.05	\$41,927.85
SemCrude	PIONEER NATURAL	\$12,262,520.87	\$8,093,263.77
SemCrude	PORTER OIL AND GAS INC	\$24,745.14	\$16,331.79
SemCrude	PRATER OIL & GAS OPERATIONS INC	\$20,268.20	\$13,377.01
SemCrude	PRIMARY NATURAL RESOURCES	\$25,196.49	\$16,629.68
SemCrude	PRIMEXX OPERATING CORPORATION	\$151,713.42	\$100,130.86
SemCrude	PRODUCERS CRUDE	\$153,883.78	\$101,563.29
SemCrude	REGENCY GAS SERVICES LLC	\$234,065.45	\$154,483.20
SemCrude	RKR EXPLORATION INC	\$65,179.82	\$43,018.68
SemCrude	ROGER DAN MORRIS OIL	\$19,734.65	\$13,024.87
SemCrude	RUDE CRUDE	\$43,967.37	\$29,018.46
SemCrude	SAGE OIL COMPANY	\$20,036.40	\$13,224.02
SemCrude	SANDY PETROLEUM	\$21,227.93	\$14,010.43
SemCrude	SHELL TRADING	\$4,687,127.78	\$3,093,504.33
SemCrude	SHORT OIL COMPANY	\$20,608.40	\$13,601.54
SemCrude	SOUTHCREEK PETROLEUM COMPANY	\$84,413.13	\$55,712.67
SemCrude	SOUTHERN UNION GAS ENERGY LTD	\$2,020,897.65	\$1,333,792.45
SemCrude	SOUTHERN UNION GAS ENERGY LTD	\$1,690,395.88	\$1,115,661.28
SemCrude	SPECIAL ENERY	\$1,284,864.47	\$848,010.55
SemCrude	STATEWIDE CRUDE	\$1,054,291.39	\$695,832.32
SemCrude	SUNDANCE OIL & GAS LLC	\$20,900.65	\$13,794.43
SemCrude	TAYLOR GAS	\$136,517.95	\$90,101.84
SemCrude	TRAYNOR OIL AND GAS INC	\$0.00	\$0.00
SemCrude	TEXON LP	\$1,268,103.60	\$836,948.38
SemCrude	TIDAL ENERGY	\$2,684,593.46	\$1,771,831.68
SemCrude	TIMBERLAND GATHERING AND PROCESSING CO	\$14,797.11	\$9,766.09
SemCrude	WARD WILLISTON CO.	\$1,324,318.38	\$874,050.13
SemCrude	WENTWORTH OPERATING COMPANY	\$21,329.12	\$14,077.22

	an assertions and approximation of the companies of the c		
	Clamant	503(b)(9) Disputed Amounti	Proposed Settlement Amount
Debtor	WFD OIL CORP	\$184,134.15	\$121,528.54
SemCrude	WHEELER ENERGY COMPANY	\$47,790.23	\$31,541.55
SemCrude	WHITING OIL AND GAS CORPORATION	\$9,207.14	\$6,076.71
SemCrude	WILLIAMS FIELD	\$352,969.56	\$232,959.91
SemCrude SemCrude Total	99 Claimants	\$101,631,141.50	\$68,691,692.28
Semcrude Total), Oamana	,	
SemFuel	AG-LAND FS INC	\$40,159.70	\$26,505.40
SemFuel	AIR BP AVIATION SERVICES	\$34,983.63	\$23,089.20
SemFuel	MARATHON PETROLEUM CO LLC	\$422,519.99	\$278,863.19
SemFuel	CENTER MARKETING CO	\$135,638.90	\$89,521.67
SemFuel	CHEVRON PRODUCTS INC	\$0.00	\$0.00
SemFuel	CITGO PETROLEUM CORP	\$290,708.90	\$191,867.87
SemFuel	EXXON MOBIL OIL CORP	\$48,165.53	\$31,789.25
SemFuel	GARROW OIL CORP	\$13,588.37	\$8,968.32
SemFuel	GLENCORE LTD	\$3,504,655.35	\$2,313,072.53
SemFuel	HARTLAND FUELS PRODUCTS LLC	\$272,376.69	\$179,768.62
SemFuel	HOFMEISTER OIL CO LLC	\$7,007.64	\$4,625.04
SemFuel	HOPSON OIL CO INC	\$4,601.85	\$3,037.22
SemFuel	LANMAN OIL CO	\$48,850.73	\$32,241.48
SemFuel	MIDWEST FUELS	\$11,745.86	\$7,752.27
SemFuel	MORGAN STANLEY CAPITAL GROUP INC	\$3,575,545.43	\$2,359,859.98
SemFuel	MURPHY OIL USA INC	\$788,071.58	\$520,127.24
SemFuel	NELSON OIL CO INC	\$3,350.80	\$2,211.53
SemFuel	WARREN OIL CO	\$2,737.01	\$1,806.43
SemFuel	WEBER OIL CO INC	\$293.70	\$193.84
Total SemFuel	19 Claimants	\$9,205,001.66	\$6,075,301.10
			#0.00
SemGas	B.M. ENERGY INC.	\$5,677.44	\$0.00
SemGas	BC OPERATING INC	\$6,796.48	\$4,485.67
SemGas	BEREN CORP	\$0.00	\$0.00
SemGas	BETTIS BOYLE & STOVALL	\$7,011.96	\$0.00
SemGas	BKM PRODUCTION	\$197,630.34	\$0.00
SemGas	BRAMMER PETROLEUM INC	\$199.21	\$0.00
SemGas	BRECK OPERATING CORP	\$8,549.36	\$0.00
SemGas	BRIDWELL OIL COMPANY	\$6,307.92	\$0.00

Debtor	Claimant	503(b)(9) Disputed Amountl	Proposed Settlement Amount
SemGas	CHARTER OAK PRODUCTION CO LLC	\$92,040.49	\$0.00
SemGas	CHEETAH GAS CO LTD	\$1,397.53	\$0.00
SemGas	CHESAPEAKE ENERGY MARKETING INC	\$773,831.76	\$0.00
SemGas	CHESAPEAKE ENERGY MARKETING INC	\$4,494.27	\$0.00
SemGas	CHESAPEAKE EXP. LTD PARTNERSHIP	\$33,287.81	\$0.00
SemGas	CHOLLA PETROLEUM INC	\$60,069.15	\$0.00
SemGas	CIMMARRON TRANSPORTATION CO, LLC	\$1,044,308.37	\$689,243.53
SemGas	ENBRIDGE GATHERING (NORTH TEXAS) LP	\$104,880.35	\$0.00
SemGas	GRAYSON PETROLEUM COMPANY	\$15,187.61	\$0.00
SemGas	HEP OIL COMPANY	\$40,628.75	\$0.00
SemGas	HEP OIL COMPANY, LTD FOR ACCT OF	\$134,187.79	\$0.00
SemGas	INDIAN OIL COMPANY INC	\$1,197.52	\$0.00
SemGas	M-B OPERATING CO INC	\$98,850.02	\$0.00
SemGas	PRECISION OPERATING INC	\$3,375.68	\$2,227.95
SemGas	THORPE OIL CORP LLC	\$6,288.18	\$0.00
SemGas	TROY JACO INC	\$5,679.32	\$0.00
SemGas Total	24 Claimants	\$2,651,877.32	\$695,957.15
SemMaterials LP	ADVANCE CHEMICAL- TN	\$1,858.98	\$1,226.93
SemMaterials LP	AKZO NOBEL CATALYSTS LLC	\$327,999.40	\$216,479.60
SemMaterials LP	AMERICAN REFINING GROUP INC	\$73,781.82	\$48,696.00
SemMaterials LP	APEX OIL COMPANY, INC.	\$27,560.43	\$18,189.88
SemMaterials LP	ARIZONA CHEMICAL	\$19,341.40	\$12,765.32
SemMaterials LP	ARIZONA CHEMICAL CO	\$38,708.60	\$25,547.68
SemMaterials LP	ARRMAZ PRODUCTS	\$43,635.60	\$28,799.50
SemMaterials LP	ARR-MAZ PRODUCTS	\$50,743.42	\$33,490.66
SemMaterials LP	ATMOS ENERGY MARKETING	\$175,801.26	\$116,028.83
SemMaterials LP	BAROID DRILLING FLUIDS INC.	\$825.40	\$544.76
SemMaterials LP	BARRETT OIL	\$68,477.87	\$45,195.39
SemMaterials LP	BASF CORPORATION	\$731,575.70	\$482,839.96
SemMaterials LP	BASIC CHEMICAL SOLUTIONS LLC	\$8,681.94	\$5,730.08
SemMaterials LP	BELLMAN OIL CO INC	\$87,319.02	\$57,630.55
SemMaterials LP	BENTON OIL CO	\$59,747.04	\$39,433.05
SemMaterials LP	BLUE RIDGE PAPER PRODUCTS	\$26,530.35	\$17,510.03
SemMaterials LP	BOMINFLOT ATLANTIC LLC	\$837,014.89	\$552,429.83

Debtör	Claimant	503(b)(9) Disputed Amount Propos	sed Settlement Amount
SemMaterials LP	BP ENERGY CO. (WASATCH)	\$172,689.56	\$113,975.11
SemMaterials LP	BP NORTH AMERICA INC	\$334,608.36	\$220,841.51
SemMaterials LP	BP PRODUCTS N AM	\$0.00	\$0.00
SemMaterials LP	BRENNTAG NORTHEAST INC	\$32,790.00	\$21,641.40
SemMaterials LP	BRENNTAG PACIFIC INC	\$13,506.59	\$8,914.35
SemMaterials LP	BRENNTAG SOUTHWEST INC	\$15,008.54	\$9,905.64
SemMaterials LP	CALUMET SHREVEPORT LUBRICANTS	\$1,244,683.00	\$821,490.78
SemMaterials LP	CALUMET SHREVEPORT LUBRICANTS & WAXES LLC	\$25,426.50	\$16,781.49
SemMaterials LP	CENEX COOP SUPPLY INC	\$86,547.48	\$57,121.34
SemMaterials LP	CENTROTRADE MINERALS & METALS INC	\$423,930.58	\$279,794.18
SemMaterials LP	CHS INC BY PRODUCTS	\$116,097.69	\$76,624.48
SemMaterials LP	CHUSEI-WAXTECH POLYMERS LLC	\$3,351.40	\$2,211.92
SemMaterials LP	CIC FRONTIER CORPORATION & FRONTIER TERMINAL, LLC	\$2,268,684.58	\$1,497,331.82
SemMaterials LP	CITY OF HALSTEAD	\$0.00	\$0.00
SemMaterials LP	CL CONSULTING & MANAGEMENT CORP	\$122,855.76	\$81,084.80
SemMaterials LP	COFFEYVILLE RESOURCES LLC	\$147,113.08	\$97,094.63
SemMaterials LP	CONOCOPHILLIPS CO	\$1,106,834.02	\$730,510.45
SemMaterials LP	CONSTELLATION NEWENERGY	\$127,805.67	\$84,351.74
SemMaterials LP	CUMMINS INVESTMENT CORP	\$296,461.95	\$195,664.89
SemMaterials LP	DESIGN PLAST	\$0.00	\$0.00
SemMaterials LP	DYNASOL LLC	\$2,898,633.13	\$1,913,097.87
SemMaterials LP	EXCEL POLYMERS LLC	\$53,099.39	\$35,045.60
SemMaterials LP	EXXONMOBIL CORPORATION	\$162,528.78	\$107,269.00
SemMaterials LP	EXXONMOBIL OIL CORP	\$5,491,812.84	\$3,624,596.47
SemMaterials LP	FAUSER ENERGY RESOURCES	\$32,958.46	\$21,752.58
SemMaterials LP	FEDERATED COOP LTD	\$56,048.98	\$36,992.33
SemMaterials LP	FEDERATED CO-OPERATES LTD	\$776,362.81	\$512,399.46
SemMaterials LP	FRONTIER OIL AND REFINING CO	\$1,972,222.58	\$1,301,666.90
SemMaterials LP	GARY WILLIAMS ENERGY CORP	\$751,392.98	\$495,919.37
SemMaterials LP	GARY-WILLIAMS ENERGY CORP.	\$169,911.36	\$112,141.50
SemMaterials LP	GEORGIA PACIFIC CHEMICALS LLC	\$21,333.67	\$14,080.22
SemMaterials LP	GLADIEUX TRADING & MARKETINC CO LP	\$8,893.80	\$5,869.91
SemMaterials LP	GLADIEUX TRADING & MARKETING	\$86.47	\$57.07
SemMaterials LP	GROWMARK, INC.	\$28,070.38	\$18,526.45
SemMaterials LP	HERCULES INC	\$80,967.60	\$53,438.62

Debtor	Claimant	503(b)(9) Disputed Amount1	Proposed Settlement Amount
SemMaterials LP	HOLLY REFINING & MARKETING CO	\$807,359.67	\$532,857.38
SemMaterials LP	HUSKY OIL MARKETING CO	\$1,466,595.88	\$967,953.28
SemMaterials LP	ICL PERFORMANCE PRODUCTS LP	\$85,699.08	\$56,561.39
SemMaterials LP	IGI RESOURCES, INC	\$0.00	\$0.00
SemMaterials LP	IKA WORKS (ASM CAPITAL III)	\$56,376.53	\$37,208.51
SemMaterials LP	IMPERIAL OIL PRODUCTS & CHEMICAL DIV	\$997,139.76	\$658,112.24
SemMaterials LP	INNOPHOS INC	\$73,599.86	\$48,575.91
SemMaterials LP	INNOPHOS INC.	\$6,679.43	\$4,408.43
SemMaterials LP	INTERNATIONAL SULPHUR INC	\$19,975.00	\$13,183.50
SemMaterials LP	JAT OIL & SUPPLY INC	\$71,671.57	\$47,303.23
SemMaterials LP	JT RUSSELL & SONS INC	\$231,948.96	\$153,086.31
SemMaterials LP	KAO SPECIALTIES AMERICAS LLC	\$12,575.90	\$8,300.09
SemMaterials LP	KOCH SULFUR PRODUCTS CO LLC	\$46,087.29	\$30,417.61
SemMaterials LP	KRATON POLYMERS US LLC	\$305,536.00	\$201,653.76
SemMaterials LP	LCY ELASTOMERS LP	\$318,864.00	\$210,450.24
SemMaterials LP	LEIGHOW OIL CO INC	\$12,852.70	\$8,482.78
SemMaterials LP	LION OIL CO	\$237,443.82	\$156,712.92
SemMaterials LP	MARATHON PETROLEUM COMPANY, LLC	\$395,050.25	\$260,733.17
SemMaterials LP	MARIE INDUSTRIES	\$115,038.79	\$75,925.60
SemMaterials LP	MARTIN LIMESTONE INC	\$4,740.33	\$3,128.62
SemMaterials LP	MOOSE JAW REFINERY INC	\$460,415.15	\$303,874.00
SemMaterials LP	NUSTAR MARKETING LLC	\$402,825.88	\$265,865.08
SemMaterials LP	PEDRONI FUEL CO	\$37,054.94	\$24,456.26
SemMaterials LP	PIONEER OIL, LLC	\$746,692.23	\$492,816.87
SemMaterials LP	REAGENT CHEMICAL RESEARCH INC	\$5,566.34	\$3,673.78
SemMaterials LP	RIVER CITY OIL	\$8,478.00	\$5,595.48
SemMaterials LP	SAFETY KLEEN SYSTEMS INC	\$115,559.38	\$76,269.19
SemMaterials LP	SHELL ENERGY	\$77,459.20	\$51,123.07
SemMaterials LP	SMURFIT-STONE CONTAINER ENTERPRISES INC	\$10,614.24	\$7,005.40
SemMaterials LP	SUNOCO INC.	\$110,130.41	\$72,686.07
SemMaterials LP	TEX-CON OIL CO	\$30,332.20	\$20,019.25
SemMaterials LP	TOTAL PETROCHEMICALS USA INC	\$446,974.22	\$0.00
SemMaterials LP	TRUMAN ARNOLD CO	\$42,994.01	\$28,376.05
SemMaterials LP	ULTRAPAVE CORP	\$111,955.58	\$73,890.68
SemMaterials LP	UNITED REFINING CO	\$1,088,941.44	\$0.00

		A CONTRACTOR OF THE PROPERTY O	and the state of t
Debtor	Clamant	503(b)(9) Disputed Amount 1 Prop	osed Settlement Amount
SemMaterials LP	UNIVAR USA INC	\$17,006.97	\$11,224.60
SemMaterials LP	VALERO MARKETING & SUPPLY	\$162,478.45	\$107,235.78
SemMaterials LP	WESTERN REFINING COMPANY, L.P.	\$6,942.09	\$4,581.78
Total SemMaterials	140 Claimants	\$30,769,976.64	\$19,294,480.25
SemStream	AKA ENERGY GROUP LLC	\$535,835.48	\$353,651.42
SemStream	AMERIGAS	\$18,904.31	\$12,476.84
SemStream	ANADARKO ENERGY SERVICES CO	\$720,753.40	\$475,697.24
SemStream	ATLANTIC ENERGY INC	\$38,110.08	\$25,152.65
SemStream	BLM-AMARILLO FIELD OFFICE	\$119,763.40	\$79,043.84
SemStream	BP CANADA ENERGY MARKETING CORP	\$54,000.00	\$35,640.00
SemStream	ENCANA MARKETING (USA) INC	\$17,247.50	\$11,383.35
SemStream	CITGO PETROLEUM CORP	\$2,721,181.68	\$1,795,979.91
SemStream	CROSSTEX NGL MARKETING LP	\$2,240,410.92	\$1,478,671.21
SemStream	CHEVRON PRODUCTS CO	\$1,522,835.29	\$1,005,071.29
SemStream	HARMATTAN GAS PROCESSING LP	\$52,462.84	\$34,625.47
SemStream	KERR MCGEE GATHERING LLC	\$355,871.64	\$234,875.28
SemStream	SEA-3 INC	\$125,898.42	\$83,092.96
SemStream	SOLAR GAS INC	\$158,913.61	\$104,882.98
SemStream	TARGA MIDSTREAM SERVICES LP	\$32,733.35	\$21,604.01
SemStream	WESTERN GAS RESOURCES INC	\$46,544.90	\$0.00
Total SemStream	16 Claimants	\$8,761,466.82	\$5,751,848.47
TOTAL	253 Claimants	\$154,633,636.62	\$100,647,442.35

These amounts were obtained from Debtors' 503(b)(9) Notices filed on July 17, 2009 [Docket No. 4660] or, for those claimants not listed in the Notices, from Debtors' 4th Amended Schedules. The entirety of these claims are subject to dispute pursuant to the 503(b)(9) global legal objections and the factual objections made by Debtors in their 503(b)(9) Notices and by the Administrative Agent and the Creditors' Committee in their objections to Debtors' 503(b)(9) Notices.

Exhibit 11.1

UNITED STATES DISTRICT COURT for the

Northern District of Oklahoma

Plainty	Action No. 08-MD-1989-GKF-FHM
	action is pending in another district, state where;
Deficials	
SUBPOENA TO PRODUCE DOCUMENTS, INFO OR TO PERMIT INSPECTION OF	
To: SemGroup Holdings, LTP.	
Two Warren Place, 6120 South Yale Avenue; Suite 700, Tulsa, OK	
documents, electronically stored information, or objects, and permit their material: See Schedule 'A'	
Place, Nelson, Roselius, Terry, O'Hara & Morton Date	and Time: See Schedule "A"
3540 Soulf Sive; Suite 300 Edmond OK 73013	atus status
ij Inspection of Premises. YOU ARE COMMANDED to permit	entry onto the designated promises, land, or 🕾
other property possessed or compolled by you at the time, date, and local may inspect, measure, survey, photograph, test, or sample the property of	ion set forth below so that the requesting party, r any designated object or operation on it.
Place:	and Time:
The provisions of Fed. R. Civ. P. 45(c), relating to your protection	
45 (d) and (e), relating to your duty to respond to this subpoens and the particular	otential consequences of not doing so, are
Date: Sully 14, 2007 CLERK OF COURT	
CLERK OF COURT	
o	Spira Wagier Felixon
Signatura of Clerk or Deputy Clerk	Atterney's signature
The name, address, e-mail, and telephone number of the attorney represe Fund Advisors, LLC	nting (name of party) Lead Plaintiff, Harvest Who issues or requests this subposus, are:
Lauren Wagner Pederson, Esq., Barroway Topaz Kessler Meltzer & Che 280 King of Prussia Road, Radnor, PA 19087	the filter filter. The factor is a little filter for the transfer in the action of the filter and the filter it
Email: icederson@bikmc.com, Tel: 610-667-7706	

Civil Action No. 108-MD-1989-GKF-FHM

PROOF OF SERVICE

(This section should not be filed with the court unless required by Fed. R. Civ. P. 45.)

This subpoena for (name of individual and title, if any)	
was received by ma on (doie)	
I personally served the subpoena on the individual.	5 Alexander
Ly I personany served the subjection of the individual	on (date) or
П.I left the subpoena at the individual's residence or u	really full of the control of the co
47 42 46 46 46 46 46 46 46 46 46 46 46 46 46	person of suitable age and discretion who resides there, he individual's last known address; or
[1] I served the subpoent to (name of individual) designated by law to accept service of process on beh	Wito 15
	On (date)
☐ I returned the subpoena tinexecuted because	i.Or
O other (specify)-	
Unless the subpoena was issued on behalf of the Unite	d States, or one of its officers or agents, I have also
tendered to the witness fees for one day's attendance,	and the miteage allowed by law, in the amount of
My fees are S for travel and \$	for services, for a total of \$ 0.00
I declare under penalty of perjury that this information	is true.
Date:	Server's signature
	Server & Segnature
	Printed name and title
	s s anca man ana
	。 《《《《《································
	Server's oddress (1988) And Andrews

Additional information regarding attempted service, etc:

SCHEDULE A

DATE AND TIME TO RESPOND:

The time for responding to this subpoens to produce documents is extended until five (5) days following the final disposition of the Motion for a Limited Modification of the Automatic Stay (Case Nos. 08-11525, 08-12504) filed by Lead Plaintiff, Harvest Fund Advisors, LLC.

L. DEFINITIONS

The following definitions apply to the enumerated terms listed below:

- 1. The terms YOU or YOUR means SemGroup Holdings, L.P., and any Person acting on behalf of or as an agent or representative of SemGroup Holdings, L.P.
- The term PERSON or PERSONS means any individual, firm, proprietorship, corporation, association, or any other type of organization or entity
- The term DOCUMENT is defined to have the same meaning and be equal in scope of the usage of the term in Rule 34 of the Federal Rules of Civil Procedure as well as the term "writings" as defined by Rule 1001(1) of the Federal Rules of Evidence, and includes all electronically stored information, whether printed or displayed, and any preliminary versions drafts; revisions thereof. Documents include, by way of example, only, any memorandium, letter, envelope, correspondence, electronic mail, report, note, handwritten materials. Post-lt, message, telephone message, voicemail message, transcription of voicemail message, telephone log, diary, journal, appointment calendar, desk calendar, pocket calendar, Palm pilot calendar or other similar handheld device or personal organizer, group schedule calendar, drawing, painting, meeting minutes, working paper, financial report, accounting report, work papers, drafts, facsimile, contract, invoice, record of purchase or sale, chart, graph, index, directory, computer

directory, computer disk, computer tape; or any other written, printed, typed, taped, filmed or graphic, matter however, produced or reproduced. Documents also include any metadata applicable to any Documents.

- A. The term "CREDITORS COMMITTEE" shall refer to the statutory committee of unsecured creditors appointed by the Office of the United States Trustee on August 1, 2008 in the Chapter 11 bankruptcy proceeding styled in re Semcrude, L.P. et al., Case No. 08-11525 (D. Del. Bankr.) (hereafter "SemGroup Bankruptcy Proceeding").
- The term "DEFENDANTS" means any defendant currently named in this action including SGLP, SemGroup Energy Partners G.P. LLC, Kevin L. Foxx; Gregory G. Wallace, Midhael J. Brochetti, Thomas L. Kivisto, W. Anderson Bishop, Brent Cooper; Brian F. Billings, Alex G. Stallings, Citigroup Global Markets, Inc., Merrill Lynch Pierce Ferner & Smith Inc. A.G. Edwards & Sons, Inc.; BOSC, Inc.; Goldman, Sachs & Co., J.P. Morgan Securities Inc., Raymond James & Associates, Inc., RBC Capital Markets Corporation, SMH Capital Inc., UBS Securities LLC, Wachovia Corporation, Wachovia Capital Markets, LLC, Wells Fargo, Bank of America, Ritchie Opportunistic Trading, Ltd., Carlyle/Riverstone, A.R. Thane Ritchie, Andrew Ward, and E. Bartow Jones.
- 6 The term DEBTOR shall refer to SenGroup L.P. ("SemGroup"); and each of several direct and indirect subsidiaries, all of which are the subject of the SemGroup Bankruptcy.

 Proceeding.
- The term EXAMINER shall mean Louis J. Freeh, in his capacity as the examiner. appointed by the United States Trustee, on September 10, 2008, in connection with the SemGroup Bankruptcy Proceeding.
 - 8. The term GOVERNMENT ENTITY shall mean any federal or state regulatory or

Scentifies and Exchange Commission, and the United States Department of Justice, or any enforcement agency, including but not lumited to, a state Attorney, Ceneral, the United States

committee of the Legislative Dranch of the United States government

The term PRODUCERS: COMMITTEE shall refer to the Official Producers!

Committee appointed by the United States Trustee, on October 28, 2008, in connection with the

SemCroup Bankrupicy Proceeding.

The term PUBLIC OFFERINGS shall mean the initial public offering in full

2007, and secondary public offering in February 2008 of SGLP common units.

The term SBC shall refer to the United States Securities, and Exchange

The term U.S. ATTORNEY shall refer to the United States Attorney, in his Commission

smodabiO to prinzid merseW and rot verrious seists befull as viranges

The term NYMEX refers to the New York Mercanifle Exchange.

The term OTC of "Over the Counter" refers to a security traded in some context.

other than on a formal exchange such as the New York Stock Exchange, American Stock

Exchange of MASDAO, for example, through a dealer network.

zuccezzorz, bareniz, divisions, affiliales, zubeidiaries, its present and former officers, directors, The term SOLP refers to SemiCroup Energy Partners, L.P., its predecessors,

employees, attorneys, accountants, advisors, agents or anyone acting or purporting to act on its

behalf,

RULES OF CONSTRUCTION

The following rules of construction shall apply to these requests for production of

documents

- 2. The word "ALL" means "any and all," and the word "any" means "any and all";
- 3. The term "INCLUDING" means "including but not limited to"
- 4. The connectives "AND" and "OR" shall be construed either disjunctively or conjunctively as necessary to bring within the scope of the discovery request all responses that might otherwise be construed to be outside of its scope;
 - 5. The use of the singular form of any word includes the plural and vice yersa; and
- 6. The use of any tense of any verb shall also include within its meaning all other tenses of that verb

III. INSTRUCTIONS

- These requests shall be deemed continuing pursuant to the provisions of Rule 26(e) of the Federal Rules of Civil Procedure and require the timely supplementation of responses and production in a timely manner if the party learns that in some material respect the disclosure or response is incomplete or incorrect; and if the additional or corrective information has not otherwise been made known to the other parties during the discovery process or in writing.
- 2. The originals of the DOCUMENTS requested herein must be produced as they are kept in the usual course of business (including the production of file folders OR other binding in which the DOCUMENTS have been filed) OR must be organized AND labeled to correspond with the specific paragraphs of this request for production.
- in responding to these requests, YOU shall produce all responsive DOCUMENTS which are in YOUR possession, custody OR control OR in the possession, custody OR control of YOUR predecessors, successors, parents, subsidiaries, divisions, OR affiliates, OR any of YOUR respective directors, officers, managing agents, employees, attorneys, accountants, OR

other representatives. A DOCUMENT shall be deemed to be within YOUR control if YOU have the right to secure the DOCUMENT OR a copy of the DOCUMENT from another person having possession OR custody of the DOCUMENT.

- If a DOCUMENT requested herein was formerly in YOUR possession, custody
 OR control but is not now, state:
 - (a) its date:
- (b) the name of each PERSON who anthored, signed OR received the DOCUMENT:
 - (c) its subject matter,
 - (d) its form;
 - (e) its current OR last known locations AND custodian; and if applicable;
- (f) the circumstances surrounding its loss OR destruction AND whether it is missing OR lost, destroyed, transferred voluntarily OR involuntarily to others. OR otherwise disposed of:
- 5. If YOU object to a request for production of DOCUMENTS on the grounds of privilege work product OR otherwise, state the grounds AND factual basis for YOUR objection with specificity, AND provide the following information in sufficient detail to permit the court to rule on YOUR claim:
- (a) the date, author, primary addressee AND secondary addressees or PERSONS copied; including the relationship of those PERSONS to the client AND/OR author of the DOCUMENT.
- (b) a brief description sufficient to identify the type (e.g., letter memorandum, notes), subject matter AND purpose of the DOCUMENT;
 - (c) ALL PERSONS to whom its contents have been disclosed; and
 - (d) the party who is asserting the privilege,

- 6. If a portion of any DOCUMENT responsive to these requests is withheld under claim, of privilege pursuant to instruction No. 4, any non-privileged portion of such DOCUMENT must be produced with the portion claimed to be redacted for privilege.
- 7. YOU are to produce each DOCUMENT requested herein in its entirety, without deletion OR excision (except as qualified by Instructions Nos. 4 AND 5 above) regardless of whether YOU consider the entire DOCUMENT within the scope of the requests.
- 8. If YOU object to a particular request for production, but the objection does not apply to ALL DOCUMENTS within the scope of that request, YOU must produce ALL requested DOCUMENTS that are outside the scope of the objection.
- 9 If a DOCUMENT requested herein has been copied, AND if any such copy is not identical because of subsequent modification OR otherwise, such non-identical copy must be produced.

CV. FORM OF PRODUCTION

ALL electronically stored information ("ESI") should be produced in native format linked to single page tagged image file format ("TIFF"). ESI in TIFF format should be identified by an Option cross-reference file. ALL metadata—data that describes the electronic files, e.g., "date last modified"—associated with ESI shall be produced in text format linked to the associated file.

V. REQUESTS

- ALL Requests for Production of DOCUMENTS from, responses thereto AND ALL DOCUMENTS that YOU have produced to, any GOVERNMENT ENTITY in response to any investigation; formal OR informal inquiry, OR request for information related to SemGroup or SGLP, INCLUDING the following:
 - a SemGroup's insolvency or liquidity problems;

- SemGroup's hedging activities and relationships with counterparties;
- 6. SemGroup's NYMEX and OTC accounts; trading; margin calls and transfer of the NYMEX account to Barclays Bank;
- d SemGroup's use of borrowed funds and funds generated from its operations or liquidation of assets and assets of entities owned and controlled by SemGroup to satisfy margin calls for SemGroup trading accounts or other accounts;
- e. Any related-party transactions involving Semicroup or SGLP and any

 DEFENDANT or spouse, affiliate or relative of any DEFENDANT.
- E. SGLP's PUBLIC OFFERINGS;
- g SGLP's credit facilities or lines of credit; and
- h. The change of control of SGLP and the removal of certain directors from the Board of Directors of SGLP's General Partner.
- 2. ALL Requests for Production of DOCUMENTS from, responses thereto AND ALL DOCUMENTS that YOU have produced to the EXAMINER.
- 3. ALL Requests for Production of DOCUMENTS from, responses thereto AND ALL DOCUMENTS that YOU have produced to the CREDITORS COMMITTEE.
- 4. ALL Requests for Production of DOCUMENTS from, responses thereto AND ALL DOCUMENTS that YOU have produced to the PRODUCERS COMMITTEE.
- 5. ALL Requests for Production of DOCUMENTS from, responses thereto AND ALL DOCUMENTS that YOU have produced to the DEBTOR.
- 6. Copies of transcripts of Rule 2004 Examinations, depositions or interviews of SGLP's or its General Partner's current and former directors, officers, executives, employees or agents.

Northern District of Oklahoma	
In re SemGroup Energy Partners, L.P. Sec. Littig	
Civil Action No. 06-MD-1989-GKF-FHM	
(If the action is pending in another district; state when	
SUBPOENA TO PRODUCE DOCUMENTS, INFORMATION, OR OBJECTS OR TO PERMIT INSPECTION OF PREMISES	
To: SemGroup, LLR. Two Warren Place: 6120 South Yale Avenue, Suite 700, Tulsa, OK 74136	
Production: YOU ARE COMM ANDED to produce at the time, date, and place set forth below the to documents electronically stored information, or objects, and permit their inspection, copying, testing, or samp material: See Schedule A.	ing of the
Place: Nelson Roselius: Terry, O'Hara & Morton Date and Time; See: Schedule: "A	
3540 South Blvd. Suite 300 Edmond/GK:73013	
Disspection of Premises. YOU ARE COMMANDED to permit entry onto the designated premises, last other property passessed or controlled by you at the time, date, and location set forth below, so that the request may inspect, measure, survey, photograph, test, or sample the property or any designated object or operation of	ung party
Place: 30 Date and Time:	
The provisions of Fed. RCiv. P. 45(c), relating to your protection as a person subject to a subpoens; 45 (d) and (e), relating to your duty to respond to this subpoens and the potential consequences of not doing sattached.	and Role o, are
Date: \square \frac{\frac{\frac{14}{200}}{200}}{200}	
LDD. THE ELECTION OF A COUNTY PROCESS AND A SHARE A CONTROL OF A COUNTY OF A C	10.
OR Jawen Ub July Signature of Clerk or Deputy Clerk Attorney's I known &	ucusi
The name, address, e-mail, and telephone number of the attorney representing (name of purby) Lead Plaintiff.	Harvest
Fund Advisors, LLC , who issues or requests this subpos	total bearing to the state of
Lauren Wagner Pederson, Esq., Barroway, Topaz Kessler Meltzer & Check, LLP 280 Knin of Pryssia Road, Radnor, PA 19087	

Civil Action No.: 06-MD-1989-GKF-FHM

PROOF OF SERVICE

(This section should not be filed with the court unless required by Fed. R. Civ. P. 45.)

	This subpoena for (name of individual and title, if any)					ionis
was rea	gived by me on (date)					
					PSE4	
	I personally served the subpoena on the indivi-	dual at (place)		V (STARSIN	i de la companya de	14/2
			on (date)		y	700
	☐ 1 left the subpoena at the individual's resident		of abade with Asses	學學學		
, Populario			uitable age and disc		College The	erior.
	on (date) and mailed a cop	- 7 - 24 - 31 - Call 11 - 11 - 11 - 11 - 11 - 11 - 11 - 1	TREES PROPERTY SELECTION AND ADMINISTRATION OF THE CO.	A TRANSPERT PROFILE		
	on (date) and manen a cop	A to the marking	iai s iast known and			
	1 A served the subpoens to (name of individual)		godravia vij		, who is	
	designated by law to accept service of process or	behalf of <i>(num</i> s	of organization)			
	的复数多数连续数据的基础多数的 经不同的分子 计自己的 计自己的 计自己的 经产品的	各种对抗于为15年20年8	on (date)		i de la companya de l	
		en en en en				
	[] I remined the subpoena unexecuted because	- an est est est est est est est. To final a la fille est est est est est est est est est es	dominato (n. 1864). Portato de Carlos (n. 1864).			
	(I) other (specify):					
Drophs South						4-47/84 4-3-4
	Unless the subpoena was issued on behalf of the tendered to the witness fees for one day's attenda	nce, and the mi	leage allowed by la	v, in the am	ount of	計画の
Mý fe	s are \$ for travel and \$		for services, for a t	otal of \$	0.00	
		કિલ્લો જુના કે જાત કરે છે. પ્રાપ્ત કર્યો કે જે જે જે જોઈ કરો છે.				કે લક્ષ્મિક ઇન્સ્થા
	I declare under penalty of perjury that this inform	etion is true				7. B
	I decisie and a periory dimensional					\$ (\$.5) \$ (\$.5)
Date:	2000,7200,720,720,720,720,720,720,720,72		Server's signoture		reneral e	
		M. W. Congress				
	gan fra General (1918) Gan Barton (1918) Gan Gan Gan Gan (1918)		Printed name and little	· Sir sir Nick	an was je i jin.	7 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
			a request from some state			
				智慧 表語		
			Server's address		ESEX.	
2 3312	and information regarding attempted service, etc.					2 3

SCHEDULE A

DATE AND TIME TO RESPOND:

The time for responding to this subpoena to produce documents is extended until five (3), days following the final disposition of the Motion for a Limited Modification of the Automatic Stay (Case Nos. 08-11525, 08-12504) filed by Lead Plaintiff, Harvest Fund Advisors, LLC.

I. DEFINITIONS

The following definitions apply to the enumerated terms listed below:

- 1. The terms YOU or YOUR means SemGroup L.P., and any Person acting on behalf of or as an agent or representative of SemGroup, L.P.
- 2 The term PERSON or PERSONS means any individual; firm, proprietorship, corporation, association, or any other type of organization or entity.
- The term DOCUMENT is defined to have the same meaning and be equal in scope of the usage of the term in Rule 34 of the Federal Rules of Civil Procedure, as well as the term "writings" as defined by Rule 1001(1) of the Federal Rules of Evidence, and includes all electronically stored information, whether printed or displayed, and any preliminary versions, drafts, revisions thereof. Documents include, by way of example, only, any memorandum, letter, envelope, correspondence, electronic mail, report, note, handwritten materials, Post-It, message, telephone message, volcemail message; transcription of voicemail message, telephone log, diary, journal, appointment calendar, desk calendar, pocker calendar, Palm pilot calendar or other similar handlield device or personal organizer, group schedule calendar, drawing, painting, meeting minutes, working paper, financial report accounting report, work papers, drafts, facsimile, contract, invoice, record of purchase or sale, chart, graph, index, directory, computer

directory, computer disk, computer tape, or any other written; printed, typed: taped. filmed or graphic; matter however; produced; or reproduced; Documents also include any metadata applicable to any Documents.

- 4. The term "CREDITORS COMMITTEE" shall refer to the statutory committee of unsecured creditors appointed by the Office of the United States Trustee on August 1, 2008 in the Chapter 11 bankruptcy proceeding styled in re-Semerade, I.P. et al., Case No. 08-11525 (D. Del. Bankr.) (hereafter "SemGroup Bankruptcy Proceeding").
- The term "DEFENDANTS" means any defendant currently named in this action including SGLF, SemGroup Energy Partners G.P. LLC, Kevin L. Foxx; Gregory G. Wallace, Michael J. Brochetti, Thomas L. Kivisto, W. Anderson Bishop, Brent Cooper, Brian F. Billings, Alex G. Stallings, Citigroup Global Markets, Inc.; Mertill Lynch Pierce, Fenner & Smith Inc.; A.G. Edwards & Sons, Inc. BOSC, Inc., Goldman, Sachs & Co., J.P. Morgan Securities, Inc., Raymond James & Associates, Inc., RBC Capital Markets-Corporation, SMH Capital Inc. UBS Securities LLC, Wachovia Corporation, Wachovia Capital Markets, LLC, Wells Fargo, Bank of America, Ritchie Opportunistic Trading, Ltd., Carlyle/Riverstone, A.R. Thane Ritchie, Andrew Ward, and B. Bartow Jones.
- 6. The term DEBTOR shall refer to SemGroup L.P. ('SemGroup'), and each of several direct and indirect subsidiaries, all of which are the subject of the SemGroup Bankruptcy Proceeding.
- 7. The term EXAMINER shall mean Louis J. Freeh, in his capacity as the examiner appointed by the United States Trustee, on September 10, 2008, in connection with the SemGroup Bankruptcy Proceeding.
 - 8. The term GOVERNMENT ENTITY shall mean any federal or state regulatory or

enforcement agency, including but not limited to, a state Attorney General, the United States Securities and Exchange Commission, and the United States Department of Justice, or any committee of the Legislative branch of the United States government.

- 9. The term PRODUCERS' COMMITTEE shall refer to the Official Producers' Committee appointed by the United States Trustee, on October 28, 2008, in connection with the SemGroup Bankruptcy Proceeding.
- 10. The term PUBLIC OFFERINGS shall mean the initial public offering in July 2007 and secondary public offering in February 2008 of SGLP common units.
- 11: The term SEC shall refer to the United States Securities and Exchange Commission.
- 12. The term U.S. ATTORNEY shall refer to the United States Attorney, in his capacity as United States Attorney for the Western District of Oklahoma.
 - 13. The term NYMEX refers to the New York Mercantile Exchange.
- 14. The term OTC or "Over the Counter" refers to a security traded in some context other than on a formal exchange such as the New York Stock Exchange, American Stock Exchange or NASDAQ, for example, through a dealer network.
- The ferm SGLP refers to SanGroup Energy Partners, L.P., its predecessors, successors, parents, divisions, affiliates, subsidiaries, its present and former officers, directors, employees, attorneys, accountants, advisors, agents or anyone acting or purporting to act on its behalf.

II. RULES OF CONSTRUCTION

1. The following rules of construction shall apply to these requests for production of

documents:

- 2. The word "ALL" means "any and all," and the word "any" means "any and all"
- The term "INCLUDING" means, including, but not limited to";
- 4. The connectives "AND" and "OR" shall be construed either disjunctively or conjunctively as necessary to bring within the scope of the discovery request all responses that might otherwise be construed to be outside of its scope;
 - The use of the singular form of any word includes the plural and vice yersa; and
- 6. The use of any tense of any verb shall also include within its meaning all other tenses of that verb.

III. INSTRUCTIONS

- These requests shall be deemed continuing pursuant to the provisions of Rule 26(e) of the Federal Rules of Civil Procedure and require the timely supplementation of responses and production in a timely manner if the party learns that in some material respect the disclosure or response is incomplete or incorrect, and if the additional or corrective information has not otherwise been made known to the other parties during the discovery process or in writing.
- The originals of the DOCUMENTS requested herein must be produced as they are kept in the usual course of business (including the production of file folders OR other binding in which the DOCUMENTS have been filed) OR must be organized AND labeled to correspond with the specific paragraphs of this request for production.
- 3. In responding to these requests, YOU shall produce all responsive DOCUMENTS which are in YOUR possession, custody OR control OR in the possession, custody OR control of YOUR predecessors, successors, parents, subsidiaries, divisions, OR affiliates, OR any of YOUR respective directors, officers, managing agents, employees, attorneys, accountants, OR

other representatives. A DOCUMENT shall be deemed to be within YOUR control if YOU have the right to secure the DOCUMENT OR a copy of the DOCUMENT from another person having possession OR custody of the DOCUMENT.

- 4. If a DOCUMENT requested herein was formerly in YOUR possession, custody.

 OR control, but is not now, state:
 - (a) its date:
- (b) the name of each PERSON who authored, signed OR received the DOCUMENT:
 - (c) its subject matter;
 - d) its form:
 - (e) its current OR last known locations AND custodian; and if applicable,
- the circumstances surrounding its loss OR destruction AND whether it is missing OR lost, destroyed, transferred voluntarily OR involuntarily to others, OR otherwise disposed of:
- 5. If YOU object to a request for production of DOCUMENTS on the grounds of privilege work product OR otherwise, state the grounds AND factual basis for YOUR objection with specificity. AND provide the following information in sufficient detail to permit the court to rule on YOUR claim:
- (a) the date, author, primary addressee AND secondary addressees or PERSONS copied, including the relationship of those PERSONS to the client AND/OR author of the DOCUMENT.
- (b) a brief description sufficient to identify the type (e.g., letter memorandum, notes), subject matter AND purpose of the DOCUMENT;
 - (c) ALL PERSONS to whom its contents have been disclosed; and
 - (d) the party who is asserting the privilege.

- 6. If a portion of any DOCUMENT responsive to these requests is Mithheld under claim of privilege pursuant to instruction No. 4, any non-privileged portion of such DOCUMENT must be produced with the portion claimed to be reducted for privilege.
- 7. YOU are to produce each DOCUMENT requested herem in its entirety, without deletion OR excision (except as qualified by Instructions Nos. 4 AND 5 above) regardless of whether YOU consider the entire DOCUMENT within the scope of the requests.
- 8 If YOU object to a particular request for production, but the objection does not apply to ALL DOCUMENTS within the scope of that request, YOU must produce ALL requested DOCUMENTS that are outside the scope of the objection.
- 9 If a DOCUMENT requested herein has been copied, AND if any such copy is not identical because of subsequent modification OR otherwise, such non-identical copy must be produced.

IV. FORM OF PRODUCTION

ALL electronically stored information ("ESI") should be produced in native format linked to single page tagged image file format ("TIFF"). ESI in TIFE format should be identified by an Opticon cross-reference file. ALL metadata – data that describes the electronic files, e.g., "date last modified" – associated with ESI shall be produced in text format linked to the associated file.

V. REQUESTS

- 1. ALL Requests for Production of DOCUMENTS from, responses thereto AND ALL DOCUMENTS that YOU have produced to, any GOVERNMENT ENTITY in response to any investigation, formal OR informal inquiry, OR request for information related to SemGroup or SGLP. INCLUDING the following:
 - a. SemGroup's insolvency or liquidity problems;

- b. SemGroup's hedging activities and relationships with counterparties.
- c SemGroup's NYMEX and OTC accounts trading; margin calls and transfer of the NYMEX account to Barclays Bank;
- d SemGroup's use of borrowed funds and funds generated from its operations of liquidation of assets and assets of entities owned and controlled by SemGroup to satisfy margin calls for SemGroup trading accounts or other accounts;
- e. Any related-party transactions involving SemGroup or SGLP and any DEFENDANT;

 DEFENDANT or spouse, affiliate or relative of any DEFENDANT;
- f SGLP's PUBLIC OFFERINGS;
- g. SGLP's credit facilities or lines of credit; and
- h. The change of control of SGLP and the removal of certain directors from the Board of Directors of SGLP's General Partner.
- 2. ALL Requests for Production of DOCUMENTS from, responses thereto AND ALL DOCUMENTS that YOU have produced to the EXAMINER.
- 3. ALL Requests for Production of DOCUMENTS from, responses thereto AND ALL DOCUMENTS that YOU have produced to the CREDITORS COMMITTEE.
- 4. ALL Requests for Production of DOCUMENTS from, responses thereto AND ALL DOCUMENTS that YOU have produced to the PRODUCERS COMMITTEE.
- ALL Requests for Production of DOCUMENTS from, responses thereto AND ALL DOCUMENTS that YOU have produced to the DEBTOR.
- 6. Copies of transcripts of Rule 2004 Examinations, depositions or interviews of SGLP's or its General Partner's current and former directors, officers, executives, employees or agents.

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re

SEMCRUDE, L.P., et al.,

Debtors.

Chapter 11

Case No. 08-11525 (BLS)

Jointly Administered

OFFICIAL COMMITTEE OF UNSECURED CREDITORS OF SEMCRUDE, L.P., et al. for and on behalf of the Debtors' Bankruptcy Estates, Plaintiff,

and

BANK OF AMERICA, N.A., as Administrative Agent for the Debtors' Pre-Petition and Post-Petition Lenders Plaintiff/Intervenor,

٧.

THOMAS L. KIVISTO, GREGORY C. WALLACE, WESTBACK PURCHASING CO., LLC, BRENT COOPER, KEVIN L. FOXX, ALEX G. STALLINGS Defendants.

Adversary No. 09-50189 (BLS).

DEFENDANT THOMAS L. KIVISTO'S FIRST SET OF REQUESTS FOR PRODUCTION OF DOCUMENTS TO PLAINTIFF, PLAINTIFF/INTERVENER, AND DEBTORS

Pursuant to Rule 34 of the Federal Rules of Civil Procedure, made applicable to this proceeding by Federal Rule of Bankruptcy Procedure 7034, Defendant Thomas L. Kivisto ("Kivisto") requests that Plaintiff, the Official Committee of Unsecured Creditors (the "Committee") of SemCrude, L.P., its parent SemGroup, L.P., and certain direct and indirect subsidiaries of SemGroup, L.P. (the "Debtors"), Plaintiff/Intervenor Bank of America, N.A., and the Debtors produce the following documents for inspection and copying at the offices of

Greenberg Traurig LLP, 300 West 6th Street, Suite 2050, Austin, Texas, 78701, on or before thirty (30) days from the date of service, or at such other location and date upon which the parties may mutually agree.

INSTRUCTIONS

- 1. All Documents within Your possession, custody or control shall be produced. Without limitation, a Document is deemed to be in Your control if You have a superior right to compel the production of the requested Document from a third party (including any agent, affiliate, authority, or representative). Pursuant to the standing orders dated March 15, 2009, June 5, 2009, and October 5, 2009, the Committee is authorized to stand in the shoes of the Debtors for the purposes of this adversary proceeding. As such, documents in the Debtors* possession, custody or control shall be considered in the Committee's possession, custody or control.
- 2. If You claim that an attorney-client privilege or any other privilege applies to any information or Document requested herein, You are requested to (a) state with specificity the claim of privilege or protection used to withhold the information, and (b) provide all requested information that is not subject to a claim of privilege or protection by redacting, excising, or otherwise protecting the portions for which a privilege or protection is asserted.
- 3. These requests are deemed to be continuing so as to require You to serve supplemental responses if You, Your attorney, or any Person acting on Your behalf obtains additional Documents or information before the time of trial. Such supplemental responses should be served promptly upon obtaining the Documents or information. See FED. R. CIV. P. 26(e).

- 4. For the purpose of interpreting or construing the scope of these requests, terms should be given their most expansive and inclusive interpretations unless otherwise specifically limited in any particular request. This includes, without limitation, the following:
 - a. construing "and" as well as "or" in the conjunctive or disjunctive as necessary to make the request more inclusive;
 - construing the singular form of a word to include the plural and the plural form to include the singular;
 - c. construing the past tense of a verb to include the present and the present tense to include the past; and
 - d. construing the terms "refer," "reflect" and "relating" to include any connection whatsoever, direct or indirect, with the requested subject matter.
- 5. In making production, produce all Documents as kept in the normal course of business and identify the file from which each document was taken. If any tape, disk, card, drive, wire, or any other form of electronic or mechanical recording or transcript or any computer program is produced, please produce such Documents as are necessary for decoding, playing back, printing and/or interpretation thereof, and any other Documents that are necessary to convert said information into a useable format. Kivisto expressly reserves the right to request that any Documents produced in paper format be produced in their electronic (*i.e.*, native) format to the extent such Documents are available in electronic format. All electronically stored information ("ESI") should be produced in native format, including all metadata.

DEFINITIONS

- 1. "Acknowledgement" means the alleged acknowledgement signed by Defendant Kivisto purporting to offer Westback's assets and Defendant Kivisto's personal assets to support the creditworthiness of the Receivable, as alleged in Paragraph 78 of the Third Amended Complaint.
- 2. "Bonuses" means the bonuses allegedly paid by the Debtors to Defendants

 Kivisto, Gregory Wallace, Kevin L. Foxx, Melinda Oven and James Coen from 2004 to 2008, as
 alleged in Paragraphs 106 and 260 of the Third Amended Complaint.
- 3. "Communication(s)" means the transmittal or exchange of any information, including but not limited to sending or receiving any kind of information by or through speech, writings, Documents, language (machine, foreign or otherwise), computer electronics or electronic data, sound, radio or video signals, telecommunication, telephone, teletype, facsimile, telegram, microfilm, microfiche, photographic film of all types or other media of any kind. The term "Communication(s)" also includes, without limitation, all inquiries, discussions, conversations, correspondence, negotiations, agreements, understandings, meetings, notices, requests, responses, demands, complaints, press, publicity, or trade releases.
 - 4. "Complaint" means the Third Amended Complaint in the above-captioned matter.
- 5. The "Examiner" means Louis J. Freeh, Esq., the examiner appointed by the United States Trustee and approved by the Bankruptcy Court by order dated October 14, 2008, and its counsel and expert consultants.
- 6. "Debtors" means SemCrude, L.P., its parent SemGroup, L.P., and certain direct and indirect subsidiaries of SemGroup, L.P. as set forth in the Third Amended Complaint.

- 7. "Defendants" means Thomas L. Kivisto, Gregory C. Wallace, Westback Purchasing Co., LLC, Brent Cooper, Kevin L. Foxx and Alex G. Stallings.
- 8. "Document" is defined to be synonymous in meaning and equal in scope with the usage of this term in Federal Rule of Civil Procedure 34 and that is now in your possession, custody or control, or in the possession, custody or control of your advisors, agents, employees, servants, representatives, trustees, counsel or other persons acting or purporting to act on your behalf. A draft of a non-identical copy is a separate document within the meaning of this term.
- 9. "Evidence" or "evidencing" means constituting, proving, reflecting, indicating, or indicative of the existence or nature of any fact, allegation or given matter.
- 10. "Financial Statements" includes but is not limited to interim, final, pro forma, actual, projected, complete, or partial, annual, quarterly, monthly, weekly or otherwise, audited or unaudited, budgets, budget plans, balance sheets, schedules of direct costs, schedules of miscellaneous income, schedules of general services, fiscal serviceman administrative services, statements of earnings and earnings per share, income statements, cash flow statements, statement of revenues and statements of expenses, all notes or other commentary concerning any of the foregoing and all underlying workpapers and all drafts used in connection with the preparation of any of the foregoing.
- 11. "Goldman Sachs" means Goldman, Sachs & Co. and J. Aron & Co., and any entity affiliated therewith, including any of their predecessors, successors, divisions, subsidiaries, partnerships, officers, directors, members, shareholders, partners, employees, agents, and/or any other affiliated entity or anyone acting or purporting to act on their behalf.
- 12. "Kivisto Affiliate" means any entity other than Westback that allegedly transacted with SemGroup for Defendant Kivisto's own personal benefit, including but not limited to any

art gallery, Cogan CD Development, LLC, Ivory Energy, L.L.C., D'Novo Lean Gourmet, L.L.C., or Quote, L.L.C., as alleged in Paragraph 114 of the Third Amended Complaint.

- 13. "Management Committee" means the Management Committee of SemGroup Energy Partners G.P., L.L.C., an Oklahoma limited liability company that is the sole general partner of SemGroup Energy Partners, L.P.
- 14. "Person" or "persons" means any natural person(s) or business, legal or governmental entity or association, including but not limited to any proprietorships, corporations, partnerships, limited liability companies, limited liability partnerships, general partnerships, limited partnerships, master limited partnerships, trusts, joint ventures, groups, associations, organizations, governmental agencies, and other agencies.
- 15. "Receivable" means the alleged accounts receivable of approximately \$289 million owed by Westback to the Debtors related to trading in commodity and derivative securities.
- 16. "Refers," "relates," "relating," "reflecting" or "referring or relating to" means, in addition to the usual and customary meaning, concern or concerning, depict or depicting, discuss or discussing, refer or referring, reflect or reflecting, support or supporting, refute or refuting, address or addressing, evidence or evidencing, and record or recording.
- 17. "Ritchie" means Ritchie Opportunistic Trading, Ltd., Ritchie Capital Management, Ltd., Ritchie Capital Management, L.L.C., A.R. Thane Ritchie, and any entity affiliated therewith, including any of its predecessors, successors, divisions, subsidiaries, partnerships, officers, directors, members, shareholders, partners, employees, agents, and/or any other affiliated entity or anyone acting or purporting to act on its behalf.

- 18. "Riverstone" means C/R SemGroup Investment Partnership, L.P., C/R Energy Coinvestment II, L.P., Carlyle/Riverstone Global Energy and Power Fund II, L.P., investment partnerships affiliated with Carlyle/Riverstone Global Energy and Power Fund II, L.P., Riverstone Holdings, LLC, The Carlyle Group, and any entity affiliated therewith, including any of its predecessors, successors, divisions, subsidiaries, partnerships, officers, directors, members, shareholders, partners, employees, agents, and/or any other affiliated entity or anyone acting or purporting to act on its behalf.
- 19. "SemGroup" means the Debtors, SemGroup Energy Partners G.P., L.L.C., SemGroup Energy Partners, L.P., Eaglwing, L.P., and any entity affiliated therewith, including any of its predecessors, successors, divisions, subsidiaries, partnerships, officers, directors, members, shareholders, partners, employees, agents, and/or any other affiliated entity or anyone acting or purporting to act on its behalf.
 - 20. "SEC" means the Securities and Exchange Commission.
- 21. "SEC filings" means all Documents filed or furnished or prepared for the purpose of filing or furnishing with the SEC and any other state or federal regulatory agency, including, but not limited to, Forms 8-K, 10-K, 10-Q, Schedules TO-T, 13-D, 14A, 14D-1, and 14D-9, and any drafts thereof or amendments thereto.
- 22. "Statement(s)" means any oral, written, stenographic or recorded declaration of any kind or description.
- 23. "Trading Book" means an accounting book or statement of position that reflects all trading positions in securities.
- 24. "VAP-IV" means VAP-IV, LLC and any entity affiliated therewith, including any of its predecessors, successors, divisions, subsidiaries, partnerships, officers, directors, members,

shareholders, partners, employees, agents, and/or any other affiliated entity or anyone acting or purporting to act on its behalf.

- 25. "Westback" means Westback Purchasing Co., LLC and any entity affiliated therewith, including any of its predecessors, successors, divisions, subsidiaries, partnerships, officers, directors, members, shareholders, partners, employees, agents, and/or any other affiliated entity or anyone acting or purporting to act on its behalf.
 - 26. "You" or "Your" refers to the Debtors, the Committee and Bank of America.

TIME PERIOD

Unless stated otherwise, the time period to which these requests relate is January 1, 2005 through the date of production. If a document prepared before this time period is necessary for a correct or complete understanding of any document covered by a request, you must produce the earlier or subsequent document as well. If any document is undated and the date of its preparation cannot be determined, the document shall be produced if otherwise responsive to any request.

REQUESTS FOR PRODUCTION

REQUEST NO. 1:

All Documents You or SemGroup received or viewed from the Examiner.

REQUEST NO. 2:

All Documents You or SemGroup produced, furnished, or made available to the Examiner.

REQUEST NO. 3:

All Documents You or SemGroup produced, furnished, or made available to the SEC, the Department of Justice, the United States Commodity Futures Trading Commission, the

International Swaps and Derivatives Association, and any other governmental or regulatory body or agency relating to the allegations in the Complaint, prior versions of the Complaint, or the matters discussed in the Examiner's Report.

REQUEST NO. 4:

All Documents produced, furnished, or made available to You by any person in response to a formal request (e.g., pursuant to an order of the Court authorized by Federal Rule of Bankruptcy Procedure 2004) or informal request that relate to the allegations in the Complaint, prior versions of the Complaint, or the matters discussed in the Examiner's Report.

REQUEST NO. 5:

All Documents constituting or reflecting Communications between SemGroup, L.P., on the one hand, and Hall Estill, Baker Botts, or any other law firm, on the other hand, that relate to the allegations in the Complaint, prior versions of the Complaint, or the matters discussed in the Examiner's Report.

REQUEST NO. 6:

All articles of incorporation, bylaws, partnership agreements, and any other Documents governing the formation, organization, or management of SemGroup, L.P., SemGroup Energy Partners G.P., L.L.C., SemGroup Energy Partners, L.P., Eaglwing, L.P., and Westback.

REQUEST NO. 7:

All Documents depicting SemGroup's organizational structure, including but not limited to organization charts reflecting SemGroup's owners, parents, divisions, subsidiaries, and affiliates.

REQUEST NO. 8:

All Documents relating to Westback's financial condition, including but not limited to any assets, liabilities, net worth, liquidity, operations, Financial Statements, tax records, bank accounts, trading accounts, and trading records.

REQUEST NO. 9:

All Documents relating to Westback's financial, economic, or similar relationship with SemGroup or Defendant Kivisto, including but not limited to any financial, economic, or other benefits obtained by or available to SemGroup from Westback, including but not limited to the following:

- a. SemGroup's ability to use, assume, or own Westback's Trading Book;
- b. SemGroup's ability to use, assume, or own Westback's strategies for trading commodity or derivative securities;
- c. SemGroup's ability to use, assume, or own any Westback profits;
- d. SemGroup's ability to use, assume, nominate, or own Westback's rights to receive crude oil or any other natural resource;
- e. SemGroup's ability to use, assume, or own Westback's rights to direct producer shipments of crude oil or any other natural resource;
- f. SemGroup's rights in the Receivable; and
- g. any related Communications involving SemGroup, the Management Committee, or any bank, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 10:

All Documents relating to Westback's contracts, agreements, arrangements, or similar relationships with SemGroup or Defendant Kivisto, including but not limited to the following:

- a. contracts, agreements, or arrangements between Eaglwing, Westback, and Defendant Kivisto;
- b. accounts receivable/payable or other assets/liabilities between Westback, SemGroup, or Defendant Kivisto;

- c. payments between Westback, SemGroup, or Defendant Kivisto, including the nature and source of funds used for such payments; and
- d. any related Communications involving SemGroup, the Management Committee, or any bank, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 11:

All Documents relating to Westback's trading in commodity or derivative securities, including but not limited to the Trading Book, trading strategies, trading risks, risk-management policies and practices, and margin requirements relating to Westback, SemGroup, L.P., SemGroup Energy Partners G.P., L.L.C., SemGroup Energy Partners, L.P., GSTRuE, or Defendant Kivisto. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 12:

All Documents relating to the Receivable, including any related Communications involving SemGroup, the Management Committee, or any bank, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 13:

All Documents relating to the Acknowledgement of the Receivable, including any related Communications involving SemGroup, the Management Committee, or any bank, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 14:

All Documents relating to, provided to, or generated by the Management Committee, including but not limited to any meeting agendas, meeting minutes, meeting notes, board

packets, correspondence, reports, consents, and any informational materials provided to or generated by the Management Committee.

REQUEST NO. 15:

All Documents relating to Defendant Kivisto's equity interests in SemGroup, L.P. and SemGroup Energy Partners G.P., L.L.C. via Eaglwing Energy, LLC, including but not limited to the availability of such units to satisfy the Receivable. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 16:

All Documents relating to Defendant Kivisto's equity interests in SemGroup, L.P. and SemGroup Energy Partners G.P., L.L.C. via the Thomas L. Kivisto Trust, including but not limited to the availability of such units to satisfy the Receivable. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, attorney, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 17:

All Documents relating to any valuation of SemGroup, L.P., SemGroup Energy Partners G.P., L.L.C., or Westback, or any valuation of the equity interests in these entities. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, investment bank, credit union, lender, consultant, appraiser, accountant, lawyer, financial or other advisor, or any other third party.

REQUEST NO. 18:

All Documents relating to any valuation of Defendant Kivisto's net worth or financial position. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 19:

All Documents relating to VAP-IV, including but not limited to the creation, ownership, and management of VAP-IV and the alleged sale of Westback's oil, gas, or other leases to VAP-IV. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 20:

All Documents relating to any proposed or actual credit facility, credit agreement, letter of credit, loan, or other debt financing of SemGroup, L.P., SemGroup Energy Partners G.P., L.L.C., SemGroup Energy Partners, L.P., Westback, or Defendant Kivisto. This request also includes any related Communications involving SemGroup, the Management Committee, or any bank, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 21:

All Documents relating to proposed or actual trading in commodities or derivative securities by SemGroup, L.P., SemGroup Energy Partners G.P., L.L.C., SemGroup Energy Partners, L.P., Westback, or Defendant Kivisto. This request includes but is not limited to any requirements or restrictions concerning the Trading Book, trading strategies, trading risks, risk-

management policies or practices, and margin requirements. This request also includes any related Communications involving SemGroup, the Management Committee, or any bank, attorney, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 22:

All Documents relating to any proposed private or public offering of SemGroup equity securities in 2007 and 2008, including but not limited to the proposed "GS TrUE" program or equity offering involving Goldman Sachs. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, attorney, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party. This request includes, but is not limited to, all due diligence materials.

REQUEST NO. 23:

All Documents relating to any proposed or actual Westback credit facility in 2008. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, attorney, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 24:

All Documents relating to any threatened, potential, or actual breach of any covenant, provision, requirement, or other obligation under any credit agreement, credit facility, letter of credit, or other debt financing, including but not limited to the Bond Indenture Agreement and Credit Agreement (as those terms are used in the Complaint). This request includes any related Communications involving SemGroup, the Management Committee, or any bank, investment

bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 25:

All Documents relating to the suspension, administrative leave, departure, resignation, or termination of Defendant Kivisto from his positions at SemGroup, L.P., SemGroup Energy Partners G.P., L.L.C., and SemGroup Energy Partners, L.P. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 26:

All Documents relating to Defendants' employment agreements with SemGroup and any Bonuses they received from SemGroup, including but not limited to the determination, approval, and payment of the Bonuses. This request includes any Communications involving SemGroup or the Management Committee.

REQUEST NO. 27:

All Documents relating to any proposed or actual due diligence performed by Ritchie and Riverstone related to their respective ownership interests in SemGroup from 2000 to the present, including but not limited to due diligence related to the acquisition of such interests and due diligence related to maintaining such interests.

REQUEST NO. 28:

All Documents relating to any profit, earnings, income, dividend, distribution, interest, proceeds, return, revenue, gain, or any other monetary benefit obtained by any entity or person as

a result of its ownership interest in SemGroup, L.P., including but not limited to Ritchie and Riverstone.

REQUEST NO. 29:

All Documents relating to any profit, earnings, income, dividend, distribution, interest, proceeds, return, revenue, gain, lien, bankruptcy right or any other pecuniary interest or benefit in SemGroup obtained by Ritchie or Riverstone relating to: (1) Goldman Sachs's trading in commodity or derivative securities with SemGroup, Westback, or Defendant Kivisto; or (2) the bankruptcy proceeding relating to the Debtors.

REQUEST NO. 30:

All Documents relating to any commercial transaction between any Kivisto Affiliate and SemGroup. This request includes any Communications involving SemGroup or the Management Committee.

REQUEST NO. 31:

All Documents relating to Defendant Kivisto's ownership or management of any Kivisto

Affiliate, including the time and other resources devoted by Defendant Kivisto, or any

SemGroup employee acting at Defendant Kivisto's direction, to any Kivisto Affiliate. This
request includes any Communications involving SemGroup or the Management Committee.

REQUEST NO. 32:

All Documents relating to the actual use or dedication of any SemGroup assets, employees, or other resources for the benefit of any Kivisto Affiliate. This request includes any Communications involving SemGroup or the Management Committee.

REQUEST NO. 33:

All Documents relating to any allegations of wrongdoing concerning Defendant Kivisto, Westback, any Kivisto Affiliate, VAP-IV, or Goldman Sachs, including but not limited to any right to turnover of property of the Debtors' estates, alleged fraudulent transfer, breach of fiduciary duty, corporate waste, breach of contract, unjust enrichment, speculation in or manipulation of commodity or derivative markets, and tortious business conduct.

REQUEST NO. 34:

All Documents relating to insurance policies or indemnification agreements that may provide coverage to Defendant Kivisto or Westback, individually or collectively, for any claims or causes of action asserted in this litigation or that may provide reimbursement for payments made in a defense of this litigation, including, without limitation, loss mitigation insurance or liability-sharing arrangements, including options to purchase such insurance or sharing arrangements, whether acquired before or after the initiation of the litigation.

REQUEST NO. 35:

All Documents concerning any presentation that SemGroup provided to any potential or actual investor or lender. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 36:

All Documents concerning any analysis or audit of SemGroup's trading in commodity or derivative securities. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, attorney, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 37:

All Documents concerning any Communications by any person that stated or suggested that SemGroup's trading in commodities or derivative securities was speculative.

REQUEST NO. 38:

All internal SemGroup policies, Credit Agreements and Bond Indentures that governed or restricted SemGroup's trading in commodity or derivative securities.

REQUEST NO. 39:

All Documents that indicate that SemGroup bought or sold "naked" options, as alleged in the Complaint at Paragraph 28. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, attorney, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 40:

All Documents that indicate that SemGroup engaged in trading in commodity or derivative securities that were not matched against physical inventory or an offsetting trading position. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, attorney, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 41:

All Documents that support Your contention that SemGroup was insolvent as of December 21, 2007. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, attorney, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 42:

All Documents that concern or discuss SemGroup's Risk Management Policy. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, attorney, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 43:

All Documents that suggest or indicate that SemGroup's trading in commodity or derivative securities violated the Risk Management Policy. This request includes any related Communications involving SemGroup, the Management Committee, or any bank, attorney, investment bank, credit union, lender, consultant, appraiser, accountant, financial or other advisor, or any other third party.

REQUEST NO. 44:

All Documents that suggest or indicate that SemGroup's trading in commodity or derivative securities violated SemGroup's lender covenants.

REQUEST NO. 45:

All of SemGroup's market-to-market ("MtM") reports, Position Reports and Borrowing Base Reports.

REQUEST NO. 46:

All Documents concerning any meeting or discussion between members of SemGroup's Risk Management Committee.

REQUEST NO. 47:

All Documents concerning Your allegation that Westback is Kivisto's alter ego.

REQUEST NO. 48:

All Documents that concern ConAgra's decision to terminate its trading relationship (over-the-counter account) with SemGroup.

REQUEST NO. 49:

All Documents that suggest or indicate that Eaglwing paid Gregory Price for services performed for Kivisto.

REQUEST NO. 50:

All Documents that suggest or indicate that Sharon Pens or Gregory Price accounted for Eaglwing trading activity outside of the SemGroup's ordinary accounting system, as alleged in the Complaint at Paragraph 68.

REQUEST NO. 51:

All Documents concerning any transfer of funds from SemGroup to Westback.

REQUEST NO. 52:

All Documents concerning any transfer of funds from Westback to SemGroup.

REQUEST NO. 53:

All Documents provided to any member of the Management Committee that mention Westback.

REQUEST NO. 54:

All Documents concerning any Communications with PricewaterhouseCoopers concerning Westback.

REQUEST NO. 55:

All Documents concerning any acknowledgment by Kivisto that Westback owed SemGroup any money as a result of trades entered into by SemGroup on behalf of Westback, as alleged in the Complaint in Paragraph 76.

REQUEST NO. 56:

All Documents concerning any acknowledgment by Kivisto that his personal assets were to be considered available to support the creditworthiness of the Westback Receivable, as alleged in the Complaint in Paragraph 78.

REQUEST NO. 57:

All Documents concerning any attempt by any person to monitor or track the Westback Receivable.

REQUEST NO. 58:

All Documents concerning Westback's transfer of its interests in oil and gas leases to VAP-IV, as alleged in the Complaint in Paragraph 90.

REQUEST NO. 59:

All Documents concerning any commission or other economic benefit paid by SemGroup to Westback for Westback acting as an agent of Eaglwing, as alleged in the Complaint in Paragraph 94.

REQUEST NO. 60:

All Documents concerning Eaglwing's receipt of \$13.1 million from Westback on or about July 25, 2007, as alleged in the Complaint in Paragraph 97.

REQUEST NO. 61:

All Documents concerning the negotiation of Kivisto's employment agreement dated April 30, 2004.

REQUEST NO. 62:

All Documents concerning SemGroup Management Committee's authorization of a \$100 million partner distribution on or about February 15, 2008.

REQUEST NO. 63:

All Documents concerning commissions paid to Kivisto that form the basis of Counts 9-12 of the Complaint.

REQUEST NO. 64:

All Documents concerning SemGroup's treatment of its trading in commodity or derivative securities for federal income tax purposes (e.g., whether SemGroup's trading qualify as hedging transactions or whether they are speculative trades).

REQUEST NO. 65:

All Documents exchanged between Weil Gotshal & Menges LLP and PA Consulting Group to analyze, quantify and document SemGroup's derivative trading, including any reports or summaries prepared by PA Consulting Group.

REQUEST NO. 66:

All Documents concerning any training in derivative trading provided to Mia Oven or James Coen.

REQUEST NO. 67:

All Documents concerning any analysis or review of SemGroup's trading activity, including Documents that identify alleged trading errors, improper trades, or trades allegedly made in violation of the Risk Management Policy.

REQUEST NO. 68:

All Documents indicating the petroleum product storage capacity of SemGroup.

REQUEST NO. 69:

All Documents concerning SemGroup's attempt to keep its Trading Book delta neutral.

REQUEST NO. 70:

All Documents used by SemGroup to manage its Trading Book, including all reports, spreadsheets or Documents provided to Coen and Oven by SemGroup's Treasury or Risk departments (e.g., Daily Position Reports, Daily Margin Reports, Price Shock Reports, Delta Position Reports, Reconciliation Schedules, risk graphs and bank reports).

REQUEST NO. 71:

All Documents that identify any person who had any access to trading records or trading position information.

Dated: October 13, 2009

GREENBERG TRAURIG, LLP

Paul R. Bessette (admitted pro hac vice)

Michael J. Biles (admitted pro hac vice)

Jesse Z. Weiss (admitted pro hac vice)

Kimberly G. Davis (admitted pro hac vice)

300 West Sixth Street, Suite 2050

Austin, Texas 78701

Telephone: 512.320.7200 Facsimile: 512.320.7210

Donald J. Detweiler (No. 3087)
Dennis A. Meloro (No. 4435)
The Nemours Building
1007 North Orange Street, Suite 1200
Wilmington, Delaware 19801
Telephone: 302.661.7000
Facsimile: 302.661.7360

COUNSEL FOR THOMAS L. KIVISTO

CERTIFICATE OF SERVICE

I hereby certify that on the 13th day of October, 2009, I caused the foregoing Defendant Thomas L. Kivisto's First Set of Requests for Production of Documents to Plaintiff and Plaintiff/Intervener to be served upon the following individuals by first-class mail, postage prepaid:

Martin A. Sosland Sylvia A. Mayer WEIL GOTSHAL & MANGES LLP 200 Crescent Court, Suite 300 Dallas, TX 75201

Mark D. Collins RICHARDS LAYTON & FINGER, PA One Rodney Square PO Box 551 Wilmington, DE 19899 Counsel for Debtors SemCrude, LP

Jane W. Parver KAYE SCHOLER 425 Park Avenue New York, NY 10022-3598

R. Stephen McNeill
Laurie Selber Silverstein
POTTER ANDERSON & CORROON LLP
1313 N. Market Street
P.O. Box 651
Wilmington, DE 19899
Counsel for Bank of America, N.A. as
Administrative Agent for the Debtors
Pre-Petition and Proposed Post-Petition
Lenders

Richard I. Werder, Jr.
R. Brian Timmons
Susheel Kirpalani
QUINN EMANUEL URQUHART
OLIVER & HEDGES LLP
51 Madison Avenue, 22nd Floor
New York, NY 10010

Bonnie Glantz Fatell
David W. Carickhoff
BLANK ROME LLP
1201 Market Street, Suite 800
Wilmington, DE 19801
Counsel for Official Committee of
Unsecured Creditors of Semcrude, L.P.,
et al.

Kimberly G. Davis